TREASURY OF THE INDIC SCIENCES series
Editor-in-Chief: Robert A.F. Thurman
Jey Tsong Khapa Professor of Indo-Tibetan Buddhist Studies,
Columbia University
President, American Institute of Buddhist Studies
Series Editor: Gary A. Tubb
Dharam Hinduja Senior Lecturer in Sanskrit Teaching
and Indic Research in the Department of Religion,
Columbia University
Executive Editor: Thomas F. Yarnall
Department of Religion,
Columbia University

Editorial Board: Ryuichi Abe, Jay Garfield, David Gray, Laura Harrington, Thubten Jinpa, Joseph Loizzo, Vesna Wallace, Christian Wedemeyer, Chunfang Yu

The American Institute of Buddhist Studies (AIBS), in affiliation with the Columbia University Center for Buddhist Studies and Tibet House US, has established the *Treasury of the Indic Sciences* series as a complement to its *Treasury of the Buddhist Sciences* series. For over fifteen hundred years the scholars representing various Buddhist traditions were actively engaged in dialogue and debate with representatives of various Vedist, Hindu, Jainist, and other religious and secular traditions. In this cosmopolitan milieu the shared arts and sciences of all these Indic traditions were constantly refined and developed. In full acknowledgement of this broader context, the AIBS is widening its Buddhistic mandate with the *Treasury of the Indic Sciences* series in which it is publishing translations of influential classics from the diversity of Indic traditions. Publications from both *Treasuries* will bring important Indic contributions into contemporary discourse, enriching and contributing to our global cultural, philosophical, spiritual, and scientific heritage.

# SCHOLASTIC SANSKRIT

5

### A HANDBOOK FOR STUDENTS

Gary A. Tubb and Emery R. Boose

by

Treasury of the Indic Sciences series

Published by The American Institute of Buddhist Studies at Columbia University in the City of New York

Co-published with Columbia University's Center for Buddhist Studies and Tibet House US

New York

X2007. 366 Universität Hamburg + Asien-Afrika-Institut

*Treasury of the Indic Sciences* A refereed series published by:

American Institute of Buddhist Studies Columbia University 80 Claremont Avenue, Room 303 New York, NY 10027

http://www.aibs.columbia.edu

Co-published with Columbia University's Center for Buddhist Studies and Tibet House US

Distributed by Columbia University Press

© 2007 by Gary A. Tubb and Emery R. Boose All rights reserved.

No portion of this work may be reproduced in any form or by any, electronic or mechanical, including photography, recording, or by any information storage and retrieval system or technologies now known or later developed, without written permission from the publisher.

Printed in Canada on acid-free paper.

ISBN 978-0-9753734-7-7 (cloth)

#### Library of Congress Cataloging-in-Publication Data

Tubb, Gary A. (Gary Alan), 1950-

Scholastic Sanskrit : a handbook for students / by Gary A. Tubb and Emery R. Boose.

p. cm. — (Treasury of the Indic sciences series) "Co-published with Columbia University's Center for Buddhist Studies and Tibet House US."

Includes bibliographical references and index.

ISBN 978-0-9753734-7-7 (hardcover : alk. paper)

1. Sanskrit prose literature—History and criticism—Handbooks, manuals, etc. 2. Sanskrit prose literature—Study and teaching—Handbooks, manuals, etc. I. Boose, Emery R. (Emery Robert), 1952–II. Title.

PK2903.T83 2007

891'.2808-dc22

2007000683

Dedicated to the memory of our teacher, Daniel Henry Holmes Ingalls (1916–1999), Wales Professor of Sanskrit, Harvard University

Se	ries F	Editor's Preface	xxi
Αι	thors	s' Preface	xxv
Ał	brev	iations	xxix
In	trodu	ction: The Sanskrit Commentary	. 1
	0.1	Importance of commentaries	1
	0.2	The five services of a commentary	3
		0.2.1 Padaccheda	4
		0.2.2 Padārthokti	· 4
		0.2.3 Vigraha	. 4
		0.2.4 Vākyayojanā	1
		0.2.5 Ākṣepasamādhāna	5
Pa		Methods of Glossing Gary A. Tubb	7
1	The	First Four Services	9
	1.1	Introduction	9
	1.2	Types of glossing	10
2	Divi	sion of Words	13
	1.3	Introduction	13
	1.4	Repetition of the words of the text	14
	1.5	Specification of word-division	14

Contents
----------

viii

	1.6	Citation of individual words	14
		1.6.1 <i>iti</i>	15
		1.6.2 - <i>śabda, -kāra</i>	15
		1.6.3 Inflected indeclinable	15
		1.6.4 Anubandhas	16
	1.7	Variant readings	16
3	Givi	ng the Meaning of Words	19
	1.8	Introduction	19
	1.9	Simple synonyms	20
	1.10	Location of glosses	20
		1.10.1 Enclitics	20
		1.10.2 Vigrahas	21
	1.11	Glossing by equivalent forms	21
		1.11.1 Periphrasis	21
		1.11.2 Participle	21
	1.12	Expansions of glosses	22
		1.12.1 -ādi	22
		1.12.2 Implications	22
		1.12.3 Implied contrasts	22
	1.13	More removed paraphrases	23
		1.13.1 ity arthah	23
		1.13.2 iti bhāvaḥ	24
		1.13.3 iti yāvat	25
		1.13.4 Other phrases	25
		1.13.5 anena, etena	25
		1.13.6 References to speech	26
	1.14	Meanings by extension	27
	1.15	Glossing of puns	27
	1.16	Definitions	28
		1.16.1 Definitions offered by the commentator	28
		1.16.2 Definitions quoted from dictionaries	29
	1.17	Treatment of special items	30
		1.17.1 Proper names	30
		1.17.2 Names of species	31
		1.17.3 Meanings of particles	31
		1.17.4 Other indeclinables	31

		1.17.5 Pronouns	32
4	Ana	lysis of Grammatical Complexes	33
	1.18		33
		Modes of complex formation	35
	1.20	General principles of analysis	35
	1.21		39
		1.21.1 Glossing the words used in the analysis	40
		1.21.2 Indicating the inflection of the original	41
5	-	ivative Verbs	43
		Introduction	43
		Desideratives	44
		Intensives	45
		Causatives	45
	1.26	Denominatives	46
		1.26.1 Desire to have	46
		1.26.2 Behavior toward	47
		1.26.3 Behavior like	47
		1.26.4 Becoming something	47
		e c	47
		1.26.6 Feeling something	48
		1.26.7 Doing or making something	48
		1.26.8 Making use of something	48
6	Prim	nary Nominal Derivation	49
	1.27	Introduction	49
		Basic bhāve meanings	50
	1.29		52
		1.29.1 kartari	53
		1.29.2 karmani	54
		1.29.3 karaņe	54
		1.29.4 sampradāne	54
		1.29.5 apādāne	55
		1.29.6 adhikarane	55
	1.30	Temporal restrictions	55
		1.30.1 <i>bhūte</i>	56

ix

		1.30.2 vartamāne	
		1.30.3 bhavişyati	7
	1.31	Other restrictions of meaning	
	1.32	Accompanying words (upapada)	3
		1.32.1 kartari krt with accusative	D
		1.32.2 kartari krt with non-accusative 6	1
		1.32.3 krt in the sense of another kāraka 62	2
		1.32.4 Temporal restrictions	2
		1.32.5 Other conditions of meaning	3
	1.33	Verbal participles	4
	1.34	Irregular forms (nipātana) 6	5
7	Seco	ndary Nominal Derivation 6'	7
'		Introduction	
		Dependent formulas of analysis	
	1.50	1.36.1 Accusative	
		1.36.2 Instrumental	
		1.36.3 Dative	
		1.36.4 Ablative	1
		1.36.5 Locative	2
	1.37		4
		1.37.1 tasyâpatyam	5
		1.37.2 tasya bhāvah	5
		1.37.3 Others	6
		1.37.4 General connection	7
	1.38	Nominative formulas of analysis	7
	1.39	Compound formulas of analysis	8
	1.40	Possessive suffixes (matvarthīya)	9
		1.40.1 Straightforward possession	0
		1.40.2 Additions of meaning	0
	1.41	Vibhaktisamjñaka suffixes	1
	1.42	Svārthika suffixes	2
		1.42.1 With no addition of meaning	
		1.42.2 Diminutives	
		1.42.3 Other additions of meaning	
		1.42.4 <i>cvi</i> , etc	
		1.42.5 Samāsânta suffixes	4

8 Nom	ninal Compounds	85
A. Ge	eneral Principles	85
1.43	Introduction	85
	Basic categories of compounds	86
	1.44.1 Avyayībhāva	86
	1.44.2 Tatpurusa	87
	1.44.3 Karmadhāraya	87
	1.44.4 Dvandva	87
	1.44.5 Bahuvrīhi	88
	1.44.6 Kevala-samāsa	88
1.45	Additional elements of meaning	89
	1.45.1 Comparison	89
	1.45.2 Aggregation	90
	1.45.3 Clarification of connection	90
1.46	Optional and obligatory compounds	90
	1.46.1 Asvapadavigraha compounds	<u>,</u> 90
	1.46.2 Conventional meanings	91
	1.46.3 Bound forms	92
1.47	Irregularities	92
B. Avy	yayībhāva Compounds	<b>9</b> 2
1.48	Introduction	<sup>:</sup> 92
	Optional avyayībhāvas	93
1.50	Obligatory avyayībhāvas	93
1.51	Irregular avyayībhāvas	95
C. Tat	tpurușa Compounds	96
1.52	Introduction	96
	Dependent tatpurusas	96
	1.53.1 Straightforward analysis	97
		100
	1.53.3 Partitive genitive as second member	101
1.54	<b>-</b>	102
	1.54.1 Adjective plus substantive	102
		103
	1.54.3 Substantive plus substantive	104

xi

	1.54.4	Additional specification of connection .						105
1.55	Karmad	<i>lhārayas</i> of comparison	•	•				106
	1.55.1	Upamānapūrvapada karmadhāraya						106
	1.55.2	Upamita-samāsa						107
	1.55.3	Rūpaka-samāsa	•					108
1.56	Dvigu c	compounds	•					109
	1.56.1	Taddhitārtha-dvigu		•		•	•	110
	1.56.2	Uttarapada-dvigu		•		•	•	110
	1.56.3	Samāhāra-dvigu						111
1.57	Tatpuru	sas with inseparable first member	•	•		•	•	112
	1.57.1	<i>ku, su,</i> etc					•	113
	1.57.2	Prādi-tatpurusas with fixed meaning .	•	•	•••	•	•	113
	1.57.3	Dhātu-lopa prādi-tatpurusa	•	•	•••	•		114
	1.57.4	Gati-tatpurusas	•	٠		•	•	116
1.58		e tatpurușas					•	118
1.59	Upapad	da compounds	•	٠	•••	•		119
1.60	Irregula	ar <i>tatpurusa</i> s					•	121
	1.60.1	•• •					•	121
	1.60.2	Mayūravyamsakādi compounds					•	123
	1.60.3	Suffix-formations resembling compound	ls	٠	•••	·	•	125
D. Dv	andva C	ompounds						125
1.61	Introdu	ction						125
1.62	Itaretar	ra dvandvas						126
1.63	Samāhā	āra dvandvas	•			•		126
E. <i>Bal</i>	huvrīhi	Compounds						127
1.64	Introdu	ction	•					127
1.65	Apposi	tional <i>bahuvrīhi</i> s		•				128
	1.65.1	Simple appositional bahuvrīhis						128
	1.65.2	Negative bahuvrīhis		•				130
	1.65.3	Prādi-bahuvrīhi compounds	•				•	131
	1.65.4	Saha-bahuvrīhi compounds	•		• •	•	•	131
	1.65.5	Anekapada-bahuvrīhi compounds						132
1.66	Non-ap	positional bahuvrīhis		•		•	•	133
	1.66.1	Locative as second member	•		• •		•	133
	1.66.2	Genitive as first member	•				•	133

### Contents

1.67	Special bahuvrīhis	135
	1.67.1 Samkhyā-bahuvrīhi compounds 🕁	135
	1.67.2 Dig-bahuvrīhi compounds	136
	1.67.3 Reciprocal combat	136
1.68	Irregular bahuvrīhis	136
	nger Compounds	137
1.69	Introduction	137
	General principles	138
	Special techniques	141
	1.71.1 Simpler formulas of analysis	141
	1.71.2 Separate analysis	141
1.72	Choice of options	142
	<i>seșa</i> Formations	147
	Introduction	
1.74	Dvandvâikaśeṣa	.147
1.75	Napumsakâikaśesa	148
0 Indi	cating the Construction	149
A. Ar	ranging the Words of the Text	<b>149</b> 149
<b>A. Ar</b> 1.76	ranging the Words of the Text	;
<b>A. Ar</b> 1.76 1.77	ranging the Words of the Text	149
<b>A. Ar</b> 1.76 1.77 1.78	ranging the Words of the Text	149 149
<b>A. Ar</b> 1.76 1.77 1.78 1.79	ranging the Words of the Text	149 149 149 149
<b>A. Ar</b> 1.76 1.77 1.78 1.79	ranging the Words of the Text	149 149 149 150
<b>A. Ar</b> 1.76 1.77 1.78 1.79 1.80	ranging the Words of the Text	149 149 149 150 151
<ul> <li>A. Ar</li> <li>1.76</li> <li>1.77</li> <li>1.78</li> <li>1.79</li> <li>1.80</li> <li>B. Oth</li> </ul>	ranging the Words of the Text	149 149 149 150 151 156
<b>A. Ar</b> 1.76 1.77 1.78 1.79 1.80 <b>B. Ott</b> 1.81	ranging the Words of the Text	149 149 149 150 151 156 161
<b>A. Ar</b> 1.76 1.77 1.78 1.79 1.80 <b>B. Ott</b> 1.81	ranging the Words of the Text	149 149 149 150 151 156 161 161
<b>A. Ar</b> 1.76 1.77 1.78 1.79 1.80 <b>B. Ott</b> 1.81	ranging the Words of the Text	149 149 149 150 151 156 161 161
<b>A. Ar</b> 1.76 1.77 1.78 1.79 1.80 <b>B. Ott</b> 1.81	ranging the Words of the Text	149 149 150 151 156 161 161 161 161
<b>A. Ar</b> 1.76 1.77 1.78 1.79 1.80 <b>B. Ott</b> 1.81	ranging the Words of the Text         Introduction         The kathambhūtinī approach         The anvayamukhī approach         Insertion of explanatory material         Variations on the two approaches         her Techniques         Introduction         Direct statement of connections         1.82.1 anvaya         1.82.2 sambandha	149 149 150 151 156 161 161 161 161 162
A. Ar 1.76 1.77 1.78 1.79 1.80 B. Otl 1.81 1.82	ranging the Words of the Text         Introduction         The kathambhūtinī approach         The anvayamukhī approach         Insertion of explanatory material         Variations on the two approaches         her Techniques         Introduction         1.82.1 anvaya         1.82.2 sambandha         1.82.3 višeṣaṇa	149 149 150 151 156 161 161 161 161 162 163

9

10

1.8	1.83.1       iti śesah	165 166 166 167 167 168 169
<b>D</b> (1		
Part ] t	I The Bhāṣya Style y Emery R. Boose	171
11 Tł	e Fifth Service	173
2.1		173
2.2		174
2.3		174
12 AI	stract Nouns	177
2.4	Introduction	. 177
2.4	Formation	. 177
	2.5.1 $t\bar{a}$ and $tva$	177
	2.5.2 -bhāvah and vrddhi derivatives	. 178
2.0	Significance	. 178
	2.6.1 Used by necessity	. 178
	2.6.2 Used by choice	. 179
2.	Common constructions	. 179
	2.7.1 With subjective genitive	. 179
	2.7.2 In compound	. 180
	2.7.3 Causal phrase within a sentence	. 180
	2.7.4 <i>iti</i> clause without abstract suffix	
	2.7.5 Two genitives	. 182
2.	Causal ablative and instrumental	. 182
2.	Other cases	. 183
2.9		. 183 . 183

		2.9.3 Dative of purpose	83
			84
			84
	2.10	Predicative instrumental	84
	2.11	Accusative with verb of motion	85
13		1	87
			87
	2.13		87
		2.13.1 Identification	87
		2.13.2 Dissection	38
		2.13.3 Grouping	38
	2.14	Relation to words outside the compound	39
	2.15	ka to mark a bahuvrīhi $\ldots \ldots 19$	90
	2.16	<i>tat</i> to mark a <i>dvandva</i>	Ð
	2.17	-bhūta to mark a karmadhāraya	91
	2.18	Pronouns in compound	<del>)</del> 2
	2.19	Idiomatic terms	93
		2.19.1 -antara	<del>)</del> 3
		2.19.2 -artha	)3
		2.19.3 -kalpa	<b>}</b> 3
		2.19.4 -gata	
		2.19.5 -garbha	)4
		2.19.6 - jana	94
		2.19.7 -jāta	)5
		2.19.8 <i>-pūrva</i>	95
		2.19.9 - <i>mātrā</i> 19	95
		2.19.10 - viśeșa 19	)6
14	Wor	d Order 19	17
	2.20	Introduction	17
	2.21	Subject and predicative nouns	17
		2.21.1 Predicative nouns	17
		2.21.2 Pronoun subjects	8
	2.22	Attributive and predicative nouns	9
		2.22.1 Word order with subjects	9
		2.22.2 Word order with other nouns	9

xv

xvi

.

		2.22.3 Markers of attributive qualifiers
	2.23	Scope of <i>iti</i>
		Analyzing long sentences
15		al Sentences 205
		Introduction
	2.26	Ablative and instrumental phrases
		2.26.1 Ablative causal phrases
		2.26.2 Series of causes
		2.26.3 <i>ca</i> in causal series
		2.26.4 Verbal nouns expressing cause
		<i>iti</i> clauses
	2.28	Relative pronouns and adverbs
	2.29	Locative absolute
		2.29.1 Expressing a cause
		2.29.2 Expressing a condition
•		2.29.3 Concessive locative absolute
		2.29.4 Abbreviated locative absolute
	0.00	Causal particle hi
	2.30	Causal particle $hi$
16		
16	Uses	of <i>iti</i> 215
16	<b>Uses</b> 2.31	of iti         215           Introduction
16	<b>Uses</b> 2.31 2.32	of iti         215           Introduction         215           Common constructions         215
16	<b>Uses</b> 2.31 2.32	of iti215Introduction215Common constructions215Quotation215
16	<b>Uses</b> 2.31 2.32	of iti       215         Introduction       215         Common constructions       215         Quotation       215         2.33.1       iti with source       216
16	<b>Uses</b> 2.31 2.32	of iti       215         Introduction       215         Common constructions       215         Quotation       215         2.33.1       iti with source       216         2.33.2       iti with ablative       217
16	<b>Uses</b> 2.31 2.32 2.33	of iti       215         Introduction       215         Common constructions       215         Quotation       215         2.33.1       iti with source       216         2.33.2       iti with ablative       217         2.33.3       iti with partial quotations       217
16	Uses 2.31 2.32 2.33 2.34	causal particle in 11111111111111111111111111111111111
16	Uses 2.31 2.32 2.33 2.34	of iti       215         Introduction       215         Common constructions       215         Quotation       215         2.33.1       iti with source       216         2.33.2       iti with ablative       217         2.33.3       iti with partial quotations       217         Explanation       218         Comment       218
16	Uses 2.31 2.32 2.33 2.34	of iti       215         Introduction       215         Common constructions       215         Quotation       215         Quotation       216         2.33.1       iti with source       216         2.33.2       iti with ablative       217         2.33.3       iti with partial quotations       217         Explanation       218         Comment       218         2.35.1       Comment with iti       218
16	Uses 2.31 2.32 2.33 2.34 2.35	causal particle in 11111111111111111111111111111111111
16	Uses 2.31 2.32 2.33 2.34 2.35	causar particle in 11111111111111111111111111111111111
16	Uses 2.31 2.32 2.33 2.34 2.35	of iti       215         Introduction       215         Common constructions       215         Quotation       215         Quotation       216         2.33.1       iti with source       216         2.33.2       iti with ablative       217         2.33.3       iti with partial quotations       217         Explanation       218         Comment       218         2.35.1       Comment with iti         2.35.2       Commonly used comments       219         Constructions with the infinitive       220         2.36.1       Infinitive with sak and yuj       220
16	Uses 2.31 2.32 2.33 2.34 2.35	of iti       215         Introduction       215         Common constructions       215         Quotation       215         Quotation       216         2.33.1       iti with source       216         2.33.2       iti with ablative       217         2.33.3       iti with partial quotations       217         Explanation       218         Comment       218         2.35.1       Comment with iti         2.35.2       Commonly used comments       219         Constructions with the infinitive       220         2.36.1       Infinitive with sak and yuj       220         2.36.2       Infinitive with arh       221
16	Uses 2.31 2.32 2.33 2.34 2.35 2.36	of iti       215         Introduction       215         Common constructions       215         Quotation       215         Quotation       216         2.33.1       iti with source       216         2.33.2       iti with ablative       217         2.33.3       iti with partial quotations       217         Explanation       218         Comment       218         2.35.1       Comment with iti         2.35.2       Commonly used comments       219         Constructions with the infinitive       220         2.36.1       Infinitive with sak and yuj       220

Contents

17			225
	2.38	Introduction	225
	2.39	Text and author of text	225
		2.39.1 Verbs of taking	225
		2.39.2 prakrta, aprakrta	226
		2.39.3 Author of text commented on	226
		L	227
		2.39.5 Participles of verbs of saying	228
		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	228
		2.39.7 Auto-commentary	229
	2.40	References to the commentary	229
		• •	229
		-	230
		2.40.3 Relative clause	230
		2.40.4 Reference to earlier topic	231
			231
			231
	2.42	Genitive in reference to particular views	232
18	Exar	nples and Comparisons	233
18			<b>233</b> 233
18	2.43	Introduction	-
18	2.43	Introduction	233
18	2.43	Introduction         Introduction<	233 233
18	2.43	Introduction	233 233 233
18	2.43 2.44	Introduction	233 233 233 233 234
18	2.43 2.44	Introduction       Introduction         Bahuvrīhis in ādi       Introduction         2.44.1       Compounds in -ādi         2.44.2       Series of compounds         2.44.3       ityādi         Suffix vat       Ityādi	233 233 233 233 234 235
18	2.43 2.44	Introduction	233 233 233 234 235 236
18	2.43 2.44	Introduction	233 233 233 234 235 236 236 236
18	2.43 2.44 2.45	Introduction	233 233 233 234 235 236 236 236 236
	<ul><li>2.43</li><li>2.44</li><li>2.45</li><li>2.46</li></ul>	Introduction	233 233 234 235 236 236 236 236 236 237 238
	<ul><li>2.43</li><li>2.44</li><li>2.45</li><li>2.46</li><li>Term</li></ul>	Introduction       Introduction         Bahuvrīhis in ādi       Bahuvrīhis in ādi         2.44.1       Compounds in -ādi         2.44.2       Series of compounds         2.44.3       ityādi         2.44.3       ityādi         2.44.3       ityādi         2.45.1       Adverbs in -vat         2.45.2       With standard of comparison         2.45.3       itivat, ityādivat         iva and yathā       2	233 233 233 233 234 235 236 236 236 236 236 237 238 239
	<ul> <li>2.43</li> <li>2.44</li> <li>2.45</li> <li>2.46</li> <li>Term</li> <li>2.47</li> </ul>	Introduction       Introduction       Introduction         Bahuvrīhis in ādi       Introduction       Introduction         2.44.1       Compounds in -ādi       Introduction         2.44.2       Series of compounds       Introduction         2.44.3       ityādi       Introduction         2.44.3       ityādi       Introduction	233 233 233 234 235 236 236 236 236 236 236 236 237 238 239 239
	<ul> <li>2.43</li> <li>2.44</li> <li>2.45</li> <li>2.46</li> <li>Term</li> <li>2.47</li> </ul>	Introduction       7         Bahuvrīhis in ādi       7         2.44.1       Compounds in -ādi         2.44.2       Series of compounds         2.44.3       ityādi         2.44.3       ityādi         2.44.3       ityādi         2.45.1       Adverbs in -vat         2.45.2       With standard of comparison         2.45.3       itivat, ityādivat         2.45.4       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2         1.1       2 </th <th>233 233 233 234 235 236 236 236 236 236 237 238 <b>239</b> 239 239</th>	233 233 233 234 235 236 236 236 236 236 237 238 <b>239</b> 239 239
	<ul> <li>2.43</li> <li>2.44</li> <li>2.45</li> <li>2.46</li> <li>Term</li> <li>2.47</li> </ul>	Introduction       3         Bahuvrīhis in ādi       3         2.44.1       Compounds in -ādi         2.44.2       Series of compounds         2.44.3       ityādi         2.44.3       ityādi         2.44.3       ityādi         2.45.1       Adverbs in -vat         2.45.2       With standard of comparison         2.45.3       itivat, ityādivat         2.45.3       itivat, ityādivat         iva and yathā       2         Introduction       2         Dialectic style       2	233 233 233 233 234 235 236 236 236 236 236 236 236 237 238 239 239 239 239

xvii

		2.48.4	Example of dialectic style	
	2.49	Objecti	on	,
		2.49.1	<i>nanu</i> and <i>atha</i>	;
		2.49.2	Interrogatives	1
		2.49.3	iti cet	
		2.49.4	ity āśankya	
		2.49.5	prāpta and āksipta 245	Ì
		2.49.6	Verbs of speaking	ŕ
		2.49.7	Clauses beginning with <i>na ca</i> 246	)
	2.50	Reply		)
		2.50.1	Verbs of speaking	j.
		2.50.2	tan na and other denials 247	
		2.50.3	prasanga and āpatti 248	i
		2.50.4	doșa	
		2.50.5	<i>syāt</i> and <i>bhavet</i>	
		2.50.6	yady api and other concessives	
		2.50.7	tarhi	
	2.51	Alterna	tive interpretation	
		0 51 1	0 11	
		2.51.1	Second interpretations	
			Citing others	
20	Dowt	2.51.2	Citing others	
20		2.51.2 icles and	Citing others	
20	2.52	2.51.2 icles and Introdu	Citing others	•
20	2.52	2.51.2 icles and Introdu Particle	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253	
20	2.52	2.51.2 icles and Introdu Particle 2.53.1	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253         atha       253	
20	2.52	2.51.2 icles and Introdu Particle 2.53.1 2.53.2	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253         atha       253         tatra       253	
20	2.52	2.51.2 icles and Introdu Particle 2.53.1 2.53.2 2.53.3	Citing others       251         d Pronouns       253         action       253         ass connecting sentences       253         atha       253         tatra       254         tāvat       254	
20	2.52	2.51.2 icles and Introdu Particle 2.53.1 2.53.2 2.53.3 2.53.4	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253         atha       253         tatra       253         tāvat       255         atra and iha       255	
20	2.52	2.51.2 icles and Particle 2.53.1 2.53.2 2.53.3 2.53.4 2.53.5	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253         atha       253         tatra       253         tāvat       254         tāvat       255         atra and iha       255         kim ca, etc.       256	
20	2.52	2.51.2 icles and Particle 2.53.1 2.53.2 2.53.3 2.53.4 2.53.5 2.53.6	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253         atha       253         tatra       253         tatra       253         tatra       254         tāvat       255         atra and iha       255         kim ca, etc.       256         tu, etc.       257	
20	2.52	2.51.2 icles and Particle 2.53.1 2.53.2 2.53.3 2.53.4 2.53.5 2.53.6 2.53.7	Citing others       251         d Pronouns       253         ction       253         atha       253         atha       253         tatra       253         tatra       253         tāvat       253         tāvat       255         atra and iha       255         kim ca, etc.       256         tu, etc.       257         tathā hi and tad yathā       257	
20	2.52 2.53	2.51.2 icles and Particle 2.53.1 2.53.2 2.53.3 2.53.4 2.53.5 2.53.6 2.53.7 2.53.8	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253 $atha$ 253 $tatra$ 255 $atra$ and $iha$ 255 $kim$ ca, etc.       256 $tu$ , etc.       257 $tatha$ hi and tad yathā       257         Words for "therefore"       258	
20	2.52 2.53	2.51.2 icles and Particle 2.53.1 2.53.2 2.53.3 2.53.4 2.53.5 2.53.6 2.53.7 2.53.8 Particle	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253         atha       253         tatra       253         tatra       254         tāvat       255         atra and iha       255         kim ca, etc.       256         tu, etc.       257         tathā hi and tad yathā       257         Words for "therefore"       258	
20	2.52 2.53	2.51.2 icles and Particle 2.53.1 2.53.2 2.53.3 2.53.4 2.53.5 2.53.6 2.53.7 2.53.8 Particle 2.54.1	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253 $atha$ 253 $atha$ 253 $tatra$ 253 $tatra$ 253 $tatra$ 253 $tatra$ 253 $tatra$ 254 $t\bar{a}vat$ 255 $atra$ and $iha$ 255 $kim$ ca, etc.       256 $tu$ , etc.       257 $tath\bar{a}$ hi and tad yath $\bar{a}$ 257         Words for "therefore"       258 $eva$ 258	
20	2.52 2.53 2.54	2.51.2 icles and Particle 2.53.1 2.53.2 2.53.3 2.53.4 2.53.5 2.53.6 2.53.7 2.53.8 Particle 2.54.1 2.54.2	Citing others       251         d Pronouns       253         ction       253         es connecting sentences       253 $atha$ 253 $tatra$ 254 $t\bar{a}vat$ 255 $atra$ and $iha$ 255 $kim$ ca, etc.       256 $tu$ , etc.       257 $tath\bar{a}$ hi and $tad$ yath $\bar{a}$ 257         Words for "therefore"       258 $eva$ 258 $api$ 259	
20	2.52 2.53 2.54	2.51.2 icles and Particle 2.53.1 2.53.2 2.53.3 2.53.4 2.53.5 2.53.6 2.53.7 2.53.8 Particle 2.54.1 2.54.2 Pronom	Citing others       251         d Pronouns       253         ction       253         atha       253         atha       253         atha       253         tatra       254         tāvat       255         atra and iha       255         kim ca, etc.       256         tu, etc.       257         tathā hi and tad yathā       257         Words for "therefore"       258         eva       258         api       259	

2.55.2 Adverbs as attributives
2.56 Correlative pronouns and adverbs
2.56.1 Repeated correlative
2.56.2 Multiple pairs
2.56.3 yath $\bar{a}$ with tath $\bar{a}$
2.56.4 yat meaning "that"
2.56.5 Repeated anya
2.57 Demonstrative pronouns
2.57.1 Scale of proximity
2.57.2 Reference to earlier and later passages
2.57.3 tad etat and tad idam
Appendix: Useful Reference Works 267
Indices 271
Commentarial Passages Quoted
Grammatical Citations
Sanskrit Index
English Index

1

•

. .

## **Series Editor's Preface**

It was only after this book was accepted for publication in the *Treasury* of the Indic Sciences series that I was called upon to become editor of the same series, and thus eventually found myself in the rather odd position of welcoming my own work into the series. I am grateful for this opportunity for two reasons. First, it affords me a place, separate from the preface and introduction that Emery Boose and I wrote together, to describe subsequent developments in the handbook that have involved only my portion of the work and not his. Second, it allows me to make a few comments, at a point near the beginning of this new series, on my hopes for the series and their connection with the subject matter of the present volume.

This is the first printed edition of a handbook for students on the reading of Sanskrit commentaries that has been circulated privately in five previous versions over a period reaching back twenty-five years. The original version was a typewritten draft containing two sections corresponding to the two parts of the present edition. The first section was mine, and dealt with commentarial techniques for explaining the meanings of words in a text, and thus provided a survey of methods of glossing. The second section, written by Emery Boose, covered the distinctive features of the type of expository prose used in discussing the contents of the text, and was thus concerned with what Renou and others have referred to as the *bhāṣya* style of Sanskrit.

Subsequent versions of the handbook have largely been the result of my own attempts to find the most useful way to deal with a complex topic treated only sporadically in the original version—the use in commentaries of the technical devices of traditional Pāṇinian grammar, a topic that is difficult to introduce concisely and simply, and impossible to treat comprehensively in any portion of a single volume. Throughout the history of these attempts, Emery Boose's portion of the handbook has remained essentially the same, with the exception of progressive changes in visual formatting and the correction of minor typographical errors. Neither I nor any of the many readers of the handbook I have heard from has discovered any way in which his contribution could be improved. I have also avoided making any changes to the preface and introduction that we wrote together, other than those required by the alterations in my own material that I shall now briefly describe.

In the second version of the handbook, the material on Paninian grammar was extracted to form the core of a separate part of the work, and at the same time the other parts of work were typed into computer files and reformatted. My treatment of this Pāninian material became progressively more unwieldy, and once it had been circulated among colleagues in the third version of the handbook, I realized that this material was too difficult and extensive to be handled with the same comprehensiveness as in the part of the handbook dealing with glosses in non-technical language, and that the same range of technical information was beginning to be covered in recent work by George Cardona and others. In the fourth version, the other portions of the handbook were reformatted separately from the material on Panini, which was then withdrawn. In the fifth version, the first part was rewritten in view of the new structure, both parts were completely reformatted using the LATEX macro system with automated cross references and indexing, and a shortened overview of Paninian technical terminology was added to the first part to compensate for the withdrawal of a separate section on this topic. The present edition is an improved reformatting of this fifth version, reshaped to fit the series in which it appears.

Along the way this handbook has been read by many students and colleagues, and in recent years here at Columbia University I have used it occasionally in courses with a focus on the study of commentarial techniques, in which we have read from samples of commentaries on poetic and philosophical works that I have annotated with cross-references to the sections of the handbook. I am grateful to all those students who have offered comments on the handbook, and I would especially like to thank Rupa Viswanath for her detailed comments and suggestions.

It was only after coming to Columbia that I was exposed to the extensive role played by Sanskrit commentarial techniques in Buddhist texts. For this I have Professor Robert A.F. Thurman to thank. He encouraged me and others to be attentive to the interplay between Brahmanical and Buddhist intellectual activities, and it was in the setting of joint meetings of classes of advanced Sanskrit and advanced Tibetan, in which we read the same texts in both languages, that the interconnections became most clear to me. In reading with our students a commentary by Candrakīrti, I realized that some of Candrakīrti's points were often lost on Tibetan-oriented readers who were not versed in the technicalities of Sanskrit grammatical explication, and at the same time that Candrakīrti had expounded detailed principles of exegesis that had remained largely unknown to readers who specialized in non-Buddhist texts.

I was therefore pleased to learn of the initiation of the *Treasury of the Indic Sciences* series, which is designed to publish works of non-Buddhist origin that are relevant to the history of Buddhism in India and beyond, and I was more particularly delighted by Professor Thurman's invitation to include this handbook on Sanskrit commentaries in the series. I am also grateful for the concurrent inclusion in the same series of the catalogue of Sanskrit manuscripts at Columbia University compiled by David Pingree. Both works are perhaps atypical of the sort of text that will constitute the bulk of the series, but both are tools for pursuing its aim of exposing the intellectual interplay between Buddhist and non-Buddhist activity within a shared intellectual world.

The opening volume in the series presented a Brahmanical work on logic, translated by Stephen H. Phillips and N.S. Ramanuja Tatachárya. The vitality of this sort of text as a participant in cross-cultural conversations was made clear once again last year in a conference on Mind and Reality held here at Columbia, in which Professor Phillips engaged in stimulating dialogue with experts on Buddhist logic. I am confident that subsequent volumes in this series will make available texts of similar interest and usefulness.

In addition to my gratitude to Professor Thurman for all of his many acts of inspiration and support, I feel a deep indebtedness to the technical editor of the series, Thomas Yarnall. His knowledge of the interacting systems of thought I have mentioned is matched by his expertise in all the departments of publishing technology, and enriched by his generosity and patience.

My longest-standing debt of gratitude is to my old friend Emery Boose. His contribution to this handbook speaks for itself in its insight and rigor,

xxii

and without him none of this would ever have been written. I hope that the publication, at long last, of this handbook might serve in some small way to begin to repay his many kindnesses.

In this edition, chapters are numbered sequentially throughout the book, but the numbering of sections is separate for each part. Cross-references are by section or subsection, and the numbers referring to these begin with the number of the part in which they appear, so that it will be immediately apparent whether the topic in question is being treated in terms of glossing or of argument.

I have typeset this book using open-source LATEX computer typesetting software running on open-source Debian GNU/Linux operating systems.

Gary A. Tubb Series Editor, Treasury of the Indic Sciences New York November 2006

## **Authors' Preface**

This handbook is intended to give students in the early years of their study of Sanskrit some help in using Sanskrit commentaries and in reading the *bhāsya* style of Sanskrit prose, which is employed not only in commentaries but in many other works as well. We try to bring together simple explanations of the things that often make it difficult for students to begin to deal with scholastic works in Sanskrit, and of the things that are most readily useful once they are understood. We hope that the handbook will serve both as a reference work—by using the index or table of contents it should be possible to find explanations of many puzzling problems as they arise—and as a general introduction to the techniques used in commentaries and in the *bhāsya* style in general.

6

In identifying the difficulties that scholastic styles of Sanskrit present we have tried to follow our own observations of the obstacles most often encountered by students—observations that include our own memories of the problems we grappled with as students. We were fortunate in having teachers whose explanations made accessible to us much that had seemed impenetrable. What has prompted us to attempt the writing of this handbook is the fact that so many things that turned out to be so simple and useful did have to be explained to us, unless we had managed to figure them out on our own in the process of struggling with the texts: in printed sources much of the information we needed was buried, scattered, or nonexistent.

Most of what we present here is not our own discovery. In particular we have made frequent and grateful use of information gleaned from remarks made in class by Professor Daniel H. H. Ingalls and from the works of Speijer and Coulson mentioned in the Appendix.

We are grateful to Professor Ingalls for reading the first drafts of Part

One and for making many detailed and helpful suggestions.

Part One of the handbook, written by Gary A. Tubb, deals with the devices used in Sanskrit commentaries to explain the words of the text being commented on and their meanings; it covers fully the more important techniques of glossing in non-technical language used by the commentators, and gives a brief introduction to the rudiments of analysis in the technical terminology of Pāṇinian grammar as employed in commentaries. Part Two, written by Emery R. Boose, deals with the devices used in commentaries to discuss the ideas contained in the text being commented on, and provides an introduction to the *bhāṣya* style and to the techniques of Sanskrit expository prose in general, whether employed in a commentary or in an independent work. In each part there are occasional references to sections in the other parts, and both parts are covered simultaneously by the indices at the end of the handbook, but neither of the parts presupposes a knowledge of the material covered in the other.

The examples used in the handbook are drawn as far as possible from works that are easily accessible and that students of Sanskrit are likely to read. In Part One they come chiefly from well-known commentaries on the great *mahākāvya* poems, especially from the commentaries of Mallinātha, and occasionally an example is borrowed from the works of the grammarians. In Part Two the examples come from texts of the Śāṅkara school of Vedānta, especially from Śaṅkara's own *Brahmasūtrabhāṣya*, and in a few instances constructed examples are given for the sake of simplicity. Although Part One thus relies primarily on literary works for its examples, and Part Two on philosophical works for its examples, the points illustrated in each part are applicable to most commentarial or expository works on any subject.

The members of Sanskrit compounds are separated by hyphens as far as possible, and a circumflex accent is used to mark a long vowel or diphthong produced by sandhi between words or members of a compound. Square brackets indicate words not present in the Sanskrit text that must be supplied from context. In Part One, words quoted from the text being commented on in an example are in bold type. Also in Part One, indicatory phonemes (*anubandhas*) in technical elements of Pāṇinian grammar are printed as uppercase letters.

Ellipsis in a passage cited in Sanskrit is marked by three dots, and is indicated only when it occurs within the words actually quoted. Words

preceding or following the quoted words in the same original sentence are often omitted without notice when the quoted words yield the same meaning that they have in the full sentence.

In the translations of the examples, parentheses indicate words that are necessary to give a clear translation in normal English and that are implied in the Sanskrit text but do not correspond directly to words actually present in the Sanskrit text. Square brackets indicate words corresponding to terms that in the Sanskrit text are not expressly present but that must be understood from the context; the English words in square brackets are often translations of Sanskrit words supplied in square brackets in the text. In Part Two a very literal translation is sometimes followed by a colon and a more natural translation, and alternative translations are separated by a semicolon.

Reference is often made to the works of Coulson, Speijer, and Whitney where these authors have additional discussion or examples for the grammatical point in question. References to Coulson are by page number; references to Speijer and Whitney are by paragraph number. See the Appendix for bibliographic information on these and other works that may be useful in learning to read scholastic styles of Sanskrit.

## Abbreviations

**Coulson** Michael Coulson, Sanskrit: An Introduction to the Classical Language (Teach Yourself Books)

 $\mathbb{C}_{\mathbb{C}}$ 

GBh Bhagavadgītābhāsya of Śańkara (Śrīvāņivilāsa)

Kāśikā Kāśikā Vrtti on Pāņini's Astādhyāyī

Kir. Kirātārjunīya of Bhāravi (Nirņaya-Sāgara Press)

**Kir.M.** Mallinātha's commentary on the *Kirātārjunīya* (Nirņaya-Sāgara Press)

Kum. Kumārasambhava of Kalidāsa (Nirnaya-Sāgara Press)

- Kum.C. Cāritravardhana's commentary on the Kumārasambhava (Gujarati Printing Press)
- Kum.M. Mallinātha's commentary on the *Kumārasambhava* (Nirņaya-Sāgara Press)
- Kum.V. Vallabhadeva's commentary on the *Kumārasambhava* (ed. M. S. Narayana Murti)

Megh. Meghadūta of Kālidāsa (ed. M. R. Kale).

Megh.M. Mallinātha's commentary on the Meghadūta (ed. M. R. Kale).

Megh.V. Vallabhadeva's commentary on the *Meghadūta* (ed. E. Hultzsch).

Pā. Pāņini's Astādhyāyī

xxx

- **PP** *Pañcapādikā* of Padmapāda (Madras Government Oriental Series 155)
- Ragh. Raghuvamśa of Kalidāsa (Nirņaya-Sāgara Press)
- **Ragh.H.** Hemādri's commentary on the *Raghuvaņśa* (ed. Rewā Prasāda Dwivedī)
- **Ragh.M.** Mallinātha's commentary on the *Raghuvaṃśa* (Nirṇaya-Sāgara Press)
- SBh Brahmasūtrabhāsya of Śankara (Svāmi Satyānanda Sarasvatī)
- SDS Sarvadarśanasangraha of Sāyaņamādhava (Bhandarkar Oriental Research Institute)
- Śiśu. Śiśupālavadha of Māgha (Kashi Sanskrit Series 69)
- **Śiśu.M.** Mallinātha's commentary on the *Śiśupālavadha* (Kashi Sanskrit Series 69)
- Śiśu.V. Vallabhadeva's commentary on the Śiśupālavadha of Māgha (Kashi Sanskrit Series 69)
- Speijer J. S. Speijer, Sanskrit Syntax (Rpt. Motilal Banarsidass)
- Whitney William Dwight Whitney, Sanskrit Grammar (Harvard University Press)

## Introduction

### The Sanskrit Commentary

## 0.1 Importance of commentaries

The Sanskrit tradition offers to each student of its texts a rich source both of potential assistance and of potential difficulties in the form of an extraordinary abundance of works of commentary. The importance that these commentaries have for students is due in general to two characteristics—the simple fact of their abundance, and the specific ways in which they offer assistance in the reading of a Sanskrit text.

5

Works of commentary pervade the history of Sanskrit thought to a degree that is unparalleled in the writings of most other traditions: it is no exaggeration to say that of all the expository works available in Sanskrit most are, at least in external form, commentaries. There are several reasons for this prevalence of commentaries in Sanskrit, and some of therm are tied to features that are peculiar to the Sanskrit tradition.

One striking feature is the frequency with which we find works in Sanskrit that seem to require, or even to presuppose, the eventual services of a commentator. In works of the *sāstra* type, this situation is an ancient one. The old *sūtras* standing at the head of several of the philosophical schools are essentially signposts in a line of oral argument (which may in itself be a form of commentary on an older text, as in the exegesis of the Upanişads underlying the composition of the *Vedāntasūtras*), and in the absence of that oral corpus the *sūtras* often have meaning for us only as they are expounded in a full scholastic commentary. Throughout the history of Sanskrit thought a similar situation is produced by authors who give a bare outline of their material in short *sūtras* or *kārikās* that are then fleshed out

1

by the author's own commentary; the *Tarkasangraha* of Annambhatta, for example, is little more than a string of lists without the explanations supplied in Annambhatta's own *Tarkadīpikā* or in the similar works by other scholars.

In these laconic works the underlying motive is clearly a desire to facilitate memorization of the chief points of a system. Another frequent source of works requiring commentary is the deliberate practice of composing works of great difficulty, a practice that is explicitly and proudly acknowledged in the poet Bhatti's famous boast, that his mahākāvya could be understood only with the aid of a commentary ( $vy\bar{a}khy\bar{a}$ -gamyam idam  $k\bar{a}vyam$ , etc., Bhattikāvya 22.34). Bhatti's verse was taken up by Bhāmaha in his work on poetics and twisted into the petulant observation that such poems are more śāstra than kāvya (Kāvyālańkāra 2.20), but we should also remember that Bhāmaha himself, when describing the characteristics of an ideal mahākāvya, does not say that the poem should require no commentary, but rather that it should not require too much commentary (nâtivyākhyeyam, Kāvyālańkāra 1.20).

Yet poems far less recondite than Bhatți's have attracted many commentators (witness the enormous number of available commentaries on Kālidāsa's mahākāvyas), and for this the reasons are obviously pedagogical. Indeed, the fact that we have old commentaries on these mahākāvyas, and only much more recent ones on great plays of equivalent age and difficulty, may be presumed to indicate that the mahākāvyas made up the standard curriculum for Indian students of Sanskrit kāvya.

Another striking feature of the Sanskrit tradition is the frequency with which works that may as well have been independent treatises are cast into the external form of a commentary on an earlier text. In this way many treatises of great originality have been made to depend, at least nominally, on earlier works that they leave far behind—one thinks of Praśastapāda's *Padārthasangraha*, which is ostensibly a commentary on the *Vaiśeṣikasūtra*, and which in turn provides the occasion for Udayana's equally original *Kiraṇāvalī*. And the *Kiraṇāvalī* stands at the head of a very long string of commentaries upon commentaries, with varying degrees of dependence on the work being commented upon in each instance. In many such works the motives for writing in the form of a commentary go beyond the aim of providing exegesis, and include the desire to associate oneself with an established authority—to present one's views as a worthy unfolding of timehonored tradition.

In many other works in which the commentarial format is freely chosen the underlying concern is, once again, pedagogical. We often find a scholar breaking a single body of material into several parts of increasing difficulty, which are then presented as a string of commentaries, one upon another. An example of this is Hemacandra's work on poetics, the  $K\bar{a}vy\bar{a}nus\bar{a}sana$ , which appears in the form of a number of brief  $s\bar{u}tras$ supporting his own commentary, the Alankāracūdāmani, which in turn is expanded upon in another commentary by the same author, the Viveka. The material in the Viveka is of an advanced nature and could easily have been presented in the form of an independent treatise, but by shaping it into the form of a commentary on his more elementary work Hemacandra allows the students in his advanced class to continue their studies using a framework with which they are already familiar.

In some ways a work like Hemacandra's is equivalent to one of our modern books in which extensive footnotes and appendices are provided, but there is an important practical difference between the two formats: in a modern work the additional material is presented in a number of separate notes whose connection with the main text is merely signaled by typographical symbols or references, but in a Sanskrit work the additional material is usually presented in a more continuous composition that relies on a large body of conventional terminology and special syntactical devices in fulfilling its role as a commentary on the original text. As a result, an understanding of the special techniques of the Sanskrit commentarial style is necessary in the reading of such works.

## 0.2 The five services of a commentary

In identifying the useful services offered by Sanskrit commentaries, we have been guided by the list of the five characteristics of commentaries given in a verse from the *Parāśarapurāna* that is quoted in the *Nyāyakośa*:

pada-cchedaḥ padârthôktir vigraho vākya-yojanā ākṣepeṣu samādhānaṃ vyākhyānaṃ pañca-lakṣaṇam

These five services correspond, in this same order, to the problems most often faced by students attempting to decipher a Sanskrit text:

#### 0.2.1 Padaccheda

Word-division, separation of the words of the text: the rules of sandhi and the peculiarities of the Devanāgarī script make this the first problem that confronts a student attempting to translate a text. Before one can begin to look up the words one must know where one word ends and the next begins. Any commentary that is not too long-winded can immediately solve this problem for even a beginning student, yet many students do not know enough about the basic nature of a commentary to take advantage of this simple service.

#### 0.2.2 Padārthokti

Stating the meaning of the words, paraphrasing: once it has been determined what the words of the text are, the next problem is to find the appropriate meaning in the dictionary. Because most Sanskrit words have so many possible meanings, the most frustrating part of the study of Sanskrit for many students in their early years is the necessity of laboriously writing down or holding in mind several definitions for each word until it can be determined which possible meaning is most appropriate for each word in the present context. While it is true that the glosses offered by commentators may often be as unfamiliar to a student as the word being glossed, even a beginning student who is confronted, for example, with the word bhāskarah in a Sanskrit text may find a simple gloss such as bhāskarah sūryah accessible enough to save some of the time spent flipping through the dictionaries. On a higher level, of course, these glosses often provide information that is difficult or impossible to find in the standard reference works, or that at least is more precise in its appropriateness to the text being commented on.

#### 0.2.3 Vigraha

Analysis of grammatical complexes (i.e., of nominal compounds and of derived stems): here the dictionaries are often of no use, especially where nominal compounds are involved. Time spent in learning to understand the basic formulas of analysis used by commentators is useful not only because it makes it possible to take advantage of this service of commentaries, but

also because it is a good way to acquire a grasp of the different types of complex formations used in Sanskrit.

#### 0.2.4 Vākyayojanā

Construing the sentences, indicating the construction of the text: once again, a student can avoid the tedious procedure of trying to figure out the text in several stages, if only the basic principles by which commentators repeat the words of the text being commented on are understood. These principles are really fairly simple, but until they have been explained to a student most commentaries appear to be masses of unconnected explanations.

#### 0.2.5 Āksepasamādhāna

The answering of objections: this part of a commentary is concerned with the ideas contained in the original text rather than with its words. Accordingly, it differs from the four preceding services in two ways. First, at this level a commentary goes beyond straightforward exegesis and becomes an argumentative treatise in its own right. Second, this portion of a commentary is based on the style of oral debate rather than on the style of oral instruction, so that it uses a different arsenal of vocabulary, syntax, and organization. An understanding of this style can be applied to the reading of any text written in expository Sanskrit, whether that text is a commentary or not.

The first four of these services form the subject of Part One of this handbook in their achievement through methods of glossing using non-technical language. Their achievement through the use of the technical elements of Pāninian grammar is touched on here in Sections 1.18-1.20, but a full treatment of the topic lies outside the scope of this handbook in its present form. The fifth service forms the subject of Part Two.

Chapter 1

## **The First Four Services**

## 1.1 Introduction

Of the five services that a full commentary is expected to provide, the first four are services of glossing and deal directly with the words of a text, while the fifth is a service of discussion and is concerned more with the ideas expressed in that text. Generally the first four services—the four principal elements in the glossing of the words of a text—consitute the bulk of a commentary on a literary text, but may appear only intermittently in a commentary on a philosophical or scientific text. In commentaries of any type, however, the basic techniques of glossing are the same. In Sanskrit commentaries these four services are achieved partly through the use of the technical elements of Pāṇinian grammar, and partly through methods of glossing using non-technical language. Part One of this handbook deals with the use of non-technical language to fulfil the first four services:

÷;

- 1. *padaccheda*—division of words, i.e., the resolution of external sandhi (Sections 1.3–1.7).
- 2. *padārthokti*—statement of the meaning of the words, i.e., the giving of synonyms, paraphrases, definitions, and examples (Sections 1.8–1.17).
- 3. *vigraha*—analysis of complex formations, i.e., breaking them down into their constituent elements (Sections 1.18–1.75).
- 4. *vākyayojanā*—construing the sentences, i.e., indicating how the words fit together syntactically (Sections 1.76–1.84).

A full treatment of the achievement of these same services through the use of the technical elements of Pāņinian grammar lies outside the scope of this handbook.

The fifth of the services provided by commentaries, that involving discussion of the ideas contained in the text, will form the subject of Part Two.

## **1.2** Types of glossing

In the following sections, methods of glossing will be taken up one by one, but in an actual commentary they often work together in explaining a particular word. As preliminary examples of the types of glossing used by commentators and of how the explanation of those types is distributed in this handbook, here are the full remarks of the commentator Mallinātha on three words from the *Raghuvamśa* of Kālidāsa.

In the first example, Mallinātha is glossing the word *saisave*, "in in-fancy":

Ragh.M. 1.8 sisor bhāvah saisavam bālyam. "prānabhrj-jāti-vayovacanôdgātr-ādibhyo 'ñ" ity añ-pratyayah. "sisutvam saisavam bālyam" ity amarah. tasmin vayasi [saisave]. Infancy is the condition of being an infant, childhood. The suffix  $a\tilde{N}$  (i.e., -a, with vrddhi strengthening of the first syllable of the stem to which it is added), by (sūtra 5.1.129 in Pāṇini's grammar) " $a\tilde{N}$  is added to words for species of animals and for the ages of life, and to udgātr, etc. (in the senses of the condition of being that or the activity of that)." "Infanthood, infancy, childhood"—Amara. In that age of life (viz., **in infancy**).

Here *sisor bhāvah* is an analysis in ordinary language of the formation *saisavam*, an example of secondary nominal derivation (Section 1.37.2). This is followed by a simple synonym, *bālyam* (Section 1.9). The next sentence gives the *taddhita* suffix by which the word is formed according to the Pāṇinian system of grammar, and quotes the specific rule by which its formation is justified. The next sentence cites the treatment of the word in the dictionary of Amara (Section 1.16.2). The final sentence, after these explanations in which the word has been treated in the nominative case, in effect gives it the locative case ending that it has in the original text

(Section 1.21). This last step is necessary so that the word will fit properly in the syntactical arrangement of the entire verse (Section 1.78).

In the examples given in this handbook, the words repeated from a text being commented on are emphasized in bold type. In this example, the word being explained never actually appears in the commentary in the same form that it has in the original text, and we have therefore supplied it in square brackets at the point at which the words of the commentary serve to reconstitute it.

In the next example, Mallinātha explains the word *prajānām*, "of the people":

Ragh.M. 1.24 prajāyanta iti prajā janāh. "upasarge ca samjňāyām" iti da-pratyayah. "prajā syāt santatau jane" ity amarah. tāsām [prajānām]. They are generated, thus progeny, people. The suffix Da (i.e., -a, with loss of the final vowel and final consonant of the root to which it is added), by (sūtra 3.2.99 of Pāṇini) "And (Da is added to the root jan) when a verbal prefix is used, if a name is being formed." "prajā in the senses of offspring and people"—Amara. Of them (viz., of the people).

Here the complex formation being analyzed is an example of primary nominal derivation (Section 1.29.1), and thus involves a krt suffix, but otherwise the elements of Mallinātha's remarks are the same as in the preceding example.

In the final example Mallinātha chooses to end his remarks by repeating in its original case the word being explained, rather than simply indicating that case by the use of a pronoun:

Ragh.M. 1.54 dhuram vahantîti dhuryā yugyāh. "dhuro yad-dhakau" iti yat-pratyayah. "dhūr-vahe dhurya-dhaureya-dhurīnāh sa-dhuramdharāh" ity amarah. **dhuryān** rathâśvān viśrāmaya vinīta-śramān kuru. They bear a yoke, thus yoke-bearers, draft animals. The suffix yaT (i.e., -ya, with acute pitch accent on the preceding syllable), by (sūtra 4.4.77 in Pāṇini) "yaT and dhaK are added to the root dhur (in the sense of 'bearing that')." "In the meaning of a yoke-bearer: dhurya, dhaureya, dhurin, and dhuramdhara"—Amara. **Rest** make quiet and rested **the yoke-bearers** the chariot horses.

#### Chapter 1. The First Four Services

Of the four services of commentaries listed in the preceding section, the first service, division of the words of the original text, is accomplished in each of these examples simply by isolating the word in order to explain it. The second service, that of explaining the meaning of the individual word, is accomplished partly by giving synonyms, partly by quoting dictionaries, and partly by performing the third service, that of analyzing complex formations, since each of the three words being explained in the comments quoted here happens to be an example of such a formation. The fourth service, that of indicating the syntactical structure of the original text, is generally achieved through the use of methods that can be glimpsed in the ways in which the commentator has returned each word to the required grammatical case at the end of each of the examples.

It is in the analysis of complex formations that the role of Pāṇinian grammar becomes most visible. The formulas used to give the meanings of these items—*sisor bhāvah saisavam, prajāyanta iti prajāh,* etc.—are based on the treatment of such formations in Pāṇini's grammar, but the formulas themselves are in ordinary language and can understood without a detailed knowledge of the technical elements of Pāṇinian grammar. The same is true of much of the terminology used by the grammarians to name important classes of complex formations. The artificial names for elements such as suffixes, however, together with the meanings of the indicatory letters added to them, require special explanation; a brief overview of this topic is given in Sections 1.18—1.20.

Chapter 2

## **Division of Words**

## 1.3 Introduction

In Sanskrit sentences the constant application of rules for the euphonic combination of words often makes it difficult to determine where one word ends and the next begins. The resolution of sandhi between words is the chief purpose of the oldest surviving exegetical works in India; to each of the collections of Vedic hymns there was added in ancient times a padapātha ("word-text") version, in which each word-isolate is exhibited separately, with terminal sandhi. This straightforward approach to the separation of words can still be found in the padaccheda feature of many of the modern editions of Sanskrit texts printed as ponies for Indian students. In most Sanskrit commentaries, however, the indication of boundaries between words is not given a distinct section reserved for that purpose, but rather is achieved naturally through three general practices: the repetition of the words with intervening explanatory material (Section 1.4), the occasional specification of word-division in doubtful instances (Section 1.5), and the citation of individual words for the purpose of making special remarks (Section 1.6). In the course of displaying the words of the text being commented on, a commentator may also mention variant readings (Section 1.7).

5<u>7</u>.

## **1.4** Repetition of the words of the text

Ordinarily a continuous commentary repeats each word found in the text. For examples of how this is done and for a description of common exceptions to this general rule, see Sections 1.76–1.80. Because each word repeated from the text is usually followed by a gloss or by other inserted material, separation of the words of the text is automatically provided in the course of presenting the words for explication. The value of the repeated words in providing this service is enhanced if they are printed in bold type; if they are not, they may be sought out and underlined (Section 1.79).

### 1.5 Specification of word-division

For purposes of word-division, the repetition of words from the text involves ambiguity only when two adjacent items in the text are repeated next to each other in the commentary without the insertion of intervening explanatory material. Because it may not be clear whether or not such items form a compound word, the commentator will sometimes explicitly state whether they are two words or one, or will specify where the division should be made.

Ragh.M. 1.17 **ā manoh**, manum ārabhyêty abhividhih. pada-dvayam câitat, samāsasya vibhāşitatvāt. Since Manu—beginning with Manu; this is an inclusive limit. And these are two words, because the compound is optional.

Kum.M. 5.55 tadêti cchedah. tadā prabhṛti. tata ārabhyêti arthaḥ. Divide after tadā. From then on, i.e., beginning then.

Kum. V. 5.63 tvayā yathā śrutam tathâiva tat. ... yathā śrutam iti dve pade. It is just as you have heard. yathā śrutam—these are two words.

## 1.6 Citation of individual words

In the course of giving a running gloss, the words repeated from the text are generally presented in the same form in which they appear in the text .....

(Section 1.9). Often a commentator will also single out a word for further remarks, and when this is done there are four principal methods of citing the word (i.e., of referring to it as a word, rather than simply using it in the course of a continuous gloss):

#### 1.6.1 iti

The word may be followed by *iti* (corresponding to our use of quotation marks):

Kum. V. 1.58 vikriyanta iti karmani. (The verb) vikriyante is passive.

Kum. V. 1.25 um iti pranaya-kope. (The particle) um is used in feigned anger.

#### 1.6.2 -sabda, -kāra

The word may be placed in compound with -*sabda* or (especially if it consists of a single syllable) with -*kāra*:

Ragh.M. 1.2 dvau kva-śabdau mahad antaram sūcayatah. The two words 'where' suggest a great difference.

Ragh.M. 1.1 pärvatī-paramêśvarau. parama-śabdah sarvôttamatva-dyotanârthah. Pārvatī and the Supreme Lord. The word 'supreme' has the purpose of revealing his preeminence above all others.

Kum.M. 1.4 kimcêti cakārârthah. The meaning of (the particle) ca is "furthermore."

### **1.6.3** Inflected indeclinable

If the word is indeclinable, it may be given an inflectional ending and be declined like a masculine noun. The presence of the case-ending will then be enough to indicate that the word is being referred to as a word rather than being used in its own meaning. Ambiguity exists only when the indeclinable ends in a long vowel and is presented in the nominative, but here

## 1.7. Variant readings

the context will usually make it clear that the word is being referred to as a word.

Ragh.M. 1.24 apih samuccaye. api (is used) in (the sense of) conjunction.

Ragh.M. 1.4 athavā pakṣântare. athavā (is used) in (the sense of) an alternative.

Kum.V. 2.42  $cah p \bar{u}rv \hat{a}pek saya \bar{a} samuccaye$ .  $h \hat{t} i p \bar{a} the c \hat{a} r the h h$ . (The particle) ca (is used) in (the sense of) conjunction with repect to what precedes. In the reading "hi," (the particle) hi (is used) in the meaning of (the particle) ca.

As the word *cârthe* in the last example shows, this device can also be used in compound, where the inflectional ending will, of course, be absent. This should not result in ambiguity as long as it is recognized that a compound is being used.

#### 1.6.4 Anubandhas

The word may be displayed along with the indicatory letters (anubandhas) used in the Pāņinian system of grammar:

Ragh.M. 1.90  $\bar{a}$  tapo-vanāt.  $\bar{a}n$  maryādāyām. As far as the penance-grove. (The preposition)  $\bar{a}$  [called  $\bar{a}n$  in Pāņini's grammar] (is used) in the sense of an exclusive limit.

Kum.M. 1.35 naño viśesanatvam ca-śabda-prayogād eva jñeyam. And that the word *na* [called *nañ* in Pānini's grammar] is a modifier can be known simply from the use of the word *ca*.

For an overview of the use of *anubandhas* and the technical citation of grammatical elements such as roots, seeSections 1.76–1.80.

## 1.7 Variant readings

Variant readings are referred to by using the noun *pātha* or any of the forms of the verb *pathati*.

Kum.V. 2.26 kṣatâhaṃ-kāra-śaṃsina iti jarat-pāțho 'tra ramyataraḥ. Here the old reading "kṣatâhaṃ-kāra-śaṃsinaḥ" iş more appealing.

Kum. V. 2.28 prārthayadhvam ity apapāţhaħ. "prārthayadhvam" is a bad reading.

Kum. V. 3.44 iyâdīnām chandasi vidhānād bhāşāyām triyambakam iti prayogo durlabha iti maheśvaram iti paṭhanīyam. Because (sandhi substitutes) such as "iy" are prescribed (only) for the Vedic language, the usage "triyambakam" is difficult to arrive at in the classical language, so that one ought to read "maheśvaram."

Kum. V. 4.36 marmârpitam iti pāthântaram. There is another reading, "mamârpitam."

Kum.M. 3.44 kecit sāhasikāh tri-locanam iti pețhuh. Certain reckless (commentators are said to have) read "tri-locanam."

## Chapter 3

## **Giving the Meaning of Words**

## **1.8 Introduction**

The basic task of giving the meaning of a word in the text being commented is carried out by placing an explanation of that meaning, at the simplest level often merely a synonym for it (Section 1.9) alongside the word in question. Ordinary this simple gloss follows the word being explained, but the location of the gloss may vary (Section 1.10). Commentators may also gloss words with equivalent forms other than simple synonyms (Section 1.11), and may expand the gloss in various ways (Section 1.12) or append more removed paraphrases (Section 1.13). Techniques are available for dealing with words that refer by extension to objects other than those directed denoted (Section 1.14) and with words that have double meanings (Section 1.15). These techniques of glossing may be supplemented by the use of separate definitions (Section 1.16). Further devices are used in glossing special items such as proper names, the names of species, particles, indeclinables, and pronouns (Section 1.17).

When the word being explained happens to be a complex formation arrived at through derivation, the formation may be analyzed using the formulas available for each of the types of derived stems (Sections 1.18 ff). Words may also be explained through the use of the technical elements of Pāninian grammar. 20

## **1.9** Simple synonyms

The words repeated from the text being commented on are most commonly glossed by the insertion of a synonym after each. Simple glosses of this sort may be provided for virtually every part of speech:

Ragh.M. 1.1 vande 'bhivādaye I honor I salute

Ragh.M. 1.2 matih prajñā mind intellect

Ragh.M. 1.4 me mama my of mine

Ragh.M. 1.29 nūnam dhruvam surely certainly

These inserted glosses are placed in grammatical apposition with the word being glossed. Thus a gloss of a verb will agree with it in person and number; a gloss of a substantive will agree with it in case and number, and a gloss of an adjective will agree with it in case, number, and gender:

 Ragh.M. 1.27
 vande 'bhivādaye
 I honor I salute

 Ragh.M. 1.2
 mohād ajñānāt
 out of folly out of ignorance

 Ragh.M. 1.36
 snigdo madhurah
 agreeable (masc.)—sweet

## 1.10 Location of glosses

Usually the gloss follows immediately after the word quoted from the text being commented on. There are two principal exceptions:

#### 1.10.1 Enclitics

Enclitics and other words closely dependent upon a preceding word must usually be separated from the words they are glossing.

Ragh.M. 1.28 auşadham yathā auşadham iva as if medicine like medicine [here iva glosses yathā]

Ragh.M. 1.24 tāsām pitaras tu janma-hetavo janma-mātra-kartārah kevalam utpādakā evâbhūvan. But their parents were causes of birth creaters merely of birth alone simply producers. [here eva glosses kevalam]

### 1.10.2 Vigrahas

When an analysis of a grammatical complex (such as a nominal compound) is given, the original word from the text is generally repeated after the analysis, rather than before it (see Section 1.21).

## 1.11 Glossing by equivalent forms

A simple gloss need not be an exact morphological copy of the original; often a different construction with equivalent meaning is used instead. Two types of glossing by equivalent forms are especially common:

## 1.11.1 Periphrasis

A periphrastic construction is used to bring out the precise meaning of a morphological feature in the original:

Ragh.M. 1.20 anumeyā anumātum yogyāh inferable able to be inferred

Ragh.M. 1.26 yajñāya yajñam kartum for sacrifice in order to perform sacrifice

Ragh.M. 1.72 mucye mukto bhavāmi I am freed I become freed

#### 1.11.2 Participle

A past active participle may be used to gloss a finite verb:

Ragh.M. 1.17 na vyatīyur nâtikrāntavatyah they did not trangress they did not go beyond

Ragh.M. 1.21 jugopa rakșitavān he guarded he protected

## 1.12 Expansions of glosses

A simple gloss is often expanded beyond the mere giving of a synonym by including illustrative or explanatory material.

#### 1.12.1 -ādi

Inserted examples are generally offered in the forms ending in  $-\bar{a}di$  ("such as," Section 2.44):

Ragh.M. 1.14 sūribhih kavibhir vālmīky-ādibhih by wise men by poets such as Vālmīki

Ragh.M. 1.16 **nrpa-guṇai** rāja-guṇais tejaḥ-pratāpâdibhiḥ kula-sīla-dākṣinyâdibhiś ca with the qualities of a king with royal qualities such as valor and prowess, and such as good family, good conduct, and politeness.

#### 1.12.2 Implications

Commentators very frequently fill out a gloss by explicitly stating what is merely implied in the original wording:

Ragh.M. 1.24 rakṣaṇād bhaya-hetubhyas trāṇāt because of protecting because of protection from sources of danger

Ragh.M. 1.24 bharanād anna-pānâdibhih poṣanāt because of supporting because of nourishing with food, drink, etc.

Ragh.M. 1.25 parinetur dārān parigrhņatah of the marrier of the wedder of a wife

#### 1.12.3 Implied contrasts

Similarly, commentators often spell out the implications of an expression by making implied contrasts explicit: Ragh.M. 1.7 yaśase kīrtaye... vijigīşūnām vijetum icchūnām, na tv artha-samgrahāya. prajāyai samtānāya grha-medhinām dāra-parigrahānām, na tu kāmôpabhogāya. Of those who were desirous of victory wishing to conquer for the sake of fame of glory, but not for the sake of amassing wealth. Of those who were householders married men for the sake of offspring of progeny, but not for the sake of the enjoyment of pleasure.

## 1.13 More removed paraphrases

When a commentator feels that glossing a word with a synonym or equivalent form is not sufficient, he may provide a freer paraphrase. Such paraphrases are usually followed by one of three phrases: (Section 1.13.1) *ity arthah*, (Section 1.13.2) *iti bhāvah*, or (Section 1.13.3) *iti yāvat*. For the first two of these phrases other expressions are sometimes substituted (Section 1.13.4), and additional formulas are used to mark comments serving to bring out the implications of the words being commented on (Section 1.13.5) and the purposes of the author's statements (Section 1.13.6).

#### 1.13.1 ity arthah

The phrase *ity arthah* (lit., "such is the meaning," "the meaning is that ...") is used to mark a paraphrase that simply restates the literal meaning of the text in another way. Such paraphrases are often added after a simple gloss:

Kum.M. 1.46 tato nirvisesam nirbhedam. tat-sadrsam ity arthah. Not distinguished from it—not different from it. In other words, like it.

Kum.M. 1.43 lolā capalā, paribhramaņa-sīlêty arthah Unsteady restless, i.e., tending to roam about

Just as frequently they are used instead of a simple gloss:

Kum.M. 1.11 mandām ... gatim na bhindanti na tyajanty ity arthah their slow pace they do not break—in other words, they do not abandon it Kum.M. 1.18 nāga-vadhûpabhogyam nāga-kanyā-pariņetāram ity arthah. who was to be enjoyed by a Nâga wife—i.e., who was to marry a Nāga bride.

## 1.13.2 iti bhāvah

The phrase *iti bhāvah* (lit. "such is the idea," "the idea is that ...") is used to mark comments that go a step further by stating the underlying idea intended by the author of the text being commented on.

Kum.M. 1.15 anvista-mrgair mārgita-mrgaih, śrāntair iti bhāvah. for deer who had been hunting deer. The idea is that they were tired.

Kum.M. 3.67 kimcit ... parilupta-dhairyah, na tu prākrta-janavad atyanta-lupta-dhairya iti bhāvah. With his calmness slightly ruffled. But not with his calmness exceedingly ruffled, as in common people—this is the idea.

Because expressions that are difficult enough to require a free paraphrase are also likely to require further explanation, it is quite common to find a remark ending with *ity arthah* being followed immediately by a comment ending with *iti bhāvah*:

Kum.M. 3.68 sācī-kṛtā ... tasthau. hriyā mukhaṃ sācī-kṛtya sthitêty arthah. na kevalaṃ harasyâiva devyā apy udito rati-bhāva iti bhāvaḥ. She stood turned aside, i.e., she stood with her face averted out of embarrassment. The idea is that the emotion of love had arisen not only in Śiva, but in the Goddess as well.

Kum.M. 5.86 kleśah phalena phala-siddhyā punar-navatām vidhatte. pūrvavad evâklistatām āpādayatīty arthah. sa-phalah kleśo na kleśa iti bhāvah. Toil, through its fruit—through the accomplishment of its fruit—restores freshness. In other words, it makes one unburdened by toil, just as one had been before. The idea is that fruitful toil is not toil.

Kum.M. 4.10 tava padavīm mārgam aham pratipatsye. tvām anugamisyāmîty arthah. ato me nâsti vicāra iti bhāvah. Your path way I shall resort to—i.e., I shall follow you. Hence you need not worry about me—this is the idea.

## 1.13.3 iti yāvat

The phrase *iti yāvat* (lit., "just this much [is meant]") is used after a paraphrase that expresses the meaning of the original text more precisely, either by substituting a more specific term for a wider one, or by reducing a picturesque expression to plain language. *iti yāvat* may thus be translated "specifically," "to be precise," "to put it plainly," etc., depending on the context.

Ragh.M. 1.25 manīsiņo vidusah. dosa-jňasyêti yāvat. Of a wise man of a learned man. Specifically, of one who can recognize faults.

Kum.M. 3.5 punar-bhavaḥ punar-utpattiḥ. saṃsāra iti yāvat. Reoccurring arising again. To be precise, transmigration.

Kum.M. 1.39 vedi-vilagna-madhyā vedivat kṛśa-madhyā. tanu-madhyêti yāvat. Altarground-narrow-waisted thin-waisted like an altarground. To put it simply, slender-waisted.

#### 1.13.4 Other phrases

Several other phrases are frequently used to mark the end of a free paraphrase. The most common of these are *ity abhiprāyaḥ* ("such is the import"), *iti tātparyaḥ* ("such is the overall sense"), and *ity āśayaḥ* ("such is the intention [of the author]"). These phrases are used in exactly the same way as *ity arthaḥ* (Section 1.13.1) and *iti bhāvaḥ* (Section 1.13.2), with similar meanings.

Kum.M. 5.73 yathä śmaśāna-sūpa-satkriyā na kriyate tathā tvam api tasmai na ghatasa iti tātparyârthah. The overall sense is that just as the consecration of the sacred post used in Vedic ritual is not performed for the executioner's stake in a cemetary, so also you are suited for him.

#### 1.13.5 anena, etena

In another group of formulas used to mark comments on the author's intentions, the remark of the commentator begins with the word *anena* or *etena* 

#### Chapter 3. Giving the Meaning of Words

("by this [expression or statement]"). The formulas anena ... vivaksitah and anena ... iti vivaksyate (both meaning "by this he means to say ...") are used with essentially the same force as the formula ity arthah (Section 1.13.1). The other formulas beginning with anena generally involve an attempt to convey the subtler implications or suggestions of the author's statement: e.g., anena ... sūcitah and anena ... iti sūcyate (both meaning "by this ... is hinted at"), anena ... iti gamyate ("by this ... is made known"), and anena ... iti dhvanyate ("by this ... is suggested"). All such formulas should be self-explanatory once their general pattern has been understood.

Ragh.M. 1.8 abhyasta-vidyānām. etena brahmacaryâśramo vivaksitah. "Of those who studied the sacred lore"—by this he means the stage of sacred studentship.

Ragh.M. 1.59 atharva-nidher ity anena purohita-krtyâbhijñatvāt tat-karma-nirvāhakatvam muner astīti sûcyate. "Of him who was the repository of the Atharva Veda"—by this it is hinted that the sage was expert in the rites of the domestic priest and was therefore capable of carrying out those rituals.

#### 1.13.6 References to speech

Similar observations are frequently made by any of the many ways of referring to the act of speaking on the part of the author of the text being commented on.

Kum. V. 1.15 bhāgīrathī-nirjhara-śīkarānām

gangā-prapāta-jala-kaņānām vodhā kseptêti śaitya-puņyatvôktih. The carrier the tosser of the sprays of the cataracts of the Bhāgīrathī of the drops of water of the waterfalls of the Ganges—this is an expression (ukti) of coolness and holiness.

Kum.M. 2.29 sahasra-grahanam āsthâtiśayārtham. The mention of the thousand (eyes of Indra simultaneously turned towards his guru) is aimed at the abundance of his regard (for the guru).

Kum.M. 1.15 kramād višesaņa-trayeņa šaitya-saurabhya-māndyāny uktāni. The three modifying (phrases) express coolness, fragrance, and gentleness, in that order.

## 1.14 Meanings by extension

The verbs *lakṣayati* and  $\bar{a}kṣipati$  are used by commentators to refer to the power of words to include in their meanings objects connected with the object directly denoted by them.

Kum.V. 1.37 athavā nīvi-sabdena nīvyā yad grathyate tad adho-vāso laksyate. Or, by the word "waistknot" there is included by extension the lower garment that is bound by that waistknot.

Kum. V. 3.10 madhuś caitras tena ca vasanto laksyate. Madhu—Caitra, and this includes by extension the springtime.

Kum. V. 1.27 dīpa-sabdenâtra jvālā-varti-taila-bhājanâtmakah samudāya āksiptah. Here the word "lamp" implies by extension the aggregate composed of flame, wick, oil, and vessel.

## **1.15 Glossing of puns**

When words of the text being commented on refer simultaneously to two objects through double meanings, the commentator may organize his references to these two sets of meanings either by using expressions meaning "in (reference to) one ... in (reference to) the other place" or by referring to the two sides with the word *paksa*.

Kum.M. 2.27 labdha-pratisthāh labdha-sthitayah. labdhâvakāśā ity anyatra. Having obtained a foundation [in reference to the gods], having obtained their positions. In the other [alternative, i.e., in reference to rules of grammar], having obtained the opportunity for operation.

#### Chapter 3. Giving the Meaning of Words

Kum.M. 1.24 labdha udayo yayā sā labdhôdayā. utpannêty arthaḥ. abhyuditêty anyatra. By whom arising has been attained, she is labdhôdayā, i.e., born [in reference to Pārvatī]. In the other [i.e., in reference to the moon], having risen.

Ragh.M. 2.73 adarśanena pravāsa-nimittena ... candra-pakṣe, adarśanam kalā-kṣaya-nimittam. By nonappearance caused by his being away from home. ... In the case of the moon, the nonappearance is caused by its waning.

Ragh.M. 4.33 **phalam** lābham. vrkṣa-pakṣe prasavam ca. ... **utkhātaḥ** sva-padāc cyāvitaiḥ. anyatra, utpāṭitaiḥ. **The fruit** the profit [in reference to the kings]. And in the case of the trees, their produce. ... **Dislodged** unseated from their positions. In the other [i.e., in reference to the trees], uprooted.

## 1.16 Definitions

When a word in the text being commented on is used in a technical or uncommon sense, the commentator may find it necessary to interrupt his running gloss in order to give a formal definition of the word. In such instances he may either offer his own definition (Section 1.16.1), or quote an authoritative dictionary (Section 1.16.2).

#### 1.16.1 Definitions offered by the commentator

Since a definition involves an independent sentence stating the meaning of a word, it cannot reproduce the case and number of the word as used in the original text. To solve this problem a commentator will sometimes add a pronoun after the definition in order to "pick up" the defined word and restore it to its original case, etc., so as to match the original text and continue the running gloss:

Ragh.M. 1.7 tyāgāya—sat-pātre viniyogas tyāgah—tasmai for donation—donation is presentation to a worthy recipient—for it

#### 1.16. Definitions

Kum.M. 4.9 upacāra-padam parasya rañjanârtham yad asatya-bhāṣaṇam sa upacāras tasya padam sthāṇam. kaitava-sthānam iti yāvat. An occasion of upacāra—an upacāra is an untrue statement made for the purpose of pleasing someone else. An occasion—an instance—of it. To put it bluntly, an instance of deceit.

Alternatively the commentator may simply insert the definition before or after the portion of the running gloss in which the word in question is presented. In such instances he will often employ the particle  $n\bar{a}ma$  ("by name," "that which is called ...") immediately after the word being defined, in order to make it clear that he is pausing to give a definition:

Ragh.M. 1.44 yūpo nāma samskrtah pašu-bandhāya dāru-višeṣah. yūpā eva cihnāni yeṣām teṣu (y**ūpa-cihneṣu**). (That which is called) a yūpa ("sacrificial post") is a special piece of wood consecrated for the tying up of the sacrificial victim. Of which the marks are yūpas, in those (y**ūpa-marked ones**).

Ragh.M. 1.4 vamśe kule. janmanâika-laksanah santāno vamśah. In the vamśa in the family line. A vamśa is a series preserving continuity by birth.

## 1.16.2 Definitions quoted from dictionaries

Most of the Sanskrit dictionaries quoted by the commentators follow one of two standard formats. Some, such as the dictionary of Amarasimha, simply string synonyms together after the fashion of a thesaurus, so that all the words appear in the nominative case:

Ragh.M. 1.7 "tyāgo vihāpitam dānam" ity amarah. "Donation, grant, gift"—Amara [cited to define tyāga].

Others, such as the dictionary of Keśavasvāmin, display the word to be defined in the nominative case and its meanings in the locative case:

Ragh.M. 1.4 "vajram tv a-strī kuliśa-śastrayoh, mani-vedhe ratna-bhede" iti keśavah. "vajra, non-feminine (i.e., found both as masc. and as neut.), in (the meanings of) thunderbolt and weapon, drill for boring gems, type of jewel"—Keśava.

#### Chapter 3. Giving the Meaning of Words

Frequently the quotations from dictionaries add little to what the commentator has already said on his own, and appear to be made simply for the weight of the lexicographer's authority or to reinforce the memory of students who were supposed to learn the dictionary by heart:

Kum.M. 2.64 **ratih** smara-priyā. "ratih smara-priyā" ity amarah. **Rati** the beloved of Kāma. "Rati, the beloved of Kāma"—Amara.

Quotations from dictionaries interrupt the running gloss when introduced near the place in the commentary at which the word to be defined is presented. Some commentators therefore save such quotations and lump them together at the end of their comments on a verse or other portion of the text.

## 1.17 Treatment of special items

Commentators use special devices to mark proper names (Section 1.17.1) and names of species (Section 1.17.2), and to explain the meanings of particles (Section 1.17.3), other indeclinables (Section 1.17.4), and pronouns (Section 1.17.5).

#### 1.17.1 Proper names

Proper names are marked as such, either by being followed by *nāma* (used as a separate, indeclinable word) or by being displayed as the first member of a *bahuvrīhi* compound ending in a word meaning "name" (*nāma*, samjñā, ākhyā, etc.).

Kum.M. 2.47 uccaihśravo nāma Uccaihśravas by name

Kum. V. 2.47 uccaihśravah-samjño 'śva-śresthah. The most excellent horse named Uccaihśravas.

Kum.M. 2.1 tārakeņa tāraka-nāmnā vajraņakha-putreņa kenacid asureņa. By Tāraka by a certain demon named Tāraka, the son of Vajraņakha.

#### 1.17.2 Names of species

When words referring to species or particular types of things are glossed, the term *visesa* is placed in compound after a word referring to a wider class of things to make it clear that the word being glossed does not apply to all members of that wider class.

Kum.M. 3.26 aśoko vrksa-viśesah aśoka a species of tree

Kum.M. 2.59 ayaskāntena maņi-višeseņa by lodestone by a particular kind of mineral

Kum.M. 2.21 pāśah rajjur āyudha-višeṣah lariat lasso, a particular type of weapon

#### 1.17.3 Meanings of particles

Many indeclinables in Sanskrit are function words whose meaning is primarily grammatical. For these items it may be difficult to provide a synonym or definition, and to explain their meaning a commentator must cite the word and make a direct statement of the force with which it is used. For examples of how such statements are made see Section 1.16. Note particularly the device of expressing the function of a particle through a word presented in the locative case (e.g., *apih samuccaye*, "*api* in the sense of conjunction"; *ān maryādāyām*, "*ā* in the sense of an exclusive limit").

#### 1.17.4 Other indeclinables

For other indeclinables, such as temporal adverbs, a commentator may simply indicate the precise significance of the word in the present context:

Ragh.M. 1.59 atha prasnantaram next after the question

Kum.M. 3.59 kathamcid atiyatnena somehow with great effort

32

#### 1.17.5 Pronouns

Similar information is inserted to explain the significance of pronouns. When the pronoun's antecedent is not in the text, the commentator may supply it:

Ragh.M. 1.18 sa rājā he the king

Kum.M. 1.2 yam himālaya whom Himālaya

And for other pronouns the special force with which they are used may be indicated:

Kum.M. 1.48 tam prasiddham keśa-pāśam that famous head of hair

Kum. V. 1.56 kenâpi kāmena kayâpy ajñātêcchayā with a certain desire with some unknown wish

Megh.M. 1 kaścid anirdistanāmā yakso a certain—unspecified by name—yaksa.

## Chapter 4

## **Analysis of Grammatical Complexes**

.

## 1.18 Introduction

The opportunity to break down grammatical complexes into their constituent parts arises constantly in commentaries on Sanskrit texts, since most words in the language are made up of readily discernible elements. Any inflected word, for example, contains both an inflectional stem and an inflectional affix, and such words can be analyzed by identifying the stem and naming the affix. The resulting analysis, however, will be a purely technical one, since the affixes can be named only through the use of the artificial terminology of Sanskrit grammar. When speaking of Sanskrit commentaries, what is more commonly meant by the term *vigraha* is an analysis in ordinary language, which is produced by using several separate words to restate the meaning of a single complex form—i.e., by "expanding" the complex form into a phrase with the same meaning.

Because ordinary language provides no entire words that can serve as direct substitutes for morphemes such as the inflectional endings, this second type of analysis is possible only for words more complex than inflected primary words, and it is therefore reserved for formations involving nominal or verbal stems that are themselves derived from simpler elements. In analyzing these complex formations, it is usually possible either to separate the constituent elements and place them in a phrase in which their relationship is explicitly stated (as in the analysis of many kinds of nominal compounds), or to produce a phrase that gives the same meaning as the complex form by using ordinary words to express the force of the grammatical elements involved in the complex formation (as in the analysis of forms derived through suffixation).

For these complex formations there are thus two kinds of analysis available: analysis in the technical language of grammar (*sāstrīya-vigraha*, "technical analysis," or *alaukika-vigraha*, "extraordinary analysis"), and analysis in ordinary, non-technical language (*laukika-vigraha*).

The śāstrīya-vigraha proceeds by naming the sequence of stems and suffixes involved, together with any inserted elements required by the sequence. For example, a genitive tatpurusa compound such as  $r\bar{a}ja$ -purusah ("king's man") will be explained in a technical analysis as being made up of the stem  $r\bar{a}jan$  with the termination of the genitive singular, -as (which is dropped in the final form), followed by the stem purusa with the termination of the nominative singular, -s. The actual analysis will take the form  $r\bar{a}jan + Nas + purusa + sU$ . The extra letters added in the names for the suffixes serve various purposes and disappear in the final form; here the N in Nas indicates that certain things are inserted before this suffix when it comes after a feminine stem, and the U in sU is simply to make it possible to pronounce the name of the suffix. Similarly, the taddhita formation steyam ("thievery," from stena, "thief"), may be analyzed in technical terms as stena + yaT + am. Here the T in the technical name of the taddhita suffix yaT indicates which sort of pitch accent the suffix carries.

For these same formations the *laukika-vigraha* proceeds by expanding each formation into a phrase having the same meaning:  $r\bar{a}j\bar{n}ah$  purusah ("a man of the king's") and *stenasya karma* ("the activity of a thief"). This sort of analysis in ordinary language occurs constantly in Sanskrit commentaries, and is obviously more readily accessible to students than analysis in technical language, since it requires no knowledge of the artificial terminology of Pāṇinian grammar. What it does require is familiarity with the modes of complex formation susceptible to such analysis (Section 1.19), the general principles of analysis involved (Section 1.20), the manner in which the formulas of analysis are inserted in a running commentary (Section 1.21), and the common formulas used in analyzing each type of complex formation (Sections 1.27–1.75).

Full coverage of the terms required in technical grammatical analysis and the procedures by which they are used by the commentators is beyond the scope of this handbook, but a brief overview of how such analyses are usually worded by commentators is given below in Section 1.20.

## 1.19 Modes of complex formation

An analysis in ordinary language is a restatement in the "phrase mode"  $(v\bar{a}kya-vrtti)$  of the meaning expressed in a more compact mode by the formation being analyzed. Sanskrit grammar recognizes five modes (vrtti) of the formation of derived words for which a *vigraha* in ordinary language may be possible:

- 1. san-ādy-anta-dhātu-vrtti, or the formation of new verbal stems (ending in the suffix san, etc.), i.e., of denominative verbs by adding suffixes to nouns, and of stems of secondary conjugation by adding suffixes to verbal roots (Sections 1.22-1.26).
- 2. krd-vrtti, or the formation of new nominal stems by primary derivation, i.e., by adding krt suffixes to verbal roots (Sections 1.27-1.34).
- 3. *taddhita-vrtti*, or the formation of new nominal stems by secondary derivation, i.e., by adding *taddhita* suffixes to nominal stems (Sections 1.35–1.42).
- 4. samāsa-vŗtti, or the formation of new nominal stems through the compounding of nouns (Sections 1.43-1.72).
- 5. ekaśesa-vrtti, or expressing the sense of several words related by meaning or syntactical concord through the retention of only one of those words, e.g., "dogs" for "dogs and bitches" (Sections 1.73–1.75).

## **1.20** General principles of analysis

For each type and subtype of complex formation susceptible to analysis in ordinary language, there exists a basic formula of analysis that, while it may be understood without a knowledge of the details of Pāṇinian grammar, is based directly on the treatment of these formations by Pāṇini and his successors. For some formations, notably those derived by the addition of *taddhita* suffixes (Section 1.35), the formula follows the exact wording of the rules of Pāṇini. For others, the formulas are designed to express the meanings assigned to the formation by Pānini.

Some words are analyzed not in terms of formulas used for types or subtypes of complex formations but in terms of the specification by the grammarians of a meaning for that particular word. Such irregular words are said to be *nipātita*, "set down," as opposed to words that are *lakṣaṇalabdha*, i.e., whose formation is arrived at by more general rules (see, for example, Section 1.34).

To some extent the formulas actually used by many commentators are drawn from a practical tradition of commentarial techniques rather than directly from a close adherence to the detailed explanations of the grammarians. Thus for some formations (e.g., *nañ-tatpuruṣa* compounds, Section 1.58) the formula usually encountered in the commentaries is more straightforward and less strict than some grammarians would like, and for others (e.g., *samāhāra-dvandva* compounds, Section 1.63) the commentators often do not bother to include in their formula of analysis the full details of the special elements of meaning involved in the formation. And for some formations the full standard formula coexists with a simplified formula used when the complex formation forms part of a larger complex (as in the analysis of *karmadhāraya* compounds serving as members of larger compounds, Section 1.71).

When some of the less common formations are to be analyzed, the degree to which the formulas used in a given commentary coincide with the formulas offered by the grammarians may also be influenced by the type of reader for whom the commentary is intended. Rudimentary lists of simple formulas of analysis, collected in short handbooks such as the *Samāsacakra*, join lists of verbal and nominal paradigms as the first items memorized by students in the traditional system of Sanskrit-medium education, and commentaries intended for students in their early years of study will naturally attempt to conform to the formulas presented in these elementary handbooks.

Careful study of the basic formulas of analysis available for each type of formation is valuable in several ways. First, a knowlege of the formulas is useful in itself as an aid to a clearer understanding of the different types of complex formations used in Sanskrit. Second, familiarity with the nature of the more important formulas is helpful in recognizing an analysis for what it is, should it show up unexpectedly in an otherwise discursive work of commentary. Finally, the ability to recognize the basic formulas easily is necessary if one is to understand the sentences in running glosses into which they are inserted, because the formula itself may often be obscured in ways described in the following section.

For all these reasons the treatment of each type of complex formation in this handbook includes a statement of the basic formula of analysis appropriate to that formation, drawn from the works of the grammarians, alongside any examples taken from commentaries.

Of the five types of complex formation listed above, the first three (derivative verbs, nouns of primary derivation, and nouns of secondary derivation) are also susceptible to analysis in terms of the technical elements of Pāṇinian grammar in addition to analysis in ordinary language. While both approaches depend upon the details of Pāṇini's treatment of the language, the technical analysis goes beyond the ordinary analysis in employing the specially coded terminology used by Pāṇini in referring to grammatical elements. In the analysis of derived forms the most important of these elements are the suffixes, and the Pāṇinian treatment of suffixes involves three special problems.

First, Pāṇini recognizes derivation by suffixation only, so that many phenomena that we might not automically associate with suffixes are described by the Pāṇinians as being due to the addition of some suffix. For example, the form  $-\bar{a}bha$  that occurs at the end of the *upapada* compound analyzed by Mallinātha and Hemādri in the examples used in Section 1.21.2 is a derivative of the verbal root  $bh\bar{a}$ , and the vowel of the root appears to have been shortened. For reasons of economy the way that Pāṇini explains such a form is to say that it is the result of the addition of a short *-a* suffix, before which the long  $-\bar{a}$  of the root has dropped. Similarly, formations that appear to consist of the bare root with no alteration (e.g., the final member of *veda-vid*, "Veda-knowing," which is simply the root *vid*, "to know") are explained by Pāṇini as the result of the addition of a "null suffix," i.e., a suffix that completely disappears.

Second, the technical names of the suffixes generally contain extra letters beyond the actual form of the suffix itself, the purpose of which is purely indicatory. For example, the short -a suffix that forms the component  $-\bar{a}bha$  is technically called Ka, and in this name the K is merely a "dummy letter" serving to convey grammatical information about the resulting form. Such an indicatory phoneme (displayed here in uppercase) is called an *anubandha* or *it*, and it does not appear in the final form. The general purpose of K in this role is to indicate that when the suffix marked

38

with it is added to a root, the guna-strengthening usually caused by the addition of a suffix doe not occur (e.g., when the suffix Ka is added to the root *budh*, the resulting form is *budha* rather than *bodha*), and there is also a special rule to the effect that the final vowel of roots in long  $-\bar{a}$  is dropped before suffixes with indicatory K (thus the form  $-\bar{a}bha$  rather than  $-\bar{a}bh\bar{a}$ ).

Third, in the technical names of some suffixes even the actual form of the suffix does not appear, being replaced by some substitute specified by Pāṇini. For example, yu in the technical name of suffixes stands for the longer form ana, and to this yu will be added indicatory letters covering the grammatical phenomena associated with the suffix in question. Thus LyuTis the name of the kT suffix -ana that produces nominal stems in which the acute accent rests on the syllable before the suffix (which phenomenon is indicated by the L in the technical name) and for which the feminine stems end in  $-\bar{t}$  rather than the usual  $-\bar{a}$  (which is indicated by the T in the technical name).

All three of these problems arise in the case of a suffix such as KviP, the technical name for one of the null suffixes mentioned under the heading of the first problem noted above. The actual suffix is vi, which is said to always disappear, so that it is in effect replaced by zero. The K, once again, indicates lack of strengthening, and the P indicates that roots ending in short vowels will add the augment -t (e.g., loka-krt, "world-making," from the root kr).

The meanings of these indicatory letters are fully explained by Otto von Böhtlingk in his *Pânini's Grammatik* and (much more accessibly for readers of English) by G. V. Devasthali in his *Anubandhas of Pānini*. Both books also give lists of the suffixes, both with and without the indicatory letters, and some similar information is provided by M. R. Kale in *A Higher Sanskrit Grammar*. A more thorough introduction to Pānini's methods is available in the works by George Cardona listed in the Appendix. We cannot reproduce this information here, but we can point out the two most common ways in which these technical names of suffixes are used.

First, the actual technical analysis of a complex formation consists of the technical name of the suffix, preceded by the ablative of the item to which the suffix is added. Thus the word *śaiśavam* ("infancy"), formed from *śiśu* ("infant") by the addition of the short -*a* suffix technically called *a*N (the N indicates vrddhi-strengthening of the first syllable of the formation) is analyzed *śiśor a*N. Further technical devices may used in identifying the base to which the suffix is added; for example, one way of referring to verbal roots is by adding the suffix *-ti*, so that the word *mānam*, formed from the root *mā* ("to measure"), may be analyzed by saying *māter LyuT* (*"LyuT* [i.e., the *taddhita* suffix *-ana*] after *māti* [i.e., the verbal root *mā*]").

Second, the meaning of these suffixes is indicated by preceding the technical name of the suffix with its meaning in the locative case (cf. Chapter 6). Thus for *śaiśavam* a commentator might say *bhāve a*Ņ ("*a*Ŋ in the sense of a state or condition"; see Section 1.28), and for *mānam*, if it is interpreted as having instrumental force (*mīyate anenêti mānam*, "one measures with it, thus a measurer," i.e., an instrument of measuring) the comment might be *karaņe LyuŢ* ("*LyuŢ* in the sense of an instrument"; see Section 1.29.3).

This same technique is used in dealing with other grammatical elements as well. Verbal roots, for example, as they appear in the traditional list (the *Dhātupāțha*, an appendix to Pāṇini's grammar), are provided with indicatory letters of their own—e.g., kr, "to do," is technically called DUkrN, with the *DU* indicating that the root may take a certain suffix, and the N indicating that the root takes ātmanepada endings with the benefit of the action accrues to the agent. The meaning of the root is indicated by a word in the locative case placed after the technical name of the root—e.g., DUkrN karaņe, "kr in the sense of doing."

Commentators frequently use such technical remarks as supplements to analyses in ordinary language or in place of them. In the quotations from commentators offered in the handbook we will often include such remarks as samples of this approach, but a full understanding of their use requires a dedicated study of Pāṇinian grammar.

## 1.21 Insertion of analyses in a commentary

The formulas of analysis described below in Sections 1.22–1.75 are inserted at the proper places in a running commentary in much the same way as the simple glosses described in Section 1.9 (for detailed examples of the arrangement of running commentaries and the insertion of explanatory material, see Sections 1.76–1.80). In their placement within a commentary the formulas of analysis differ from simple glosses only in that, as the examples given in the following sections indicate, an analysis is usu-

1.21. Insertion of analyses in a commentary

#### 40

Chapter 4. Analysis of Grammatical Complexes

ally placed before the complex being analyzed, while synonyms and other simple glosses usually follow the words they explain (Section 1.10).

In practice commentators often do not bother to repeat the full original form of the complex being analyzed (especially when it is a nominal compound); in such instances the analysis itself replaces the repetition of the original formation, or the analysis may conclude with a simple reference to the original form rather than with its actual restatement (as in the phrases *tat tathā, tat tathôktam, tat tathā-bhūtam*, etc., used at the end of analyses of *bahuvrīhi* compounds; Section 1.70).

The basic formula of analysis may often be obscured by the addition of other material when the commentator finds it necessary either to gloss the words used in the analysis (Section 1.21.1), or to indicate the inflection of the complex being analyzed as it appears in the original text (Section 1.21.2).

#### 1.21.1 Glossing the words used in the analysis

Within the basic framework of a standard formula of analysis, commentators may insert glosses of the words making up that analysis at any point, using the same techniques that are employed in glossing separate words in the original text. For example, the word *marut-sakhah*, an epithet of fire, is a straightforward genitive tatpurusa compound, meaning "friend of the wind," and its analysis, using the standard formula (Section 1.53.1) is therefore simply *marutah sakhā marut-sakhah*, "friend of the wind: windfriend." In employing this standard formula, however, the commentator Mallinātha inserts synonyms after the first member in the analysis and after the repetition of the entire compound:

## Ragh.M. 2.10 marutah vāyoh sakhā marut-sakho 'gnih. Friend of the wind, of air: wind-friend, fire.

This technique is taken a step further in the practice of using synonyms in place of the parts of the original complex when analyzing the complex. For example, the analysis given by Mallinātha of the compound *marut-prayuktāh* ("impelled by the wind") in the same verse is not *marutā prayuktāh* but *vāyunā preritāh*—synonyms for the original members of the compound, displayed in the analysis with the same inflection and position that the original members would have in the standard formula of analysis.

## 1.21.2 Indicating the inflection of the original

The formulas for the analysis of derivative and compound nouns as offered by the grammarians generally present the noun that is being analyzed in the nominative case. Of course such nouns may appear in any case in a text being commented on, and in order to make the transition from the standard formula to the inflection found in the original text a commentator may follow either of two procedures in analyzing the noun.

One procedure is to modify the basic formula by inflecting the syntactically predominant part of the analysis to match the inflection of the noun being commented on. For example, where the text has  $r\bar{a}ja$ -putr $\bar{a}t$ , "from the king's son," the commentator may give the analysis  $r\bar{a}j\bar{n}ah$  putr $\bar{a}t$ , based on the nominative-case formula  $r\bar{a}j\bar{n}ah$  putrah but reflecting the ablative case of the word being explained. Such an analysis is called a parinisthita-vibhakti-vigraha, "analysis (using) the case-ending of the finished (form)."

The other procedure is to use the standard formula without modification, and then to add a pronoun to pick up the formation and convert it to the required inflection. Using this method the analysis of  $r\bar{a}ja$ -putr $\bar{a}t$ would be  $r\bar{a}j\bar{n}ah$  putras tasm $\bar{a}t$ , "the king's son, from him." Such an analysis is called a prathamânta-vigraha, "analysis ending in the nominative case-ending."

Where a pronoun already occurs at the end of a standard formula, as for example in the analysis of *bahuvrīhi* compounds, the same two options are still available. Here the pronoun already present at the end of the standard formula is the syntactically predominant part of the analysis, as it refers to the same thing that the formation as a whole refers to. Thus the analysis of the compound *dīrgha-karņāt*, "from Long-ear," may be either *dīrghau karņau yasya tasmāt (tathôktāt)*, "whose ears are long, from him (who is so-called)," or, with the addition of a further pronoun to pick up the analysis, *dīrghau karņau yasya sa (tathôktas) tasmāt*, "whose ears are long, he (is so-called), from him."

Just as parts of a formula of analysis may be separated by intervening glosses (Section 1.21.1), this "pick-up" pronoun added at the end of a nominative-case analysis may be separated from the body of the analysis by any amount of intervening explanatory material. For example, the compound *marut-sakhah* used as an example in Section 1.21.1 is actually the

#### Chapter 4. Analysis of Grammatical Complexes

first member of a longer compound, *marut-sakhâbham* ("shining like fire," masculine accusative singular), which Mallinātha interprets as an *upapada* compound (Sections 1.32, 1.59). He therefore uses a pronoun to pick up *marut-sakhaḥ* for the rest of the analysis of the longer compound, and then he must add another pronoun at the end to place the whole compound in the accusative case. But before adding this last pronoun he pauses to explain which *krt* suffix is used to form the last part of the compound:

Ragh.M. 2.10 maruto vāyoh sakhā marut-sakho 'gnih. sa ivâbhātîti marut-sakhâbhah. "ātoś côpasarge" iti ka-pratyayah. tam. Friend of the wind, of the air: wind-friend, fire. (He) shines as he does, thus wind-friend-shiner. The suffix Ka (i.e., -a, without strengthening of the root to which it is added) by (Pā. 3.1.136) "And (there is the suffix) ka after (a root ending in) long  $-\bar{a}$ , when a verbal prefix is used." Him.

Note that insertions such as this do not alter the basic formula of analysis: they simply interrupt it. The same problem arises when a commentator offers more than one explanation of a part of the analysis, for he will carry on with the basic structure of the formula no matter how many alternatives he inserts in a part of the formula. For example, when the commentator Hemādri analyzed this same compound *marut-sakhâbham*, he not only inserted a quotation from a dictionary, but also could not decide whether *marut-sakhaḥ* means Agni or Indra, and whether the longer compound is an *upapada* compound, as Mallinātha thought, or a *madhyamapadalopabahuvrīhi* compound (Section 1.66.2):

Ragh.H. 2.10 maruto vāyoh sakhā iti marut-sakho 'gnis tasyâbhêvâbhā yasya. indrasyêti vā, "pulomārir marutsakhā" iti dhanamjayah. tadvad ābhātîti vā. tam. Friend of the wind, of the air, thus wind-friend, fire. Whose appearance is like the appearance of his [i.e., of Agni's]. Or [like the appearance] of Indra's: "Pulomāri, Marutsakha [among the epithets of Indra]," says [the lexicographer] Dhanamjaya. Or, (he) shines as he does, thus [wind-friend-shiner]. Him.

In reading the examples quoted from commentaries in the sections that follow, take care to separate the features of expression required in order to match the inflection of the word being explained from those features proper to the formula of analysis being exemplified.

## Chapter 5

## **Derivative Verbs**

## 1.22 Introduction

Of the five modes of complex formation recognized in Pāninian grammar, four result in derived nominal stems. The fifth involves the addition of suffixes to verbal roots and to nominal stems to create formations that are treated as derived verbal roots. The first of these "root-forming suffixes" taken up by Pānini is the desiderative suffix *-sa*, called *san* by Pānini (the final *-n* serves only to indicate that the resulting formation will have the acute pitch accent on its first syllable). The formations so derived are therefore called *san-ādy-anta-dhātu* ("roots ending in *san*, etc.," Pā. 3.1.5–3.1:32; corresponding to "derivative or secondary conjugation," Whitney 996 ff.).

÷.,

Some of the suffixes are added to items already technically considered roots without producing any change in meaning that requires analysis in ordinary language. This is true, for example, of verbs that use the causative suffix without clear causative meaning, including the items viewed by Pāṇini as roots of the *cur-ādi* or tenth-class and by Western grammarians as denominatives with causative rather than denominative accent (*corayati*, *cintayati*, etc., Pā. 3.1.25; Whitney 607, 775, 1041 b, 1056), and it is also true of the roots that use the desiderative suffix without clear desiderative meaning (Pā. 3.1.6–3.1.7; Whitney 1040)—*jugupsate*, *titikṣate*, *cikitsati*, *mīmāṃsate*, *bībhatsate*, *dīdāṃsate*, *sīsāṃsate*. But these suffixes also produce four large groups of formations susceptible to analysis in ordinary language: desiderative verbs (Section 1.23), intensive verbs (Section 1.24), causative verbs (Section 1.25), and denominative verbs (Section 1.26).

## 1.23 Desideratives

These are the sann-anta-dhātus, "roots ending in (the suffix) san," Pā. 3.1.7; "desiderative verbs," Whitney 1026–1040. A straightforward desiderative formation is analyzed in ordinary language by a formula using the verb  $i_s$ (*icchati*) after the infinitive of the root in question:

Kāśikā 3.1.7 kartum icchati cikīrsati. He wishes to do.

The same principle applies to the analysis of desiderative formations in which the addition of a further suffix has produced a participle, an adjective in -u, or a noun in  $-\bar{a}$ ; the same suffix will be added to the root is in the analysis:

Kum.M. 3.14 *ipsitam aptum istam* desired sought to be obtained

Ragh.M. 1.2 titīrsus taritum icchuh desirous of crossing seeking to cross

Ragh.M. 1.7 vijigīşūņām vijetum icchūnām of those desirous of conquering of those seeking to conquer

Kum.M. 2.7 sisrksayā srastum icchayā through the desire of creating through the wish to create

Certain idiomatic uses of the desiderative require special analysis:

Kāśikā 3.1.7 śanke patişyati kulam [pipatişati kulam] I fear the bank is going to fall—the bank is likely to fall

Kāśikā 3.1.7 [śanke mariṣyate śvā] śvā mumūrṣati I fear the dog is going to die—the dog is likely to die

## 1.24 Intensives

These are the *yan-anta-dhātus*, "roots ending in [the suffix] *yan*," Pā. 3.1.22–3.1.24; "intensive or frequentative verbs," Whitney 1000–1025. In the basic formulas for the analysis of intensives and frequentatives the finite simple verb is used with the addition of *punah punah* ("again and again") for the frequentative meaning or *bhrśam* ("intensely") for the intensive meaning:

Kāśikā 3.1.22 punah punah pacati pāpacyate he cooks repeatedly

Kāśikā 3.1.22 bhrśam jvalati jājvalyate it blazes intensely

As in the analysis of desideratives (Section 1.23), the same basic formula applies even when further suffixation occurs:

Kum.M. 1.23 jangamyante bhrśam gacchantîti jangamā deva-tiryan-manusyîdayah. They move all around—they go intensely—thus moving creatures—gods, animals, humans, and so forth.

Here a *krt* suffix used in the sense of the agent (Section 1.29) is added to the intensive formation in order to form a nominal stem.

Again as in the analysis of desideratives, certain idiomatic uses of the intensive require special analysis:

*Kāśikā 3.1.23 kuțilam krāmati cankramyate* he moves crookedly (i.e., he goes in zig-zag fashion).

Kāśikā 3.1.24 garhitam lumpati lolupyate he breaks it in a reprehensible way (i.e., he busts it all to pieces).

## 1.25 Causatives

These are the *nij-anta-dhātus*, "roots ending in [the suffix] *nic*," Pā. 3.1.26; "causative verbs," Whitney 1041–1052. Straightforward causative formations may theoretically be analyzed by using the present participle of the root in question, followed by some verb of instigation; thus *gamayati*, "he

#### 1.26.2 Behavior toward

(Pā. 3.1.10):

5

Kāśikā 3.1.10 putram ivâcarati putrīyati he treats (him) like a son

#### 1.26.3 Behavior like

(Pā. 3.1.11):

Kāśikā 3.1.11 śyena ivâcarati śyenāyati it acts like a falcon

Ragh.M. 2.61 amrtam ivâcaratîty amrtāyamānami it acts like the nectar of immortality, thus acting like the nectar of immortality

### **1.26.4** Becoming something

(Pā. 3.1.12-3.1.13):

Kāśikā 3.1.12 (abhrśo) bhrśo bhavati bhrśāyate (not having been intense) it becomes intense—it itensifies

Ragh.M. 2.17 śyāmāni bhavantīti śyāmāyamānāni they become dark, thus darkening

For the basic formula and the optional omission of its first part, cf. the analysis of *cvi* formations (Section 1.42.4).

### 1.26.5 Making something so

(Pā. 3.1.21):

Kāsikā 3.1.21 muņdam karoti muņdayati he makes it shaven—he shaves it

Ragh.M. 3.35 laghayisyatā laghum karisyatā by the one being about to lighten it—by the one being about to make it light

Ragh.M. 5.73 malinayanti malināni kurvanti they dirty them—they make them dirty

causes him to go," may be analyzed gacchantam prerayati, "he sets him going." In practice, however, such formations are only very rarely analyzed in ordinary language; instead a commentator will offer the causative form of a synonymous verb as a gloss or provide an analysis in the technical terminology of Pāṇinian grammar.

As in the analysis of the derivative verbs described in the previous two sections, certain idiomatic uses of the causative require special analysis. An example:

Kāśikā 3.1.26 kamsa-vadham ācaste kamsam ghātayati He tells the story of the slaying of Kamsa—he has Kamsa slain.

### **1.26** Denominatives

These are the *nāma-dhātus*, "roots [formed from] nouns," Pā. 3.1.8 ff.; "denominative verbs," Whitney 1053–1068. Denominative verbs are formed by the addition of suffixes to substantives and adjectives. They fall into several groups according to the precise meaning imparted by the suffix. For each group the standard formula of analysis in ordinary language is based directly upon the wording of the portions of the rules of Pānini describing those meanings; the commentators follow these basic formulas of analysis quite closely, but sometimes omit the elements referring to finer details of meaning. In the following examples, the basic formulas are taken from the  $K\bar{a}sik\bar{a}$ , and the elements sometimes omitted by commentators have been placed in parentheses.

#### 1.26.1 Desire to have

(Pā. 3.1.8-3.1.9):

Kāśikā 3.1.8 (ātmanaḥ) putram icchati putrīyati he desires a son (for himself)

Ragh.M. 1.35 putrakāmyayā ātmanaḥ putrecchayā through desire for a son through the wish for a son for themselves

In the last example the denominative stem has been subjected to further suffixation in order to produce a krt formation (cf. Section 1.23).

#### Chapter 5. Derivative Verbs

48

# 1.26.6 Feeling something

(Pā. 3.1.18):

Kāśikā 3.1.18 sukham vedayati sukhāyate he feels happiness

### 1.26.7 Doing or making something

(Pā. 3.1.17, etc.):

Kāśikā 3.1.17 śabdam karoti śabdāyate he makes sound-he sounds

Kāśikā 3.1.19 namaḥ karoti namasyati he does honor—he honors Often a particular noun will call for a more specific verb:

Kāśikā 3.1.25 satyam ācaste satyāpayati he tells the truth

Kāśikā 3.1.15 tapaś carati tapasyati he performs austerities

Such meanings shade off into the next category.

### 1.26.8 Making use of something

(Pā. 3.1.15, etc.):

In this category belong many specific analyses applying to particular words. A few examples follow.

Kāśikā 3.1.15 romantham vartayati romanthāyate it turns over the cud—it chews the cud

Kāśikā 3.1.20 [puccham paryasyati] paripucchayate it tosses its tail around—it wags its tail

Kāśikā 3.1.25 ślokair upastauty **upaślokayati** he praises with eulogies—he eulogizes

Kāśikā 3.1.25 lomāny anumārsty anulomayati he strokes the hairs in the direction in which they lie—he curries

Kāśikā 3.1.25 varmaņā samnahyati samvarmayati he girds with armor-he armors

# **Chapter 6**

# **Primary Nominal Derivation**

# 1.27 Introduction

The formations called *krt* or *krt*-anta ("ending in *krt* suffixes," Pā. 3.1.91– 3.4.117; "primary derivatives," Whitney 1143–1201) are the nominal stems (including participles and verbals) derived through the addition of suffixes to verbal roots. In general terms the category of *krt* suffixes includes all suffixes added directly to verbal roots with the exception of the *san-ādi* suffixes ("root-forming suffixes," Sections 1.22–1.26) and the *tin* suffixes (the inflectional endings of finite verbs).

<u>77</u>

Akin to the krt suffixes are the suffixes called  $un-\bar{a}di$  ("beginning with un"), which serve the same function of deriving nominal stems from verbal roots, but often do so through forced and improbable etymologies. They are preserved in collections outside the grammar of Pānini, and although Pānini makes reference to this group of suffixes (Pā. 3.3.1–2, 3.4.75), his general policy is to take the nouns derived by  $un-\bar{a}di$  suffixes as ready-made forms, so that he does not concern himself with their derivation. The commentators mention them rather infrequently in comparison with the suffixes taught by Pānini, and when they do refer to them their techniques of analysis are the same as those used in analyzing krt formations. For example, the first of the  $un-\bar{a}di$  suffixes, un (the purpose of the final -n is to indicate that the vowel of the root to which it is added is subject to vrddhi strengthening), is said to form the word  $k\bar{a}ru$  from the root kr, and this formation is analyzed karotîti kāruh silpī, "he makes, thus a maker, an artisan," exactly as in the analysis of kartari krt formations (Section 1.29.1).

### Chapter 6. Primary Nominal Derivation

As in the analysis of the other types of complex formations, the formulas of analysis in ordinary language used by the commentators to explain krt formations follow very closely the treatment of these formations in Pānini's grammar. Thus the formulas of analysis may be divided by their being applicable to either of two major categories of meaning: that of the verbal action itself (Section 1.28), and that of the various participants in an action (Section 1.29). Beyond these basic varieties of meaning, the precise meaning of a formation may be narrowed by restrictions of relative time (Section 1.30) or by any of a number of other restrictions of meaning (Section 1.31). Furthermore, many krt formations are possible only when compounded with an accompanying word (Section 1.32). The various verbal participles form special subsets of krt formations, and some of these require special treatment (Section 1.33). Finally, the explanation of certain formations depends not upon the basic formulas of analysis but upon the specification by the grammarians of particular meanings for particular words (Section 1.34).

# 1.28 Basic bhāve meanings

A great many  $k_{P}$  suffixes are used  $bh\bar{a}ve$ , i.e., in the sense of the verbal action itself. Most of the resulting formations are nouns of action corresponding in meaning to the English gerund in '-ing'. Since the only straightforward method of duplicating the meaning of such formations in ordinary language in Sanskrit is through the use of another verbal noun formed with a different suffix, the explanation of these items in commentaries is usually achieved either by giving an analysis in technical terminology or by simply offering a synonym:

Ragh.M. 5.45 bhange bhanjane in breaking—in (the action of) breaking

Ragh.M. 5.61 *āgamena āgamanena* by the arrival—by the arriving Kum.V. 1.59 saņmārjanaņ saņmārgaņ (the action of) sweeping—sweeping

The offering of a synonym in the form of another verbal noun using a different suffix is also employed when the word being glossed happens to

be a participle with simple *bhāve* meaning:

Ragh.M. 2.54 prahrtam tu prahāras tu but the stroke--but the striking.

When, however, the formation being explained is an indeclinable verbal form, more complicated formulas are required.

Continuatives are generally analyzed by using a verbal noun derived from the same root, followed by a continuative form of the root *kr*:

Ragh.M. 3.51 prahasya prahāsam krtvā laughing performing the action of laughing.

When this technique is used, it is often the continuative made from kr, rather than the verbal noun, that indicates the presence of secondary meanings such as that of the causative:

Ragh.M. 5.62 pravesya pravesam kārayitvā showing in causing him to perform the action of entering

The rarer *namul* form of continuative is usually glossed simply by offering the commoner form of the continuative as a synonym:

Kum.M. 4.26 stanau sambādhya stana-sambādham striking the ; breasts—striking the breasts

Infinitives may often be explained through the use of a verbal noun in the dative case, but even where this is possible other more roundabout expressions are also allowed, which generally use verbal formations referring to the agent rather than to the verbal act itself:

*Kāśikā 3.3.11 [paktum vrajati] pākāya vrajati* [he goes in order to cook] he goes for cooking

Kāśikā 3.3.11 bhoktum vrajati bhojako vrajati he goes in order to eat—he goes as an eater

Kāśikā 3.3.12 [aśvān dātum vrajati] aśva-dāyo vrajati [he goes in order to give horses] he goes as one intending to give horses

Kāśikā 3.3.13 [kartum vrajati] kariṣyāmîti vrajati [he goes in order to do] he goes (with the thought), "I shall do"

### **1.29** Basic *kāraka* meanings

The krt suffixes that do not refer to the verbal action itself refer to a participant in that action. In the Pāninian system there are six categories of such participation, called kārakas. With few exceptions (Sections 1.30.1, 1.31, 1.32.4, 1.32.5) formations made with such suffixes are analyzed using formulas containing finite forms of the verb in question. When the suffix is used in the sense of the agent (Section 1.29.1) or the object (Section 1.29.2) of the action, an active or passive finite verb is sufficient to indicate the thing to which the formation refers, since by Panini's rules the verbal termination of an active verb is itself used kartari (in the sense of the agent), and that of a passive verb is used karmani (in the sense of the object). The thing referred to by the formation will then be the grammatical subject of the finite verb. But where the suffix is prescribed by Pānini in the sense of some other kāraka (Sections 1.29.3-1.29.6), the category of participation through which the thing referred to is connected with the verbal action must be indicated by using some form of the pronoun idam, exhibited in the grammatical case appropriate to the kāraka in question.

None of these formulas, unlike those used in the analysis of derivative verbs and of nominal compounds, can be substituted directly for the word that it analyzes, because in each of these formulas the syntactically predominate element is a verb, while the word being analyzed is a noun. Instead the formula in itself merely offers a clear explanation of the manner in which the thing referred to particpates in the verbal action. Accordingly, each of these formulas must be concluded with the particle *iti* followed by the word being explained. For example, in order to explain that the word  $p\bar{a}caka$  refers to the agent in the action of cooking, the basic formula is simply *pacati*, "he cooks"; but the full analysis is *pacatîti pācakaḥ*, "he cooks, thus [he is] a cooker." Much less frequently a commentator will avoid the need for the particle *iti* by using a relative pronoun: *yaḥ pacati sa pācakaḥ*, "he who cooks is a cooker." The same technique is occasionally used in analyzing formations involving any of the *kārakas*; compare the following analyses:

Kum.C. 7.53 apidhīyate 'nenêty apidhānam kapātam (something) is covered by this, thus a covering, a leaf of a door

Kum.V. 7.53 apidhīyate sthagyate yena tad apidhānam kavātah that

by which (something) is covered—concealed—is a covering, a leaf of a door

Notice also that, as in the analysis of other types of complex formations, the commentator may insert glosses at any point in the analysis.

#### 1.29.1 kartari

Formations made with krt suffixes used in the sense of the agent of an action are analyzed with finite active forms of the verb in question. Where the word being analyzed is plural, a plural form of the finite verb will be used. The same basic formula applies whether the word being analyzed is a substantive or an adjective.

Ragh.M. 4.12tapatîti tapanah sūryahit heats, thus heater, the sunRagh.M. 4.42yudhyanta iti yodhāhthey fight, thus fightersRagh.M. 2.72sukhayatîti sukhahit pleases, thus pleasant

When such formations contain a verbal prefix, two methods of analysis are available. The prefix may be expanded, so to speak, into a separate adverbial form:

Ragh.M. 12.68 viśesena bhīsayate śatrūn iti vibhīsanah he especially frightens his enemies, thus terrifying

Kum.V. 1.15 niyatam jharati sravatîti nirjharah it flows--streams--continuously, thus a cascade

Or the prefix may simply be retained with the finite verb used in the analysis:

Ragh.M. 1.38 utkiranti viksipantīty utkirāh they scatter upwards—toss upwards, thus scattering up

Kum.V. 5.44 upayacchate svīkarotīty **upayantā** pati he takes—accepts as his own, thus **taker**, bridegroom

54

#### 1.29.2 karmani

Formations made with *krt* suffixes used in the sense of the object of an action are analyzed in much the same way as *kartari* formations, except that the finite verb used in the analysis will be passive rather that active.

Ragh.M. 2.65 kāmyata iti kāmo varaķ it is desired, thus a desire a choice

Ragh.M. 1.20 prārabhyanta iti prārambhāh they are undertaken, thus undertakings

Ragh.M. 2.4 prasūyata iti prasūtiķ santatiķ it is procreated, thus procreation offspring

#### 1.29.3 karane

Formations made with *krt* suffixes used in the sense of an instrument involved in an action are analyzed with a finite verb, usually passive, and a form of the pronoun *idam* exhibited in the instrumental case.

Ragh.M. 5.57 sammohyate 'nenêti sammohanam (one) is infatuated by means of this, thus infatuating

Kum.M. 2.4 vibhajyate 'nenêti vibhāgaḥ (something) is divided by this, thus a division

Kum.M. 3.40 vihanyanta ebhir iti vighnāh pratyūhāh (people) are obstructed by these, thus obstructions obstacles

#### 1.29.4 sampradāne

Formations made with  $k_{rt}$  suffixes used in the sense of the one for whose sake an action are analyzed with a finite verb and a form of the pronoun *idam* exhibited in the dative case.

Kum. V. 6.87 [6.88] arpyate 'smā ity arpaņaķ (something) is offered to this, thus recipient

### 1.29.5 *apādāne*

Formations made with krt suffixes used in the sense of a point of departure involved in an action are analyzed with a finite verb and a form of the pronoun *idam* exhibited in the ablative case.

Ragh.M. 1.2 prabhavaty asmād iti prabhavah kāraņam (something) arises from this, thus a source a cause

### 1.29.6 adhikarane

Formations made with kpt suffixes used in the sense of a location involved in an action are analyzed with a finite verb and a form of the pronoun *idam* exhibited in the locative case.

Kum.V. 6.43 samcaranty asminn iti samcaro mārgah (people) travel on this, thus a road a street

Ragh.M. 5.40 dhīyate 'syām iti dhānī (something) is put in it, thus a receptacle

Kum.M. 1.7 likhyate yeşu te lekhāh patrikāh those on which writing is done are writing sheets leaves

Notice that in the last example the commentator has used the alternative approach described in the introduction to Section 1.29.

### **1.30** Temporal restrictions

While most *krt* suffixes are used with present or temporally indefinite force, a great many suffixes are assigned specific temporal meaning, and this specification will be reflected in the analysis of the resulting formations. Most common is the specification of past time (Section 1.30.1), but one also often finds present time assigned to participles that usually refer to past time (Section 1.30.2), and some suffixes have future meaning (Section 1.30.3).

### 1.30.1 bhūte

In addition to the various past participles, many other words are formed with krt suffixes expressly assigned past meaning by Pāṇini; the specification  $bh\bar{u}te$  ("in the sense of what is past") applies throughout a large segment of Pāṇini's rules for krt suffixes (3.2.84–3.2.122) and appears sporadically elsewhere as well. Formations to which this specification applies are conventionally glossed through the use of past participles (either active or passive) rather being analyzed through the use of of finite verbs. In such glosses the particle *iti* is not necessary, since the participle, by virtue of its own nominal character, can be substituted directly for the noun being explained. For the same reason it is possible to indicate the case of the word being glossed by the case of the participle used, so that the formation need not be exhibited in the nominative case, as is necessary with formulas of analysis using finite verbs and *iti*.

Ragh.M. 1.44 yajvanām vidhinêstavatām Of those who have sacrificed of those who have offered sacrifice in accordance with the rules

Kum.M. 2.46 yajvabhih vidhivad istavadbhih By those who have sacrificed by those who have offered sacrifice in accordance with the rules

Here the past active participle is used. For further examples and for examples of past passive participles in the explanation of krt formations, see Section 1.32.4.

#### 1.30.2 vartamāne

Specification of present meaning is necessary only for formations that by more general rules would otherwise have past or future meaning. By far the most common instance is the assignment of present meaning to the "past passive participle." When this occurs the commentator will either use the usual formula for the analysis of a *krt* suffix used *karmani* (Section 1.29.2), or will gloss the item with a .

Ragh.M. 5.2 śrūyata iti śrutam vedaśāstram it is heard, thus śruta, the Vedic scripture

1.31. Other restrictions of meaning

Kum.M. 1.58 arcitam ... pūjyamānam worshiped ... being worshiped

### 1.30.3 bhavisyati

Formations made with *krt* suffixes used with future sense, like those to which present sense is expressly assigned, may either be analyzed with a finite future verb and *iti*, or glossed with a future participle.

Kum V. 1.49 [1.50] bhavisyatîti bhavitrī she will be, thus future

Ragh.M. 5.32 samprasthitah prasthāsyamānah san going forth being about to go forth

# **1.31** Other restrictions of meaning

In addition to the temporal specifications just described, there are other conditions of meaning  $(up\bar{a}dhi)$  that are applied to various krt suffixes by Pāņini. Often the condition amounts to a restriction on which suffixes may be used with a particular root in particular meanings. For example, the suffixes mentioned in Pā. 3.1.145–147 carry the restriction *silpini*, "when an artisan is being referred to," and as a result words such as *nartaka*, "dancer," and *gāthaka* or *gāyana*, "singer," should be used only when speaking of professional dancers and singers, and not when speaking of just anyone who happens to dance or sing. Such restrictions of meaning are not usually reflected in the analyses offered by commentators; although they may often refer to the rule that lays down such a restriction, they will generally be content with the usual formula of analysis—*gāyatîti gāyanaḥ*, etc.

Some more general conditions of meaning, however, are regularly incorporated in the formulas of analysis used by commentators. By far the most frequently encountered of these is the specification *tac-chīla* (i.e., *tatsīla*), "having that (action) as one's nature or habit, characterized by that (activity)," a condition that applies throughout a lengthy sequence of rules in Pāṇini (3.2.134–3.2.179) and in some other rules as well. For example, 58

### Chapter 6. Primary Nominal Derivation

the word  $satya-vad\bar{i}$ , "truth-speaking" (an *upapada* compound; see Section 1.32), may be used accurately in reference to a man who habitually or characteristically speaks the truth, even if he does not happen to be speaking at the moment, but it may not be used accurately in referring to an habitual liar, even if he happens to be speaking the truth at the moment.

In explaining formations involving this condition of meaning, some commentators simply supply Pāṇini's specification *tac-chīla* after a finite verb giving the general meaning of the formation:

Ragh.H. 2.3 dayāluh dayate tac-chīlah compassionate—he has compassion, one who is characterized by that

Much more frequently, commentators follow the model of  $P\bar{a}nini's$  specification by glossing the formation with a word in which  $-s\bar{z}la$  is placed in compound with a verbal noun denoting the action in question.

Ragh.M 4.85 jisnur jaya-śīlah victorious characterized by victory

Ragh.M. 2.72 sahisnuh ... sahana-silah forbearing ... characterized by forbearance

Kum.M. 1.23 sthiti-śīlāḥ sthāvarāḥ śaila-vṛkṣîdayaḥ characterized by standing still—stationary things, mountains, trees, etc.

Slightly different formulas may be used when the krt formation being analyzed occurs at the end of a compound (Section 1.32.5).

# 1.32 Accompanying words (upapada)

Many krt formations occur only at the end of nominal compounds when expressing a given sense, and thus cannot be used in that sense without a subordinate word (*upapada*) accompanying them as the first member of the compound. To take as an example the formation that has provided the name for the entire class of primary compounds, the word krt itself, formed from the root kr, "to do, to make," occurs at the end of compounds in the sense of "who has done or made (something)"—e.g., *mantra-krt*, "one who has made mantras." The word krt is not available in this meaning outside such compounds, although it does appear as an uncompounded word in another meaning (viz., as the technical term for suffixes of primary derivation). It is therefore not possible to analyze the compound as a straightforward *tatpuruşa* compound using the same words that make up the compound, since one cannot say \**mantrānām krt*, as one could say, for example, *mantrānām sraṣtā* in analyzing the genitive tatpuruşa compound *mantra-sraṣtā*. Instead the analysis of an *upapada* compound is usually achieved by analyzing the *krt* formation at the end of the compound with the standard formulas used for uncompounded *krt* formations and exhibiting the *upapada* separately in the case that indicates its relationship to the action expressed by the verbal form in that formula, rather than its relationship to a noun that expresses the sense of the final member of the compound:

# Ragh.M. 1.61 mantrān krtavān mantra-krt One who has made mantras—a mantra-maker

When, however, a compound ends with a krt formation for which the rules of Pānini provide a derivation as an independent word and do not restrict it to use in compound with an *upapada*, we have a regular *tatpurusa* compound, and a commentator may thus analyze the *krt* formation in the usual way outside the compound, and then analyze the compound itself as a straightforward *tatpurusa*:

Ragh.M. 2.3 dharantīti dharāḥ.... payasām dharāḥ payo-dharāḥ stanāḥ they bear (something), thus bearers.... bearers of milk—milk-bearers, breasts

Occasionally it may not be clear which interpretation of a given compound is more appropriate:

Kum.M. 1.20 jānātîti jñah.... vedanānām jñah [vedanā-jñah]. athavä vedanām jānātîti vedanā-jñah. He knows, thus a knower.... a knower of pain [a pain-knower]. Or: he knows pain, thus a pain-knower.

For *upapada* compounds in which the *krt* suffix is used *bhāve*, i.e., in the sense of the verbal action, the only way of analyzing the compound in non-technical language is to substitute a freestanding synonym for the

# 1.32. Accompanying words (upapada)

#### Chapter 6. Primary Nominal Derivation

krt formation and proceed with the standard analysis for a *tatpurusa* compound (usually a genitive *tatpurusa*, with the first member displayed in the objective genitive case in the analysis), just as the substitution of a synonym is the usual way of explaining uncompounded *bhāve krt* formations (Section 1.28). For example, Pā. 3.1.108 prescribes the formation of the noun *hatyā*, "the action of killing," from the verbal root *han*, provided that the form is accompanied in compound by an *upapada*. The compound *brahma-hatyā* means "the killing of a brahmin," and is analyzed as a straightforward genitive *tatpurusa* compound—*brahmano hananam* with the substitution of the verbal noun *hananam* for *hatyā*. Because *hatyā* cannot occur without an *upapada*, the analysis \**brahmano hatyā* is impossible.

Most *upapada* compounds, however, do not involve *bhāve krt* formations as their final member, and are therefore analyzed with formulas using finite verbs. The most common situation is a *kartari krt* formation as the final member (i.e., a formation referring to the agent, Section 1.29.1), with an *upapada* appearing in the accusative case in the analysis (Section 1.32.1), but the *upapada* may also appear in some other case (Section 1.32.2), and many other compounds have a *krt* formation in the sense of some other *kāraka* as the final member (Section 1.32.3). *Krt* formations that appear at the end of *upapada* compounds, like those functioning as freestanding words, may involve temporal restrictions (Section 1.32.4) or other conditions of meaning (Section 1.32.5), which will sometimes require formulas of analysis that differ slightly from those used in analyzing uncompounded *krt* formations.

### 1.32.1 kartari krt with accusative

Most *upapada* compounds have as their final member a *krt* formation referring to the agent of the action expressed by the verbal root to which the *krt* suffix is added, and as their first member a word referring to the direct object of that action. These compounds are analyzed by displaying the first member in the accusative case, and using the standard *kartari krt* formula (Section 1.29.1) in analyzing the final member.

Ragh.M. 2.71 hutam aśnātîti hutîśo 'gnih it eats the oblation, thus the oblation-eater, fire

Ragh.M. 1.85 mahīm kşiyate īsta iti mahī-kşit. tasya mahī-kşitah He rules—controls—the earth, thus earth-ruler. Of him, of the earth-ruler.

Kum. V. 5.22 mahāntam utkrstam arhatīti mahîrhā bahudhanā it deserves a great (price)—a high (price), thus costly, worth much money

As in the analysis of uncompounded forms, relative pronouns may be used to avoid the use of *iti*. Compare the following two analyses:

Kum.M. 2.51 senām nayatīti senā-nīš camū-patiķ... tam senā-nyam. He leads an army, thus army-leader, general ... [we wish to create] him, [we wish to create] an army-leader.

Kum.V. 2.51 senām nayati yas tam senā-nyam who leads an army, him—an army leader

### 1.32.2 kartari krt with non-accusative

When the first member of the *upapada* compound does not refer to the direct object of the verbal action involved in the second member, it will appear in the analysis in some case other than the accusative.

Kum.M. 1.50 kāmena svêcchayā caratîti kāma-carah he roams at will—at his own desire—thus roaming-at-will

Kum.M. 3.37 pankāj jāyata iti panka-jam it is born from the mud, thus mud-born

Ragh.M. 4.72 guhāsu śerata iti guhā-śayāh they lie in caves, thus cave-lying

This class of compounds includes many in which the first member retains its case ending within the compound, a phenomenon that does not affect the basic formula of analysis:

Ragh.M. 5.72 stambe ramanta iti stambe-ramāh hastinah they delight in a clump of grass, thus stambe-ramas, elephants

Kum.M. 1.10 vane carantîti vane-carāh kirātāh they roam in the forest, thus forest-roamers, mountain tribesmen

1.32. Accompanying words (upapada)

62

# Chapter 6. Primary Nominal Derivation

# 1.32.3 krt in the sense of another kāraka

When the krt formation appearing as the final member of an *upapada* compound is passive, the standard formula for the analysis of a *karmani krt* formation (Section 1.29.2) is used, and the *upapada* is exhibited in the case needed to express its connection with the verbal action. In the following example the *upapada* refers to the instrument, and thus appears in the instrumental case.

Kum. V. 1.10 tailena pūryanta iti taila-pūrāh they are filled with oil, thus oil-filled

The krt formation may have the sense of any of the other kārakas as well, and for each the standard formula of analysis described for uncompounded krt formations (Section 1.29) will be used, with the upapada appearing in whatever case is appropriate to connect it with the verbal action. In the following example, the krt suffix forming the final member of the compound is used karane, in the sense of the instrument (Section 1.29.3), and the upapada refers to the object of the action. Since a passive verb is used in the analysis, the upapada shows up in the nominative case.

Ragh.M. 3.10 pumān sūyate 'nenêti pumsavanam a male is produced by means of this, thus a male-producing (rite)

# 1.32.4 Temporal restrictions

As in uncompounded kpt formations, restrictions of the time referred to may apply, the most common instance being restriction to past time (Section 1.30.1). These are ordinarily glossed with a past active participle, with which the particle *iti* is not strictly necessary, and may be either included or omitted.

Ragh.M. 3.62 vrtram hatavān iti vrtra- $h\bar{a}$  he has slain Vrtra, thus Vrtra-slayer

Kum.V. 6.3 smaram säsitavän smara-säsanah who has chastised Smara—the Smara-chastiser

Kum. V. 1.19 paksāms chinnavān iti **paksa-cchit** he has clipped the wings, thus **wing-clipper** 

Ragh.M. 1.23 pāra-drśvanaḥ pāram antam drṣṭavataḥ limit-seers who have seen the limit, the extremity

The past passive participle may appear instead, when the verb is passive or has a "past passive participle" with active meaning. As in other *upapada* compounds, in the analysis of compounds involving intransitive *krt* formations the *upapada* may appear in some case other than the accusative.

Ragh.M. 5.69 sarasi jātah sarasi-jah kamalah born in the water-water-born lotus

Ragh.M. 1.31 magadha-vamśe jātā magadha-vamśa-jā born in the Magadha dynasty---Magadha-dynasty-born

Ragh.M. 1.85 tīrthîbhişekeņa jātām tīrthîbhişeka-jām produced by bathing in the sacred ford—sacred-ford-bathing-produced

### 1.32.5 Other conditions of meaning

As with uncompounded krt formations, the most common condition of meaning applied to krt formations in *upapada* compounds is that of  $t\bar{a}cch\bar{i}lya$ , "having that as one's characteristic or habitual action" (Section 1.31). However, if the compounds ending in  $-s\bar{s}la$  that are used to gloss uncompounded krt formations are to used in glossing *upapada* compounds then one must include the *upapada* as the first member of the compound gloss, or, if the *upapada* has been exhibited separately in order to make a remark, one must either include the *upapada* in the compound gloss or use *tat*- as a substitute for it there:

Ragh.M. 1.3 kavi-yaśah-prārthī. kavīnām yaśah kāvya-nirmāņena jātam tat-prārthanā-sīlah. Seeking the fame of poets: the fame of poets, produced by the making of poetry--characterized by the seeking of it.

More commonly, such *upapada* compounds are analyzed by using the infinitive of the verb in question, with the *upapada* as a separate object in the accusative, followed by *sīlam asya*, "whose nature it is to ...," and *iti*.

# 1.34. Irregular forms (nipātana)

#### 64

#### Chapter 6. Primary Nominal Derivation

Ragh.M. 1.43 sva-niḥśvāsam anukartum śīlam asyêti sva-niḥśvāsânukāriṇam whose nature it is to imitate their own breath, thus imitating their own breath

Ragh.M. 1.49 samidhaś ca kuśāmś ca phalāni cîhartum śīlam yeṣām iti samit-kuśa-phalîharāh, taih samit-kuśa-phalîharaih whose nature it is to bring fuel, kuśa grass, and fruits---bringers of fuel, kuśa grass, and fruits; by them, by bringers of fuel, kuśa grass, and fruit

Other conditions of meaning are analyzed with formulas dependent upon the specification of meaning provided in the rules of Pāṇini. For example, the suffix *-in* that is prescribed with the condition *tācchīlye* for compounds such as *svanihśvāsânukārin* in the example given above (by Pā. 3.2.78) is prescribed in the sense of the agent with the condition *kartary upamāne*, "when (the *upapada*) is (another) agent serving as a standard of comparison" for some other compounds (by Pā. 3.2.79). Such compounds will be glossed by exhibiting the *upapada* in the nominative case, followed by *iva* and the standard formula for the analysis of a *kartari krt* formation (Section 1.29.1):

Ragh.M 4.4 dvirada iva ... gacchatîti dvirada- $g\bar{a}m\bar{i}$  he walks ... like an elephant, thus walking like an elephant

By a subsequent rule (Pā. 3.2.83), the same suffix can be added to the root *man*, "to consider," in the sense of an agent when accompanied by an *upapada*, with the condition of meaning  $\bar{a}tma-m\bar{a}ne$  (considering oneself to be so). Accordingly, the formula of analysis for such compounds adds the object  $\bar{a}tm\bar{a}nam$ :

Kāsikā 3.2.83 darsanīyam ātmānam manyate ... darsanīya-mānī he considers himself good-looking ... (thus,) considering himself good-looking

# 1.33 Verbal participles

The category of *krt* formations includes verbal participles, which are treated in several ways by commentators.

Most participles are simply glossed with participles of the same form from a root of equivalent meaning.

Ragh.M. 12.74 samsatī bruvāņā indicating telling

Ragh.M. 11.10 uhyamāna iva vāhanena prāpyamāna iva as if being carried along as if being transported by a vehicle

The same procedure is followed in general for a past passive participle, but other synonyms of different form are chosen if the participle is used with active meaning, with present meaning, or in the sense of the verbal action itself ( $bh\bar{a}ve$ ).

Ragh.M. 3.5ādṛta ādṛtavānwith respect having felt respectRagh.M. 12.75lajjitā lajjāvatīembarrassed with embarrassmentKum.M. 4.17upagūdhāni ālinganāniembraces embracings

Gerundives are analyzed with several formulas designed to bring out the specific meaning of the participle in the context, most often by using other gerundives together with infinitive forms, but sometimes with compounds beginning with verbal nouns.

Ragh.M. 1.71 asahyā sodhum aśakyā unbearable unable to be borne

Ragh.M. 1.86 yāyayitum yogyam yājyam worthy of sponsoring a sacrifice—fit for sacrificing

Ragh.M. 4.6 stutyam stotrârham praiseworthy deserving praise

Ragh.M. 1.3 upahāsyatām upahāsa-viṣayatām being laughable being an object of laughter

# **1.34** Irregular forms (*nipātana*)

Formations that diverge from the general rules of the class to which they belong, either in their meaning or in their form, are often singled out by the grammarians for an explicit individual specification of the formation

### Chapter 6. Primary Nominal Derivation

as irregular (*nipātana*). Such words are said to be *nipātita*, "set down.". Commentators frequently mention the rules specifying these forms, and explain them either with simple synonyms or with a formula of analysis based on the meaning of the form rather than on the general formula for the class to which the form belongs. The following are examples of irregular gerundives:

Kum. V. 3.18 **bhavya eva** bhavitâiva. ... bhavya-sabdah kartari nipātitah. **is surely to be** will certainly be. ... The word bhavya is set down as an irregular form (by Pāṇini 3.4.68) in the sense of the agent (rather than in the sense of the object, as one would expect).

Ragh.M. 11.8 bhinatti kūlam iti bhidyah. "bhidyôddyau nade" (Pāṇini 3.1.115) iti kyab-antau nipātitau. It bursts its bank, thus a rushing river. By Pāṇini 3.1.115 (the words bhidya and uddya) are set down as irregular gerundive forms (with active meaning, rather than passive meaning as one would expect).

Kum.M. 3.57 an-avadyam a-garhyam. "avadya-paṇya-" iti nipātaḥ. uncensurable not deserving of censure. By Pāṇini 3.1.101 (the word avadya) is set down as an irregular form (in the meaning "censurable," although the regular form in other senses of the word is anudya).

# Chapter 7

# **Secondary Nominal Derivation**

# 1.35 Introduction

The formations called *taddhita* or *taddhitânta* ("ending in *taddhita* suffixes," Pā. 4.1.76–5.4.160; "secondary derivatives," Whitney 1202–1245) are the nominal stems derived through the addition of suffixes to other nominal stems. In general terms the category of *taddhita* suffixes includes all suffixes added directly to nominal stems with the exception of the suffixes (which convert nominal stems to the feminine gender) and the *sup* suffixes (the inflectional endings of nouns).

 $\mathcal{C}_{\mathcal{C}}$ 

Most of the formulas used to analyze *taddhita* formations are taken directly from Pāṇini's specifications of the meanings for each suffix. In Pāṇini's rules the nominal stem to which a suffix may be added is generally represented by the pronoun *tat* or an equivalent, and the analysis of any particular formation simply substitutes the noun that serves as the base in that formation for the form of *tat* prescribed by Pāṇini. To take as an example the meaning that has given the *taddhita* class of suffixes its name, Pā. 5.1.5 states "*tasmai hitam*," "good for that, beneficial to that," and any formation derived by the addition of a suffix added to a base to produce a word with this meaning can be analyzed by exhibiting the base in the dative case, followed by the word "*hita*":

Kāśikā 5.1.5 vatsebhyo hito (go-dhuk) vatsīyah beneficial to calves—(a dairyman) good to calves

#### Chapter 7. Secondary Nominal Derivation

#### Kāśikā 5.1.10 sarvasmai hitam sārvam beneficial to all-good for all

Formulas of this type may be called dependent formulas of analysis (Section 1.36). Other formulas that follow the specifications of Pāņini in precisely the same way are those in which the base to which the suffix is added is exhibited in the genitive case (Section 1.37), in the nominative case (Section 1.38), or as the first member of a nominal compound (Section 1.39). Important subdivisions of *taddhita* suffixes are the possessive suffixes (*matvarthīya*, Section 1.40), the suffixes that have the force of particular case endings (*vibhaktisamjñaka*, Section 1.41), and the suffixes that produce formations referring to the same object as the base to which the suffix is added (*svârthika*, Section 1.42).

As in other categories of grammatical complexes (cf. Sections 1.20, 1.34), some particular *taddhita* formations are individually specified by Pāņini as exceptional forms (*nipāta*), rather than being derived through more general rules. Such specification is called *nipātana* by the commentators, and the words so specified are said to be *nipātita*. Typically they involve either unusual forms or, more often, very specific meanings, that could not be known from the general rules for the suffixes with which they are formed. For example, the word *varņin*, although made by adding the possessive suffix *-in* (called *inI* by Pāņini) to the noun *varņa*, has the very specific meaning of a member of the first *varņa* or social class, i.e., a *brahmacārin* or student. This meaning is specified in Pā. 5.2.134, "After *varņa* in the sense of a *brahmacārin*," in the section of his rules dealing with this suffix.

Ragh.M. 5.19 varnī brahmacāri. "varņino brahmacāriņah" ity amarah. "varnād brahmacāriņi" itīni-pratyayah. A varņin a brahmacārin. Amara says "varņins are brahmacārins." The suffix inI by (Pā. 5.2.134) "After varņa in the sense of a brahmacārin."

# **1.36** Dependent formulas of analysis

In these the suffix is treated as though it were the second member of a dependent *tatpurusa* compound. Thus in the analysis the word to which the suffix is added is placed in whatever case is required by its relation to the word expressing the meaning of the suffix:

### 1.36. Dependent formulas of analysis

(instrumental)

kulālena krtam—kaulālakam "made by a potter" dadnā samskrtam—dādhikam "prepared with yogurt" nāvā tāryam—nāvyam "crossable by boat" (dative)

ātmane hitah-ātmanīnah "good for oneself"

gave hitam-gavyam "good for a cow"

(ablative)

nyāyād anapetam—**nyāyyam** "not deviating from the law" manor jātah—**manuşyaḥ** "born from Manu, human"

vidusah īşan-nyūnah—vidvatkalpah "slightly less than learned, i.e., fairly learned" (cf. Section 1.42.3), Section 2.19.3

(genitive)

sisor bhāvaḥ--saisavam "the state of a child, childhood"
pṛthor bhāvaḥ--pārthavam "the state of being wide, wideness"
kāṣthasya vikāraḥ--kāṣṭhamayam modification of wood, wooden
gavāṃ samūhaḥ--gotrā collection of cows, herd of cattle"
mātuḥ bhrātā-māturaḥ "mother's brother, materal uncle"
rājñāṃ yogyaṃ--rājanyakam "appropriate for kings"
(note the use of ayam in the following genitive examples:)
devasyâyam--daivaḥ "this is a god's, belonging to a god"
janasyâyam--janakīyaḥ "this is the people's, belonging to the people"

70

#### (locative)

atithisu sādhuh—ātitheyah "good toward guests, hospitable"
sabhāyām sādhuh-sabhyah "good in the court, polite"
ustre bhavah-austrakah "relating to a camel"
grisme bhavah-graismakah "produced in summer"
kurusu jātah—kauravakaķ "born among the Kurus"

#### 1.36.1 Accusative

When the word to which the suffix is added is displayed in the accusative case in the analysis, the formula used is like that employed in analyzing *upapada* compounds (Sections 1.32, 1.59):

vyākaraņam adhīta iti vaiyākaraņaņ "he studies grammar, thus grammarian"

dharmam caratîti dhārmikaņ "he practises righteousness, thus righteous"

And this formula may be used with the first element in other cases as well:

udupena taratîty audupikah "he crosses by boat, thus boatman"

Examples from the commentaries:

Ragh.M. 1.25 daṇḍam arhantîti daṇḍyāḥ. "daṇḍâdibhyo yat" iti ya-pratyayaḥ.... tān daṇḍyān They deserve punishment, thus punishable. The suffix -ya by (Pāṇini 5.1.36) "(The suffix) yaT after daṇḍa, etc." .... Those criminals.

# 1.36.2 Instrumental

Where the other oblique cases are involved, the formulas are akin to those used in analyzing case-*tatpurusas*.

Kum.M. 3.64 patangena tulyam **patangavat** śalabhavat... "tena tulyam kriyā ced vatih" iti vati-pratyayah. equivalent to a moth **moth-like** like a moth... The suffix vatI by (Pāṇini 5.1.115) "There is (the suffix) vatI in the meaning of 'like this,' if an action is referred to."

Ragh.M. 4.31 nāvyā naubhis tāryā nadīķ navigable crossable by boats rivers

### 1.36.3 Dative

Kir.M. 13.69 ātmane hitam ātmanīnam.

*"ātma-viśva-jana-bhogôttarapadāt khah."* beneficial to oneself **good for oneself**. By (Pā. 5.1.9) *"*(The suffix) *kha* (i.e., *-īna*) after *ātman*, *viśva*, *jana*, and a word ending in *bhoga."* 

Śiśu.M. 1.41 viśvasmai janāya hitam viśvajanīnam.

" $\bar{a}tma$ -viśva-jana-bhogôttarapadāt khaḥ." beneficial for all people good for all people. By (Pā. 5.1.9) "(The suffix) kha (i.e., - $\bar{n}a$ ) after  $\bar{a}tman$ , viśva, jana, and a word ending in bhoga."

Śiśu.M. 14.4 he sārva sarvahitatvāt sārvah, tatsambodhane. "sarva-puruṣābhyām ṇa-dhañau." O (you who are) good for all good-for-all because of being beneficial for all; (the vocative) in addressing that. (The suffix is by Pā. 5.1.10) "After sarva and puruṣa there are (the suffixes) Na and  $dha\tilde{N}$  (i.e., -a and -eya, respectively)."

#### 1.36.4 Ablative

Kum.M. 1.26 abhijanād āgatena ābhijanena come from a good family of good family

Kum.V. 5.29 [M 30] brahmana āgatam brahmamayam, tena [brahmamayena] vedotthena tejasā coming from brahma brahmanic, with that [brahmanic]—arising from the Vedas—splendor

71

#### Chapter 7. Secondary Nominal Derivation

Ragh.M. 4.4 pitur āgatam pitryam. "pitur yat" iti yat-pratyayah. simhāsanam coming from the father paternal—the suffix yaT (i.e., -ya) by (Pā. 4.3.79) "yaT after pitr (in the sense of coming from that")—throne.

Ragh.M. 5.9 janapadebhya āgatair jānapadaiķ. "tata āgata" ity aņ. kadangarīyaiķ with country from the country—the suffix aN by (Pā. 4.3.74) "come from that"—straw

Ragh.M. 1.60 daivīnām devebhya āgatānām durbhikṣâdīnām, mānuṣīnām manuṣyebhya āgatānām caura-bhayâdīnām. ubhayatrâpi "tata āgatah" ity aņ. "tid-dhân-añ-"ity-ādinā nīp. āpadām of disasters (both) divine coming from the gods, such as famine, and human coming from humans, such as the danger of theft. In both there is the suffix aN by (Pā. 4.3.74) "come from that." There is (the feminine suffix)  $N\overline{n}P$  (i.e.,  $-\overline{i}$ ) by (Pā. 4.1.15).

Kum.V. 2.3 arthād anapetā acalitā **arthyā** anadhyāropitârthāh. "dharma-pathy-artha-nyāyād anapete." not deviating—not moving—from the facts **factual** without hyperbole. (The suffix -ya is added by Pā. 4.4.92) "After dharma, pathin, artha, and nyāya in the sense of not deviating."

Kum.M. 2.2 arthād anapetābhiḥ arthyābhiḥ. artha-yuktābhir ity arthaḥ. "dharma-pathy-artha-nyāyād anapete" iti yat-pratyayaḥ. not deviating from meaning meaningful provided with meaning. (The suffix -ya is added by Pā. 4.4.92) "After dharma, pathin, artha, and nyāya in the sense of not deviating."

Ragh.M. 2.55 **nyāyyā** nyāyā anapetā. yuktêty arthaḥ. dharma-pathy-arthâdinā yat-pratyayaḥ. **lawful** not deviating from the law; i.e., proper. The suffix yaT by (Pā. 4.4.92.) "After dharma, pathin, artha, ...."

#### 1.36.5 Locative

Perhaps the most common locative formula of analysis is *tatra bhavam*, "occurring in that, pertaining to that."

Ragh.M. 1.20 " $pr\bar{a}k$ " ity avyayena p $\bar{u}rva$ -janmôcyate. tatra bhavāh p $r\bar{a}ktan\bar{a}h$ . " $s\bar{a}yam$ -ciram-"ity- $\bar{a}din\bar{a}$  tyul-pratyayah. By the indeclinable  $pr\bar{a}k$  ("before") a former lifetime is referred to. Pertaining to that former. The suffix tyuL (i.e., -ana, before which -t- is inserted) by (Pā. 4.3.23) "after s $\bar{a}yam$ , ciram, ...."

Ragh.M. 1.36 prāvrsi bhavah prāvrseņyah. "prāvrsa eņyah" ity eņya-pratyayah. tam **prāvrseņyam**. Occurring in the monsoon—monsoonal. The suffix -eņya by (Pā. 4.3.17) "eņya after prāvrsa (in the sense of occurring in that)." That **monsoonal**.

Ragh.M. 1.88 vane bhavam vanyam kanda-mūlâdikam occurring in the forest forest (food), such as bulbs and roots.

Kum.M. 3.23 haimavatam himavati bhavam. Himalayan occurring in the Himalaya.

When the relevant rule of Pāṇini specifies any special condition of meaning (called an  $up\bar{a}dhi$ ), the commentator will include that specification in analyzing the formation. Here, for example, are two refinements on the *tatra bhavam* formula:

Ragh.M. 2.70 **prāsthānikam** prasthāna-kāle bhavam. tat-kālôcitam ity arthah. "kālāṭ ṭhaā" iti ṭhaā-pratyayah. **pertaining to starting a journey** occurring on the occasion of starting a journey; i.e., appropriate for that occasion. The suffix  $tha\tilde{N}$  (i.e., -ika, by Pā. 4.3.11) "(The suffix)  $tha\tilde{N}$  after (a word referring to) an occasion."

Ragh.M. 5.15  $\bar{a}ranyak\bar{a}$  aranye bhavā manuşyā muni-pramukhāh. "aranyān manuşye" iti vuñ-pratyayah. wilderness peoplehumans occurring in the wilderness, such as sages. The suffix  $vu\tilde{N}$  (i.e., -aka) by (Pā. 4.2.129) "After aranya when referring to humans."

Another common locative formula is *tatra sādhuh*, "good in that, good at that."

Ragh.M. 1.55 sabhāyām sādhavah sabhyāh. "sabhāyā yah" iti ya-pratyayah. Good at court courteous. The suffix -ya by (Pā. 4.4.105) "(The suffix) -ya after sabhā."

### Chapter 7. Secondary Nominal Derivation

Kum.M. 2.57 samyuge sādhum sāmyugīnam. "pratijanādibhyah khañ" iti khañ-pratyayah. good at battle good at battle. The suffix kha $\tilde{N}$  (i.e., -īna) by (Pā. 4.4.99) "After pratijana, etc., (in the meaning 'good at that.'"

Kum.V. 2.57 sāmyugīnam raņa-paņditam. sādhv-arthe "pratijanādibhyaḥ khañ." good at battle expert in battle. (The suffix -īna) in the sense of "good (at that)" by (Pā. 4.4.99) "After pratijana, etc., (the suffix)  $kha\tilde{N}$  (in the meaning 'good at that')."

Ragh.M. 2.30 śarane sādhuḥ śaranyaḥ. "tatra sādhuḥ" iti yat-pratyayaḥ. Good at refuge **providing good refuge**. The suffix yaT (i.e., -ya) by (Pā. 4.4.98) "Good at that."

Another locative formula is based on the meaning of enrollment or enlistment:

Ragh.M. 4.45 senāyām samavetāh sainyāh. ... "senāyā vā" ity nya-pratyayah. Enrolled in the army soldiers. The suffix Nya (i.e., -ya) by Pā. 4.4.45 senāyā vā, "After senā there is optionally (the suffix Nya[i.e., -ya] rather than thaK [i.e., -ika] in the meaning of enrollment)."

Ragh.M. 4.74 sainikāh senāyām samavetāh prāgvahatīyas thak-pratyayah soldiers enrolled in the army. The suffix thaK (i.e., -ika) from the section extending (from Pā. 4.4.1) up to "vahati" (in Pā. 4.4.76).

# 1.37 Genitive formulas of analysis

The single largest topic in the section on *taddhita* suffixes in Pānini is that of patronymics and related formations referring to descent. These are analyzed with a formula placing the base in the genitive case: *tasyâpatyam*, "offspring of his" (Section 1.37.1). Genitive formulas are also used in analyzing derivatives with abstract meaning (Section 1.37.2), and formations with many other meanings as well (Section 1.37.3), as well as formations expressing a general connection (Section 1.37.4).

#### 1.37.1 tasyâpatyam

Ragh.M. 1.1 parvatasyâpatyam pārvatī. "tasyāpatyam" ity aņ, "tid-dhân-añ-" ity-ādinā nīp. offspring of the mountain (parvata) **Pārvatī**. (The suffix) aŅ (i.e., -a by (Pā. 4.1.92) "Offspring of that"; (the feminine suffix)  $N\overline{P}$  (i.e.,  $-\overline{i}$ ) by (Pā. 4.1.15).

Ragh.M. 3.5 magadhasya rājāo 'patyam strī māgadhī. "dvyaā-magadha-kalinga-sūramasād aņ" ity aņ-pratyayah. Female offspring of the king Magadha Māgadhī. The suffix aN (i.e., -a) by (Pā. 4.1.170) "(The suffix) aN after names of two syllables and after Magadha, Kalinga, and Sūramasa."

Ragh.M. 4.87 rājā o' patyāni rājanyāh kṣatriyās tān**rājānyān** "rāja-śvaśurād yat" ity apatyârthe yat-pratyayah. Offspring of aking—royal men, kṣatriyas—them**royal men**. The suffix yaT (i.e., -ya) inthe meaning of offspring by (Pā. 4.1.137) "(The suffix) yaT after rājanand śvaśura."

#### 1.37.2 tasya bhāvah

Ragh.M. 1.8 śiśor bhāvah śaiśavam bālyam.

"prāņabhrjjāti-vayovacanôdgātrâdibhyo 'ñ" ity añ-pratyayah. The state of an infant **infancy**. The suffix  $a\tilde{N}$  (i.e., -a) by (Pā. 5.1.129) "(The suffix)  $a\tilde{N}$  after names of animal species, words for stages of life, and udgātr, etc."

Ragh.M. 1.8 yūno bhāvo yauvanam tāruņyam. yuvâditvād an-pratyayah. tasmin vayasi [yauvane]. The state of a young man—youth, adolescence. The suffix aN (i.e., -a) because of belonging to the yuvâdi group (mentioned in Pā. 5.1.130). In that stage of life in youth.

Kum.V. 1.4 dhätavo 'tra sindūrâdayo raktāh, te vidyante yasya sa dhātumān; tad-bhāvo dhātumattā, tām [dhātumattām]. Minerals here are red ones such as red lead. That which has them is mineral-rich. The state of being that is mineral-richness. That mineral-richness.

Kum.M. 2.11 prakāmasya bhāvah prākāmyam yathā-kāmatvam. The condition of being willful willfulness, doing as one wishes.

#### 1.37.3 Others

Similar genitive formulas are used with many other specific meanings. For example:

#### activity (tasya karma):

Kum.M. 2.35 stenasya bhāvah karma vā steyam cauryam. "stenād yan nalopas ca iti yat-pratyayah na-lopas ca." The condition or activity of a thief **theft**, stealing. By (Pā. 5.1.125) "After stena there is (the suffix) yaT (i.e., -ya) and the dropping of na."

Kum. V. 4.9 kitavasya karma kaitavam The activity of a deceiver deception.

Ragh.M. 4.1  $r\bar{a}jyam$ ,  $r\bar{a}j\bar{n}ah$ , karma prajā-paripālanâtmakam. purohitāditvād yak. rule the activity of a ruler, consisting of protecting his subjectsrule the activity of a ruler, consisting of protecting his subjects. (The suffix) yaK (i.e., -ya) because of belonging to the purohitâdi group (mentioned in Pā. 5.1.128).

modification, product (tasya vikārah):

Ragh.M. 2.29 dhātor gairikasya vikāro dhātumayī tasyām [dhātumayyām] adityakāyām ūrdhva-bhūmau. a modification of a mineral, of red chalk—made of mineral, on it on the highland the upland made of mineral.

Ragh.M. 5.2 hiranyasya vikārah hiranmayam. dāndināyana-"ity-ādinā nipātah A modification of gold made of gold. An exceptional form (see Sections 1.20, 1.35), by (Pā. 6.4.174) "dāndināyana ...."

ruler (tasya rājā, etc.):

Ragh.M. 4.40 kālingah kalingānām rājā.

" $dvya\overline{n}$ -magadha-kalinga-"ity- $\overline{a}din\overline{a}$  an-pratyayah. The Kālinga king of the Kalingas. The suffix aN (i.e., -a) by (Pa. 4.1.170) "(The suffix) aNafter names of two syllables and after Magadha, Kalinga, and Sūramasa."

Ragh.M. 2.20 **pārthivena** prthivyā īśvareņa. "tasyeśvaraḥ" ity añ-pratyayaḥ. **By the ruler** by the lord of the earth. The suffix aN by (Pā. 5.1.42) "Lord of that." collection (tasya samūhah):

Kum.V. 7.57  $m\bar{a}l\bar{a}n\bar{a}m$  samūho  $m\bar{a}lyam$ . " $p\bar{a}s\hat{a}d\bar{a}bhyo$  yah" ity anena yah. A collection of garlands garland wreath. The suffix -ya by (Pā. 4.2.49) "ya after  $p\bar{a}sa$ , etc. (in the sense of a collection of that)."

### 1.37.4 General connection

In analyzing formations expressing general connection, the base is displayed in the genitive case and a form of the pronoun *idam* is added: *tasyêdam*, "this belongs to it, this is connected with it."

Ragh.M. 5.75 gangāyā idam gāngam this is of the Ganges Gangetic

Kum.M. 1.41 tasyā imau tadīyau bāhū These two are hers her two arms

Kum.M. 2.1 svayam-bhuvo brahmana idam svāyambhuvam This belongs to the Self-born—Brahmā of the Self-born

Kum. V. 2.1 svayam-bhūr brahmā tasyêdam svāyambhuvam The Self-born is Brahmā; this is his of the Self-born

Occasionally the pronoun *idam* is omitted and a form of the word *sambandhin*, "connected to," is used instead.

Ragh.M. 4.35 vaitasīm vetasah sambandhinīm vrttim. pranatim ity arthah. The behavior associated with a reed of a reed; i.e., bending.

Ragh.M. 1.13 kṣātraḥ kṣatra-sambandhī warriorly connected with a warrior

# 1.38 Nominative formulas of analysis

In these the base to which the suffix is added appears in the nominative case together with a further specification of meaning drawn from Pānini's rules, and the object referred to by the resulting formation is indicated by a pronoun in the genitive case. The genitive pronoun may be either a form of *idam* or a relative pronoun picked up by some form of *tat*. These formulas are thus akin to those used in analyzing *bahuvrīhi* compounds.

Kum.M. 4.18 *rtur asya prāptah* **ārtavaņ** vāsantam. "*rtor aņ*" *ity an-pratyayah* Its season has arrived **seasonal** vernal. The suffix *aN* by ( $P\bar{a}$ . 5.1.105) "(The suffix) *aN* after *rtu* (in the sense of having arrived for this)."

Kum.M. 4.27 duhkham asyāh samjātam duhkhitā. samjāta-duhkhêty arthah. tārakâditvād itac. grief has occurred for her **aggrieved** griefstricken. (The suffix) itaC (in the sense of having occurred for this) because of belonging to the tārakâdi group (mentioned in Pā. 5.2.36).

Kum. V. 2.33 [yāvan-mātrena] yat parimāņam asya yāvān. tatah svârthe mātrāc. [By just which much.] It has which as its extent—which much. After that there is (the suffix) mātraC in the sense of the base itself.

Kum. V. 6.92 vivāhah prayojanam yasyāh sā vaivāhikī That which has a wedding ceremony as its occasion is **nuptial** 

Ragh.H. 5.58 [M 5.57] gāndharvaṃ gandharvo devatā yasya tat That which has a Gandharva as its deity is Gāndharva.

Kum.V. 2.17 catvāro 'vayavā yasyāh sā catustayī. "sankhyāyā avayave tayap." That which has four parts is **four-part**. (By Pā. 5.2.42) "(The suffix) tayaP after a numeral, in the sense of (having that many) parts."

# 1.39 Compound formulas of analysis

In addition to the use of compounds ending in *-sambandhin* as a substitute for a genitive pronouns (Section 1.37), there are some meanings which are expressed through compounds to begin with in Pānini's rules, and commentators preserved those formulas in analyzing formations. The most common of these are *tad-artham* ("for the purpose of that") and *tad-arham* ("appropriate to that," "deserving that").

Ragh.M. 1.44 arghah  $p \bar{u} j \bar{a}$ -vidhih, tad-artham dravyam arghyam. " $p \bar{a} d \hat{a} rgh \bar{a} bhy \bar{a} m$  ca" iti yat-pratyayah. Honoring is the rite of hospitality, an offering for that purpose an offering of honor. The suffix yaT by (Pa. 5.4.25) "and after  $p \bar{a} d a$  and argha (there is yaT in the sense of being for that purpose)." Ragh.M. 5.2 arghârtham idam **arghyam**. " $p\bar{a}d\hat{a}rgh\bar{a}bhy\bar{a}m$  ca" iti yat-pratyayah.  $p\bar{u}j\hat{a}rtham$  dravyam This is for the purpose of honoring **an offering of honor**. The suffix yaT by (Pā. 5.4.25) "and after  $p\bar{a}da$  and argha (there is yaT in the sense of being for that purpose)." An offering for the purpose of hospitality.

Ragh.M. 5.3 vidhivad vidhy-arham. yathā-sāstram ity arthaḥ. "tadarham" iti vati-pratyayaḥ. properly appropriately to the rules; i.e., in accordance with the prescriptions. The suffix vatl by (Pā. 5.1.117) "appropriate to that."

Kum.M. 3.76 anukampitum arhām **anukampyām**. "*r*-halor nyat" iti nyat-pratyayah. Deserving to be pitied **pitiable**. The suffix NyaT by (Pā. 3.1.124) "(The suffix) NyaT to (words ending in) -*r* or a consonant."

# 1.40 Possessive suffixes (matvarthīya)

Formations made with possessive suffixes are glossed with *asya vidyate* or *asyâsti* ("of him there is, he has"), placed after the word to which the suffix is added, and followed by *iti*:

daṇḍo 'syâstîti <b>daṇḍikaḥ</b>	"he has a staff, thus <b>staff-bearer</b> "	
dhanam asyâstîti <b>dhanī</b>	"he has wealth, thus wealthy"	
gāvo 'sva santîti <b>gomän</b>	"he has cows, thus <b>cow-having</b> "	

The last word could also be analyzed *gāvo 'smin santīti gomān*, "there are cows in it, thus **cow-having**"

Similar formulas are used in conjunction with the additon of a word in the analysis to bring out the full meaning of certain other suffixes:

*ghṛtaṃ pracuram asminn iti ghṛtamayaḥ* "ghee is abundant in it, thus made (mostly) of ghee"

*jānu pramāņam asyêti jānu-dagdham* "the knee is its extent, thus **knee-deep**"

80

### Chapter 7. Secondary Nominal Derivation

vīņā-vadanam silpam asyêti vaiņikaņ "his art is vīņā-playing, thus vīņā-player"

Less commonly, any of these possessive formulas of analysis can be replaced with formulas using relative pronouns, as in the analysis of *bahu* $vr\bar{i}hi$  compounds (Sections 1.64–1.68); e.g., *ghrtam pracuram yasmin sa ghrtamayah*, "that is *ghrtamaya* in which ghee is abundant."

#### 1.40.1 Straightforward possession

Kum.V. 3.69 balam vidyate yatra tad balavat. vaso 'syâstîti vasī sāntah. That in which strength is found is strong. He who has control is self-controlled peaceful.

Kum.V. 5.10 pranayah prītir vidyate yasyāsau pranayī That which has love—affection—is a lover

Ragh.M. 1.12 śuddhir asyâstîti śuddhimān. tasmin śuddhimati. tad-anvaye tasya manor anvaye. It has purity, thus pure. In that in the pure lineage of his in the lineage—the dynasty, of him—of Manu.

Ragh.M. 1.53 āhuti-gandho yeṣām astîty āhuti-gandhinaḥ of which there is the aroma of offerings, thus having the aroma of offerings

Kum.M. 3.65 tapo 'syâstîti tapasvī. "as-māyā-medhā-srajo vinih" iti vini-pratyayah. tasmai tapasvine. He has asceticism, thus an ascetic. The suffix vinI by (Pā. 5.2.121) "(The suffix) vinI after words ending in as and after māyā, medhā, and sraj."

Ragh.M. 5.52 vāco 'sya santîti vāgmī vaktā. "vaco gminiḥ" iti gmini-pratyayaḥ. He has speeches, thus eloquent a speaker. The suffix gminI by (Pā. 5.2.124) "(The suffix) gminI after vac."

#### 1.40.2 Additions of meaning

Ragh.M. 1.91 putriņām sat-putravatām. prašamsāyām ini-pratyayah. having sons having good sons. The suffix inI expressing praise. *1.41.* Vibhaktisamjñaka suffixes

Kum.M. 1.55 sparśavatīh sukha-sparšāh. mrdvīr ity arthah. prašamsāyām matup. having (a good) feel having a pleasant feel; i.e., soft. (The possessive suffix) matUP expressing praise.

Kum.M. 1.4 dhātavaḥ sindūra-gairikâdaya 'sya santīti dhātumān. nitya-yoge matup. tasya bhāvo dhātumattā. tām [dhātumattām]. It has minerals—red lead, red chalk, etc.—thus mineral-rich. (The possessive suffix) matUP expressing constant connection. The condition of being that, mineral-richness. It mineral-richness.

Kum.M. 1.27 putrāś ca duhitaraś ca putrāh. "bhrātŗ-putrau svasŗ-duhitŗbhyām" ity ekaśeṣah. te 'sya santîti putravān. bhūmârthe matup. tasya putravato 'pi bahv-apatyasyâpîty arthah. Sons and daughters—children. An ekaśeṣa formation (see Chapter 9) by (Pā. 1.2.68) "(When) bhrātṛ and putra (come together) with svasŗ and duhitṛ (respectively, the first alone is retained)." He has them, thus having children. (The possessive suffix) matUP expressing the sense of abundance. of him although having children, i.e., although having many offspring.

# 1.41 Vibhaktisamjñaka suffixes

These are indeclinable suffixes having the meaning of a particular case ending. Most common are *-tas*, with ablative meaning, and *-tra*, with locative meaning. They are glossed by replacing them with the appropriate case ending.

Ragh.M. 1.66 mattah param mad anantaram. "pañcamyās tasil." After me subsequent to me. (By Pā. 5.3.7) "(The suffix) tasIL in place of the fifth case ending."

Kum.M. 1.46 tatah tasyāh pārvatyāh. "pañcamyās tasil." From her from that one—from Pārvatī. (By Pā. 5.3.7) "(The suffix) tasIL in place of the fifth case ending."

# 1.42 Svārthika suffixes

These suffixes are added to a base "in the sense of its own meaning," i.e., to produce a formation meaning the same thing as the base to which the suffix is added. Often these suffixes result in no change in meaning at all (Section 1.42.1), but more broadly the category also includes those suffixes that impart a some connotation, resulting in diminutives (Section 1.42.2) or related formations (Section 1.42.3) such as pejoratives and honorifics. Also included here are the suffixes allowing nouns to be used in conjunction with verbs in the senses of becoming that or making something that (Section 1.42.4), the *samasânta* suffixes (Section 1.42.5), which are added to end of nominal compounds without changing the meaning of those compounds.

### 1.42.1 With no addition of meaning

In glossing these forms a commentator will simply use the base word followed by *eva* to emphasize that the word formed with the suffix means just the same thing as the base to which the suffix was added.

Kum.V. 6.43 tama tamisram darkness darkness

Kum. V. 2.44 sampraty eva sāmpratam. prajāditvād aņ. Now now. (The suffix aŅ because of belonging to the prajādi group (mentioned in Pā. 5.4.38).

Kum. V. 2.33 [yāvan-mātreņa] yat parimāņam asya yāvān. tatah svârthe mātrāc. [By just which much.] It has which as its extent—which much. After that there is (the suffix) mātraC in the sense of the base itself.

#### 1.42.2 Diminutives

Kum. V. 5.14 alpā hrasvā vā vrksā vrksakāņ meager or short trees saplings

Ragh.M. 1.51 vrksakāh hrasva-vrksāh ... "hrasvârthe ka-pratyayah. saplings short trees. ... The suffix -ka in a diminutive sense.

### 1.42.3 Other additions of meaning

Kum. V. 3.52 atisayena bahu bhūyistham. exceedingly much mostly.

Ragh.M. 1.12 atiśayena śuddhimān śuddhimattarah.

"dvivacana-vibhajyopa-"ity-ādinā tarap-pratyayah. exceedingly pure purer. The suffix taraP by (Pā. 5.3.57) "(The suffixes taraP and  $\bar{i}yasUN$  in the sense of exceeding) when speaking of two things," etc.

Kum. V. 4.35 soma iva somyah. "sākhâdibhyo yat." Like the moon gentle. (By Pā. 5.3.103) "yaT after sākhā, etc. (in the sense of 'like')."

Ragh.M. 5.36 *īşad-asamāptaņ kumāraņ kumārakalpam* skanda-sadršam. "*īşad-asamāptau*" ity-ādinā kalpap-pratyayah. A slightly incomplete Kumāra virtually a Kumāra. The suffix kalpaP by (Pā. 5.3.67) "In the sense of slighty incomplete ...."

### 1.42.4 cvi, etc

*Cvi* is the Pāṇinian name for a particular suffix that converts nouns into items functioning as verbal prefixes before the verbs  $bh\bar{u}$  and kr in the meanings of "becoming that, not having been it before" (*abhūta-tadbhāve*, Pā. 5.4.50) and "making something that." Several other *taddhita* suffixes are used in much the same way.

Kum.V. 4.13 akathā kathā krtam kathīkrtam. cvih. "asya cvau" itîtvam. Not (having been) a story, make (into) a story reduced to story. (The suffix) cvih. There is  $\bar{i}$ - (in place of the final - $\bar{a}$  before this Cvi, which is then dropped) by (Pā. 7.4.32) "(There is  $\bar{i}$ ) in place of -a or - $\bar{a}$  when Cvi follows."

Kum.M. 4.13 kathī-krtam akathā kathā sampadyamānam krtam sabda-mātrâvasistam. reduced to story made to become a story, not having been a story—remaining in report alone.

Ragh.M. 1.30 aparikhāh parikhāh sampadyamānāh krtāh **parikhī-krtāh** ... abhūta-tadbhāve cvih made to become moats, not having been moats **made moats** ... (The suffix) Cvi in the sense of becoming something, not having been it before. Kum.M. 3.68 asāci sāci sampadyamānā sācī-kṛtā tiryak-kṛtā becoming oblique, not having been oblique made oblique cast sidelong

#### 1.42.5 Samāsânta suffixes

These suffixes are added at the end of a compound, without affecting the meaning of the compound. The most commonly mentioned is -ka at the end of a *bahuvrīhi* (see Section 2.15), but there are others, including -a at the end of certain *tatpurusas* in which the final member is a stem ending in a consonant.

Ragh.M. 1.13 vyūdham vipulam uro yasya sa vyūdhôraskah. "urah-prabhrtibhyah kap." He who has a broad—wide—shoulder is broad-shouldered. (By Pā. 5.4.151) "(The suffix) kaP after (a bahuvrīhi) ending in uras, etc."

Ragh.M. 1.81 saha patnyā vartata iti sa-patnīkaḥ san. "nady rtaś ca" iti kap-pratyayaḥ He is together with his wife, thus being with his wife. The suffix kaP by (Pā. 5.4.153) "And (there is kaP after a bahuvrīhi) that has the feminine ending  $-\overline{i}$  or  $-\overline{u}$  or that ends in r."

Ragh.M. 1.63 purusasyâyuh purusâyuşam. varşa-satam ity arthah. "satâyur vai puruşah" iti śruteh. "acatura-"ity-ādi-sūtreņâc-pratyayânto nipātah... brahmaņo varco brahma-varcasam. "brahma-hastibhyām varcasah" ity ac-pratyayah. The life span of a man a man's life span; i.e., a hundred years, because the Veda says, "A man has a lifetime of a hundred years." An exceptional form [see Sections 1.20, 1.35] ending in the suffix aC, by (Pā. 5.4.77) "acatura-," etc. The splendor of brahma the brahma-splendor. The suffix aC by (Pā. 5.4.78) "(The suffix aC) after varcas, when it follows brahma or hastin." Chapter 8

# **Nominal Compounds**

# **A. General Principles**

# 1.43 Introduction

Most Sanskrit compounds, no matter how many words they are composed of, can be analyzed in the first instance as compounds containing only two members (cf. Section 2.13). In the longer compounds, of course, either or both of these members may itself be a compound of any number of words; it is then analyzed in precisely the same way, beginning by breaking it down into two members. The only compounds that must be analyzed from the beginning as containing more than two members are certain *dvandva* and *bahuvrīhi* compounds (Sections 1.62, 1.65.5), but for these the basic techniques of analysis are exactly the same as those used in analyzing twomember compounds of these types. It is therefore convenient to begin by describing the standard formulas used to analyze two-member compounds in each important variety of Sanskrit compound (Sections 1.44–1.68), before explaining how these same formulas are applied to the more complicated business of analyzing long compounds (Sections 1.69–1.72).

Which formula is used in analyzing a given compound depends upon several factors—which of the four basic categories the compound belongs to (Section 1.44), whether the compound form conveys additional elements of meaning beyond those included in the general rules for each category (Section 1.45), and whether the compound is an optional one or an oblig-

85

atory one—a distinction determining whether the compound can be analyzed using the same words that make up the compound (Section 1.46). On the basis of these distinctions the grammarians have evolved a series of standard formulas to be used in analyzing each particular type of compound; we describe the formulas for the most important types in Sections 1.48-1.68.

# 1.44 Basic categories of compounds

In what follows we assume that you are familiar with the basic differences separating the four major categories of Sanskrit compounds:

- 1. avyayībhāva compounds (Section 1.44.1; treated in detail in Sections 1.48–1.51),
- tatpurusa compounds (1.44.2; details in Sections 1.52–1.60), including the important subclass of karmadhāraya compounds (1.44.3; Sections 1.54–1.55),
- 3. dvandva compounds (1.44.4; Sections 1.61–1.63), and
- 4. bahuvrīhi compounds (1.44.5; Sections 1.64-1.68).

Only occasionally is a compound analyzed as falling outside these groups (Section 1.44.6). The English names given to these categories in Whitney's grammar are noted at the beginning of the sections listed after each category.

For each of these basic categories there is a standard formula of analysis that can be used to produce a *vigraha* for most of the regular compounds within that category. The formulas differ from each other to reflect the differences in the location of the syntactical predominance (*prādhānya*) in compounds of different categories.

### 1.44.1 Avyayībhāva

The most straightforward formulas are those that apply to compounds in which one member adds some specification to the other simply by standing in some oblique (i.e., other than nominative) case relationship to it; in analyzing such compounds all that is necessary is to separate the two words and give them their proper case-endings.

In *avyayībhāva* compounds it is the first member that is syntactically predominant and the second member that simply adds some specification to the notion expressed by the first member; in the analysis it is therefore the second member that is exhibited in some oblique case (namely, the case governed by the first member): e.g., *bahir-grāmam*, "outside the village," is analyzed *bahir grāmāt*.

### 1.44.2 Tatpurusa

In simple *vibhakti-tatpurusa* compounds ("case-*tatpurusa*," "dependent determinative compound") it is the second member that adds the specification; in the analysis the first member is therefore placed in some oblique case: e.g., *svarga-patitah*, "fallen from heaven," is analyzed *svargāt patitah*.

### 1.44.3 Karmadhāraya

For *karmadhāraya* compounds, the variety of *tatpurusa* in which the first member adds specification to the second member by standing in apposition to it ("descriptive compounds"), it is often possible to proceed just as straightforwardly, by simply exhibiting the first member in the nominative case; e.g., *rakta-latā*, "red vine," may by analyzed *raktā latā*.

But for some karmadhāraya compounds—particularly those composed of two substantives, such as  $r\bar{a}jarsi$ , "king sage"—such an analysis may leave the connection between the two members vague. To make it clear that both members refer to one and the same thing, such a compound is analyzed  $r\bar{a}j\bar{a}$  ca sa rsis ca, "he is both a king and a sage," and for the sake of consistency this fuller formula is usually extended to apply to all karmadhārayas of simple apposition. Thus even rakta-latā will generally be analyzed raktā ca sā latā ca, "it is both red and a vine."

### 1.44.4 Dvandva

For the other two categories of compounds—*dvandvas* and *bahuvrīhis* the situation is more complicated than the furnishing of some specification to the predominant member by the other member: in both categories some

1.45. Additional elements of meaning

additional element of meaning is involved in the compound and must be reflected in the analysis by the addition of an extra word.

In dvandva compounds this additional element of meaning is the notion of conjunction. Syntactical predominance belongs to both members equally, and both are therefore displayed in the nominative case, but to express the idea of conjunction the word *ca* must be added in the analysis: e.g., *rāma-kṛṣṇau*, "Rāma and Kṛṣṇa," is analyzed *rāmaś ca kṛṣṇaś ca*, "(both) Rāma and Kṛṣṇa." Note carefully the difference between this formula and the fuller form of the analysis sometimes used for *karmadhāraya* compounds: the pronoun that always appears in the formula for *karmadhārayas* in order to make it clear that both members refer to the same thing is missing in the formula for *dvandvas*, in which the two members refer to two separate objects.

#### 1.44.5 Bahuvrīhi

In *bahuvrīhi* compounds the additional element of meaning is the notion of some relationship—usually that of possession—through which the compound refers not to the object expressed by either member, but to some other object outside the compound. Thus neither member has syntactical predominance; the predominance belongs to the outside object, and that object must be reflected in the analysis of the compound by the addition of a relative pronoun, the case of which will convey the relationship thorugh which the compound refers to the outside object: e.g., *dīrgha-karṇaḥ*, "Long-ear," is analyzed *dīrghau karṇau yasya saḥ*, "of whom there are long ears, he," "he who has long ears."

#### 1.44.6 Kevala-samāsa

A few compounds are not included by the grammarians under any of these four categories. Occasionally an author will produce an ad hoc compound that, while its meaning is perfectly clear, cannot conveniently be explained by any of Pāṇini's particular rules for the formation of compounds in each of the four categories (e.g., *ājanma-śuddhaḥ*, "pure since birth,"  $R\bar{a}gh$ . 1.5), and occasionally words that could as well have been considered separate are viewed as a compound for technical reasons of grammar (e.g.,  $v\bar{a}garth\bar{a}v$ -iva, "like sound and sense,"  $R\bar{a}gh$ . 1.1). Such compounds are called *kevala-samāsas* ("simply compounds") or *sup-sup-samāsas* ("nounnoun-compounds," in reference to Pāṇini's general rule for compounding, 2.1.4: "[*sup*] saha supā," "[An inflected noun] (may be compounded) with an inflected noun"). Commentators often content themselves with merely noting the fact of such a compound:

Kum.M. 4.21 vinā-kṛtā. viyojitā satīty arthaḥ. sup-supêti samāsaḥ. bereft i.e., being deprived of. (This is) a compound, by the sup-supā (rule).

Fortunately, when such compounds are analyzed, their analysis is usually trivial, since it consists simply of separating the two members of the compound—thus  $\bar{a}janma-\hat{s}uddhah$  is analyzed  $\bar{a}janma \hat{s}uddhah$ , and  $v\bar{a}g$ *arthāv-iva* is analyzed  $v\bar{a}garth\bar{a}v$  *iva*—so that they need not detain us further.

Most compounds, however, fall within the major groups, and more often than not the basic formulas of analysis are adequate. Yet for some types of compounds further formulas are necessary, either because the compound involves the addition of a further element of meaning (Section 1.45) or because one of the members of the compound is a word that cannot be used outside a compound (Section 1.46).

# 1.45 Additional elements of meaning

The conveying of special elements of meaning through composition, with the resulting necessity of adding words in the analysis to convey that meaning when the compound is dissolved, operates not only throughout the general categories of *dvandva* and *bahuvrīhi* compounds (as explained in the previous section), but also in particular types of compounds within the general categories. Each of these special types of compound has its own standard formula of analysis; the details are given in the lists of formulas below.

### 1.45.1 Comparison

For example, certain kinds of *karmadhāraya* compounds involve the notion of comparison, so that the word *iva* ("like") must be inserted in the analysis even though it does not appear in the compound (Section 1.55).

### 1.45.2 Aggregation

Similarly, certain types of *dvandva* and *tatpurusa* compounds involve the notion of aggregation, so that the *samāhāra* ("aggregate") is often employed in the analysis (Sections 1.56, 1.63).

### 1.45.3 Clarification of connection

In particular instances it is necessary to insert a word offering clarification of the precise connection between two members of a compound, without abandoning the basic formula of analysis (e.g., Sections 1.53.2, 1.54.4).

# 1.46 Optional and obligatory compounds

A further source of multiplication of formulas for the analysis of compounds is the fact that not all compounds can be analyzed using the same words that make up the compound. Traditionally, this problem is approached through the distinction between optional and obligatory compounds. In the examples of a simple avyayībhāva and a simple tatpurusa given in Section 1.44, the words that serve as members of the compound are capable of being used as separate words in a phrase to express the same meaning that they express when in compound. In order to express that meaning the compounding of the words is therefore optional. In many compounds, however, the members of the compound cannot be used to express the same meaning in a phrase; such compounds are therefore called obligatory (nitya-samāsa, "always a compound," in the sense that the words must always be in compound if they are to express that particular meaning), and are described as a-sva-pada-vigraha ("having no analysis in their own words"). There are several general reasons why a compound might be obligatory:

### 1.46.1 Asvapadavigraha compounds

In practice the designation of *a-sva-pada-vigraha* is applied only to compounds of the types mentioned in the following subsections. Strictly speaking, however, the compounds that involve additional elements of meaning and thus require that other words be added to the members of the compound when an analysis is given (Section 1.45) are obligatory, in the sense that the members of the compounds are not capable *in themselves* of expressing the full meaning of the compound through their use in a phrase. By this interpretation, all *dvandvas*, *bahuvrīhis*, *karmadhārayas* of comparison, and other such compounds are obligatory compounds. There is evidence that this was the view of Pāṇini himself, and that the later tradition misunderstood his treatment of optionality (Paul Kiparsky, *Pāṇini As a Variationist* [co-published by the Poona University Press and the M.I.T. Press, 1979], p. 43). In any case, the tradition does not regard such compounds as incapable of analysis in their own words: the standard formula of analysis for each of these types of compounds does use the words that make up the compound, even if it has to use something else in addition.

When a compound is considered a *nitya-samāsa* any rewording of it is technically not a true *vigraha* analysis but simply a description of the meaning of the compound. After offering such rewording a commentator may then point out that the compound is considered *nitya*:

Ragh.M. 1.18 bhūtyā arthāya bhūty-arthām vrddhy-arthām eva. "arthena saha nitya-samāsah sarva-lingatā ca vaktavyā." only for prosperity's sake for the sake of prosperity, for the sake of income. (By the Vārtika rule) "It should be stated that (in compound) with artha there is a nitya-samāsa (and it may be used to modify words) in every gendet."

### 1.46.2 Conventional meanings

Some compounds have acquired by convention a meaning that is more specific than that expressed in a standard analysis. Just as the English compound "bluebird" refers not to any blue bird buy only to certain species, so the Sanskrit compound *krsna-sarpa* refers not to any black snake, but only to a particular kind of cobra. Thus an analysis such as *krsnas* ca sa sarpas ca, "it is a snake and it is black," while it may indicate the etymological formation of the compound, does not accurately restate the compound's full meaning. Commentators frequently point out that such compounds are *nitya-samāsa*s; strictly speaking their full meaning can never be reproduced in the form of a standard analysis, but can only be conveyed through a synonym or definition of the compound as a unit.

#### 1.46.3 Bound forms

In practical terms the most important reason for the obligatory status of a compound is that one of its members is an element that simply cannot stand as a separate word within a phrase, or at least cannot do so with the same meaning that it has within the compound. In such instances some substitute must be found for this element in the phrasal analysis, and these compounds can therefore not be analyzed using their own words. Several important groups of Sanskrit compounds are obligatory compounds in this sense: *avyayībhāvas* in which the first member is incapable of conveying the meaning in question as a separate preposition (Section 1.50), *tatpuruṣas* in which the first member is a prefix or preposition used adjectivally (Section 1.57), *tatpuruṣas* in which the final member is a verbal noun that can exist only at the end of a compound (Section 1.59), and a few others. For each of these groups the formulas of analysis that are required differ markedly from the standard formulas for the broad category within which the group falls; specific details are given under each section.

# 1.47 Irregularities

Irregularities in compounds usually involve oddities in the form of the compound rather than in the formulas required in analyzing them. They will be touched on in separate sections under each type of compound (e.g., Sections 1.51, 1.60, 1.68).

# B. Avyayībhāva Compounds

# 1.48 Introduction

Avyayībhāva compounds (Pā. 2.1.5–2.1.20; "secondary adjective compounds used adverbially," Whitney 1313) are indeclinable adverbial compounds with a preposition or other indeclinable as first member.

For *avyayībhāva* compounds the distinction between optional (Section 1.49) and obligatory compounds (Section 1.50) is particularly noticeable: the two are analyzed in quite different ways.

# 1.49 Optional avyayībhāvas

For these compounds the analysis simply separates the two members and places the second member in the case that is governed by the first: bahirgrāmam, "outside the village," for example, is analyzed bahir grāmāt, "outside of the village." But things are usually not this simple, because in order for the compound to be optional-and thus capable of being analvzed in its own words-the preposition or indeclinable that constitutes the first member of the compound must be able to stand as a separate word governing another word in a particular case-words that can do so are technically known as karmapravacaniya-and furthermore it must be capable of doing so with the same meaning that it has in the compound. Some prepositions that occur as the first member of avyayibhāvas are karmanravacaniya in some meanings but not in others; e.g., anu-vanam ("near the forest") and anu-gangam ("along the Ganges") are optional compounds and may be analyzed anu vanam and anu gangām, but anu-ratham ("behind the chariots") and anu-jyestham ("in order of age") are obligatory compounds that require special treatment. Similarly, for certain meanings there are some prepositions that are karmapravacaniya and others that are not: e.g., anu-vanam ("near the forest") is an optional compound, but upakumbham ("near the pot") is obligatory.

When the preposition beginning the compound is available as a separate preposition governing a particular case, the analysis is straightforward:

Ragh.M. 1.5 [ā-janma] ā janmanah. janmârabhyêty arthah. "ān maryādâbhividhyoh" ity avyayībhāvah. [From birth] from birth, i.e., beginning with birth. This is an avyayībhāva by (Pā. 2.1.13) "ā in the sense of limit or inclusion (may be compounded with a following noun)."

# 1.50 Obligatory avyayībhāvas

For each of these compounds in which the first member cannot stand outside the compound while expressing the same meaning, some other way to convey the meaning of that member must be found. The forms of analysis used for such compounds differ greatly depending upon the specific

1.51. Irregular avyayībhāvas

# Chapter 8. Nominal Compounds

form and meaning of the first member. To take the obligatory *avyayī-bhāvas* mentioned in the preceding subsection: *anu-ratham* ("behind the chariots") is analyzed *rathānām paścāt* ("at the rear of the chariots"), *anu-jyeṣṭham* ("in order of age") is analyzed *jyeṣṭhānupūrvyeṇa* ("according to the sequence from the oldest"), and *upa-kumbham* ("near the pot") will be analyzed *kumbhasya samīpe* ("in the vicinity of the pot"). Obviously no general rule for the treatment of these can be given; usually the precise wording used in analyzing these forms is drawn directly from specifications found in individual rules in Pāṇini's grammar. Yet the more common of these obligatory *avyayībhāvas* occur rather frequently, so that the formulas used to analyze them show up often in commentaries; it is therefore useful to be familiar with a few of them (of course the second member will vary, but whenever the first member is used with the same meaning as in the compounds mentioned here, the basic formula of analysis will be the same):

kumbhasya samīpe: upa-kumbham, "near the pot"

ātmany adhikṛtya: adhy-ātmam, "concerning the self" rathānām paścāt: anu-ratham, "behind the chariots" śaktim anatikramya: yathā-śakti, "according to one's ability" jyeṣṭhānupūrvyeṇa: anu-jyeṣṭham, "in order of age" cakreṇa ekakāle: sa-cakram, "simultaneously with the wheel" jalasya madhye: antar-jalam, "within the water"

dine dine: prati-dinam, "on each day"

Some examples from commentaries:

Ragh.M. 1.6 vidhim anatikramya yathā-vidhi. "yathâsādrśye" ity avyayībhāvah.... yathā-kāmam abhilāşam anatikramya ... yathâparādham aparādham anatikramya ... yathā-kālam kālam anatikramya not transgressing the rule in accordance with the law; an avyayībhāva by (Pā. 2.1.7) "yathā (may be compounded with a following noun) in a meaning other than similarity." ... in accordance with their wishes not transgressing their desires ... in accordance with the crime not going beyond the crime ... in accordance with the time not going beyond the (proper) time.

Ragh.M. 1.44 padasya paścād anu-padam. paścād-arthe 'vyayībhāvaḥ. after the foot following upon. An avyayībhāva in the meaning of "behind" [by Pāṇini 2.1.6].

Ragh.M. 1.91 **a-vighnam** vighnasyâbhāvo 'stu. "avyayam vibhakti-" (Pā. 2.1.6) ity-ādinâbhāvârthe 'vyayībhāvah. May there be no obstacle an absence of an obstacle. An avyayībhāva in the meaning of "absence" (by Pāņini 2.1.6).

Kum.M. 1.48 samsayabhāvah asamsayam. sandeho nāstīty arthah. An absence of doubt without doubt no doubt; i.e., no doubt exists.

Kum.M. 1.60 ahany ahani praty-aham. on each day daily

Kum.M. 5.1 akṣṇoḥ samīpe samakṣaṃ purataḥ. "avyayaṃ vibhakti-samṛddhi-" ity-ādinâvyayībhāvaḥ. in the vicinity of her eyes before her eyes, in front of her. An avyayībhāva by (Pā. 2.1.6) "An indeclinable (is compounded with a following noun) in the senses of ....."

# 1.51 Irregular avyayībhāvas

Most of the common irregularities are in the construction of the compound itself rather than in the formula by which it is analyzed. There are, for example, *avyayībhāvas* in which the order of the members is reversed, so that they end up with a governed prior member and thus appear to be *tatpuruṣas*; these are nevertheless analyzed like regular *avyayībhāvas*: e.g., *gangā-pāram*, "across the Ganges," is analyzed *gangāyāh pāram*. Conversely, it should be noted that there are *tatpuruṣas* that have governed final members and thus appear to be *avyayībhāvas*, either because the order of the members is irregularly reversed—e.g., *bhūta-pūrvah*, "having occurred previously," analyzed *pūrvaṃ bhūtah*—or because it is logically more convenient to interpret them as having reversed members (Section 1.53.3), or because they really do have governed final members but are nevertheless traditionally regarded as *tatpurusas* because they are used adjectivally rather than adverbially (Section 1.57.3).

# C. Tatpurusa Compounds

### 1.52 Introduction

The category of *tatpuruşa* compounds (Pāṇini 2.1.22–2.2.22; "determinative compounds," Whitney 1262–1291) has several important subtypes. The varieties of *tatpuruṣa* distinctive enough to require separate formulas of analysis will be treated under the following headings:

1.53 dependent compounds (vibhakti-tatpurusa)

descriptive compounds (karmadhāraya):

- 1.54 simple karmadhārayas
- 1.55 comparative karmadhārayas
- 1.56 dvigu compounds
- 1.57 prepositional *tatpurușas* and related compounds (*prādi-tatpurușa*, etc.)
- 1.58 negative compounds (nañ-tatpurușa)
- 1.59 upapada compounds
- 1.60 irregular tatpuruşas

# 1.53 Dependent tatpurusas

Dependent *tatpurusas*, often called case-*tatpurusas* ("*vibhakti-tatpurusa*," Pāņini 2.1.22–2.1.48; "dependent compound," Whitney 1264–1278), are those in which the first member of the compound stands in some oblique case relationship to the second.

1.53. Dependent tatpurusas

In these compounds the first member simply adds some specification to the second member by standing in some case-relationship to it. Accordingly the analysis is usually quite straightforward (Section 1.53), but there are slight complications when additional clarification is necessary (Section 1.53.2) and when the second member is most conveniently interpreted as a partitive genitive (Section 1.53.3).

### 1.53.1 Straightforward analysis

The two members are simply separated, with the second member appearing in the nominative case and the first member in any oblique case. The names by which commentators refer to these compounds indicate the case in which the first member appears in the analysis:

dvitīyā-tatpurusa (accusative tatpurusa):

grāmam gatah—grāma-gataḥ ("gone to the village") muhūrtam sukham—muhūrta-sukham ("pleasure for a moment")

trtīyā-tatpurusa (instrumental tatpurusa):

asinā kalahaḥ—asi-kalahaḥ ("a fight with a sword") paraśunā chinnah—paraśu-cchinnah ("cut with a hatchet")

caturthī-tatpurusa (dative tatpurusa):

kundalāya hiranyam—kundala-hiranyam ("gold for an earring")

kuberāya balih—kubera-balih ("an offering for Kubera")

pañcamī-tatpurusa (ablative tatpurusa):

svargāt patitah—svarga-patitah ("fallen from heaven") vrkebhyo bhayam—vrka-bhayam ("fear of wolves")

sasthī-tatpurusa (genitive tatpurusa):

candanasya gandhah—candana-gandhah ("fragrance of sandalwook")

*idhmasya pravraścanah—idhma-pravaścanah* ("cutter of fuel," i.e., a hatchet) saptamī-tatpurusa (locative tatpurusa):

akṣeṣu nipuṇaḥ—akṣa-nipunaḥ ("skilled in dice") māse deyam—māsa-deyam ("payable in a month")

Note that the first member need not appear in the singular in the analysis, but may have the grammatical number required.

In these formulas, the choice of cases is of course governed by the Sanskrit idiom rather than by what one might expect in English or any other language: thus *vrksa-bhaya*, "wolf-fear," "fear of wolves," is an ablative *tatpurusa*, not genitive, because in Sanskrit the ablative case is used with words expressing the source of fear.

Examples from the commentaries:

accusative (dvitīyā):

Kum.M. 1.5 adhah-sānūni megha-maņdalād adhas tatāni gatām prāptām [adhah-sānu-gatām]. "dvitīyā

*śritâtīta-patita-gatâtyasta-prāptâpannai*," *iti samāsa*, gone to—reaching—the below-slopes—slopes below the circle of clouds **below-slope-gone**. The compound is by (Pā. 2.1.24) "A word in the accusative case is compounded with *śrita, atīta, patita, gata*," etc.

Kum.M. 2.23 astam nāšam itah prāptah [astam-itah]. astam iti makārântam avyayam. tasya "dvitīyā

*śritâtīta-patita-gatâtyasta-prāptâpannai*,<sup>m</sup>*iti samāsa*,<sup>h</sup>gone to—reached to—setting—destruction [setting-gone]. *astam* is an indeclinable ending in *m*. Its compound is by (Pā. 2.1.24) "A word in the accusative case is compounded with ...."

Kum.M. 3.41 latāgrhadvāram gatah [latāgrhadvāra-gatah] gone to the vine-hut-door [vine-hut-door-gone]

instrumental (trtīyā):

Kum.V. 1.27 [M 28] prabhayā mahatī prabhā-mahatī iti trtīyā-samāsah. great with splendor splendor-great; this is an instrumental compound. Kum.V. 1.56 [M 57]samidbhir edhaih samiddham dīptam[samit-samiddham]kindled—lighted—with firewood—with fuelfirewood-kindled

Kum.M. 1.51 mantraih pūtam samskrtam [mantra-pūtam] purified—refined—with mantras mantra-purified

dative (*caturthī*): Dative *tatpuruṣa* compounds are relatively infrequent because of the limited scope of the dative case in Sanskrit. When they do occur commentators tend to explain them not by displaying the first component in the dative case, but by using the equivalent (and perhaps less ambiguous) compound ending in *-artha* (see the example on p. 91 and cf. Section 2.19.2):

Ragh.M. 2.10 *ācārârthair lājair ācāra-lājai*, with puffed rice for the sake of custom with custom-puffed-rice

Kum.V. 1.60 [avacita-bali-puspā] avacitāny uccitāni baly-artham  $p\bar{u}j\hat{a}rtham$  puspāni yayā sā [collected-offering-flowered] she by whom flowers for the sake of offering—for the sake of worship—had been collected—accumulated

ablative (*pañcamī*):

Kum.V. 1.37 [M38] sitä itarah sitêtara iti "pañcamī" iti yoga-vibhāgāt samāsah. other than white white-other (i.e., black). The compound is by separating "a word in the ablative case (is compounded)" (from the rest of the sūtra in Pā.2.1.37).

Kum.M. 2.30 sahasrān nayanebhyo 'dhikam sahasra-nayanâdhikam. more than thousands of eyes eye-thousand-more

Kum.M. 2.45 tasya tārakasyâpātāt samāpatter bhayāt [tad-āpāta-bhayāt] from fear of his—Tāraka's—attack—encounter from his-attack-fear

genitive (sasthī):

Ragh.M. 1.9 tesām raghūnām guņais tad-guņaih. by the virtues of theirs—of the Raghus by their-virtues

Ragh.M. 1.12 tad-anvaye tasya manor anvaye kule. in his—Manu's—lineage in the Manu-lineage

Ragh.M. 1.24 vinayasya śikṣāyā ādhānāt karaṇāt [vinayâdhānāt] from the imparting—doing—of education—instruction from education-imparting

Kum.M. 1.58 adrīņām nāthah adri-nāthah lord of mountains mountain-lord

Kum.M. 2.25 [ogha-samrodhah] oghasya samrodhah pravāha-pratibandhah current-obstruction obstruction—flow-obstacle—of the current

locative (saptamī):

Kum.M. 1.44 pravāle bāla-pallava upahitam nihitam [pravālôpahitam] laid—put down—on a bud—young leaf bud-laid

Kum.M. 2.45 bhuvanānām ālokane prītiķ [bhuvanâlokana-prītiķ] joy in the seeing of the worlds world-seeing-joy

Kum. V. 3.8 pādânataḥ pādayor [patita] ānataś caraṇa-patitaḥ fallen—bowing down—at the feet foot-fallen

Kum.M. 4.10 tvad-adhīnam tvayy adhīnam dependent on you you-dependent

Ragh.M. 4.5 sāmrājya-dīksitam sāmrājye sāmrājya-karmaņi maņdalâdhipatye dīksitam abhisiktam suzerainty-consecrated consecrated—anointed—in suzerainty—being suzerain, overlordship **这些世界影响的**100

### 1.53.2 Additional specification of connection

Occasionally the connection between the first and second member is not adequately indicated merely by the case of the first member as it appears in the analysis, so that an additional word of clarification must be supplied. This occurs, for example, in a group of instrumental *tatpurusas* dealing with combinations of foods (Pāṇini 2.1.34–35): thus *dadhy-odanah* ("yogurt-rice") is analyzed *dadhnôpasikta odanah* ("rice *sprinkled* with

yogurt"), and guda-dhānāh ("molasses-barley") is analyzed gudena miśrā dhānāh ("barley mixed with molasses"). Commentators call such comnounds madhyama-pada-lopa (see Section 1.54.4).

Kum. V. 3.46 akşa-pūrņaņ sūtram akşa-sūtram bead-filled string bead-string

Kum.V. 3.56 bimba-tulyo 'dharo bimbâdhara iti śāka-pārthivâditvāt samāsah. cherry-like lip cherry-lip: the compound is (justified) by belonging to the saka-pārthiva group.

### 1.53.3 Partitive genitive as second member

In *tatpurusa* compounds with partitive genitive as second member (*ekadeśi-tatpurusa*, *avayava-tatpurusa*, *amśi-tatpurusa*, Pä. 2.2.1–2), in which the first member of a *tatpurusa* compound is a word delimiting a part of a single whole expressed by the second member, a logical problem arises. The object referred to by a *tatpurusa* is supposed to be a variety of the thing expressed by the second member, with the first member simply adding some specification, but (to give an English example) a halfpenny, for instance, is not really any sort of penny at all. Accordingly, the second member in such compounds is construed as a partitive genitive and the compound is considered to have its members reversed (because the syntactical predominance ends up belonging to the first member). Thus  $p\bar{u}rva-k\bar{u}yah$  ("fore-body") is analyzed  $p\bar{u}rvam k\bar{u}yasya$  ("front of the body"), and *arddha-pippalī* ("half-pepper") is analyzed *arddham pippalyāh* ("half of a pepper").

This same procedure—making the first member predominant and exhibiting the second member in the genitive case in the analysis—is also used for several other rather obscure groups of *tatpurusas* (see Pāṇini 2.2.3–5), and for a few irregular compounds (Pāṇini 2.2.31) typified by the word rāja-dantah ("king-tooth," i.e., front tooth), which might appear to be a karmadhāraya but instead is analyzed dantānām rājā ("king of teeth").

Some commentators interpret the  $r\bar{a}ja$ -dantâdi category rather broadly, to include compounds with reversed members that other commentators interpret as falling outside all categories (i.e., as *sup-sup-samāsas*; see Section 1.44.6):

Ragh.H. 3.55 [M 54] [an-āsvāditapūrvam] pūrvam āsvāditam āsvādita-pūrvam, tato nan-samāsah, rājadantādih.

[not-tasted-before:] previously tasted—before-tasted, then a negative compound. (The compound *āsvādita-pūrva* belongs to) the *rāja-dantâdi* group.

Ragh.M. 3.54 anāsvādita-pūrvam pūrvam anāsvāditam, sup supêti samāsah. untasted-before previously untasted. A compound by (the general rule) "noun with noun."

金融市業になる

# 1.54 Karmadhārayas of simple apposition

*Karmadhārayas* (Pāņini 2.1.49–2.1.72; "descriptive compounds," Whitney 1279-1291) are *tatpuruṣa* compounds in which the two members appear in grammatical apposition in the analysis. Although some important types of *karmadhāraya* involve the additional notion of comparison and thus require special formulas of analysis (Section 1.55), in most *karmadhārayas* the first member simply adds some specification to the thing expressed by the second member.

For all *karmadhāraya* compounds of simple apposition a single basic formula of analysis applies, but minor differences sometimes occur depending upon whether the compound contains an adjective as the first member only (Section 1.54.1), as both members (Section 1.54.2), or as neither member (Section 1.54.3). For the latter group, in which both members of the compound are substantives, special treatment is sometimes required when further specification of the connection between the members is required (Section 1.54.4). Note that *karmadhārayas* in which both members are substantives often involve the notion of comparison (Sections 1.55.2, 1.55.3), and *karmadhārayas* composed of substantive plus adjective nearly always involve comparison (Section 1.55.1).

### 1.54.1 Adjective plus substantive

The karmadhāraya compound composed of an adjective followed by a substantive (viśeṣaṇa-[pūrva-pada]-karmadhāraya, "karmadhāraya with an

#### 1.54. Karmadhārayas of simple apposition

adjective [as first member]," Pāṇini 2.1.57 ff.; Whitney 1280) is the simplest and most common form of karmadhāraya, and its analysis is done in the manner already explained in Section 1.44: while compounds such as  $n\bar{\imath}l\delta tpalam$  ("blue waterlily") and rakta-latā ("red vine") may be analyzed simply by separating their members— $n\bar{\imath}lam$  utpalam and raktā latā—they are more commonly analyzed with the fuller formula that uses a pronoun to make it clear that both members refer to the same thing:  $n\bar{\imath}lam$  ca tad utpalam ca ("it is both blue and a waterlily") and raktā ca sā latā ca ("it is both red and a vine"). Again, note that the presence of this pronoun (usually some form of tad or adas) in the analysis serves to distinguish these compounds from dvandvas.

In practice, commentators often use the regular formula but sometimes simply separate the two words and place them in apposition:

#### Kum.M. 1.2 sarve ca te śailāś ca sarva-śailāh.

"pūrva-kālâika-sarva-jarat-purāṇa-nava-kevalāḥ samānâdhikaraṇena" iti samāsaḥ. they are both all and mountains **all-mountains**. The compound is by (Pā. 2.1.47) "When preceding, the words *eka*, *sarva*, *jarat*, *purāṇa*, *nava*, and *kevala* (are compounded) with a word in apposition."

#### Kum.M. 2.32 mahān asuro mahâsurah.

"san-mahat-paramôttamôtkṛṣṭāḥ pūjyamānaiḥ" iti tatpuruṣaḥ. a great demon great-demon. The tatpuruṣa is by (Pā. 2.1.61) "The words san, mahat, parama, uttama, and utkṛṣṭa (are compounded) with things to be praised."

### 1.54.2 Adjective plus adjective

Karmadhāraya compounds composed of an adjective followed by an adjective (viśeṣanôbhaya-pada-karmadhāraya, "karmadhāraya with adjectives as both members") are usually analyzed using the formula just described. Thus the analysis regularly presents both members as adjectives describing the same thing, even though in most of these compounds (e.g., those formed by Pāṇini 2.1.68) the first member actually functions as an adverb modifying the second member. For example, *tulya-śvetah* ("equally white") is analyzed *tulyaś câsau śvetaś ca* ("it is both equal and white"), and *bhojyôṣṇam* ("palatably warm") is analyzed *bhojyaṃ ca tad uṣṇaṃ ca* ("it is both palatable and warm").

As with other simple *karmadhārayas*, commentators may use the regular formula or simply place the two words in apposition:

Ragh.M. 1.83 pallavavat snigdhā câsau pāṭalā ca [pallava-snigdha-pāṭalā] it is both glossy like a sprout and pink [sprout-glossy-pink]

Ragh.M. 1.16 bhīmaiś ca kāntaiś ca [bhīma-kāntaiḥ] both awesome and lovable awesome-lovable

When both members of the compound clearly do function as adjectives describing the same object, they are often past participles indicating consecutive actions involving that object (Pāṇini 2.1.49); for these compounds a special formula of analysis is used: for example, *snātânuliptaḥ* ("bathed and oiled") is analyzed *ādau snātaḥ paścād anuliptaḥ* ("first bathed, afterwards oiled"), and *pītôdgīrṇam* ("drunk and vomited") is analyzed *ādau pītam paścād udgīrṇam* ("first drunk, afterwards vomited").

Another common type of karmadhāraya combines two color adjectives.

Kum.M. 3.36 kṛṣṇaś câsau sāraḥ śabalaś ca kṛṣṇa-sāraḥ kṛṣṇa-mṛgaḥ. "varṇo varṇena" iti samāsaḥ. it is both black and speckled—spotted, a black-spotted antelope. The compound is by (Pā. 2.1.69) "a color with a color."

Kum.M. 4.14 haritam ca arunam ca [haritârunam]. "varno varnena" iti tatpurusah. both green and pink [green-pink]. The compound is by (Pā. 2.1.69) "a color with a color."

### 1.54.3 Substantive plus substantive

Many such compounds involve the added notion of comparison (Sections 1.55.2, 1.55.3). When merely apposition is involved, the standard formula for a *karmadhāraya* is used: e.g.,  $r\bar{a}jarsih$  ("king-sage") is analyzed  $r\bar{a}j\bar{a}$  ca sa rsis ca ("he is both a king and a sage"). Such compounds are not very frequent in Sanskrit, except when a title is compounded with a proper name

(in which case the title usually comes last—e.g., Śańkarâcārya, Bhojarāja—even though it might logically be expected to come first), because it may be unclear from the compound in what sense the first member applies to the thing expressed by the second.

In practice commentators often analyze noun-noun *karmadhārayas* simply by putting the particle *eva* after the first member (cf. *rūpaka-samāsas*, Section 1.55.3) rather than using the *câsau*... *ca* formula:

Kum.V. 3.27 navam cūtam eva bāṇaḥ śaras tasmin [nava-cūta-bāṇe] the arrow—shaft—that is a fresh mango blossom, on it on the fresh-mango-arrow

Kum.M. 3.27 navam cūtam cūta-kusumam tad eva bāņas tasmin nava-cūta-bāņe. a fresh mango—mango blossom, the arrow that it is, on it on the fresh-mango-arrow

Ragh.M. 1.58 rājyam evâśramas tatra munim [rājyâśrama-munim] the life-stage that is kingship, a sage in it [a kingship-lifestage-sage]

Ragh.M. 4.5 padmâtapatrena padmam evâtapatram tena with a lotus-parasol a parasol that was a lotus, with it

Kum.M. 3.36 kusumam evâikam sādhāraņam pātram tasmin [kusumâika-pātram]. a single—common—cup that was a flower, in it in the flower-single-cup

Kum.M. 3.53 vasanta-puspāņy evâbharaņam [vasanta-puspâbharaņam] an ornament that is springtime-flowers a springtime-flower-ornament

### 1.54.4 Additional specification of connection

This last problem is particularly acute in *karmadhārayas* composed of two substantives when the first member cannot be applied, as it stands, directly to the thing denoted by the second member. In analyzing loose compounds of this sort it is necessary to add some explanation of the precise connection between the members. This is done by placing the explanatory word in compound with the first member in the analysis: for example, *deva-brāhmaṇaḥ* ("deity-brahmin") is analyzed *deva-pūjako brāhmaṇaḥ* 

104

("deity-worshipping brahmin"). Commentators frequently refer to compounds of this type as "*sākapārthivâdi* compounds," since they are grouped together (by a *vārtika* on Pāņini 2.1.60) in a list beginning with the compound *sāka-pārthivaḥ* ("vegetable-king" or "era-king," analyzed *sāka-bhojī pārthivaḥ* ["vegetable-eating king"] or *sāka-priyaḥ pārthivaḥ* ["a king fond of vegetables," or, according to some, "a king beloved of an era"]).

Note that these compounds are similar to those such as  $guda-dh\bar{a}n\bar{a}h$ (Section 1.53.2): in both groups the connection between the two members requires some explanation, and in both groups this explanation is supplied by adding a word between the members ( $guda-misra-dh\bar{a}n\bar{a}h$ , deva $p\bar{u}jaka-brlahmanah$ ) that can be interpreted as standing in compound with the first member ( $guda-misr\bar{a}$   $dh\bar{a}n\bar{a}h$ ,  $deva-p\bar{u}jako$   $br\bar{a}hmanah$ ). Accordingly, commentators often refer to such compounds as uttara-pada-lopa $sam\bar{a}sas$  ("compounds in which the second word [of the first member] has been dropped"), or, much more commonly, as madhyama-pada-lopa $sam\bar{a}sas$  ("compounds in which the middle word has been dropped").

Megh.M. 1 chāyā-pradhānās taravaś chāyā-taravah. sākapārthivâditvāt samāsah. shade-rich trees shade-trees. The compound is because of being in the sāka-pārthivâdi group.

# 1.55 Karmadhārayas of comparison

These are of two basic types. In one type (Section 1.55.1) the first member is a substantive representing the standard of comparison (the *upamāna*) and the second member is a substantive representing the common property (*sāmanyavacana*). In the second type both members are substantives: the first represents the object of comparison (*upamīta*) and the second represents the standard of comparison (*upamāna*). This second type can be analyzed in two ways, depending upon whether the underlying comparison is viewed as a simile (Section 1.55.2) or as a metaphor (Section 1.55.3).

### 1.55.1 Upamānapūrvapada karmadhāraya

These formations (*upamānapūrvapada-karmadhāraya*, "*karmadhāraya* in which the first member is a standard of comparison," Pāṇini 2.1.55), are

analyzed by simply separating the members of the compound and inserting between them the particle *iva* ("like"):

ghana iva śyāmah-ghana-śyāmah ("dark as a cloud-cloud-dark")

*kumudam iva śyenī—kumuda-śyenī* ("white like a waterlily—lilywhite")

Examples from the commentaries:

Ragh.M. 1.13 śālo vrksa iva prāmsur unnatah sāla-prāmsuh. tall—lofty—like a sal tree sal-tall

Kum.M. 1.13 candra-marīcibhir iva gauraiķ šubhraiķ [candra-marīci-gauraiķ] .... "upamānāni sāmānya-vacanaiķ iti samāsaķ. pale—white—as moon-beams moon-beam-pale. The compound is by (Pā. 2.1.55) "Standards of comparison [are compounded] with words expressing shared properties."

Kum.M. 3.29 [bālêndu-vakrāņi] bālêndur iva vakrāņi [young-moon-crooked] crooked as a young-moon

A commentator may express the comparison through the use of the suffix *-vat* rather than the particle *iva*:

Ragh.M. 1.83 pallavavat snigdhā câsau pāṭalā ca [pallava-snigdha-pāṭalā] it is both glossy like a sprout and pink [sprout-glossy-pink]

### 1.55.2 Upamita-samāsa

In analyzing these (upamita-samāsa, upamānottarapada-karmadhāraya, "karmadhāraya in which the second member is a standard of comparison,"  $P\bar{a}$ . 2.1.56), the particle *iva* is inserted after the second member. Because these compounds are made up of two substantives, a pronoun may optionally be used in the analysis to make it clear that both nouns refer to the same object.

purușo 'yam vyāghra iva—purușa-vyāghrah ("[this is] a man like a tiger--man-tiger")

mukham candra iva---mukha-candrah ("a face like the moon--moon-face")

Examples from the commentaries:

Ragh.M. 1.12  $r\bar{a}j\bar{a}$  indur iva  $r\bar{a}j\hat{e}nd\bar{u}$   $r\bar{a}ja$ -śresthah. upamitam vyāghrâdinā samāsah. a king like the moon **moon-king** best of kings. The compound is by (Pā. 2.1.56) "The object of a simile [compounded with] a word such as 'tiger.'"

Kum.M. 2.47 hayo ratnam iva haya-ratnam aśva-śresthah. a horse like a jewel a horse-jewel best of horses

Ragh.M. 4.61 mukhāni padmānîva mukha-padmāni. upamita-samāsah. faces like lotuses face-lotuses. A simile compound.

Ragh.M. 5.73 dantāh košā iva danta-košāh teeth like buds teeth-buds

### 1.55.3 Rūpaka-samāsa

In karmadhārayas in which one substantive is equated with another, either metaphorically or literally ( $r\bar{u}paka$ -samāsa, "compound of metaphorical identification"; avadhāranapūrvapada-karmadhāraya, "karmadhāraya in which the first member is a limitation"), what is involved here is simply a different interpretation of the compounds in the preceding group, by viewing the comparison involved as a metaphorical equation ( $r\bar{u}paka$ ) rather than as a simile ( $upam\bar{a}$ ). The formula of analysis inserts eva ("actually," "nothing but...") after the first member:

purusa eva vyāghrah—purusa-vyāghrah ("a tiger that is actually a man—man-tiger")

mukham eva candrah—mukha-candrah ("a moon that is nothing but a face—face-moon")

Examples from the commentaries:

Kum.M. 3.30 cūta-pravāla evâustham cūtapravālâustham. a lip that is actually a mango-shoot a mangoshoot-lip

Kum.M. 3.39 latā eva vadhvas tābhyaḥ [latā-vadhūbhyaḥ] wives that are actually vines, from them from vine-wives

# 1.56 Dvigu compounds

And Minister

The category of *dvigu* compounds (Pā. 2.1.52; Whitney 1312) includes several types of compounds in which the first member is a numeral. There are special restrictions on the use of numerals and words signifying directions as the first member of a *karmadhāraya*. These items may form the first member of simple two-member descriptive compounds only when the compound constitutes a proper name (Pāṇini 2.1.50). Such compounds may be broken down using the standard formula for the analysis of *karmadhārayas*, although the resulting analysis will simply indicate the grammatical formation of the compound rather than restating its full meaning, since the analytical phrase is not in itself a proper name:

sapta ca te, rsayaś ca-saptarsayah ("they are seven and are sagesthe Seven Sages")

*pūrvā câsau isukāmašamī ca—pūrvêsukāmašamī* ("it is both eastern and Isukāmašamī—East Isukāmašamī")

Example from a commentary:

Kum.M. 1.16 sapta ca te rsayaś ca saptarsayah. "dik-sankhye sañjñāyām" iti samāsah. they are seven and are sages seven-sages. The compound is by (Pā. 2.1.50) "Direction and number words (are compounded with a noun in apposition) in the sense of a proper name."

These items may also be used as the first member of a karmadhāraya compound under three more complex conditions (Pānini 2.1.51): when the compound involves an additional element of meaning equivalent to one of the senses of the suffixes used to form words by secondary derivation (Section 1.56.1), when the whole compound is itself the first member of a longer compound (Section 1.56.2), and (c) when the compound involves the added notion of aggregation by functioning as a collective noun (Section 1.56.3). Compounds beginning with numerals under any of these three conditions are called *dvigu*, and special methods of analysis are required for each of the three conditions:

108

### 1.56.1 Taddhitārtha-dvigu

In the "dvigu involving the sense of a taddhita suffix" the two members of the dvigu compound are placed in apposition, and something is added to indicate the additional element of meaning, using the techniques of analysis appplicable to words derived by the use of taddhita suffixes (Sections 1.35 ff.):

pañcasu kapāleșu samskrtah—pañca-kapālah ("prepared in five pots —five-pot")

Here the added word *samskrtah*, together with the locative case of the other words in dependence on it, serves to indicate the additional element of meaning. The word "dvigu" ("two-cow") that is used as the name of this whole class of compounds is in itself an example of this particular subclass and is analyzed dvābhyām gobhyām krītam ("bought with two cows"). Note that the same sort of analysis is used when the compound begins with a word signifying a direction:

pürvasyām śālāyām bhavah—paurva-śālah ("produced in the eastern hall, pertaining to the eastern hall—east-haller")

### 1.56.2 Uttarapada-dvigu

The "dvigu with a second member" is analyzed like any other longer compound (Section 1.69 ff.); the two members of the dvigu itself are therefore placed in apposition, using the shorter formula for  $karmadh\bar{a}rayas$  commonly employed in the analysis of longer compounds (Sections 1.54.1, 1.69):

pañca gāvo dhanam yasya: pañca-gava-dhanah ("whose wealth is five cows: five-cow-wealth")

The same principles apply when the first element is a word signifying a direction:

*pūrvā śālā priyā yasya—pūrva-śālā-priyaḥ* ("to whom the eastern hall is dear—east-hall-lover")

Examples from the commentaries:

Kum M. 2.17 [catur-mukha-samīritā] caturbhir mukhaiḥ samīritā satī. "taddhitârtha-"ity-ādinôttara-pada-samāsaḥ. samāhāre caturmukhîti syāt. four-mouth-uttered being uttered from four mouths. (This is) a compound with another word following in compound, by (Pā. 2.1.51) "(Number and direction words are compounded with a noun in apposition) to form a taddhita derivative, or when another word follows in composition, or when an aggregate is denoted." If an aggregate were denoted [the form] would be caturmukhī."

Ragh.M. 3.45 trayānām lokānām nāthas tri-loka-nāthah. "taddhitârtha-" ityādinôttara-pada-samāsah. tena [tri-loka-nāthena]. lord of the three worlds—three-world-lord. (This is) a compound with another word following in compound, by [Pā. 2.1.51] "To form a taddhita derivative," etc. By him, by the three-world-lord.

### 1.56.3 Samāhāra-dvigu

In the "dvigu of an aggregate," or collective dvigu, once again the two members of the dvigu appear in apposition in the analysis. The added notion of aggregation can be indicated in either of two ways:

 pañca gāvah samāhrtāh—pañca-gavam ("five cows taken together five-cow[-aggregate]")

trīņi bhuvanāni samāhrtāni: tribhuvanam ("the three worlds taken together: three-world[-aggregate]")

or (2) pañcānām gavām samāhārah--pañca-gavam ("an aggregate of five cows--five-cow")

*trayāņām bhuvanānām samāhārah—tribhuvanam* ("the aggregate of the three worlds—three-world")

Compare with this the analysis of collective *dvandva* compounds (Section 1.63).

In this particular subclass of compounds, words signifying directions cannot be used as the first member.

Examples from the commentaries:

110

#### Chapter 8. Nominal Compounds

Kum. V. 2.17 caturņām mukhānām samāhārah catur-mukham. pātrâdih. an aggregate of four mouths four-mouth. (But for a different interpretation see Mallinātha's comment on this word, quoted under Section 1.60 a.)

Kum.V. 2.30 sahasram nayanāni samāhrtāni sahasra-nayanam. "taddhitârthôttarapada-samāhāre ca." "dvigur ekavacanam." A thousand eyes collectively **thousand-eye**. (Pā.2.1.51:) "To form a *taddhita* derivative, or when another word follows in compound, or in the sense of an aggregate." (Pā. 2.4.1:) "A dvigu is singular in grammatical number."

Kum.M. 6.93 trayānām ahnām samāhāras try-ahah.

"taddhitârthôttarapada-samāhāre ca" iti samāsah. "rājâhah-sakhibhyaş tac" iti țac-pratyayah. dvigutvād ekavacanam. "rātrâhnâhāh pumsu" iti pum-lingatā. tasmāt try-ahād ūrdhvam an aggregate of three days—triduum. The compound is by (Pā. 2.1.51) "to form a taddhita derivative, or when another word follows in compound, or in the sense of an aggregate." There is the suffix -a (in place of the final -an of ahan) by (Pā. 5.4.91) "After rātri, ahan and sakhi there is TaC (at the end of a compound)." There is singular number because it is a dvigu. There is masculine gender because of (Pā. 2.4.29) "rātri, ahna, and ahah (at the end of a tatpurusa are) in the masculine." After it **after a triduum**.

# 1.57 Tatpurusas with inseparable first member

istatpuruşa

Many *tatpurusa* compounds—most of them *karmadhārayas*—cannot be analyzed in their own words because the first member of the compound is a prefix or other item that cannot stand independently as a separate word (Whitney 1281 ff.). These inseparable words fall into three broad groups: individual prefixes with prescribed meanings (Section 1.57.1), words included in the list of verbal prefixes ( $pr\bar{a}di$ ) (Section 1.57.2), and other items placed in compound with a following verbal form (Section 1.57.3).

### 1.57.1 ku, su, etc.

Most of the inseparable prefixes that are not included in the list of verbal prefixes occur only in certain specific meanings, so that in the analysis of compounds in which such prefixes are used a standard synonym may be used in place of the prefix. For example, the prefixes  $k\bar{a}$ - and kad- are replaced by *kutsita* ("censured") when they mean "bad," and by *īsad* when they mean "slightly, partly" (Pāṇini 6.3.101 ff.):

kutsitah purusah—kā-purusah ("bad person")

īşat puruşah—kā-puruşah ("semi-human")

kutsito 'śvah-kad-aśvah ("bad horse")

īşad uşnam-kad-uşnam ("lukewarm")

Similarly, the standard synonym *kutsita* is also used for the prefixes ku-(Pāṇini 2.2.18) and kim- (Pāṇini 2.1.64) when they mean "bad," and the standard synonym *īṣad* is also used for the prefix *kava*- (Pāṇini 6.3.107) when it means "slightly", e.g., in analyzing the compounds ku-purušaḥ ("bad person"), kim-sakhā ("bad friend"), and kavôsnam ("lukewarm").

Ragh.M. 1.2 dustaram taritum asakyam. "īṣad-duḥ-suṣu-"ity-ādinā khal-pratyayaḥ. hard-to-cross unable to be crossed. The suffix KHaL (i.e., -a), by (Pā. 3.3.126) "In connection with īṣad, duḥ, and su ...."

Ragh.M. 1.67 kavôṣṇam īṣad uṣṇaṃ yathā tathā ... kavôṣṇam iti ku-śabdasya kavâdeśaḥ. luke-warm so as to be slightly warm. ... In kavôṣṇam there is the substitution of kava for the word ku.

Ragh.M. 1.84 kôṣṇena kimcid uṣṇena. "kavam côṣṇe" iti ca-kārāt kâdeśah. luke-warm somewhat warm. There is the substitution of ka [for kava] because of the word "also" in (Pā. 6.3.107) "And kava also, when usna follows."

### 1.57.2 Prādi-tatpurusas with fixed meaning

The traditional list of verbal prefixes ( $pr\bar{a}di$ ,  $P\bar{a}nini$  1.4.58) includes a few items that are rarely prefixed to finite verbs; these are treated like the pre-

1.57. Tatpurusas with inseparable first member

fixes in Section 1.57.1. The prefix dur- ("bad"), for example, is often replaced by the standard synonym dusta in the analysis of a simple descriptive karmadhāraya: e.g., dusto janah—dur-janah ("a depraved person bad-person"). However, because these prefixes frequently have a rather vague meaning, the word that replaces them in the analysis may be chosen from a large list of synonyms: thus su-, when it means "good" (i.e., when it is used as an adjective modifying a following noun), may be replaced by  $s\bar{a}dhu$ ,  $sam\bar{i}c\bar{i}na$ , sobhana, subha,  $p\bar{u}jita$ , etc., and when it means "well" (i.e., when it is used as an adverb modigying a following adjective) it may be replaced by samyak, susthu,  $at\bar{i}va$ , etc.

Often more than one explanation is possible in analyzing these compounds. In the first example below, the commentator first interprets the compound as an *upapada tatpurusa*, then decides it is a *prādi tatpurusa*.

Kum.V. 1.1 adhirājata ity adhirājaḥ prabhuḥ. pacādy-ac. adhir aiśvarye. [nagâdhirāja iti] saṣthī-samāsaḥ. [yadi vā] adhikaś câsau rājā [ca] adhirājaḥ. "rājâhaḥ-sakhibhyaṣ ṭac." He rules over, thus overlord—master. (The krt suffix) aC [i.e., -a, which is added after the roots] pac, etc. (in the sense of an agent, by Pā. 3.1.134). (The prefix) adhi (is used) in the sense of lordship. mountain-overlord is a genitive (tatpuruṣa) compound. [Or rather] he is both superior [and] a king—over-king. (The suffix) TaC (i.e., -a, in place of the final -an of -rājan, by Pā. 5.4.91) "TaC after rājan, ahan, and sakhi (at the end of a compound)."

Kum.M. 1.1 adhiko rājā adhirājaķ. superior king over-king

Megh.M. 4 sobhanam ägatam svägatam. good coming welcome

Kum.M. 4.45 susthu caritârthāni [su-caritârthāni] ... well expressed well-expressed

# 1.57.3 Dhātu-lopa prādi-tatpurusa

A similarly straightforward analysis is sometimes used for compounds beginning with a prefix that *is* commonly used with finite verbs (e.g., *adhi* $r\bar{a}jah$ , "over-king," may be analyzed *adhiko*  $r\bar{a}j\bar{a}$ , "a higher king"), but more usually such compounds are analyzed by replacing the prefix with the past passive participle of some verb, joined together with the prefix in question. The particular verb chosen may differ even in two analyses of the same compound; e.g., *pra-vātah* ("strong wind") may be analyzed *prakrsto vātah* or *prabhūto vātah*. Such compounds are called *dhātu-lopa* ("involving the loss of a verbal root"—cf. Section 1.54.4).

When the first member of a prepositional compound is expanded by using the past passive participle of a verb in this way, the second member of the compound may appear in any of the first five cases in the analysis (*vārtikas* on Pāṇini 2.2.18):

(nom.) *pragatah pitāmahah—pra-pitāmahah* ("previous grandfather—great-grandfather")

(acc.) *atikrānto mālām—ati-mālaḥ* ("gone beyond the necklace [i.e., surpassing it in beauty]—beyond-necklace")

(instr.) *avakrustah kokilayā—ava-kokilah* ("called down to by the koil—down-koil")

(dat.) *parimlāno 'dhyayanāya—pary-adhyayanaḥ* ("[too] exhausted for study—beyond-study")

(abl.) nişkrāntah kausāmbyāh----nişkausāmbih ("gone out from Kausāmbi---outside-Kausāmbi")

Note that in the analysis of such compounds ("prepositional compounds with governed final member," Whitney 1310) the first member is syntactically predominant, despite the general rule that in *tatpurusas* the syntactical predominance belongs to the second member. In construction these compounds are thus akin to *avyayībhāvas*, but they are used as adjectives rather than as indeclinable adverbs. Whitney and some other Western scholars therefore group them together with *bahuvrīhis* under the heading of secondary adjective compounds (Whitney 1292 ff.), but because they do not fulfil the basic requirement that a *bahuvrīhi* must refer to something other than what is referred to by either of the members of the compound (something that is *ati-māla*, for example, is *ati* in a particular way), the Sanskrit tradition regards them as *tatpurusas* that form exceptions to the general rule by which it is the first member that adds some specification to the second.

#### Chapter 8. Nominal Compounds

Note also that, unlike other *tatpurusas* with an inseparable first member,  $pr\bar{a}di$ -tatpurusas with a governed final member cannot be regarded as *karmadhārayas*; once a substitute is provided for the inseparable members, they are analyzed exactly like ordinary dependent compounds (*vibhakti*tatpurusas, Section 1.53.1), except that the order of their members is reversed.

Ragh.M. 12.73 ati-śastrāḥ śastrāŋy atikrāntāḥ gone beyond weapons beyond-weapon

Ragh.M. 2.9 unmadānām utkața-madānām high-impassioned highly impassioned

Ragh.M. 3.41 atîndriyeşv indriyāny atikrānteşu. "atyādayah krāntādyarthe dvitīyayā" iti samāsah. super-sensory gone beyond the senses. The compound is by (the Vārtika rule) "ati, etc. (are compounded) with a word in the accusative, when the sense is 'advanced,' etc."

Different commentators may choose different ways of expanding the prefix:

Kum.V. 1.45 [M 46] pravāte prakrste vāte. in a strong-wind in a protracted wind

Kum.M. 1.46 *pravāte* prabhūta-vāta-sthale in a strong-wind in a powerful-wind-place

Kum.C. 1.46 pragato vāto yatra tat pravātam. where there is an advanced wind, that is strong-wind(ed)

### 1.57.4 Gati-tatpurusas

These formations (Pāṇini 2.2.18, Whitney 1090 ff.) involve prefixes compounded with verbal forms. When the prefixes *pra*-, etc. (Section 1.57.2) re used in conjunction with a verbal form they are called *gati* (Pāṇini 1.4.60), and this same designation also applies to other items used in a similar way (Pāṇini 1.4.61–79); e.g.,  $\bar{u}r\bar{i}-kr$  ("to accept"), *astaṃ-gam* ("to set, to go down"), *alaṃ-kr* ("to adorn"). For practical purposes nominal forms derived from verbs with a *gati* as the first member are regarded as

#### 1.57. Tatpurusas with inseparable first member

derivatives of the unit composed of *gati* plus verb, rather than as nominal compounds—e.g., *alam-krtah*, "adorned," is viewed as the past participle of the verb *alam-kr* rather than as a compound of *alam* and *krtah*—so that commentators regularly explain such items by giving a synonym or a technical analysis rather than an analysis in ordinary language. When it is necessary to bring out the meaning of the prefix, the formation may be analyzed by using some substitute (usually an adverb or a word in an oblique case functioning as an adverb) for the prefix; this technique is used to explain the meaning of the prefix occurs in conjunction with a finite verb, a participle, or a noun derived from a verbal root:

prakarsena jvalati-pra-jvalati ("it blazes strongly")

madhye hitam-antar-hitam ("placed in the middle-interposed")

viśesatah jñānam-vi-jñānam ("knowledge in terms of particulars-particular knowledge")

A gerund may also be used for this purpose:

Kum.M. 2.28 sambhūyâgatāh samāgatāh coming having gathered come together

In dealing with derivative nouns, however, the prefix may also be replaced with an adjective rather than an adverb:

visistam jñānam-vi-jñānam ("special knowledge").

One gati-formation that requires special analysis and that appears quite frequently is the compounding of a word ending in  $-\bar{i}$  (technically said to be formed with the suffix called *cvi*, Pāņini 5.4.50) and some form of the verb kr or  $bh\bar{u}$ . The analysis of such formations brings out the full meaning of a thing's becoming or being made something that it had not been before:

akrsnah krsnah sampadyate-krsnī-bhavati ("[having earlier been] non-black, it becomes black")

akṛṣṇaḥ sampadyamānaḥ-kṛṣṇī-bhūtaḥ ("having become black")

akrsnah sampadyate tam karoti-krsnī-karoti ("he makes it black")

# akrsnah sampadyamānah krtah-krsnī-krtah ("made black")

Similar formulas of analysis are used for other formations involving the same notion; e.g., *mithyā-karoti* ("he falsifies") may be analyzed *mithyā na bhūtvā mithyā sampadyate tat karoti* ("he makes it become false, not having been false [before]").

Analyses of *cvi*- formations have already been described under the heading of *taddhita* suffixes; see Section 1.42.4.

# 1.58 Negative tatpurusas

Negative *tatpuruşas* (*nañ-tatpuruşa*, Pāņini 6.3.73–77; Whitney 1288.1), sometimes called "alpha-privative" compounds, begin with *a*- or (before vowels) *an*-. They constitute a subclass of *karmadhārayas* with inseparable prefixes as the first member, and they require special formulas of analysis. Commentators frequently content themselves with a direct statement of the compound's meaning by using the sentence "[it is] not ...," with or without the verb "to be," followed by *iti*:

```
na brāhmana ity a-brāhmaņaļ ("not a brahmin, thus non-brahmin")
```

nâśva ity an-aśvah ("not a horse, thus non-horse")

*brāhmaņo na bhavatîty a-brāhmaņaḥ* ("[he] is not a brahmin, thus non-brahmin")

aśvo na bhavatîty an-aśvah ("[it] is not a horse, thus non-horse")

Actually, however, such compounds have a more precise meaning than is indicated in this formula of analysis: a non-brahmin, for example, is not anything that is not a brahmin, but rather someone who is like a brahmin (by being a member of the same more general category—i.e., human) and yet different from a brahmin, and similarly a non-horse is an animal that is other than a horse. To bring out this meaning commentators just as frequently use a stricter formula in analyzing these compounds:

*brāhmaņād anyah—a-brāhmaṇaḥ* ("other than a brahmin—non-brahmin")

119

aśvād anyah—an-aśvah ("other than a horse—non-horse")

Note that, as this formula indicates, negative *tatpuruşas* are anomalous in that the syntactical predominance belongs to the first member rather than to the second (cf. *prādi-tatpuruşas* with governed final members, Section 1.57.2).

Commentators often simply use the negative particle *na* in analyzing such compounds, especially when it is a verbal activity that is being negated:

*Kum.V. 1.10 na taila-pūrā a-taila-pūrā akrtrimatvāt* not oil-filled non-oil-filled nonartificial

Kum.M. 3.50 na kṣaratîty akṣaram avināśinam it does not perish, thus non-perishing neverending

# 1.59 Upapada compounds

These compounds (*upapada-tatpurusa*, Pāṇini 2.2.19, etc.; Whitney 1286– 1287) have as their last member a bound form, i.e., a word that occurs only at the end of a compound. In *tatpurusa* compounds that cannot be analyzed in their own words because the final member cannot stand as a separate word with the same meaning outside the compound, the usual situation is that the second member is a verbal derivative (i.e., a noun formed from a verbal root by primary derivation—see the sections on *krt* suffixes, Sections 1.27–1.34) that can appear with the meaning in question only when placed in compound with a preceding subordinate word (*upapada*). For example, *kumbha-kāraḥ*, "pot-maker," is an obligatory *upapada*-compound, because there is no independent word *kāraḥ* with the general meaning of "maker," whereas *payo-dharaḥ*, "water-bearer (i.e., cloud)" or "milkbearer (i.e., breast)," is an ordinary (and optional) case-*tatpuruṣa*, because *dharaḥ* can stand as a separate word meaning "bearer."

Note that a compound such as *satya-vādin*, "truth-speaker," is also an *upapada* compound even though there is an independent word  $v\bar{a}din$ , because  $v\bar{a}din$  as a separate word does not have the general meaning of "speaker," but rather more specific meanings, such as "debater." Thus a

1.60. Irregular tatpurusas

compound like *payo-charah* can be analyzed using the standard formula for ordinary *tatpurusas* (*payasām charah*, "bearer of water [or of milk]"), while compounds like *kumbha-kāraḥ* and *satya-vādin* require more round-about modes of analysis.

The usual formula involves replacing the second member with a finite verb, and displaying the first member separately in the required case. Because this produces a sentence rather than a phrase in apposition with the compound being analyzed, the analysis is followed by *iti*, just as in the analysis of negative *tatpurusas* (Section 1.58):

kumbham karotîti kumbha-kārah ("he makes a pot, thus pot-maker")

satyam vadatîti satya-vādī ("he speaks the truth, thus truth-speaker")

nagaram gacchatîti nagara-gāmī ("he goes to the city, thus city-goer")

As the examples show, the final member in an *upapada*-compound usually has active meaning, and the first member usually represents the direct object of that action. Where the second member has passive meaning, the first member may appear in an oblique case in the analysis, and the second member may be replaced either by a finite passive verb or by the equivalent participle (in which case no *iti* is needed):

aśvena krīyate sêty aśva-krītī ("she is bought with a horse, thus horse-bought")

manoh jāyata iti manu-jah ("[he] is born from Manu, thus Manuborn")

aśvena krītā—aśva-krītī ("bought with a horse—horse-bought")

manoh jätah-manu-jah ("born from Manu-Manu-born")

An alternative and simpler way of analyzing *upapada* compounds is to replace the second member with a synonym that can stand as a separate word; the compound can then be analyzed using the standard formula for ordinary dependent *tatpurusas*. For example, *mantra-krt* (*mantram karotîti*) can be analyzed *mantrānām* srastā, "creator of *mantras*."

Note that the basic formulas most commonly used in analyzing upapada-compounds are equivalent to the formulas employed in analyzing *krt*-suffix formations in general (Section 1.27 ff.). One deficiency in these formulas is that *upapada*-compounds often involve an additional element of meaning, in that the action referred to in the second member must be a characteristic or habitual one—a *nagara-gāmin*, for example, is not someone who just happens to be going to the city at this moment (as the analysis *nagaram gacchatîti* might imply), but rather someone who habitually or regularly goes to the city, whether he is going there now or not. To meet this problem, commentators sometimes prefer to analyze these compounds by using the word *sīlam* ("[characteristic or habitual] conduct, practice"), which is either employed in conjunction with an infinitive in a formula akin to that used in analyzing *bahuvrīhi* compounds (Section 1.64 ff.), or is simply placed in a *bahuvrīhi* compound with a verbal noun:

nagaram gantum sītlam asyêti nagara-gāmī ("his practice is to go to the city, thus city-goer")

nityam snātum sīlam asyêti nitya-snāyī ("his practice is to continually bathe, thus continual-bather")

*nagara-gāmī: nagara-gamana-śīlah* ("city-goer: having the practice of going to the city")

*madhura-bhāşiņī: madhura-bhāşaņa-sīlā* ("sweet-speaking: having the practice of speaking sweetly")

For examples of commentarial analyses of upapada compounds see the treatment of this topic under the heading of krt suffixes in Section 1.31.

# 1.60 Irregular tatpurusas

The common irregularities affecting the analysis of *tatpurusas* fall into two general groups.

### 1.60.1 Prsodarādi compounds

One group contains the compounds that are analyzed as being composed of words that appear in the compound in an inexplicably mutilated form.

1.60. Irregular tatpurusas

Once the supposedly original form of the mutilated member of the compound has been restored, the analysis is straightforward along the lines of the usual formulas. For example: *balāhakaḥ* ("cloud") is said to be a compound of *vāri* (which has collapsed into *ba*) and *vāhaka* (with its initial *v* changed into *l*), and is analyzed *vāriņāṃ vāhakaḥ* ("bearer of waters"); *śmaśānaḥ* ("cremation grounds") is said to be a compound of *śava* (which is replaced by *śma*) and *śayana* (changed into *sāna*), and is analyzed *śavānāṃ śayanaḥ* ("resting-place of corpses"). Commentators call such compounds *pṛṣodarâdi*, because they are grouped together (Pā. 6.3.109) under the example *pṛṣodaram*, a *bahuvrīhi* compound said to be made up of *pṛṣad* and *udaram*.

Kum.M. 1.4 vārīņām vāhakā **balāhakā**, prsôdarâditvāt sādhuh. carriers of water **clouds**. (The replacement of vāri by bala and of vāhaka by āhaka) is all right because (the compound belongs to) the prsôdarâdi group.

Ragh.M. 3.19 dyaur oko yeşām te divâukaso devāh. pṛṣôdarâditvāt sādhuh. teṣām [divâukasām] those whose dwelling is heaven are heaven-dwellers—gods. (The replacement of dyauh by diva) is all right because (the compound) belongs to the pṛṣôdarâdi group. Of them of the gods.

Ragh.M. 3.57 āśişi damştrāyām vişam yeşām te āsī-vişāh sarpāh. prşôdarâditvāt sādhuh. Those who have poison in the fang—tooth—are **fang-poisons** serpents. (The replacement of āśih by āśī) is all right because (the compound) belongs to the prsôdarâdi group.

Ragh.M. 3.59 veveşti vyāpnotîti vid vyāpakam ojo yasya sa tasya bidaujasah indrasya. prsodarâditvāt sādhuh it is active all over—it pervades—thus ubiquitous, pervasive; of him whose strength is (pervasive)—of Bidaujas of Indra. (The replacement of vid by bida) is all right because (the compound) belongs to the prsôdarâdi group.

Ragh.M. 1.11 manasa īsiņo manîşiņo dhīrāh, vidvāmsa iti yāvat. prsôdarâditvāt sādhuh. Having the wish of the mind intelligent—wise. To be precise, learned. (The compound is) all right because of its being (in the group) prsodarâdi. The Pāninian system of grammar speaks of other types of irregularities as well. For example, a Vārtika on Pā. 6.1.94, which deals with situations in which the second of two vowels is the single substitute for both, establishes a group of irregular compounds such as *śakandhu* (supposedly from *śaka* and *andhu*) in which this replacement occurs unexpectedly. Commentators may sometimes disagree even with themselves on how to explain irregular sandhi combinations in compounds; so Mallinātha, who explained the word *manīsin* as falling within the *prsodarâdi* group in the last example, places it (probably more exactly) within this *śakandhv-ādi* group in a commentary on another poem:

Kum.M. 1.28 manasa īṣā manīṣā sâsyâstîti manīṣī ... śakandhv-āditvāt sādhuḥ. The desire of the mind is intelligence; he who has it is intelligent ... (The compound is) all right because of its being (in the group) śakandhv-ādi.

Phonetic alterations within a compound can occur in many categories of compounds, and commentators make mention of other lists of irregularities as well, including the  $r\bar{a}ja$ -dantâdi group mentioned above in Section 1.53, which is used in the following example to justify a dvandva with supposedly reversed members and an irregular replacement:

Ragh.M. 1.35 **dam-patī** jāyā-patī. rāja-dantâdisu jāyā-sabdasya dam iti nipātanāt sādhuh. **married couple** husband-and-wife. (The compound) is all right (even though one would expect the form *pati-jāye*) because of the exceptional use of dam as a substitute for the word jāyā within the rāja-dantâdi group.

### 1.60.2 Mayūravyamsakādi compounds

The second group involves the compounds that are analyzed as having the syntactical predominance in the first member rather than the second, so that the order of the two members seems to have been reversed. We have already mentioned a few of these (see Section 1.51 and the sections referred to there, and Section 1.53.3). In the same category belong compounds beginning with *prāpta* or *āpanna*—e.g., *prāpta-yauvanā* ("[a girl] who has reached adolescence")—which might logically appear to be *bahuvrīhis* 

(analyzed, for example, *prāptaṃ yauvanaṃ yayā sā*, "she by whom adolescence has been reached") but which are traditionally (following Pāṇini 2.2.4) analyzed as accusative *tatpuruṣas* in which the order of the members has been reversed and in which the second member is the direct object of the first: thus *prāpta-yauvanā* is analyzed *prāptā yauvanam*, "(being) arrived at adolescence."

Compounds with reversed members that do not fall under specific rules (as the last example does) are called *mayūravyamsakādi* by the commentators, because they are grouped together (Pānini 2.1.72) under the example of the irregular *karmadhāraya* compound *mayūravyamsakah* ("peacock-cunning"), which is supposed to mean "cunning peacock." This group includes many other irregularities as well.

Of the types of compounds in this group involving *tatpurusas* in which the first member is syntactically predominant, two are especially noteworthy because of their great frequency: compounds ending in *-antaram* (cf. Section 2.19.1)—e.g., *anyo rājā—rājântaram* ("another king")—and compounds ending in *-mātram* (cf. Section 2.19.9—e.g., *cid eva—cin-mātram* ("just consciousness, pure consciousness"). For these two types of compounds the syntactical predominance of the first member is indicated by the standard formulas of analysis using *anya* and *eva*.

Ragh.M. 1.49 vanântarād anyasmād vanāt from other-forest from another forest

Kum. V. 1.56 [M 57] anyā mūrtir mūrty-antaram another form other-form

Among the other oddities included in the *mayūravyamsakādi* category are compounds containing indeclinables of various kinds:

Kum. V. 1.14 yad-rcchā-śabda icchârtho mayūra-vyamsakâdih. [The compound] yad-rcchā ("what(ever)-wish," meaning a whim, belongs to the mayūra-vyamsakâdi group.

Ragh.M. 5.15 na vidyate kimcana yasyêty **a-kimcana**h. mayūra-vyamsakâditvāt tatpurusah of whom there is not anything **anything-less**. (This is) a tatpurusa (which is all right) because of belonging to the mayūra-vyamsakâdi group.

### 1.60.3 Suffix-formations resembling compounds

Finally, it should be noted that, in some formations that look like compounds, what appears to be the second member is treated by the grammatical tradition as a suffix rather than as a noun in compound. This is true, for example, of *-rūpya* in words such as *samarūpya* ("formerly in the possession of an honest man") and *devadattarūpya* ("formerly belonging to Devadatta") (Pāṇini 4.3.81), and of *-maya* (technically called *mayaT*, Pāṇini 4.3.82, 4.3.143 ff., 4.3.148 ff.) in words such as *aśmamaya* ("made of stone") and *vrīhimaya* ("filled with rice"). In analyzing such formations, commentators use the formulas appropriate for *taddhita*-formations (Section 1.42 ff.) rather than the formulas applicable to nominal compounds (cf. Section 2.19).

Ragh.M. 1.17 rekhā pramānam asyêti rekhāmātram. īsad apîty arthah. "pramāņe dvayasac-"ity-ādinā mātrac-pratyayah. it has the size of a line, thus a mere line. The meaning is "even slight amount." The suffix mātraC (is added by Pā. 5.2.37) "-dvyasaC, daghnaC, and mātraC in the sense of size.

Ragh.M. 1.50 bhāga eva bhāga-dheyo 'mśah. "rūpa-nāma-bhogebhyo dheya-pratyayo vaktavyah" iti vaktavya-sūtrāt svâbhidheye dheya-pratyayah. A portion (bhāgadheya) is simply a portion (bhāga), a part. The suffix dheya (is added) in the sense of the base itself (i.e., without changing the meaning of the word to which it is added), by the Vārtika rule "It should be stated that the suffix -dheya (may be added) after rūpa, nāma, and bhoga."

# D. Dvandva Compounds

# **1.61 Introduction**

*Dvandva* compounds (Pāņini 2.2.29, etc.; "copulative compounds," Whitney 1252–1258) are divided into two major sub-categories of: simple copulative compounds (Section 1.62) and copulative compounds that involve the added notion of aggregation (Section 1.63).

1.64. Introduction

# 1.62 Itaretara dvandvas

Itaretara dvandva compounds (dvandva of "one and the other"; Whitney 1253.1) are cumulative compounds in which each member contributes equally to the total grammatical number (dual or plural) of the compound. For these compounds all that is necessary to accomplish an analysis is to separate the members and add ca after each to express the notion of conjunction; the standard formula is that already mentioned in Section 1.44: rāmaś ca kṛṣṇaś ca---rāma-kṛṣṇau ("Rāma and Kṛṣṇa"). The same formula applies no matter how many members the compound has: devāś ca gandharvāś ca mānuṣāś côragāś ca rākṣasāś ca--deva-gandharva-mānuṣôragarākṣasāḥ ("gods and fairies and humans and serpents and ogres").

Ragh.M. 1.1 pārvatī ca parameśvaraś ca pārvatī-parameśvarau. both Pārvatī and the Lord **Pārvatī-and-the-Lord** 

Kum.M. 2.41 chedāś ca pātāś ca cheda-pātās teṣām [cheda-pātānām] both cuttings and fallings, of them of cuttings-and-fallings

Kum. V. 2.7 strī ca pumāņš ca strī-puņsau. acaturâditvād ac. Woman and man woman-and-man. (There is the suffix) aC (i.e., -a, which replaces the final vowel and consonant of pumān) because of belonging to the list beginning with *a*-catura (in Pā. 5.4.77).

# 1.63 Samāhāra dvandvas

In this type of compounds ("dvandva of a combination," Pā.2.4.2; Whitney 1253.2), the two members are thought of as combining to operate as a unit, so the compound is always neuter singular (whereas the *itaretaradvandva* takes as its grammatical number the combined numbers of its members). More often than not commentators analyze these compounds using the simple formula applied to *itaretara-dvandvas*: e.g., *hasta-pādam* may be analyzed simply *hastaś ca pādaś ca* ("hand and foot"). When it is felt desirable to indicate in the analysis the added notion of aggregation, the techniques used are similar to those that are applied in the analysis of collective dvigu compounds (Section 1.56.3), so that two options are available: hastaś ca pādaś ca tayoh samāhārah ("hand and foot, the combination [or aggregate] of the two")

hastaś ca pādaś ca samāhrtau ("hand and foot taken together")

Examples from the commentaries:

Kum.M. 1.11 śronayaś ca payodharāś ca śroni-payodharam both hips and breasts hips-and-breasts

Kum. V. 2.5 caram câcaram ca carâcaram both moving and unmoving moving-and-unmoving

# E. Bahuvrīhi Compounds

# 1.64 Introduction

Compounds of the Bahuvrīhi category (Pānini 2.2.23-2.2.28; "possessive compounds," Whitney 1293-1308), are exocentric compounds referring to something outside the compound itself that is described by the compound. The distinctive feature of the formulas used to analyze bahuvrihi compounds is the relative pronoun that serves to express the relationship through which the compound refers to that outside thing denoted by neither of the compound's members individually. The other relationship that must be indicated in the analysis is that between the members of the compound, but in this the compound is nearly always closely analogous to one of the types of tatpurusas. To take the example used in Section 1.44dīrghau karnau yasya sa dīrgha-karnah ("he who has long ears is Longear")-the relative pronoun yasya indicates that the thing referred to by the whole compound is masculine in gender and singular in number, and is connected with long ears through a relationship expressed by the genitive case, while the relationship between dirgha and karna, as shown by the placing of the two words in apposition in the analysis, is exactly parallel to the relationship between the two members of a simple viśesana-karmadhāraya (Section 1.54.1). In terms of the ways in which they are analyzed,

### Chapter 8. Nominal Compounds

*bahuvrīhi* compounds can be divided into three basic groups: those that correspond in structure to *karmadhāraya* compounds (Section 1.65), those that correspond in structure to dependent *tatpurusas* (Section 1.66), and those special *bahuvrīhis* that cannot be directly compared in structure with any type of *tatpurusa* (Section 1.67).

# **1.65** Appositional *bahuvrīhis*

In these compounds ("samānādhikaraņa-bahuvrīhis," "appositional possessive compounds"), the relationship between the members is parallel to that between the members of a karmadhāraya, so that in the analysis the two members will usually be displayed in simple apposition (Section 1.65.1), with modifications for bahuvrīhis that correspond to special varieties of karmadhāraya: negative compounds (Section 1.65.2), prepositional compounds (Section 1.65.3), and a special form of the latter, compounds with saha or sa as the first member (Section 1.65.4). The basic principle of apposition applies also to bahuvrīhis analyzable as having more than two basic members (Section 1.65.5).

### 1.65.1 Simple appositional bahuvrīhis

In the analysis of these compounds the two members are simply placed in apposition, as in the analysis of *karmadhārayas* of the adjective-plussubstantive type (Section 1.54.1), or, more rarely, of the simple substantiveplus-substantive type (Section 1.54.3). differences in the analysis arise not in the treatment of the two members themselves, but in the case of the relative pronoun used to indicate the relationship thorugh which the entire compound refers to some outside object; depending upon the precise nature of this relationship the pronoun can appear in any of the six oblique cases (in order to make the meaning of the compounds clearer in the following examples, a word denoting the outside object to which the entire compound refers is added in square brackets):

- (acc.) prāptā atithayo yam sa prāptâtithih [grāmah] ("that [village] which guests have reached is reached-guested")
- (instr.) *ūdho ratho yena sa ūdha-rathaḥ [anadvān]* ("that [ox] by which a cart has been pulled is pulled-carted")

1.65. Appositional bahuvrīhis

- (dat.) upahrtah paśur yasmai sa upahrta-paśuh [rudrah] ("He [Rudra] for whom a beast has been sacrificed is sacrificedbeasted")
- (abl.) uddhrta odano yasyāḥ sā uddhrtâudanā [sthalī]
   ("that (dish) from which the rice has been removed is removed-riced")
- (gen.) pītam ambaram yasya sa pītâmbarah [harih] ("He [Hari] of whom there is a yellow garment is yellow-garmented")
- (loc.) vīrāḥ puruṣā yasmin sa vīra-puruṣaḥ [grāmaḥ] ("that [village] in which there are heroic men is hero-manned")

Note that except when the pronoun is in the genitive or locative the first member of the compound is usually a past passive participle. The genitive is most common, followed by the instrumental and locative; in *bahuvrīhis* of most types other than this simple appositional type only the genitive occurs.

genitive pronoun:

Ragh.M. 1.15 sadrśa ārambhah karma yasya sa tathôktah [sadrśârambhah] he who has commensurate undertaking---activity---is so-called [commensurate-activitied]

Ragh.M. 1.55 nayah sāstram eva caksus tattvâvedakam pramāņam yasya tasmai naya-caksuse. to him for whose eye—authority for determining the truth—is policy—theory to the policy-eyed one

Kum.M. 1.53 śobhanā dantā yasyāh sā su-datī she who has beautiful teeth is beautiful-toothed

Kum.M. 2.20 kunthitä aśrayo yasya tat kunthitâśri kunthita-koți. that which has blunted corners is blunt-cornered blunt-edged

instrumental pronoun:

Kum.M. 1.6 hatā dvipā gajā yais tesām hata-dvipānām kesariņām simhānām of those by whom elephants—pachyderms—have been killed of the killed-elephant ones Kum.M. 1.1 [kampita-devadāruḥ] kampitā devadāravo yena sa tathôktah. shaken-pine: he by whom the pines are shaken is so-called

Kum.M. 1.25 labdha udayo yayā sā labdhodayā. utpannêty arthah she by whom birth is obtained is **obtained-birthed**, i.e., born

Kum.M. 2.22 apaviddhā tyaktā gadā yena saḥ apaviddha-gadaḥ. he by whom the mace was discarded—abandoned—is abandoned-maced

### locative pronoun:

Ragh.M. 1.18 sahasram gunā yasmin karmaņi tad yathā tathā sahasra-guņam sahasradhā. in which action there are a thousand repetitions, that; in such a way, a thousand-fold a thousand times over

### 1.65.2 Negative bahuvrīhis

Negative *bahuvrīhis* (*nañ-bahuvrīhi* compounds) are related to the negative *tatpuruṣas* (Section 1.58). The standard formula of analysis uses the present participle *avidyamāna*:

avidyamānā bhāryā yasya so 'bhāryaḥ ("he is a-bhārya [wifeless] of whom a wife does not exist," "he is wifeless who does not have a wife")

avidyamānah putro yasya so 'putrah ("he is a-putra [sonless] of whom a son does not exist," "he is sonless who has no son")

Other formulas of equivalent meaning are also used; thus *a-putrah* ("sonless") may also be analyzed *putro nâsti yasya* or *putro na bhavati yasya* ("of whom there is no son").

Ragh.M. 1.30 avidyamānam anyasya rājnah sāsanam yasyās tām an-anya-sāsanām urvīm her for whom the rule of another king does not exist the un-other-ruled earth

### 1.65. Appositional bahuvrihis

### 1.65.3 Prādi-bahuvrīhi compounds

prepositional *bahuvrīhis* correspond to *prādi-tatpuruṣas* (Section 1.57.2), and in their analysis the prefix that constitutes the first member of the compound is replaced by a past passive participle with the prefix in question, just as in the analysis of prepositional *tatpuruṣas*:

prapatitam parnam yasya sa pra-parnah ("that is away-leaved whose leaf has fallen away")

udgatā kandharā yasya sa ut-kandharah ("that is up-necked whose neck is lifted up")

Examples from the commentaries:

Ragh.M. 2.22 purah-phalāni purogatāni pratyāsannāni phalām yesām tāni hi. ahead-fruited those whose fruits are located ahead—in the near future

Kum. V. 4.1 vigato nasto dhavah patir yasyāh sā vi-dhavā she whose husband—spouse—is gone away—perished—is husbandless

Kum.M. 3.75 ucchirasah unnata-siraso mahatah high-headed uplifted-headed—great

Note in the last example that the commentator simply inserted the expansion into the compound without bothering to offer an analysis.

### 1.65.4 Saha-bahuvrīhi compounds

Bahuvrīhis beginning with saha ("with") or its substitute sa- might be expected to be analyzed along the lines of tatpurusas with inseparable prefixes as the first member (thus sa-putrah, "with his son," might be analyzed sahitah putro yasya, "whose son is together [with him]"), and some commentators use such a formula. But many such compounds are technically limited to instances in which both the thing referred to by the the entire compound and the thing denoted by the second member are equally involved in some action—e.g., sa-putro gatah, "He went with his son" so that some commentators prefer the stricter and simpler formula saha putrena, "together with his son." Although this analysis makes the compound look like a dependent *tatpurusa*, such compounds are still classed as *bahuvrīhis*.

In practice commentators using the *saha* formula usually employ it in connection with a form of the verb *vartate*:

Ragh.M. 1.81 saha patnyā vartata iti sa-patnīkah san. "nady-rtaś ca" iti kap-pratyayah he is with his wife, thus being with-wifed. The suffix kaP (i.e., -ka, is added by Pā. 5.4.153) "and (after a bahuvrīhi) ending in the feminine endings i or  $\check{}$ , or in r."

Ragh.M. 2.23 sa-dārasya dārair arundhatyā saha vartamānasya with-wifed being with his wife Arundhatī

These compounds must be distinguished from those bahuvrīhis in which sa-stands not for saha but for samāna ("same," "shared") or sadrša ("similar"). Examples are sa-varņaḥ "having the same social class" (samāno varņo yasya sa sa-varṇaḥ) and sa-tīrthaḥ "having the same teacher" (lit., "having the same font").

# 1.65.5 Anekapada-bahuvrīhi compounds

Aneka-pada or "multiword" bahuvrihis must be analyzed as consisting of more than two members. Ordinarily compounds containing many wordswith the exception of certain dvandvas (see end of Section 1.62)-can be analyzed in the first instance as compounds of only two members, by combining words in such a way that either or both of the two basic members is itself a compound. But because of the numerous rules governing the retention of feminine endings and the addition of suffixes in bahuvrihi compounds, it is sometimes necessary to analyze an appositional bahuvrīhi compound as being composed of three or more basic members. In the compound citra-jarati-guh ("having a spotted old cow"), for example, we cannot combine the first two words ("having a spotted-and-old cow") because the form of the compound would then have to be jarac-citra-guh, and we cannot combine the last two words ("having a spotted old-cow") because the form would have to be citrā-jarad-gavīkah. We are therefore stuck with three members, but the resulting analysis is fortunately straightforward-the three are simply put in apposition: citrā jaratī gaur yasya, "of whom there is a spotted old cow." Similarly *dīrghā tanvī jarighā* yasya sa *dīrghā-tanvī-jarighaḥ*, "he is Long-thin-shank who has a long thin shank."

# 1.66 Non-appositional bahuvrīhis

A few types of *bahuvrīhi* compounds are analyzed as *vyadhikaraņa* or nonappositional. For these few *bahuvrīhis* the precise way in which the first member of the compound adds some specification to the second member is most conveniently demonstrated by displaying one of the two members in an oblique case in the analysis. This sort of analysis is used for two particular types of compound:

### 1.66.1 Locative as second member

*śaśī śekhare yasya sa śaśi-śekharah* ("he is moon-crested who has a moon on his crest")

cakram pānau yasya sa cakra-pānih ("he is discus-handed who has a discus in his hand")

Example:

Kum.M. 3.10 pinākaļp pāņau yasya sa pināka-pāņiļ.... tasya [pināka-pāņeļi]. in whose hand is (the bow) Pināka, he is Pināka-handed.... Of him [of the Pināka-handed one].

## 1.66.2 Genitive as first member

ustrasyêva mukham yasya sa ustra-mukhah ("he is camel-faced who has a face like a camel's")

candrasyêva kāntir yasya sa candra-kāntih ("he is moon-splendored who has splendor like the moon's")

In analyzing this last type of compound, many commentators prefer a fuller formula:

ustrasya mukham iva mukham yasya sa ustra-mukhah ("he is camel-faced who has a face like a camel's face")

candrasya käntir iva käntir yasya sa candra-käntih ("he is moon-splendored who has splendor like the moon's splendor")

Such compounds are considered another form of *uttara-pada-lopa-samāsa* (see end of Section 1.54.4), since the compound can be interpreted as standing for *ustramukha-mukhaḥ*, *candrakānti-kāntiḥ*, etc.

Ragh.M. 1.8 muninām vrttir iva vrttir yeşām teşām [muni-vrttīnām] of those who have a life like the life of sage of the sage-lived ones

Ragh.M. 1.13 vrsasya skandha iva skandho yasya sa tathā [vrsa-skandhaḥ].... uttara-pada-lopī bahu-vrīhiḥ. he whose shoulder is like the shoulder of a bull is so [bull-shouldered.... A bahuvrīhi with loss of the latter member (of the first word).

Kum.M. 1.11 aśvānām mukhāniva mukhāni yāsām tāh aśva-mukhyah kinnara-striyah. uṣṭra-mukhavat samāsah. those who have faces like the faces of horses are **horse-faced** Kinnara women. The compound is like "camel-faced."

Kum.M. 3.54 tarunârkasya rāga iva rāgo yasya tat [tarunārka-rāgam] bālârkârunam ity arthah. upamānapūrvapado bahuvrīhir uttarapadalopaś ca. that which has redness like the redness of the young sun is young-sun-rednessed, i.e., as pink as the young sun. A bahuvrīhi with a standard of comparison as the first member, and with loss of the latter member (of the first word).

Kum.M. 4.3 purusasyâkrtir ivâkrtir yasya tat purusâkrti that which has a shape like the shape of a man is man-shaped

Ragh.M. 3.55 kumārasya skandasya vikrama iva vikramo yasya sa tathoktaḥ [kumāra-vikramaḥ]. He whose power is like the power of Kumāra—of Skandha—is so-called [Kumāra-powered].

# 1.67 Special *bahuvrīhis*

The standard technique of using a relative pronoun in an oblique case while analyzing *bahuvrīhis* does not apply in three peculiar types of *bahuvrīhis*, involving numerals (Section 1.67.1), intermediate points of the compass (Section 1.67.2), and indeclinable descriptions of reciprocal combat (Section 1.67.3). The other peculiarities observed in the analysis of *bahuvrīhis* correspond for the most part to the irregularities found in *tatpurusa* compounds (Section 1.68).

### 1.67.1 Samkhyā-bahuvrīhi compounds

*Saṃkhyā-bahuvrīhis* ("*bahuvrīhi* of numerals, Pāṇini 2.2.25) fall into three main groups, for each of which the peculiar form of the standard analysis is self-explanatory:

 (1) daśānām āsannāh—āsanna-daśāh ("around ten") daśānām adūrāh—adūra-daśāh ("not far from ten") daśānām adhikāh—adhika-daśāh ("more than ten") daśānām samīpe ye vartante te upa-daśāh ("those which are in the vicinity of ten are 'around ten'")

The first member of this sort of compound must be one of the four listed here. Note that these are akin to genitive *tatpuruṣas* with reversed members (with the last, "*upa*-..." being like an *avyayībhāva*, and glossed in the same way as *upa-kumbha*, Section 1.50).

(2) dvau vā trayo vā—dvi-trāh ("two or three")
 trayo vā catvāro vā—tri-caturah ("three or four")

Note that these are akin to *dvandva* compounds. Commentators sometimes leave out one of the  $v\bar{a}$  words:

Ragh.M. 5.25 dvi-trāni dve [vā] trīni vā. two-or-three [either] two or three.

(3) dvih (āvrttāh) daša—dvi-dašāh ("ten [repeated] twice," i.e., twenty)

These are like our compounds "two score," etc.

1.69. Introduction

Kum.M. 3.1 trir āvŗttā daśa parimāņam eṣām iti **tridašān** devān. "sankhyayâvyayâsannâdūrâdhika-sankhyāḥ sankhyeye" iti bahuvrīhi. their extent is ten repeated thrice, thus **thirty** gods. A bahuvrīhi by (Pā. 2.2.25) "An indeclinable, *āsanna, adūra, adhika*, and a numeral [is compounded] with a numeral, in the meaning of a number."

### 1.67.2 Dig-bahuvrīhi compounds

*Dig-bahuvrīhis* ("*bahuvrīhis* of direction," Pāņini 2.2.26) form names of the intermediate points of the compass:

daksinasyah purvasyas ca disor (yad) antaralam (sa) daksina-purva (dik) ("[that direction is] southeastern [which is] the interval between the southern and eastern directions")

uttarasyāh pūrvasyās ca disor antarālam—uttara-pūrvā ("the interval between the North and the East—Northeast")

### 1.67.3 Reciprocal combat

*Bahuvrīhis* describing reciprocal combat (Pāṇini 2.2.27) are indeclinable compounds; in their analysis the members are displayed in the locative when the word  $grhītv\bar{a}$  ("grasping") is added, and in the instrumental when the word *prahrtya* ("striking") is added:

keśesu keśesu ca grhītvêdam yuddham pravrttam—keśâkeśi ("this battle proceeded grasping each other by the hair—hair-to-hair")

dandair dandaiś ca prahrtyêdam yuddham vrttam—dandâdandi ("this battle occurred striking each other with staves—staff-against-staff")

# 1.68 Irregular bahuvrīhis

These correspond in general to the types of irregularities found in the form of *tatpuruşa* on which the *bahuvrīhi* may be said to be based. Reversal of members is most common in *bahuvrīhi*s in which what would normally

be the first member is a past passive participle; those compounds of this type in which the reversal is optional are said by the commentators to belong to the *āhitâgny-ādi* group (see the last example below). Mutilation of members (e.g., in *prsodarādi* compounds—see Section 1.58) also does not affect the basic formula used in analyzing the compound, and the same is true of the numerous suffixes added at the end of *bahuvrīhi* compounds.

Ragh.M. 1.23 dharme ratir yasya tasya [dharma-rateh] of him whose love was for the law of the law-love one

Kum.V. 2.18 yuga-pramānā bāhavo yesām tebhyo yuga-bāhubhya ājānu-bhujebhyah. to those whose arms are pole-length to the pole-armed ones to those whose arms reach to their knees.

Kum.M. 3.72 bhavasya netrāj janma yasya sa bhava-netra-janmā. "avarjyo bahuvrīhir vyadhikaraņo janmâdy-uttarapadaḥ" iti vāmanaḥ. He whose birth was from the eye of Bhava is **Bhava-eye-birthed**. Vāmana says that a nonappositional bahuvrīhi need not be avoided if the last member is a word such as "birth."

Kum.M. 2.47 **deha-baddham** baddha-deham [i.e., baddho deho yena]. mūrtimad ity arthah. āhitâgny-āditvān nisthāyāh para-nipātah. **body-assumed** assumed-bodied [i.e., by whom a body has been assumed]. The meaning is embodied. Irregular reversal of members because of : belonging to the group āhitâgny-ādi (āhitâgni, "established-fired," i.e., by whom the ritual fire is maintained, is a compound that also appears in the reversed form agny-āhita).

# F. Longer Compounds

# 1.69 Introduction

Compounds containing more than two words can usually be viewed as consisting of two members, one or both of which happens to be a compound in itself. The principle exceptions are *dvandva* compounds, where

1.70. General principles

the number of members in the compound does not fundamentally alter the method of analysis (Section 1.62), and certain rare *bahuvrīhi* compounds (Section 1.65.5). Longer compounds are analyzed by breaking them down into their parts, and analyzing those parts as well, using essentially the same techniques as in the analysis of simple compounds containing only two words. The differences from the treatment of simpler compounds consist primarily of techniques used to avoid the tedious repetition of parts of the overall compound. The general principle is one of omitting as much as possible of this repetition with the help of relative pronouns (Section 1.70), and this principle is supplemented where useful by the use of simpler formulas of analysis for some types of compounds or by other special techniques (Section 1.71). Commentators often show individual preferences in their choice of the available options (Section 1.72).

# 1.70 General principles

In analyzing a longer compound the standard formulas of analysis, if used in their full form for each of the compounds forming part of the overall compound, would generate a tedious amount of repetition, and commentators often manage to omit much of this repetition, using relative pronouns where necessary to preserve an indication of how the parts fit together. We may take as a simple example a compound in *Kumārasambhava* 1.60, *tacchiraś-candra-pādāh* (i.e., *tat-śiras-candra-pādāh*), meaning "the beams of the moon on his (Śiva's) head." While an English translation of longer compounds such as this usually starts from the end of the compound, the analysis in Sanskrit usually starts from the beginning; thus *tat* is the first member of the genitive *tatpuruṣa* compound *tac-chirah*, which in turn is the first member of the locative *tatpuruṣa* compound *tacchiraś-candrah*, which in turn is the first member of the entire genitive *tatpuruṣa* compound *tacchiraścandra-pādāh*. A full analysis of the compound, using the standard formulas for these types of compounds, would be as follows:

 tasya śirah: tac-chirah, tacchirasi candrah: tacchiraś-candrah, tacchiraścandrasya pādāh: tacchiraścandra-pādāh

his head: his-head,

the moon on his-head: his-head-moon, the beams of the his-head-moon: his-head-moon-beams

Of course such an analysis is unnecessarily cumbersome, and scarcely any commentator would be so tedious (but see the first example in Section 1.72). The repetition of the compounds arrived at in each step of the analysis can be avoided by using a pronoun to replace each compound when marking its case as the first member of the next larger compound:

(2) tasya śirah: tac-chirah, tasmin candrah: tacchiraś-candrah, tasya pādāh: tacchiraścandra-pādāh

his head: his-head, the moon on it: his-head-moon, its beams: his-head-moon-beams

But even this single mention of the form of each intermediate compound is more than most commentators find necessary. It can be omitted by relying on these pronouns to indicate the connection between the parts of the entire compound. When this is done the only further necessity is to make it clear what each such pronoun refers to, and this may be done by supplying relative pronouns. If the part being referred to by the pronoun is itself one of those types of compounds analyzed by a formula ending with a pronoun whose antecedent is clear—e.g., a *bahuvrīhi* ending with "... yasya saḥ"—this problem is automatically taken care of. In our example, each of the constituent compounds is a *tatpuruşa* in which the last member is syntactically predominant; for these it is therefore possible simply to insert a relative pronoun in apposition with the last member of each successive part:

(3) tasya yac chiras tasmin yaś candras tasya pādāh

the beams of that moon which is on that head which is his

This process of simplification is carried to its extreme in the analysis of the compound offered by the commentator Mallinātha:

(4) tasya giriśasya śirasi candrasya pādāh

### the beams of the moon on his—Śiva's—head

(For the inserted gloss giriśasya see Section 1.21.) The only problem with this simplest form of analysis is that, while it indicates the syntactical connections between the individual words in the whole compound, it leaves unclear the precise arrangement of these words into subordinate compounds within the whole—in this instance, it may be unclear whether the final compound is composed of *tac-chiraḥ* plus *candra-pādāḥ*, or of *tac-chiraḥ* plus *pādāḥ*. The analysis offered by the commentator Cāritravardhana avoids this problem by using a single relative pronoun:

(5) tasya śambhoh śirasi yaś candras tasya pādāh

the beams of that moon which is on his-Śiva's-head

This makes it clear that he divides the compound into *tac-chiraś-candra*<sup>h</sup> plus *pādā*<sup>h</sup>.

Relative pronouns of this sort are frequently used to avoid what would otherwise result in uncertainty or misapprehension:

Kum.M. 1.16 [agra-saro-ruhāņi] agre upari yāni sarāņsi teşu ruhāņi padmāni [highest-pond-growing:] growing in those ponds that are highest—on top—lotuses. [Here the pronouns are necessary to prevent the reader from taking saroruha as a compound.]

### Ragh.M. 1.58 [ātithya-kriyā-śānta-ratha-kşobha-pariśramam]

*ātithyasya* [yā] kriyā tayā śānto ratha-kşobhena yah pariśramah sa yasya sa tam tathoktam [hospitality-rite-pacified-chariot-jolt-fatigue:] he is so-called who has that fatigue, due to chariot-jolts, which is pacified by that [which is] the rite of hospitality. [Here the tayā after kriyā is to prevent the reader from taking *ātithyasya* with anything other than kriyā.]

Ragh.M. 1.51 [ālavālâmbu-pāyinām] ālavālesu jalâvāpa-pradešesu yad ambu tat-pāyinām [trough-water-drinkers:] drinkers of that water which is in troughs—in watering places.

Ragh.M. 1.4 krta-vāg-dvāre krtam rāmāyaņâdi-prabandha-rūpā yā vāk sâiva dvāram pravešo yasya tasmin. in the one having a made-speech-door: in that which has a door—entrance—made that is nothing other than that which is speech, in the form of the *Rāmāyaņa*, etc. Kum.M. 1.47 [sva-cāpa-saundarya-madam] sva-cāpa-saundaryeņa yo madah tam [own-bow-beauty-pride:] that pride which is due to his own-bow-beauty

# **1.71** Special techniques

The sort of straightforward analysis just described works well for compounds made up of a string of dependent *tatpuruşas*, but it can be interrupted when an element within the larger compound is one that requires a more complicated formula. Commentators do what they can to streamline the process of analysis, usually by employing a simpler formula of analysis than that applied to compounds in isolation (Section 1.71.1), and if this is not possible they will analyze a constituent compound separately from the larger compound (Section 1.71.2).

### **1.71.1** Simpler formulas of analysis

By far the most common instance of simplified formulas is in the analysis of *karmadhārayas* forming part of a larger compound; in this situation they are often analyzed simply by placing the members of the *karmadhāraya* in apposition. In *Kumārasambhava* 4.29, for example, the bow of Kāma is described as *pelava-puspa-patri*, "having tender blossoms as its arrows," and each of the major commentators on the poem gives as the analysis of this *pelavāni puspāny eva patriņo yasya tat*, "that which has arrows that are tender blossoms," even though in isolation the *karmadhāraya* at the beginning of the compound would always be analyzed *pelavāni ca tāni puspāni ca* (Section 1.54.1), rather than *pelavāni puspāni*.

### **1.71.2** Separate analysis

Often the formula required for the analysis of a constituent compound cannot be fit smoothly in the analysis of the larger compound, and some part of the larger compound must then be analyzed separately, before beginning the continuous analysis of the whole. In *Kumārasambhava* 1.11, the *kinnara* women are described as *durvaha-śroni-payodharârtāh*, "tormented by their burdensome hips and breasts," where *śroni-payodharam*  is a *samāhāra-dvandva* compound (Section 1.63). The commentators first analyze this part separately, and then treat it as a unit in the analysis of the larger compound:

Kum.M. 1.11 śronayaś ca payodharāś ca śroni-payodharam; durvahena durdharena śroni-payodharenârtāh pīditāh hips and breasts: hips-and-breasts; tormented—oppressed—by burdensome—heavy—hips-and-breasts 「「「「「「「「」」」

Notice that the first member of this whole compound, *durvaha* ("burdensome, difficult to carry") is itself a compound, and could be given an analysis of its own—e.g., *krcchrenôhyata iti durvaham*, "it is carried with difficulty, thus difficult-to-carry"—which would then be converted to the instrumental case for the analysis of the rest of the larger compound by adding *tena* after its analysis. But commentators very frequently choose not to bother with the analysis of items whose construction is obvious, and will often simply provide a synonym instead, as Mallinātha has done here (*durdharena*).

# 1.72 Choice of options

The analyses of any longer compound offered by different commentators may differ, aside from actual differences of opinion on the grammatical construction of the compound, depending on the frequency with which each commentator chooses not to analyze constituent parts of the whole or to insert explanatory glosses, and also depending on the extent to which each commentator chooses to use the inserted relative pronouns described in Section 1.70. A simple example is *Kumārasambhava* 1.10, where phosphorescent plants (*oṣadhayah*) are described as *darī-gṛhôtsanga-niṣaktabhāsaḥ*, "having light clinging in the interior of cave-houses." An anonymous commentary (inserted in a conflated version of Vallabhadeva's commentary, BORI Devanāgarī MS No. 379 of 1887–91) analyzes this compound in the most tedious way possible, with no inserted glosses or explanatory material:

kim-visistä osadhayah? darī-grhôtsanga-nisakta-bhāsah. darya eva grham darī-grham. darī-grhānām utsango darī-grhôtsangah. 1.72. Choice of options

*darī-grhôtsange nişaktā bhāso yāsām tāḥ.* What sort of plants? **Having cave-house-interior-clinging-light**. Houses that are actually caves: cave-houses. The interior of the cave-houses: cave-house-interior. Whose light is clinging in the cave-house-interior, those.

The major commentators use much more streamlined methods of analysis, with a judicious number of synonyms thrown in:

Kum.C. 1.10 kīdrśya oṣadhayaḥ? darī-grhôtsanga-niṣakta-bhāsaḥ. darya eva guhā eva grhās teṣām utsange niṣaktāḥ praviṣtā bhāso yāsām tāḥ. What sort of plants? Having cave-house-interior-clinging-light. Those that had light clinging—penetrating—in the interior of those houses that were actually caves—actually grottoes.

Kum.M. 1.10 daryah kandarā eva grhās tesām utsangesv abhyantaresu nisaktāh samkrāntā bhāso yāsām tāh. Those that had light clinging—reflected—in the interiors—insides—of those houses that were actually caves—grottoes.

Kum.V. 1.10 daryah kandarā eva grhās tad-utsange nisaktā lagnā bhāso yāsām tāh. Those that had light clinging—adhering—in the interior of those houses that were actually caves—grottoes.

Here the three commentators differ somewhat in the choice of synonyms they offer, but in the actual analysis of the compound the commentator Cāritravardhana differs from Mallinātha only in interpreting *utsanga* as singular rather than plural. Vallabhadeva also prefers the singular here, and deviates in a minor way from the other two commentators in using the more compressed form *tad-utsange* in place of *tesām utsange*.

The style of each commentator is often marked by consistent preferences in these matters. For example, a preference for singular or plural, wherever there is a choice, characteristically separates Vallabhadeva from Mallinātha throughout their commentaries: when they differ on this point, it is consistently Vallabhadeva who gives the singular and Mallinātha who gives the plural.

To illustrate further these choices and to introduce several other aspects of the analysis of long compounds, here are three long compounds with analyses offered by these two commentators. In the quotations from

1.72. Choice of options

the commentaries, all inserted explanatory material and glosses have been deleted, so as not to distract attention from the analyses themselves. Notice that in the analysis of longer compounds such as these, Vallabhadeva differs from Mallinātha in another way as well: Vallabhadeva consistently uses the technique of inserting relative pronouns with following correlatives, while Mallinātha consistently avoids the use of pronouns wherever they are not absolutely necessary.

First, a long tatpuruşa compound (*Śiśupālavadha* 4.20): vilambi-ghaņţādvaya-parivārita-vāraņêndra-līlā, "the grace of a mighty elephant girded by a hanging pair of bells." Vallabhadeva analyzes this as vilambi yad ghanţā-dvayam tena parivārito yo vāraņêndras tasya līlā, "the grace of that elephant-lord which is girded by that bell-pair which is hanging." The analysis that Mallinātha's gives is vilambinā ghanţā-dvayena parivāritasya vāraņêndrasya līlā, "the grace of an elephant-lord girded by a hanging bellpair." Note that neither commentator bothers to analyze ghanţā-dvaya or vāranêndra.

Second, a long bahuvrīhi compound (Šisupālavadha 3.80): uttāla-tālīvana-sampravrtta-samīra-sīmantita-ketakīkāḥ, "(regions) where the screw pines are parted by the breeze moving forth from the tall grove(s) of palms." Vallabhadeva's analysis is uttālam yat tālī-vanam tasmāt sampravrtto yah samīras tena simantitāh ketakyo yatra te tathôktāḥ, "Those (regions) are so-called in which (there are) screw pines parted by that breeze which is moving forth from that palm-grove which is tall." Mallināthā's analysis is uttāleṣu tālī-vaneṣu sampravrttena samīreṇa sīmantitāḥ ketakyo yeṣu te tathôktāḥ, "Those (regions) are so-called in which (there are) screw pines parted by the breeze moving forth in tall palm-groves." Again, neither commentator bothers to analyze tālī-vana. Notice also that while Vallabhadeva interprets vana as ablative singular, Mallinātha presents it as locative plural—an instance of a characteristic difference already described.

The formula te tath $\hat{o}ktah$  ("they are so-called"), and the simpler te tath $\bar{a}$  ("they are thus," see the next example), are used very frequently at the end of an analysis requiring a pronoun to restitute the whole compound after the use of a relative pronoun (as in the analysis of bahuvrīhi compounds) or *iti* (as in the analysis of some upapada compounds). These formulas are designed to avoid the necessity of repeating the whole compound at the end, and are used even with simple two-member compounds. Thus the bahuvrīhi compound dīrgha-karņa ("Long-ear") can be analyzed dīrghau

karnau yasya sa dīrgha-karnah, or dīrghau karnau yasya sa tathôktah, or simply dīrghau karnau yasya sa tathā.

Finally, an example of a long upapada compound (Sisupālavadha 5.12): vanyêbha-mastaka-nikhāta-nakhâgra-mukta-muktāphala-prakara-bhañji, "(cave-houses) possessing [-bhañji, neut. pl. of -bhaj] heaps of pearls released by claw-tips that had been dug into the heads of wild elephants." Vallabha's analysis is vanyêbha-mastakeşu nikhatāni yāni nakhâgrāni tair muktāni yāni muktāphalāni teṣām prakarās tān bhajantîti tathā tāni, "those (cave-houses are) so, as they possess those heaps of those pearls which have been released by those claw-tips dug into wild-elephant-heads." Mallinātha's analysis is vanyêbhānām mastakeşu nikhātair nakhâgrair muktān muktāphala-prakarān bhajantîti tathôktāni, "those (cave-houses are) socalled, as they possess pearl-heaps released by claw-tips dug into the heads of wild-elephants." Here Vallabha does not analyze vanyêbha-mastaka, Mallinātha does not analyze muktāphala-prakara, and neither analyzes vanyêbha or nakhâgra.

# Chapter 9

States and

の行きたいたいたい

# **Ekaśesa** Formations

# 1.73 Introduction

*Ekasesa* ("remainder of one") formations (Pā.1.2.64–1.2.73) are those in which one noun is used in the dual or plural to stand for more than one noun of identical form but different signification (e.g., "Smiths" for "[Bob] Smith and [Tom] Smith"), one noun (the masculine form) is used to stand for both male and female members of a pair (Section 1.74), or a neuter adjective is used to modify both a neuter noun and a non-neuter noun (Section 1.75). For every kind of *ekasesa* formation the formula of analysis is exactly parallel to that of a simple *itaretara dvandva* compound (Section 1.62).

1

# 1.74 Dvandvâikaśeşa

Here the name of the male of a pair stands for both the male and the female (e.g., "dogs" for "dogs and bitches").

Ragh.M. 1.1 mātā ca pitā ca pitarau mother and father---parents.

Ragh.M. 1.40 mrgyaś ca mrgāś ca mrgāh does and bucks-deer

Kum.M. 1.27 putrāś ca duhitaraś ca putrāh sons and daughters—children

147

# 1.75 Napumsakâikaśeşa

Here a neuter adjective applies simultaneously to a neuter noun and a masculine or feminine noun.

Kum.M. 1.2 bhāsvanti ca [ratnāni] bhāsvatyaś ca [mahâuṣadhīḥ] bhāsvanti dyutimanti shining [jewels (neuter)] and shining [herbs (feminine)]—shining (neuter), radiant.  $\mathbb{C}_{\mathbb{C}}$ 

Chapter 10

他也能能對於自己的自己的意思。

and and a summary of the sound and the second s

# **Indicating the Construction**

# A. Arranging the Words of the Text

# 1.76 Introduction

Sanskrit commentaries are built up on a framework consisting of a rearrangement of the words of the text being commented on. The original text is called the  $m\bar{u}la$  ("root"). In arranging the words of the  $m\bar{u}la$  the commentators use either of two basic approaches, representing radically different ways of indicating the construction of the text: the *kathambhūtinī* approach (Section 1.77), and the *anvayamukhī* approach (Section 1.78). The explanatory material offered by the commentator is superimposed upon the framework made up by the repeated words of the  $m\bar{u}la$ , using techniques that are the same in both approaches (Section 1.79). Several variations on the two approaches are possible (Section 1.80).

# 1.77 The kathambhūtinī approach

In this approach the commentator begins by giving the basic "skeleton sentence" of the original text, and then gradually fills in each of the remaining parts of the  $m\bar{u}la$  by asking questions that bring out the ways in which those parts fit into the construction of the original text. A commentary using this approach is called a *kathambhūtinī* ( $t\bar{u}k\bar{a}$ ) because one of the most

### Chapter 10. Indicating the Construction

frequent questions used in this style is "katham-bhūta?" ("of what nature?", "being how?", "what kind of ...?"), or its equivalent "kim-visista?" ("distinguished by what?"). To illustrate this approach we may use as the mūla verse 2.1 of the Kumārasambhava of Kālidāsa:

tasmin viprakṛtā kāle tārakeṇa divâukasaḥ turāsāhaṃ purodhāya dhāma svāyaṃbhuvam yayuḥ

At that time, being oppressed by Tāraka, the gods (lit., "heaven-dwellers"), placing Turāṣāt (Indra) before them, went to the abode of the Self-born (Brahmā).

An anonymous commentary on the poem (inserted in the conflated recension of Vallabhadeva's commentary in the BORI Devanāgarī MS 379 of 1887–91), using the *kathambhūtinī* approach, arranges these words as follows:

divâukasah... svāyambhuvam dhāma... yayuh.... kim-višistāh? viprakrtāh.... kena? tārakeņa. kva? tasmin kāle.... kim krtvā? purodhāya.... kam? turāsāham....

The gods went to the abode of Svayambhū. How characterized? Oppressed. By whom? By Tāraka. At what? At that time. Doing what? Placing before them. Whom? Turāṣāt.

Notice that the "skeleton sentence" need not be restricted to just the subject and verb. The commentator will generally include at least the direct object, and may also include items that are only loosely connected with the basic sentence (e.g., interjections and vocatives), as well as any simple items that can be dealt with easily and unobtrusively.

# 1.78 The anvayamukhī approach

In this approach the commentator goes straight through all the words of the  $m\bar{u}la$  in one long string, merely rearranging them so as to place them in the most easily understandable prose order. In practice this usually results in some version of a "standard prose word-order," the essentials of which are recorded in a verse in the *Samāsacakra*, an elementary handbook used at the beginning stage in the traditional system of Sanskrit education:

### 1.79. Insertion of explanatory material

viśesanam puraskrtya viśesyam tad-anantaram kartr-karma-kriyā-yuktam etad anvaya-laksanam g

**小学生**の (1995)

The word order is characterized by subject, object, and verb, placing modifiers before what they modify.

In addition, vocatives usually appear first in such an arrangement, and relative clauses precede correlative clauses. A commentary using this approach is called an *anvayamukhī* ( $t\bar{t}k\bar{a}$ ) because it arranges the words "by means of the normal construction" (*anvaya-mukhena*). For example, the commentator Mallinātha uses this approach to arrange the words of the verse quoted above as follows:

tasmin kāle ... tārakeņa ... viprakrtāh ... divâukasah ... turāsāham ... purodhāya ... svāyambhuvam ... dhāma ... yayuh.

At that time, by Tāraka oppressed, the gods, Turāṣāt placing before, to Svayambhū's abode went.

## 1.79 Insertion of explanatory material

The basic arrangement of the words of the  $m\bar{u}la$ , arrived at by either of the two approaches just described, serves as a framework upon which the commentator superimposes a mass of explanatory material. The glosses, analyses, and other comments are simply placed alongside the words of the  $m\bar{u}la$  in their appropriate locations, using the techniques explained in preceding sections. The resulting interspersions often obscure the basic framework made up of the words repeated from the  $m\bar{u}la$ . To illustrate this problem, here is the actual beginning of Mallinātha's commentary on the verse we have been using as an example (the words repeated from the  $m\bar{u}la$  are in bold type here):

tasminn iti. **tasmin kāle** pārvatī-susrūsā-kāle **tārakeņa** tāraka-nāmnā vajraņkha-putreņa kenacid asureņa **viprakrtā** upaplutā divam okah sthānam yesām te **divaukaso** devāh...

"At that," (etc.)—At that time at the time of Pārvatī's service by Tāraka by a certain demon named Tāraka, the son of Vajraņakha oppressed distressed, of whom heaven is the abode—place—those heaven-dwellers gods ....

In the Sanskrit the disjoining effect of all this inserted matter is lessened somewhat by the connective powers of inflectional endings, and in giving this example we have helped matters further by underlining the words of the  $m\bar{u}la$ . In many printed editions of Sanskrit commentaries, however, the customary lack of typographical emphasis and detailed punctuation makes the basic framework blend in with the mass of comments, so that the obstacles to an awareness of the underlying framework are intensified. Despite these difficulties, it is essential to remember that the words of the  $m\bar{u}la$  (or at least readily discernible substitutes for them) have been arranged in a particular way by the commentator. If the commentary is viewed merely as a collection of unconnected clusters of explanation, the value of the underlying framework as a guide to the construction of the  $m\bar{u}la$  will be lost. As a practical device it is often helpful to seek out the words repeated from the  $m\bar{u}la$  and underline them, if the edition being used does not already set them apart in some way.

The explanatory portions of a commentary are inserted in precisely the same way in a commentary using the *kathambhūtinī* approach. For example, the anonymous commentator on *Kumārasambhava* 2.1 cited above fills out his basic framework as follows (note that in either approach the commentator first identifies the passage about to be commented upon by citing the *pratīka*, the first word or two, of the passage in question):

tasminn iti. divâukaso devāḥ svāyambhuvam dhāma sthānam yayur gatāḥ. kim-višiṣṭāḥ? viprakṛtā apakṛtāḥ. kena? tārakeṇa. kva? tasmin kāle īśvara-tapo-vidhāna-kāle. kim kṛtvā? purodhāya agre kṛtvā. kam? turāsāham indram.

"At that," (etc.)—The heaven-dwellers gods to Svayambhū's abode place went had gone. How characterized? Oppressed mistreated. By whom? By Tāraka. At what? At that time at the time of the Lord's performance of austerities. Doing what? Placing before putting in front. Whom? Turāṣāț Indra.

This is followed in the anonymous commentary by a mass of grammatical argument, including the citation of rules of Pānini and the analysis of the complex formations *svāyambhuvam*, *divâukasah*, and *turāsāham*. While most commentators place analyses in ordinary language alongside the formations being analyzed, comments using technical language and arguments on technical matters are treated like quotations from dictionaries (see Section 1.13): some commentators insert them in the relevant place in the running commentary, and others bunch them together at the end of the commentary on the verse or section of the original text. Comments on technical matters concerning the verse or section as a whole (e.g., the explanation of the meter in which the verse is written or of a poetic figure involving the whole verse) are nearly always placed at the end, unless they are worked into the introduction (*avatarana*) to the comment on that verse or section.

Bunching the technical comments together at the end of the commentary on a passage has the effect of reducing the interruption of the flow of the running commentary. Even when this is done, the indication of the construction of the *mūla* can still be considerably obscured by the insertion of explanatory material, and this problem is more acute in an *anvayamukhī* commentary than in a *kathambhūtinī* commentary, because a commentary using the *kathambhūtinī* approach sets forth the construction of the original text in a much more obvious way to begin with. While the *anvayamukhī* approach thus produces a commentary that is more difficult to read until one has had some experience with its characteristic procedures, it is important to learn how to deal with its intricacies, because in practice most commentators rely primarily on this style in preference to the *kathambhūtinī* approach, which, in addition to being rather more tedious, came to be regarded as old-fashioned.

As a further example to help make clear the essential difference between the two approaches and the ways in which explanatory material may be inserted in the basic framework, we may take as our  $m\bar{u}la$  the opening verse of the *Śiśupālavadha* of Māgha:

śriyah patih śrīmati śāsitum jagaj jagan-nivāso vasudeva-sadmani vasan dadarśâvatarantam ambarād dhiranya-garbhânga-bhuvam munim harih

Hari, the husband of  $\hat{S}r\bar{n}$ , the abode of the world, dwelling in the glorious house of Vasudeva in order to govern the world, saw descending from the

sky the sage (Nārada) who is the son of Hiranyagarbha.

The earliest of the commentators on the great Sanskrit  $k\bar{a}vyas$  whose works have survived, Vallabhadeva, comments on this using the old *kathambhūtinī* approach. His remarks begin as follows (the numbers in square brackets indicate the sections of this handbook in which you will find explanations of some of the devices being used here):

*śriya iti. harir munim dadarśa iti sambandhah* [Section 1.82.2]. nanu [Section 2.49] *harayo daśa samkhyātā* 

vartante--- "arka-markata-mandūka-visnu-vāsava-vāyavah/ turanga-simha-śītāmśu-yamāś ca harayo daśa." ko 'sau harir ity āha—śriyah patih laksmī-nātho visnur ity arthāh [Section 1.13.1]. munim kam ity āha—hiranya-garbhânga-bhuvam brahma-tanayam nāradam ity arthah. vady api brahma-[tanaya-]tvenâsau tadā hariņā [na] jñāyi, yena kenacic chabdena nirdeśe kartavye muni-śabdah [Section 1.6.2] pravuktah kavinā jñāta eva, tasva câitad vacanam. kadâdrāksīt kva vasudeva-sadmani sūrasena-putra-grhe śrīmati sarva-samrddhi-yukte vasan tistan. kali-dvāpara-sandhāv ity arthah. nanu bhagavān purātana-munir badary-āśrama-nivāsī kim-iti tatrôvāsêty āha-kim kartum jagac chāsitum tri-bhuvanam raksitum pālayitum. etena karmādhīnatvam bhagavato nirastam. kīdršo harih jagan-nivāsah bhuvanānām nivāsah ādhārah. nivasanty asminn iti nivāsah [Section 1.29.6]. kīdršam ambarād ākāšād avatarantam patantam, muner viśesanam [Section 1.82.3]. angabhū-śabda upacārān mānase 'pi putre vartate. yathā sarasija-śabdah sthala-kamale 'pi. ...

"Of Śrī," (etc.)—The construction is "Hari saw the sage." But there are ten Haris who are enumerated—"The sun, a monkey, a frog, Viṣṇu, Indra, the wind, a horse, a lion, the moon, and Yama are the ten Haris." Which one is this? He tells us—the husband of Śrī the spouse of Lakṣmī, i.e., Viṣṇu. What sage? He tells us—the son of Hiraṇya-garbha the offspring of Brahmā, i.e., Nārada. Although at that time Hari did not recognize him as Nārada, since he had to be designated with some word or other the word 'sage' is used, which the poet knew him to be, and this is his (i.e., the poet's) utterance. When did he see him? Where? Dwelling situated in the house of Vasudeva in the home of the son of Śūrasena, (which was) glorious endowed with all prosperity. What is meant is (that it was) at the juncture of the Kali and Dvāpara ages. But the Lord is the ancient sage who resides in the āśrama at Badarī, so why would he dwell there? He tells us—in order to do what? **In order to govern the world** in order to protect, to rescue, the universe. This (statement) refutes (the notion) that the Lord is dependent upon actions. What sort of Hari? **The abode of the world** the abode, receptacie, of the worlds. They abide in him, thus (he is) an abode. What sort (of sage)? **Descending from the sky** flying down from the heavens—this modifies the sage. The word *anga-bhū* (lit., "arising from the body") is metaphorically applied to a mind-born son as well, just as the word *sarasi-ja* (lit., "born in water") is applied even to a lotus growing on dry land. ...

The same verse is subjected to the *anvayamukhī* treatment by the more recent commentator Mallinātha, who begins as follows:

síriya iti—... śriyo lakşmyā patih. anena rukmiņī-rūpayā śriyā sameta iti sūcitam [Section 1.13.5]. "rāghavatve 'bhavat sītā rukmiņī kṛṣṇa-janmani" iti viṣṇu-purāṇāt. jagan-nivāso jagatām ādhāra-bhūtah [Section 1.84.3]. kukṣi-sthâkhila-bhuvana iti yāvat [Section 1.12.3]. tathāpi jagal lokam sāsitum duṣṭa-nigraha-siṣṭānugrahābhyām niyantum śrīmati lakṣmī-yukte vasudeva-sadmani vasudeva-rūpiṇaḥ kasyapasya veśmani vasan kṛṣṇa-rūpeṇa tiṣṭhan harir viṣṇur ambarād avatarantam. indra-sandeśa-kathanârtham iti bhāvaḥ [Section 1.13.2]. hiraṇyasya garbho hiraṇya-garbho brahmâṇḍa-prabhavatvāt, tasyāṇga-bhuvam tanū-jam. athavā tasyângād avayavād utsaṅgākhyād bhavatîti hiraṇya-garbhāṅga-bhūs [Section 1.32.2, Section 1.59] tam [hiraṇya-garbhāṅga-bhuvam] munim. nāradam ity arthaḥ. "utsaṅgān nārado jajñe dakṣo 'nguṣṭhāt svayambhuvaḥ" iti bhāgavāt. dadarśa. kadācid iti śeṣaḥ [Section 1.83]....

"Of Śrī," (etc.)—... The husband of Śrī of Lakṣmī. By this it is suggested that he was with Śrī in the form of Rukmiņī, because the Viṣṇupurāṇa says, "She was Sītā when he was Rāghava, and Rukmiņī in his birth as Kṛṣṇa." The abode of the world the receptacle of the universe. To be precise, the one in whose belly the entire universe is situated. Nevertheless in order to govern the world to control the world by restraining the evil and favoring the good in the glorious endowed with prosperity house of Vasudeva home of Kaśyapa in the form of

1.80. Variations on the two approaches

### Chapter 10. Indicating the Construction

Vasudeva **dwelling** residing in the form of Kṛṣṇa, **Hari** Viṣṇu, **descending from the sky**—the idea is that (he was coming) for the purpose of delivering a message from Indra. The fetus (*garbha*) of the golden one (*hiraṇya*)—Hiraṇyagarbha, Brahmā, because he was produced from the golden egg. His son, offspring [**the son of Hiraṇyagarbha**]. Or—arising from an *aṅga*—a part of the body, namely the lap—of him, born from a part of the body of Hiraṇyagarbha. That **sage**, i.e., Nārada. Because the Bhāgavatapurāṇa says, "Nārada was born from the lap of the Selfborn, and Dakṣa from the thumb." He **saw**. Supply "at a certain time." ...

Here the words of the *mūla*, shown in bold type in the text and translation, are presented in the normal prose word order of Sanskrit, although the continuity of this word order is impossible to convey in the English translation. Notice that Mallinātha first analyzes *hiraŋya-garbhânga-bhuvam* as a genitive *tatpuruṣa*, and then decides that it is an *upapada* compound. When a commentator gives more than one possibility in such instances, the last alternative mentioned is ordinarily the one that he prefers.

# **1.80** Variations on the two approaches

Commentators frequently use one of three variations on the basic ways of arranging the words of the  $m\bar{u}la$ .

The first variation involves a combination of the two basic approaches. Quite often a commentator will use something akin to the *kathambhūtinī* approach to break the  $m\bar{u}la$  into two or more large parts, but will use the *anvayamukhī* method in dealing with each of these parts.

In the second variation, which occurs most often in conjunction with the *anvayamukhī* approach, the commentator arranges the whole commentary in standard fashion, but omits the repetition of the words in the *mūla*. When this is done the glosses of the omitted words appear where they would be if the words of the original text were repeated in the commentary to put it another way, the gloss often takes the place of the repeated word from the *mūla* in giving the basic construction of the original text.

In the third variation, the commentator begins by giving the bare *an-vaya* of the original—i.e., the arrangement of the words of the  $m\bar{u}la$  in the order that forms the framework of an *anvayamukhī* commentary—without

inserting any explanatory material. Once this has been done, the commentator begins again and gives a full commentary using any of the approaches or variations mentioned above.

As examples of the approaches and variations just described, here are seven actual commentaries on a single text. The  $m\bar{u}la$  is verse 4.3 of the *Kumārasambhava*:

ayi jīvita-nātha jīvasîty abhidhāyôtthitayā tayā puraḥ dadṛśe puruṣâkṛti kṣitau hara-kopânala-bhasma kevalam

[When the love-god Kāma was burned to death by the fire from the third eye of Śiva, Kāma's wife Ratī fainted. Now she has regained consciousness, and—]

"Ah, lord of my life, are you alive?" Speaking thus, by her, rising up, was seen before her on the ground only the ashes from the fire of Śiva's anger, in the form of a man.

(1) Example of a *kathambhūtinī* commentary (once again, the anonymous commentary inserted in the conflated recension of Vallabhadeva's commentary):

ayîti. tayā ratyā kāmo hara-kopânala-bhasma dadrše drṣṭam, kimviśiṣṭam? puruṣâkṛti puruṣâkāram. kasyām? kṣitau pṛthivyām. katham? puro 'grataḥ. kim̥-viśiṣṭayā ratyā? utthitayā. kim̥ kṛtvā? ity abhidhāya uktvā. iti iti kim? he jīvita-nātha tvam̯ jīvasi kim̥ na? ayi iti komalâmantrane.

"ayi," (etc.)—By her by Ratī, Kāma, the ash of the fire of Śiva's anger, was seen was seen. Characterized by what? Having the form of a man having the shape of a man. On what? On the ground on the earth. How? Before in front. By Ratī, being characterized by what? Rising up. Doing what? Thus speaking uttering. What is "thus"? Oh lord of my life you are living, aren't you? "Ah" (is used) in tender address.

(2) Example of an *anvayamukhī* commentary (again, Mallinātha's Samjīvinī):

ayîti. ayi iti praśne—"ayi praśnânunayayoh" iti viśvah—**ayi jīvita-nātha jīvasi** prāņiși kaccid **ity abhidhāyôtthitayā tayā** ratyā **puro** 'gre kşitau

### Chapter 10. Indicating the Construction

puruşasyâkrtir ivâkrtir yasya tat **puruşâkrti kevalam** ekam hara-kopânala-bhasma dadrśe drstam. na tu puruşa iti bhāvah.

"ayi," (etc.)—"Ah" (is used) in the sense of a question: "ayi in (the senses of) interrogation and conciliation"—Viśva. Ah lord of my life, you are alive? You are living, I hope? Speaking thus, rising up, by her by Ratī before in front on the ground that whose form is like the form of man in the form of a man, only alone the ash of the fire of Śiva's anger was seen was seen. But not a man—this is the idea.

(3) Another example of an *anvayamukhī* commentary, this time with a bit less explanatory material added (the *Dhīrarañjikā* of Rāma Govinda Siddhāntavāgīśa):

ayi jīvitêti. ayi iti sambodhane. he jīvita-nātha tvam jīvasîty abhidhāya uktvā utthitayā tayā puro 'gre puruşâkṛti puruṣasyêvâkṛtiḥ saṃsthānam yasya tādṛśam hara-kopânala-bhasma kevalam dadṛśe ity anvayaḥ.

"ayi jīvata-," (etc.)—ayi (is used) in address. Oh lord of my life you are alive? Thus speaking uttering rising up, by her before in front in the form of a man of which the form, the configuration, is like a man's, of that sort only the ash from the fire of Śiva's anger was seen—this is the construction.

Notice that this commentator ignores the word ksitau in the mūla.

(4) Example of a commentary using a mixture of the *kathambhūtinī* approach and the *anvayamukhī* approach (Cāritravardhana's Śiśuhitaisinī):

ayîti. **tayā** ratyā **puro** 'gre **puruṣâkṛti** puruṣasyêvâkṛtir ākāro yasya tat. **hara-kopânala-bhasma** harasya kopâgner bhasma **kevalam dadṛśe** dṛṣtam. na tu kānta iti śeṣah. kim-bhūtayā tayā? **ayi** iti komalâmantrane bho **jīvita-nātha** prāṇa-nātha **jīvasîty abhidhāya** uktvā utthitayā.

"ayi," (etc.)—By her by Ratī before in front in the form of a man that whose form, shape, is like a man's. The ash of the fire of Siva's anger the ash of the anger-fire of Siva only was seen was seen. Supply "but not her beloved." By her, being of what sort? Ah used in tender address, Oh lord of my life lord of my life are you alive? Thus speaking uttering rising up. (5) Example of an *anvayamukhī* commentary with omission of many of the words of the  $m\bar{u}la$  (Vallabhadeva's  $Pa\bar{n}jik\bar{a}$ , with the omitted words supplied in square brackets):

ayîti. [ayi] kaccit [jīvita-nātha] prāņêśvara jīvasîty [abhidhāya] uktvā utthitayā tayā kevalam [puruṣâkṛti] puruṣa-saṃsthānam [hara-kopānala-bhasma] rudra-roṣâgni-bhasmi dadṛśe dṛṣṭam. na tu priyaḥ. matsyasyêva dagdhasya tasyâkāro na naṣṭaḥ.

"*ayi*," (etc.)-[**Ah**] I hope [lord of my life] lord of my life you are living? Thus [speaking] uttering rising up, by her only [in the form of a man] in the shape of a man [the ash from the fire of Śiva's anger] the ash from the fire of Śiva's anger was seen was seen. But not her beloved. When he was burned, his shape, like that of a fish, was not destroyed.

(6) Example of a sophisticated commentary that offers explanations of only a few of the words in the *mūla* (the *Prakāśikā* of Aruņagirinātha). Note that Aruņagirinātha reads *puruṣâkrti* [feminine, and thus a genitive *tatpuruṣa*] rather than *puruṣâkrti* [neuter, and thus a *bahuvrīhi* modifying *-bhasma*]:

avadhānasya kāryam āha—ayîti. "**ayi** praśne sānunaya" iti halāyudhah. **puruşâkṛtir** iti pāthah. tad-viśeṣaṇam **hara-kopânala-bhasmê**ti. hara-kopânale bhasmī-bhūtā. **kevalam** iti, na kaścid apy abhasmī-bhūtas tad-avayavas tat-paricchado vêty arthah. etad vibhāvôdbodha-param vākyam.

He tells us the effect of her attentiveness—"ayi," (etc.). "ayi in conciliatory inquiry"—Halāyudha. The (correct) reading is "**the form of a man.**" "**The ash of the fire of Śiva's anger**" modifies this. "**Entirely**"—the idea is that not a single limb of his, and not a single garment, had not been reduced to ashes. This is a statement dealing with the awakening of the factors (connected with the mood of compassion).

(7) Example of a commentary giving the bare *anvaya* without comment before proceeding to a full commentary (the *Vivarana* of Nārāyanapandita, who closely follows Arunagirinātha both in his readings and in his comments):

vilocanayor avadhāna-paratva-vidhānasya phalam āha—ayîti. jīvita-nātha tvam jīvasy ayîty abhidhāya, utthitayā tayā purah ksitau purusâkrtir dadrše.

jīvita-nātha prāņêśvara tvam jīvasi. ayi iti sânunaya-praśna-vişayam avyayam. "ayi sânunaye praśne" iti halāyudhah. jīvasi kaccid ity arthah. ity abhidhāya uktvā utthitayā krtôtthānayā. purusâkrti-mātra-darśanam jīvana-sandehe hetuh. ata eva sa-tvaram utthānam. tayā ratyā [purah] agre ksitau bhūmau purusâkrtih purusasyākrtir ākāro dadrśe drstâbhūt.

ayam abhiprāyaḥ—hara-kopânala-niṣpatanânantaram eva moham upagatā ratir īṣan-moha-vigamanânantaram priya-śarīra-samsthānam pūrvavad evâvalokya satvaram utthāya tat-samīpam upagamya purastād eva puruṣâkṛtim dadarśêti.

kīdrsī punah sā purusâkrtir ity āha—kevalam hara-kopânala-bhasma iti. kevalam tac-charīrâvayavesu tat-paricchadesu vā na kaścid amśas tatrâbhasmasād-bhūto drsto 'bhūd iti bhāvah. purusâkrtîty eva kecit pathanti. tan-mate purusâkrti hara-kopânala-bhasma dadrsa ity anvayah, anena ślokena karunasya vibhāvôdbodho darsitah.

He tells us the result of her making her eyes attentive—"ayi," (etc.). Lord of my life, you are living, aren't you? Thus speaking, by her, rising up, before her on the ground a man's form was seen.

Lord of my life lord of my life, you are living? *ayi* is an indeclinable used in the context of conciliatory inquiry. "*ayi* in conciliatory inquiry"—Halāyudha. The meaning is "You are alive, I hope?" Thus speaking uttering rising up performing the action of rising up. That she sees only the form of a man is the cause for her doubt concerning whether he is alive or not. Hence her rising up quickly. By her by Ratī [before] in front on the ground on the earth a man's form the form, shape, of a man was seen was seen.

What is intended is this—immediately after the fire of Siva's anger had gone forth, Ratī fainted. After regaining consciousness somewhat, she saw the shape of the body of her beloved, just as before, and rising quickly she went near him and saw up close the form of a man.

Of what further nature was that form of a man? He tells us—entirely ash from the fire of Śiva's anger. Entirely—the idea is that not a single portion was seen, either in the limbs of his body or in their garments, that had not been completely reduced to ashes. Some read "*puruṣâkṛti*." In their opinion the construction is "the ash from the fire of Śiva's anger was seen in the form of a man.

This verse displays the awakening of the factors connected with the mood of compassion.

# **B.** Other Techniques

# 1.81 Introduction

Although the methods of arranging the words of the text described in Sections 1.76–1.80 constitute the primary means of indicating the construction of the text being commented on, commentators also make use of several other techniques to indicate features of syntax and construction. Among these are various ways of directly pointing out connections between words (Section 1.82), formulas for supplying words needed to complete the meaning of a sentence (Section 1.83), and devices for marking words belonging to particular syntactical categories (Section 1.84). Much syntactical information is also conveyed through the use of the technical terminology of Pāņinian grammar.

# 1.82 Direct statement of connections

### 1.82.1 anvaya

The word *anvaya* ("going together," construction) can refer not only to the construction of an entire passage (as in the name of the *anvayamukhī* style of commentary, Section 1.78), but also to the construction of any part of the whole, and commentators frequently mark a specification of the construction of the original text by adding the formula *ity anvayah*, "thus (is) is the construction." For an example, see the next-to-last sentence in the commentary of Nārāyaṇapandita quoted at the end of Section 1.80.

The portion of a passage most commonly singled out for this treatment is the basic "skeleton sentence" of the original text, so that the formula *ity anvayah* is often used in connection with the *kathambhūtinī* style of commentary (Section 1.77). For example, in explaining his statement *tatra dravyāni prthivy-ap-tejo-vāyv-ākāśa-kāla-dig-ātma-manāmsi navâiva* ("Among these the substances are nine only—earth, water, fire, wind, space, time, direction, soul, and mind," *Tarkasangraha 3*), Annambhaṭta makes the following remark in his own commentary on the text, the *Tarkadīpikā*: tatra dravyâdi-madhye dravyāņi navâiva ity anvayah. kāni tāni? ity ata āha---pṛthivī ity-ādi. The basic construction is: "Among these among the substances, etc. the substances are nine only." What are they? In reply he says---earth, etc.

### 1.82.2 sambandha

The term *sambandha* ("connection") and related forms (*sambadhyate, sambandhanīya*, etc.) are used in expressly stating the connection of one word with another word or words. These terms appear in commentaries in either of two situations.

One situation is that in which the commentator must explain in his remarks on one portion of a work that a word is to be construed with a word or phrase occurring in an earlier or later portion of the work (cf. the formula *iti śeṣaḥ*, Section 1.83.1). This often happens when several individual verses of a long poem are syntactically connected (Section 1.82.4):

Ragh.M. 1.5 so 'ham. "raghūnām anvayam vakṣye" ity uttarena sambandhah . I, being so. The connection (of this) is with what follows (four verses later in the poem), "shall proclaim the dynasty of the Raghus."

Ragh.M. 1.36 *āsthitav* ārūdhau. "jagmatuh" iti pūrveņa sambandhah. Having seated themselves having taken their seats (in the vehicle). The connection is with what came before (in the preceding verse), "they went."

The other situation is that in which a word in the portion of the work being commented on is to be construed with more than one word in that portion:

Kum.M. 2.10 tathā **ātmānam ātmanā eva**—ātmany evêty atrâpi sambadhyate—svasminn eva s**rjasi**. Likewise your self by the self alone—"in the self alone" (occurring in a later clause in the same verse) is to be construed here as well—in yourself alone you create.

Ragh.M. 11.11 sisevire iti sarvatra sambadhyate. (The verb) "served" is to be construed (with the subject in each clause) throughout (this verse).

Kum. V. 2.9 apir virodhe sarvatra sambandhanīyah. (The particle) api in the sense of contradiction is to be construed with (the first word of each clause) throughout (this verse). The term yojya ("to be connected") is used in much the same way.

### - S.

### 1.82.3 viśesana

The word *visesana* ("modifier") is used in explaining which word an adjective or adverb is to be construed with:

Ragh.M. 1.83 sandhyāyām apy etad višesaņam yojyam. This modifier (viz., "glossy and pink as a fresh shoot") is to be construed with the dawn as well.

Ragh.M. 1.14 merāv api viśeṣaṇāni tulyāni. The modifiers apply equally to Meru as well.

Ragh.M. 1.18 **bhūty-arthaṃ** vṛddhy-artham **eva** .... grahaṇa-kriyā-viśeṣaṇaṃ câitat. **Only for the sake of prosperity** for the sake of well-being.... And this is a modifier of the action of taking.

As the first two examples indicate, this term is used in reference to adjectives chiefly when one adjective is to be construed with more than one noun.

### 1.82.4 Connections between verses

When a work composed in verse contains a sequence of verses that are syntactically connected (either by grammatical opposition or by the use of relative clauses), a commentator may call attention to such a sequence in two ways. In the introduction to his comments on the first verse in the sequence the commentator will usually specify the number of verses so connected through such statements as *śloka-dvayena*...  $\bar{a}ha$  ("through the use of a pair of verses [the author] says ..."), *śloka-trayena*...  $\bar{a}ha$ , etc.

The commentator may also make use of the technical terms for connected sequences of various lengths. These terms are used most frequently at the end of the comments on the final verse of the sequence, in order to mark the close of the group of connected verses, but they may also be used in the introduction to the first verse in the sequence as well. The most frequent of these technical terms are *yugmaka* or *yugma* (a pair of syntactically connected verses), *samdānitaka* (a group of three such verses), *kalāpaka* (four verses), and *kulaka* (five or more verses).

### Chapter 10. Indicating the Construction

Kum.M. 4.42–4.43 śāpâvadhir api tenâivôkta ity āha

*sloka-dvayena*..., *yugmakam*. [In the introduction to verse 42:] That he himself also stated a limit to the curse (the poet) tells us in a pair of verses....[At the end of the comments on verse 43:] (this has been a) *yugmaka*.

Ragh.M. 1.5–1.9 evam raghu-vamśe labdha-praveśas tad-varnanam pratijānānah "so 'ham" ity-ādibhih pañcabhih ślokaih kulakenâha—... kulakam. [In the introduction to verse 5:] Having thus been granted entry into the dynasty of the Raghus, (the poet) tells us that he proposes to describe it, with a kulaka in five verses beginning "I, being so"—... [At the end of the comments on verse 9:] (this has been a) kulaka.

### 1.82.5 Connections between statements

In addition to specifying the syntactical construction of a passage, commentators often try to help the reader understand the connections between the parts of the passage by making small asides designed to indicate the logical chain of thought involved. In the following example the commentator neglects to repeat many of the words being commented on, being content to let the synonyms he supplies for them take their place in revealing the structure of the original sentence [see Section 1.80, example (5)]; here the original words are inserted in square brackets.

Kum.V. 3.63 kṛta-praṇāmā ca [sā] asau [bhavena] hareṇa an-anya-bhājam a-sādharaṇam [patim] bhartāram [āpnuhi] labhasva ity [abhihițā] agadyata. etac ca [tathyam] satyam eva. devena paścād udvāhitatvāt. na câitad āścaryam. [hi] yasmād ĩśvarāṇām āḍhyānāṃ vyāhṛtaya uktayaḥ pāke phala-kāle viparītam anyam eva arthaṃ na puṣyanti na phalanti. devaś cêśvaraḥ.

And [she] that woman, having bowed to him, [was addressed] was spoken to [by Bhava] by Śiva, "[Obtain] acquire [a husband] a spouse enjoyed by no other not shared." And this quite [accurately] truly. Because afterwards Śiva married her. Nor was this surprising. [For] because the utterances the statements of lords of rich persons do not foster do not result, at ripeness at the time of fruit, in an opposite an entirely different sense. And Śiva is a lord. For further examples see the passage from *Sisu.V.* 1.1 quoted in Section 1.79. This technique is essentially another way of achieving the goal of the questions asked in the *kathambhūtinī* style of commentary (Section 1.77).

# **1.83** Supplying words needed to complete the sense

When a word must be supplied in the original text, two situations are possible: if the word is simply to be understood in the passage without actually occurring elsewhere in the text, a commentator will use the formula *iti śesah* when the supplying the word (Section 1.83.1), but if the word is to be carried over from a preceding part of the text, formulas involving several technical terms are used (Section 1.83.2). See also Section 1.82.2.

### 1.83.1 iti sesah

The formula *iti śeṣaḥ* ("thus the remainder," "this remains [to be supplied]") is placed after words understood in the original text and supplied by the commentary. It may be translated "supply ...":

Ragh.M. 12.71 tena pathā setu-mārgêņa uttīrya sāgaram iti šeṣaḥ. By that path by the route of the bridge having crossed over. Supply "the sea."

Kum.M. 2.56 tat-tapah varena vara-dānena śamitam hi. mayêti śeşah. For by a boon by the granting of a boon his ascetic power was quelled. Supply "by me."

### Ragh.M. 1.35 vasisthasyâśramam jagmatuh.

putra-prāpty-upāyâpekṣayêti śeṣaḥ. They both went to the hermitage of Vasistha. Supply "because of their need for a means of obtaining a son."

### 1.83.2 Words carried over

化能 的复数动物

Works of the  $s\bar{u}tra$  type achieve much of their conciseness by assuming that a word used in any  $s\bar{u}tra$  will be supplied in each succeeding  $s\bar{u}tra$  in which it is needed. This "carrying over" of a word is called *anuvrtti*, and a word so carried over is said to be *anuvrtta*; finite verbal forms are used

as well: *anuvartate* ("it remains over"), or simply *vartate* ("it is present"). The cessation of this process of carrying over, which occurs at the point at which the word is no longer needed to complete the meaning of a *sūtra*, is called *nivrtti*, and a word that has ceased being carried over is said to be *nivrtta*. The authors of commentaries on works in which *anuvrtti* and *nivrtti* operate frequently use these technical terms, most often at the very beginning of their commentary on the *sūtra* in question:

Kāśikā 1.2.9 sann ity anuvartate. ktvêti nivrttam "san" is carried over (from a previous rule). "ktvā" is no longer carried over (from this point on).

Kāśikā 1.2.26 vêti vartate, sed iti ca. "vā" is present (in this sūtra by anuvrtti), and also "set."

Kāsikā 8.1.54 pūrvam sarvam anuvartate gaty-artha-lotam varjayitvā. Everything previous (i.e., everything in the previous sūtra, whether actually read there or present by anuvrtti) is carried over, except "gaty-artha-lot."

Other terms are occasionally used to point out similar phenomena even in works other than those of the  $s\bar{u}tra$  type:

Kum.M. 2.11 tvam ity anusajyate. "You" is carried over (from the previous verse in the poem).

# **1.84** Devices for marking features of syntax

Commentators use special devices to mark the subject and predicate (Section 1.84.1), adjectives that are used attributively rather than as predicates (Section 1.84.2), substantives used attributively (Section 1.84.3), locative absolute constructions (Section 1.84.4), words used adverbially (Section 1.84.5), and vocatives (Section 1.84.6).

### 1.84.1 yat ... tat marking subject and predicate

The relatively free word order of Sanskrit, together with the frequent omission of linking verbs, often makes it difficult to tell which of two nouns or adjectives is the subject and which is the predicate. Commentators sometimes meet this difficulty by marking the subject with a relative pronoun and the predicate with the correlative pronoun. For example, the sentence lokantara-sukham punyam (Ragh. 1.69) could theoretically mean either "pleasure in the next world is merit" or "merit is pleasant in the next world (i.e., leads to happiness there)." Mallinātha's commentary marks punyam as the subject:

### Ragh.M. 1.69 yat punyam, tal lokântare para-loke sukham

*sukha-karam.* That which is **merit** is **pleasant** pleasure-producing **in the other world** in the next world.

### 1.84.2 sat marking attributive adjectives

Another way of distinguishing attributive adjectives from predicate adjectives is through the use of the present participle *sat* (masc. *san*, fem. *satī*), which is placed after adjectives that are used attributively. For example, the sentence  $pr\bar{t}t\bar{a}$  kāmadughā hi sā (Ragh. 1.32) could theoretically be interpreted in several ways; Mallinātha's commentary makes it clear that  $pr\bar{t}t\bar{a}$ goes with the subject  $s\bar{a}$ :

Ragh.M. 1.32 hi yasmāt kāraņāt sā prītā tustā satī... kāma-dughā bhavati. For from which as cause she, being pleased gratified, ... is a granter of wishes.

Similarly, the words  $vrstibhir udvejit\bar{a}h \dots siddh\bar{a}h$  (Kum. 1.5) might be taken to mean "the siddhas are tormented by rains," until this device makes it clear that the adjective  $udvejit\bar{a}h$  is not a predicate but serves as a modification of the subject:

Kum.M. 1.5 vrstibhir udvejitāh klesitāh santah ... śrngāny āśrayante. Being tormented troubled by rains, they resorted to its peaks.

# 1.84.3 -bhūta marking attributive substantives

The word *bhūta* may be placed in compound after a substantive to make it clear that it is being placed in apposition with some other noun.

### Chapter 10. Indicating the Construction

This device is often used avoid any difficulties the reader may have in linking two nouns of different gender, as the addition of  $-bh\bar{u}ta$  converts an attributive substantive into an adjective that may be inflected in any gender; *Bhagavadgītā* 7.5, for example, has *prakṛtiṃ viddhi me parām jīva-bhūtām*, "know my other nature, which is the soul." For more on this, see Section 2.17.

In commentaries this device is particularly useful when a pronoun of the same gender as the attributive noun might seem to refer to it, but in fact refers to some other object that is being equated with the thing denoted by that substantive:

Kum.M. 3.74 vighnam antarāya-bhūtam tam kāmam. Him Kāma the obstacle being an impediment.

Kum.M. 3.19 surāņām vijayâbhyupāye jayasyôpāya-bhūte tasmin hare. Against him against Śiva the means of victory being the means of winning for the gods.

Without this device the phrases commented on in these examples might be taken to mean "that obstacle" and "against that means of victory."

### 1.84.4 sati marking the locative absolute

The present participle *sat*, when used in the locative case, serves to mark locative absolute constructions. For example, *mahaty avarodhe mahaty api* (*Ragh.* 1.32) could mean either "even in his large harem" or "even though his harem was large"; Mallinātha's commentary makes it clear that the latter is the intended meaning:

Ragh.M. 1.32 avarodhe antahpura-varge mahati saty api even on his harem—on the group of women in his seraglio— being large

The participle may be in any gender:

Ragh.M. 1.89 asyām ... pītâmbhasi satyām apaḥ pibeḥ. you must drink water ... on her ... being one by whom water is drunk.

For beginning students this device is especially useful in marking items that they might otherwise interpret as finite verbs in the third personal singular present indicative: Ragh.M. 1.78 srotasi pravāhe nadati sati the current---the flow of water---being roaring

### 1.84.5 yathā tathā marking adverbs

The formula yathā (bhavati) tathā is used to mark words used adverbially:

Kum.M. 1.32 catasro'srayo yasya ac catur-asram anyūnânatiriktam yathā tathā šobhata iti caturasra-śobhi. That which has four corners is four-cornered (i.e., symmetrical)—having neither too little nor too much (in any part)—it is beautiful in that way, thus symmetrically beautiful.

Ragh.M. 1.18 sahasram gunā yasmin karmani tad yathā tathā sahasra-gunam sahasradhā. In which there are a thousand repetitions, that, in such a manner: in a thousand ways a thousandfold.

Ragh.M. 1.40 adūram samīpam yathā bhavati tathā. not far vicinity, in a way that is so.

### 1.84.6 he marking vocatives

Commentators regularly use the particle *he* before words in the vocative case to mark them as such, even if the word is used with no accompanying interjection in the original text:

Kum.M. 2.51 he vibho svāmin O lord master

Kum.M. 5.36 he pārvati O Pārvatī

by Emery R. Boose

# The Bhāṣya Style

Part II

ç.

# Chapter 11

# **The Fifth Service**

# 2.1 Introduction

Those portions of a Sanskrit commentary that are not specifically devoted to glossing the words of the text are usually concerned with discussing the contents and implications of the text. Often these portions constitute the major part of a commentary on a philosophical or scientific text, but only a small part of a commentary on a literary text. Traditionally, discussion of meaning takes the form of a dialogue in which the commentator allows opponents to raise various objections against the text or against his interpretation of it and then proceeds to refute these objections. This format is based both historically and stylistically on oral debate (see Section 2.48), and its traditional importance may be seen in the term chosen for the fifth service that a full commentary is expected to provide, i.e., *āksepasamādhāna*, "the answering of objections." A distinctive style of Sanskrit prose evolved for such discussion, a style that became the standard both for commentaries and for independent works of philosophy and science.

<u>;</u>]

This *bhāsya* style presents many features of syntax and terminology that may be unfamiliar to beginning students. The present chapter deals with those features that seem to pose the most problems for students, i.e., those that occur fairly often and have been found by experience to be the most puzzling. Some general observations on the discussion of meaning are given in Sections 2.2 and 2.3; the rest of the chapter has been arranged by bringing together (as far as possible) those items that are similar in form or function. Particular words or constructions may be sought in the index.

Note: although Part Three deals specifically with commentaries, all but Section 2.39 is applicable also to independent works composed in the  $bh\bar{a}sya$  style.

### 2.2 Content

Discussion of the meaning of the text is less limited in content than the analysis of the words themselves. Ordinarily discussion proceeds by a series of objections and refutations that lead to confirmation of the commentator's view. In this discussion the commentator may examine the implications of the text, defend the views represented in the text against those of rival schools, defend his interpretation of the text against other interpretations, and introduce his own insights (though the last are rarely claimed by the commentator as his own, and are usually argued to be implicit in the text itself). He may support his arguments by citing examples and comparisons (Chapter 18) or by appealing to the authority of other texts, other teachers, and common experience (Chapter 17). In learning to read such discussion it is essential to learn the conventional terms used to mark different sections of the commentary. For example, certain terms are used to raise an objection, to reply to an objection, or to introduce an alternative interpretation (Chapter 19). Familiarity with these terms often permits a reader to follow the flow of an argument even when the details of the argument are not yet understood. It is also important to remember that glosses of the text may be interspersed among long sections of discussion. These glosses may be more easily recognized as such if the words of the text are kept in mind as the commentary is read.

# 2.3 Style

The prose of commentaries is characterized by nominal constructions employing abstract nouns (Chapter 12), long compounds (Chapter 13), and relatively few finite verbs. These constructions permit the concise expression of complex ideas and so are well suited to the writing of commentaries. They also create the greatest single difficulty for the beginning student who is not yet accustomed to this style. Other characteristic features that the student must learn include conventions of word order (Chap2.3. Style

ter 14) and of causal sentences (i.e., sentences that express a reason or cause; Chapter 15), and peculiarities in the use of *iți* (Chapter 16) and in the use of various particles and pronouns (Chapter 20).

# Chapter 12

# **Abstract Nouns**

# 2.4 Introduction

Nominal abstract nouns (i.e., abstract nouns derived from other nouns) are widely used in commentarial prose. This section discusses the formation and significance of such nouns, the common constructions that employ them, the uses of the different cases in these constructions, and two id-iomatic constructions that involve the instrumental and the accusative. (See also: Coulson 140–42, 154–58; Speijer 235–39; Whitney 1237, 1239.)

5

1

# 2.5 Formation

### 2.5.1 tā and tva

The most common nominal abstract nouns are those formed with the suffixes  $t\bar{a}$  (feminine) and tva (neuter). These may be added to any adjective or substantive to form abstract nouns meaning "being such, the quality of being so and so" (Pā. 5.1.119).

*śuklatā, śuklatvam* being white, whiteness

aśvatā, aśvatvam being a horse, horseness

Note that  $t\bar{a}$  and tva are identical in meaning.

### 2.5.2 -bhāvah and vrddhi derivatives

Nominal abstract nouns also include *tatpuruşa* compounds formed with *-bhāvaḥ*, "state, being," and neuter *vrddhi* derivatives. For example, from *guru*, "heavy," are formed *guru-bhāvaḥ* and *gauravam*, both meaning "being heavy, heaviness," and both synonymous with *gurutvam* and *gurutā*. Abstracts in  $t\bar{a}$  or tva are more common in commentarial prose, and so they are the subject of the following paragraphs. However, *tatpuruṣas* in *-bhāvaḥ* and *vrddhi* derivatives are occasionally used in the same constructions described for abstracts in  $t\bar{a}$  and tva.

*indrasya lāghavāt* = *indrasya laghutvāt* because Indra is swift (causal ablative)

*maitryena gacchati = mitratayā gacchati* he goes as a friend (predicative instrumental)

tasya putra-bhave' pi = tasya putratve' pi although he is a son (locative absolute)

*deva-bhāvaṃ prapadyate = devatāṃ prapadyate* he becomes a god (accusative with verb of motion)

# 2.6 Significance

### 2.6.1 Used by necessity

In some cases the abstract notion conveyed by the abstract noun is essential to the meaning of the sentence.

SBh 2.2.17 paṭasya tu guṇāḥ śuklatvâdayaḥ paṭa-deśā abhyupagamyante na tantu-deśāḥ but the qualities of the cloth, whiteness and so on, are admitted to reside in the cloth, not in the threads

Here the gunah is śuklatvam and the sentence requires an abstract noun that means "being śukla." Note that abstract nouns formed by adding  $t\bar{a}$ , tva, or *-bhāvah* to a *dvandva* compound may denote relationship, e.g., *pitrputratvam*, "being father and son," i.e., "the relation of father and son." Here too the abstract notion is essential to the meaning of the sentence.

SBh 2.1.4 na ca vilakṣaṇatve prakṛti-vikāra-bhāvo dṛṣṭaḥ and the relation of material cause and effect is not seen (in two things) when they are different from one another

SBh 1.1.1 seşa-seşitve 'dhikrtâdhikāre vā pramāņâbhāvād dharma-brahma-jijāāsayoh because there is no proof that the inquiries into dharma and into brahman are related as subordinate and principal or as qualified and qualification

### 2.6.2 Used by choice

But in most cases the abstract notion conveyed by the abstract noun is not essential to the meaning of the sentence. Instead abstract nouns are used to form nominal constructions when the same meaning could be expressed by a different construction that does not require the abstract noun. It is this syntactical use of the abstract noun that is taken up in the remaining paragraphs of this section.

# 2.7 Common constructions

### 2.7.1 With subjective genitive

Abstract nouns in  $t\bar{a}$  or tva are usually construed with a separate subjective genitive. This construction may provide a complete statement, where the abstract noun expresses the predicate.

indrasya rājatā of Indra there is kingness: Indra is king

patasya nīlatvam of the cloth there is blueness: the cloth is blue

A Sanskrit sentence of equivalent meaning can be obtained by dropping the abstract suffix and making the subjective genitive the new subject.

indrasya  $r\bar{a}jat\bar{a} = r\bar{a}j\hat{e}ndrah$  Indra is king

patasya  $n\bar{l}latvam = n\bar{l}lah patah$  the cloth is blue

It is important to realize that these constructions are simply different ways of expressing the same idea. The abstract suffix does not imply an abstraction of thought lacking in the alternative construction.

### 2.7.2 In compound

Abstract nouns in  $t\bar{a}$  or tva may be used as a member of a compound.

*tad-rājatā-hetuļ* his-kingness-reason: the reason for his being king *ātma-nityatvam* self-eternalness: the eternality of the self But more often  $t\bar{a}$  and tva are added to whole compounds in the sense of "being that which the compound denotes." The compound may be a *tatpuruşa*, *bahuvrīhi*, or *dvandva*.

*rāja-putratā* king-son-ness: being the son of a king (*tatpuruşa*)

*mahā-balatvam* great-strength-ness: being possessed of great strength (*bahuvrīhi*)

moha-śokatvam delusion-grief-ness: being delusion and grief (dvandva)

In analyzing such constructions it is essential to determine the type of the original compound (i.e., *tatpurusa*, *bahuvrīhi*, or *dvandva*), and it is helpful to keep in mind the alternative constructions without the abstract suffix.

 $r\bar{a}masya r\bar{a}ja$ -putrat $\bar{a} = r\bar{a}ja$ -putro  $r\bar{a}ma$  Rāma is the king's son

 $indrasya mah\bar{a}$ -balatvam = mah\bar{a}-bala indrah Indra is possessed of great strength

hetor moha-śokatvam = moha-śokau hetuh the cause is delusion and grief

### 2.7.3 Causal phrase within a sentence

The nominative of an abstract noun in  $t\bar{a}$  or tva may be used (as above) with a subjective genitive to form a complete statement. More often such a phrase is employed as part of a longer sentence where the case ending of the abstract noun indicates the relationship between the phrase and the other words of the sentence.

2.7. Common constructions

*rāmasya rāja-putratā viśrutā* Rāma's king-son-ness is well-known: it is well-known that Rāma is the king's son

*deva-patir indro mahā-balatvāt* Indra is lord of the gods because of [his] great-strength-ness: Indra is lord of the gods because he possesses great strength

avidyāyā duḥkha-hetutvam darśayati he shows ignorance's suffering-cause-ness: he shows that ignorance is the cause of suffering

duhkhasyâvidyā-hetutvam darśayati he shows suffering's ignorance-cause-ness: he shows that suffering has ignorance as its cause

The nominal construction consisting of abstract noun and subjective genitive is usually translated into English by a clause. The subjective genitive is often omitted in the Sanskrit when it is easily understood from the context; in the second example above we must supply *tasya* or *indrasya* with *mahā-balatvāt*. Note the importance of correctly analyzing the original compound. In the third example above the compound is a *tatpurusa*; in the fourth it is a *bahuvrīhi*. Sometimes an author will distinguish a *bahuvrīhi* compound by adding the optional suffix *ka* (e.g., in the fourth example one could say *avidyā-hetukatvam*; see Section 2.15), but usually the compound must be analyzed on the basis of context.

# 2.7.4 iti clause without abstract suffix

Because the word *iti* may be used to quote a statement or cite a reason (see Chapter 16), the above constructions can be expressed equivalently by means of an *iti* clause without the abstract suffix.

rāmasya rāja-putratā viśrutā = rāja-putro rāma iti viśrutam

deva-patir indro mahā-balatvāt = sa mahā-bala iti deva-patir indrah

avidyāyā duḥkha-hetutvam darśayati = duḥkha-hetur avidyêti darśayati

duḥkhasyâvidyā-hetutvaṃ darśayati = avidyā-hetuṃ duḥkham iti darśayati

An understanding of these equivalent constructions can be very helpful in trying to understand long and complex sentences using abstracts in  $t\bar{a}$  or tva.i

### 2.7.5 Two genitives

As was mentioned above, abstract nouns in  $t\bar{a}$  or tva are usually construed with a separate subjective genitive that may have to be supplied from the context. If another genitive is present, it is usually placed in compound with the abstract noun.

 $r\bar{a}masya r\bar{a}ja$ -putrat $\bar{a} = r\bar{a}j\tilde{n}ah$  putram  $r\bar{a}mah$  Rāma is the king's son

Here the subjective genitive is  $r\bar{a}masya$ , and  $r\bar{a}j\tilde{n}ah$  ( $r\bar{a}ja$ - in compound) is a possessive genitive. But occasionally the other genitive appears separately, particularly if emphasis is placed on its case and number, since these are obscured in compounds.

SBh 1.1.1 sarvasyâtmatvāc ca brahmâstitva-prasiddhih and the existence of brahman is well known because it is the self of everyone

Here *sarvasya* is a partitive genitive. and the subjective genitive (which must be supplied) is *brahmanah*.

# 2.8 Causal ablative and instrumental

Sanskrit permits the use of both the ablative and the instrumental cases to express a reason or cause, and abstract nouns in  $tv\bar{a}t$ , tvena, and  $tay\bar{a}$  are very common in this sense. The ablative of  $t\bar{a}$  is rare, probably because it is identical in form with the genitive (its use is prohibited by Pā. 2.3.25).

SBh 1.1.2 tatrâivam sati brahma-jñānam api vastu-tantram eva bhūta-vastu-viṣayatvāt this being the case, knowledge of brahman also depends only on the thing itself, since it has for its object an existing thing

SBh 1.1.2 na, vedānta-vākya-kusuma-grathanârthatvāt sūtrāņām no, since the sūtras have the purpose of stringing together the flowers that are the Vedānta passages

2.9. Other cases

SDS p. 3 na câsya [sukhasya] duhkha-sambhinnatayā purusârthatvam eva nâstîti mantavyam nor should it be thought that it [pleasure] is not the aim of man because it is mixed with pain

SBh 1.1.4 mantrāņām cêşe tvêtyādīnām kriyā-tat-sādhanâbhidhāyitvena karma-samavāyitvam uktam and it has been said that the mantras, işe  $tv\bar{a}$  and so on, are based on action because they declare action and the means to action

Several such abstract nouns may be combined in one sentence to express coordinate or serial causes; see Section 2.26 for a discussion of such constructions. Note also the idiomatic use of the instrumental discussed in Section 2.10.

# 2.9 Other cases

The following are fairly typical examples of abstract nouns in  $t\bar{a}$  or tva used in cases other than the ablative or instrumental.

### 2.9.1 Nominative as subject of sentence

SDS p. 2 atha katham parameśvarasya nihśreyasa-pradatvam abhidhīyate but how can it be said that the Supreme Lord bestows nihśreyasa?

### 2.9.2 Accusative as object of the sentence

SBh 1.1.1 yasmād veda evâgni-hotr-ādīnām śreyah-sādhanānām anitya-phalatām darśayati because the Veda itself shows that the means to success, the agnihotra and so on, have transient fruits

### 2.9.3 Dative of purpose

GBh 2.20 hantir atra vipariņāmârthe drastavyo 'punar-uktatāyai the verb hanti should be regarded here as signifying "transformation," to avoid redundancy

### 2.9.4 Genitive to modify another noun

SBh 1.1.7 tan na [ $\bar{a}$ tma-śabdasya] anekârthatvasyânyāyyatvāt that is wrong, because of the impropriety of [the word  $\bar{a}$ tman's] having more than one meaning

### 2.9.5 Locative in a locative absolute construction

(See Section 2.29.)

SBh 1.1.2 nanu bhūta-vastutve brahmaņaḥ pramāņântara-viṣayatvam eva but if brahman is an existing thing, then it must be the object of another means of knowledge

# 2.10 Predicative instrumental

The instrumental of an abstract noun in  $t\bar{a}$  or tva is often used idiomatically to complete the predicate of a sentence. The instrumental expresses "in what quality or capacity" and can usually be translated by "as."

 $r\bar{a}jatvena vrta indrah$  Indra was chosen with kingness: Indra was chosen as king

*mitratvena rāmaņ jānāmi* I know Rāma with friendness: I know Rāma as a friend (i.e. to be a friend)

Usually the context will distinguish this construction from the use of the instrumental abstract to state a reason (see Section 2.8). Further examples:

SBh 1.1.1 yadi tarhi loke brahmâtmatvena prasiddham asti if in that case brahman were well known in the world as the self

SBh 1.1.2 śrutyâiva ca sahāyatvena tarkasyâbhyupetatvāt and because śruti itself accepts reasoning as an aid

SBh 1.1.4 na hi śāstram idantayā viṣaya-bhūtam brahma pratipipādayiṣati for the śāstra does not wish to teach brahman as this, as an object 2.11. Accusative with verb of motion

In the last example above note the addition of the suffix  $t\bar{a}$  to the demonstrative pronoun *idam* to form *idantā* "this-ness, being this."

# 2.11 Accusative with verb of motion

The accusative of an abstract noun in  $t\bar{a}$  or *tvam* may be used idiomatically with a verb of motion to mean "to become."

rājatām gacchati he goes to kingness: he becomes king

nīlatvam yāti it goes to blueness: it becomes blue

In the same way the accusative may be used with the causative of a verb of motion or with a verb signifying "to lead," "to bring," etc., to mean "to cause to become, to make into."

*rājatām indram gamayati* he makes Indra go to kingness: he makes Indra a king

*nīlatvam paṭam nayati* he brings the cloth to blueness: he makes the cloth blue

# Chapter 13

an and strate when

金田 一 三 一 一 一 一 一

# Compounds

# 2.12 Introduction

Long and complicated compounds are common in most Sanskrit commentaries. This section deals with how to analyze long compounds, how a compound may be related to the other words of a sentence, three devices used to clarify ambiguous compounds, the use of *tat*, *yat*, and *kim* in compound, and several common terms used idiomatically at the end of a compound. (See also: Coulson 107–11; Speijer 193.3, 197, 213 b, 228–31; Whitney 1222 h.)

5

ę

# 2.13 Analyzing long compounds

(See also Sections 1.69-1.72.)

### 2.13.1 Identification

First one must determine the type of the original compound and its relation to the other words of the sentence. Long and difficult compounds are usually *tatpurusas* or *bahuvrīhis* (*avyayībhāvas* are rare and *dvandvas* are usually straightforward). Sometimes a compound can be identified on grammatical evidence alone: The gender and number of a *tatpurusa* are those of its final member. A *bahuvrīhi* is an adjective and its gender and number are those of the word (either expressed or understood) that it modifies. *bahuvrīhis* are sometimes distinguished by the suffix *ka* (see Section 2.15). When such evidence is inconclusive, the compound must be identified on the basis of context.

### 2.13.2 Dissection

Long compounds can often be analyzed as compounds of two elements, where the second element is the last member of the compound and the first element is everything that precedes the last member. By a series of such dissections, a long compound can often be reduced to several shorter compounds. For example, *dharmâdharma-jñāna-phalâbhyudayah*, "success that is the fruit of the knowledge of *dharma* and *adharma*," can be analyzed as follows:

dharmâdharmajñānaphala-abhyu	ıdaya karmadhāraya
dharmâdharmajñāna-phala	genitive tatpurusa
dharmâdharma-jñāna	genitive tatpurusa
dharma-adharma	dvandva

The shorter compounds from which the original long compound is constructed are usually *dvandvas* or *tatpuruṣas*; rarely are they *bahuvrīhis* (except for a few very common types of *bahuvrīhis*, e.g., those ending in  $-\bar{a}di$ ). Thus adjoining words within a long compound are usually to be construed as coordinate (*dvandva*: e.g., *dharmâdharma*) or as appositional (*karmadhāraya*: e.g., *phalâbhyudaya*) or as connected by a case relationship (*vibhakti-tatpuruṣa*: e.g., *jñāna-phala*).

### 2.13.3 Grouping

The analysis outlined above must be modified for some compounds. Sometimes two or more members form a natural group and must be so treated when the compound is resolved into its two basic elements.

indraputrapautrah = indra-putrapautrah the sons and grandsons of Indra

 $r\bar{a}masitasar\bar{a}h = r\bar{a}masitasar\bar{a}h$  the sharp arrows of Rāma

The original compound can still be resolved into two elements; however, the second element here is not a single word but a compound that forms a natural unit. Such units should be looked for when the simple procedure described above does not yield a satisfactory sense. A longer example is *janmâdi-sarva-bhāva-vikāra-varjitam*, "devoid of all transformations of state, such as birth and so on":

janmâdisarvabhāvavikāra-var	jita instrumental tatpurusa
janmâdi-sarvabhāvavikāra	karmadhāraya
sarva-bhāvavikāra	karmadhāraya
bhāva-vikāra	genitive tatpurusa
janma-ādi	bahuvrīhi

# 2.14 Relation to words outside the compound

In theory, a word standing outside of a compound may be construed with the compound as a whole and not with some prior portion of it. A word that is to be construed with a prior portion must be incorporated into the compound.

*jagaj-janmâdi-kāraṇaṃ brahmâdhigamyate* brahman is known to be the cause of the origin and so on of the world

Here *brahman* is construed with the entire compound and *jagat* is construed with *janmâdi* within the compound. But in actual practice this rule is often broken, and a separate word or phrase (especially one in the genitive case preceding the compound) is construed with a prior portion of the compound.

### SBh 1.1.3 jagato janmâdi-kāraņam brahmâdhigamyate

Such constructions are a mixture of the sentence mode of expression ( $v\bar{a}kya$ -vrti) and the compound mode of expression ( $sam\bar{a}sa$ -vrti). For example, "knowledge of the eternality of the self" might be expressed:

ātma-nityatva-vidyācompound modeātmano nityatvasya vidyāsentence modeātma-nityatvasya vidyāmixed mode: theoretically acceptableātmano nityatva-vidyāmixed mode: theoretically unacceptable

The last construction occurs less often than the preceding three, but it occurs often enough to warrant consideration when the usual interpretations of a passage do not yield a satisfactory sense.

# 2.15 ka to mark a bahuvrīhi

With a few minor exceptions the suffix ka may be added optionally to any bahuvrihi compound (Pa. 5.4.154). ka does not alter the meaning of the compound and is often added to obtain a stem more easily inflected, e.g.,  $mah\bar{a}$ -yaśaska from  $mah\bar{a}$ -yaśas. ka is also used to mark a compound as a bahuvrihi, especially where, because the gender and number of the compound are inconclusive and the context might admit alternative interpretations, the bahuvrihi might be mistaken for a *tatpurusa*.

*bhakti-nimittam jñānam* knowledge is the cause of devotion (*tatpurusa*); or, knowledge has devotion as its cause (*bahuvrīhi*)

*bhakti-nimittakam jñānam* knowledge has devotion as its cause (*bahuvrīhi*)

In the first example above only the context can determine whether the compound is a *tatpuruşa* or a *bahuvrīhi*; in the second example the compound must be a *bahuvrīhi*. *ka* may be added to a *bahuvrīhi* before the abstract suffixes  $t\bar{a}$  and tva; in this case *ka* is particularly helpful because the gender and number of the original compound are no longer evident.

*jñānasya bhakti-nimittatvam prasiddham* it is well-known that knowledge is the cause of devotion (*tatpuruṣa*); or, it is well-known that knowledge has devotion as its cause (*bahuvrīhi*)

*jñānasya bhakti-nimittakatvam prasiddham* it is well-known that knowledge has devotion as its cause (*bahuvrīhi*)

A further example:

GBh 1.1 hīyamāna-viveka-vijītāna-hetukenâdharmeņâbhibhūyamāne dharme when dharma is being overcome by adharma of which the cause is a lessening of discriminatory knowledge

# 2.16 *tat* to mark a *dvandva*

The pronoun *tat* may be used in a compound to form a *dvandva* (either of the entire compound or more often of part of the compound) where otherwise a *tatpurusa* would ordinarily be understood. In this case *tat* repeats a prior member of the compound. For example, *karma-phala-tyāgah* would ordinarily be understood as "renunciation of the fruits of actions," and *karma-phala* taken as a genitive *tatpurusa*. But *karma-tat-phala-tyāgah* means "renunciation of actions and their fruits." Here *karma-tat-phala* is a *dvandva* compound of *karma* and *tat-phala*, and *tat* refers to the preceding word *karma*. Further examples:

SBh 1.1.1 adhyāso vişaya-tad-dharmāņām superimposition of the object and its qualities

*PP p. 33 adhyāsa-svarūpa-tat-sambhāvanāya* for the sake of the nature of superimposition and the possibility of it

# 2.17 -bhūta to mark a karmadhāraya

The past participle  $-bh\bar{u}ta$  may be used within a compound to show that two adjoining members are to be construed as a *karmadhāraya* (i.e., as appositional) where otherwise one might construe them as a *vibhakti-tatpuruşa* (i.e., as connected by a case relationship) or as a *dvandva* (i.e., as coordinate).

*avidyā-kāraņa-vināśah* destruction of the cause of ignorance; or, destruction of the cause that is ignorance

avidyā-bhūta-kārana-vināśah destruction of the cause that is ignorance

In the first example only the context can determine whether  $avidy\bar{a}$ - $k\bar{a}rana$  means "the cause of ignorance" (genitive *tatpurusa*) or "the cause that is ignorance" (*karmadhāraya*); in the second example  $avidy\bar{a}$ - $bh\bar{u}ta$ - $k\bar{a}rana$  must mean "the cause that is ignorance." Further examples:

### Chapter 13. Compounds

GBh 2.10 prāņīnām śoka-mohâdi-samsāra-bīja-bhūta-dośôdbhavakāraņa-pradarśanârthatvena vyākhyeyo granthah this section should be interpreted as serving to show the cause of the origin of that defect that is the seed of samsāra, i.e., grief, delusion, and so on, for living beings

SBh 1.1.19 puccha-pratișțhā-bhūta-brahma-prāpteḥ phalasya nirdiștatvāt because the fruit that is the obtaining of brahman---who is the tail, the support----is indicated (in the text)

# 2.18 Pronouns in compound

The pronouns *tat*, *yat*, and *kim* appear frequently in compound (*yat* and *kim* as first member, *tat* as any member except last). In compound *tat*, *yat*, and *kim* (the stem forms) may stand for any gender and number. Thus *tad*- $r\bar{u}pah$  may be "his color," "her color," "its color," "their color," "the color of those two," etc. The compounds may be of various types; for example:

tad-rūpah its color (tatpurusa); or, having that as its color (bahuvrīhi)

yad-rūpah the color of which (*tatpuruṣa*); or, having which as its color (*bahuvrīhi*)

*kim-rūpa*<sup>*h*</sup> the color of what (*tatpurusa*); or, having what as its color (*bahuvrīhi*)

A special use of *tat* is mentioned in Section 2.16. Note that when *yat* is in compound, its correlative pronoun need not be in compound.

GBh 2.16 yad-visaya buddhir na vyabhicarati tat sat having-which-as-its-object a cognition is not in error, that is sat: that thing is sat when a cognition having it as its object is not in error

bahuvrihi compounds containing kim are often used to ask a question.

SBh 1.1.2 kim-lakṣaṇaṇ punas tad brahma but having-what-as-its-definition is that brahman; but what is the definition of that brahman?

#### 2.19. Idiomatic terms

# 2.19 Idiomatic terms

The following words are common in an idiomatic sense at the end of a compound. See Section 2.44 for *bahuvrīhis* in  $-\bar{a}di$ .

#### 2.19.1 -antara

*-antara*, n., "interval, difference," often forms neuter *tatpurusa* compounds meaning "a different, another."

rājântaram a different king, another king

grhântarāni different houses, other houses

Commentators usually gloss such compounds with the word *anya*; e.g., *rājântaram* would be glossed *anyo rājā*.

### 2.19.2 -artha

*-artha*, m.. "purpose," often forms masculine *tatpurusa* compounds used adverbially in the accusative singular to mean "for the sake of, in order to."

*dhanârtham* for the sake of money

ātmâikatva-pradarśanârtham in order to show the oneness of the self

Note that the same meaning may be conveyed by the dative case of the word or compound to which *-artha* is added.

tat-sādhanârtham = tat-sādhanāya in order to prove that

#### 2.19.3 -kalpa

-kalpa, m., "manner, way," forms bahuvrīhi compounds with the meaning "almost, nearly."

sa mrta-kalpah he is nearly dead

rāja-kalpo devadattah Devadatta is almost a king

Note: Pā. 5.3.67 explains -kalpa as a taddhita suffix (kalpaP in Pāņinian terminology) and commentators may identify it as such.

194

#### 2.19.4 -gata

-gata, the past participle of gam, may be used in a tatpurusa compound to mean "being in or on," where no previous motion is implied.

ksetra-gatā vrksāh the trees in the field

Such compounds may also mean "relating to, concerning."

bhāryā-gatā cintā worry for his wife

In both instances *-gata* conveys a meaning usually expressed in other constructions by the locative case.

### 2.19.5 -garbha

-garbha, m., "womb, foetus," forms bahuvrīhi compounds meaning "having inside, containing, filled with."

ksīra-garbham ghatam a pot filled with milk

ratna-garbhah syūtah a sack containing jewels

#### 2.19.6 -jana

-*jana*, m., "person, people," forms masculine *tatpurusa* compounds that may denote either an indefinite individual or a collection.

pitr-janah a father; or, fathers

In longer compounds *-jana* often pluralizes a preceding member that might otherwise be understood as singular.

mitrânugrhitah helped by a friend; or, helped by friends

mitra-janânugrhitah helped by friends

### 2.19.7 -jāta

**新新的社会。1994年中央。19**14年1946年

**不是最低位于的最新的的** 

 $-j\bar{a}ta$ , n., "collection," forms neuter *tatpurusa* compounds denoting a collection of things.

kārya-jātam the aggregate of effects

In longer compounds *-jāta* may pluralize a preceding member that might otherwise be understood as singular.

deva-prasādah the grace of the god; or, the grace of the gods

deva-jāta-prasādah the grace of the gods

### 2.19.8 -pūrva

 $-p\bar{u}rva$ , "previous," when modifying a past participle in a *tatpurusa* compound may be placed after the participle.

 $sankalpita-p\bar{u}rva = p\bar{u}rva-sankalpita$  previously imagined, already. imagined

#### 2.19.9 -mātrā

*-mātrā*, "measure," forms *bahuvrīhi* compounds whose literal meaning is "having ... as its measure, having the size of."

grha-mätro gajah an elephant the size of a house

Often such compounds have the special meaning "measured by, being no more than, mere, only." In this case the compound often appears as a neuter singular substantive.

SBh 1.1.1 deha-mātram caitanya-visistam ātmā the self is just the body qualified by consciousness

### Chapter 13. Compounds

When *-mātra* has this meaning commentators usually gloss the compound with the word *eva*; e.g., *deha-mātram* would be glossed *deha eva*. Occasionally compounds in *-mātra* have the opposite meaning: "the full measure of, the whole class of, any."

guna-mātra-visaya buddhih an idea referring to any quality

Thus -mātra can generalize as well as particularize.

Note: Pā. 5.2.37 explains *-mātra* as a *taddhita* suffix (*mātraC* in Pāṇinian terminology) and commentators may identify it as such. See Section 1.60.3.

### 2.19.10 -viśesa

-viśesa, m., "distinction, difference, peculiarity," may form masculine tatpurusa compounds meaning "a particular" or "a distinguished."

giri-visesah a particular mountain

rüpa-viśesah distinguished beauty

tatpurusas in -visesa and in -bheda, m., "difference" may also mean "a kind of, a variety of."

puspa-viśesah a kind of flower

mani-viśesah a variety of jewel

# Chapter 14

# Word Order

# 2.20 Introduction

There are relatively few constraints on the order of words in Sanskrit prose, and some commentators of a more literary style vary their sentence structure often. But some observations about word order can be helpful, especially in reading long sentences. This section discusses the order of subject and predicative nouns, the order of attributive and predicative nouns, the scope of the word *iti*, and the underlying syntactical construction of long sentences. See Section 2.26 for the conventions of word order in causal sentences. (See also: Coulson 46–49; Speijer 16–17, 31.)

1

# 2.21 Subject and predicative nouns

### 2.21.1 Predicative nouns

on several months and one of

Predicative adjectives agree with the subject in number, gender, and case, while predicative substantives agree in case but not necessarily in gender or number.

udāro rāmah Rāma is noble

mitram rāmah Rāma is a friend

A predicative noun usually precedes the subject, especially in short sentences or when the predicate is emphasized.

SBh 1.1.4 ato na purușa-vy $\bar{a}p\bar{a}ra$ -tantr $\bar{a}$  brahma-vidy $\bar{a}$  therefore knowledge of brahman does not depend on the activity of man

SBh 1.1.4 nityaś ca mokṣaḥ sarvair mokṣa-vādibhir abhyupagamyate and mokṣa is accepted as eternal by all those who uphold the doctrine of mokṣa

The same order is usually followed in sentences with predicative accusatives.

SBh 1.1.4 evam-ādyāh śrutayo mokṣa-pratibandha-nivṛtti-mātram evâtma-jñānasya phalam darśayanti scriptural passages such as these show that the fruit of self-knowledge is simply the cessation of obstacles to mokṣa

But it is also quite common for the subject to precede a predicative noun, particularly if the subject is emphasized or if it has just been mentioned in the previous sentence.

SBh 1.1.4 jñānam tu pramāņa-janyam, pramāņam ca yathā-bhūta-vastu-vişayam. But knowledge arises from a pramāņa. And a pramāņa has for its object a thing as it really is.

Thus the general rule that a predicative noun precedes the subject unless the subject is emphasized is often but not always followed. Subject and predicative noun are best distinguished on the basis of context: usually the subject has been mentioned or implied in previous sentences and the predicative noun contains some new information about the subject.

### 2.21.2 Pronoun subjects

Pronoun subjects deserve special mention. The pronoun often agrees in gender and number with a predicative substantive.

tan mitram he is a friend

The pronoun may either precede or follow the predicative noun.

Sometimes the pronoun is placed in the middle of a predicate of two or more words.

PP p. 79 nanu mahad etad indra-jälam but this is a great deception

# 2.22 Attributive and predicative nouns

## 2.22.1 Word order with subjects

In general, attributive nouns express something already known or established about the subject, while predicative nouns assert something new. Thus the context is usually sufficient to distinguish the one from the other when both occur in the same sentence. But sometimes word order is also helpful, especially in long sentences. Some authors characteristically place attributive nouns before and predicative nouns after the subject.

SBh 1.1.4 ata evânustheya-karma-phala-vilakṣaṇam mokṣâkhyam aśarīratvam nityam iti siddham therefore it is proven that bodilessness, which is different from the fruit of actions to be performed and is called mokṣa, is eternal

SBh 1.1.1 evam ayam anādir ananto naisargiko 'dhyāso mithyā-pratyaya-rūpah kartrtva-bhoktrtva-pravartakah sarva-loka-pratyaksah thus this beginningless, endless, natural superimposition takes the form of false knowledge, causes [the self] to be an agent and enjoyer, and is evident to everyone

In these examples attributive nouns precede and predicative nouns follow the respective subjects, *aśarīratvam* and *adhyāso*.

# 2.22.2 Word order with other nouns

学習に見つき

This same order, i.e., attributive qualifiers—substantive—predicative qualifiers, may also be used for substantives that are not the grammmatical subject of the sentence.

SBh 1.1.1 asmat-pratyaya-gocare vişayini cid-ātmake ... adhyāsah superimposition on the vişayin, which is the object of the notion of I and whose nature is intelligence ...

Here asmat-pratyaya-gocare expresses a quality of the visayin that was mentioned previously, while *cid-ātmake* expresses a new quality. The distinction might be indicated in English as follows: "the visayin, which is the object of the notion of I, has intelligence as its nature; superimposition on it..."

### 2.22.3 Markers of attributive qualifiers

Attributive qualifiers may be marked as such by the present participle *sant* placed after the qualifier (compare the use of *sant* to gloss attibutive qualifiers; see Section 1.84.2), or by the past participle *bhūta* placed in compound with the qualifier (Section 1.84.3. Usually *sant* and *bhūta* serve to make the construction clearer and can be omitted without changing the meaning of the sentence.

SBh 2.1.4 tathêdam api jagad acetanam sukha-duḥkha-mohânvitam sad acetanasyâiva sukha-duḥkha-mohâtmakasya kāraṇasya kāryam bhavitum arhati in the same way this world also which is unconscious and possessed of pleasure, pain, and delusion must be the effect of a cause that is unconscious and whose nature is pleasure, pain, and delusion

SBh 1.1.5 tri-guņatvāt tu pradhānasya sarva-jñāna-kāraņa-bhūtam sattvam pradhānâvasthāyām api vidyate but since the pradhāna consists of the three guņas, sattva, which is the cause of all knowledge, exists also in the pradhāna state

Here sad shows that acetanam sukha-duhkha-mohânvitam qualifies jagad and bhūta shows that sarva-jñāna-kāraņa qualifies sattvam. In many cases sant has the special meaning "although being."

SBh 1.1.5 pradhānasyâcetanasyâiva sataḥ sarvajītātvam upacaryate vedānta-vākyeṣu in the Vedānta texts omniscience is figuratively ascribed to the pradhāna, although it is really unconscious

Often this meaning is confirmed by adding the particle api to sant; see Section 2.54.2. Occasionally  $bh\bar{u}ta$  marks the predicate of the sentence instead of an attributive qualifier.

PP p. 183 avidyā ca samsāra-hetu-bhūtā and ignorance is the cause of samsāra

# 2.23 Scope of iti

The word *iti* is normally placed immediately after the word or words that are quoted by it. Thus *iti* marks the end of a quotation, but in most cases the

beginning must be determined on the basis of context. Sometimes words placed at the beginning of a sentence for the sake of emphasis are to be construed with words after the *iti* clause; in this case care must be taken not to include these beginning words in the *iti* clause. Negatives such as *na ca* and *nahi* are especially common in this construction. In the following examples quotation marks have been added to show the extent of each *iti* clause.

SBh 1.1.1 tam ke cid "anyatrânya-dharmâdhyāsa" iti vadanti some say that it is the superimposition of the qualities of one thing on another thing

PP p. 356 na ca "sa na pratīyata" iti yuktam vaktum nor is it reasonable to say that it is not known

SBh 1.1.4 nahi "'śarīrâdy-ātmâbhimānino duḥkha-bhayâdimattvam dṛṣṭam' iti tasyâiva veda-pramāṇa-janita-brahmâtmâvagame tad-abhimāna-nivṛṭtau tad eva mithyā-jñāna-nimittam

duhkha-bhayâdimattvam bhavati" iti śakyam kalpayitum for it cannot be imagined on the ground that someone who wrongly identifies his self with his body, etc. is seen to possess pain, fear, etc., that when the same person has knowledge of the self as *brahman* produced by the *pramāna* which is the Veda and when the mistaken identification has ceased, he should have that same possession of pain, fear, etc. which was caused by false knowledge

The last example shows that one *iti* clause may include another: The first *iti* quotes the words *śarīrâdy-ātmâbhimānino*... *drṣṭam* and this clause expresses a reason. The second *iti* quotes all the words between *nahi* and *śakyam kalpayitum* and this clause expresses what it is that cannot be imagined.

# 2.24 Analyzing long sentences

Unusually long and difficult sentences are best approached by first identifying their underlying syntactical construction. Once the skeletal construction is understood, the remaining words of the sentence can be construed

2.24. Analyzing long sentences

correctly. Sometimes it is helpful (when possible) to underline the words comprising the skeletal sentence. For example:

SBh 1.1.3 kim u vaktavyam "aneka-sākhā-bheda-bhinnasya deva-tiryan-manuşya-varnāśramâdi-pravibhāga-hetor rg-vedâdy-ākhyasya sarva-jñānâkarasya aprayatnenâiva līlā-nyāyena puruşa-nihśvāsavad yasmān mahato bhūtād yoneh sambhavah 'asya mahato bhūtasya nihśvāsitam etad yad rgveda' ityādi-śruteh tasya mahato bhūtasya niratiśayam sarvajñatvam sarva-śaktimattvam ca" <u>iti</u>

The skeletal construction of this sentence (underlined above) is:

kim u vaktavyam ... iti how much more must it be said that ...

Thus the final *iti* quotes every word of the sentence after *vaktavyam*. Within this long quotation, the following skeletal construction (in boldface above) may be seen:

sarva-jñānâkarasya... yasmān mahato bhūtād yoneḥ sambhavaḥ... tasya mahato bhūtasya niratiśayaṃ sarvajñatvaṃ sarva-śaktimattvaṃ ca from which great being as the source there arose that mine of all knowledge, that great being possesses supreme omniscience and omnipotence

To this skeletal construction the remaining elements of the sentence may be added. The first is a series of qualifiers of *sarva-jñānâkarasya*:

#### aneka-śākhā-bheda-bhinnasya

deva-tiryan-manusya-varnaśramadi-pravibhaga-hetor

*rg-vedâdy-ākhyasya* divided according to various branches, the cause of the distinctions of gods, animals, men, class and stage of life, etc., and known as the Rig Veda, etc.

The second is an adverbial phrase modifying sambhavah:

aprayatnenâiva līlā-nyāyena purusa-niķśvāsavad without any effort at all, in the manner of sport, like a man's breathing out

The third is substantiation of the entire statement by a quotation from scripture:

asya mahato bhūtasya niḥśvāsitam etad yad rgveda ityādi-śruteḥ because of scriptural passages such as "from that great being this was breathed forth which is the Rig Veda"

The whole sentence may now be translated:

How much more must it be said that that great being possesses supreme omniscience and omnipotence, from which great being as the source there arose—without any effort at all, in the manner of sport, like a man's breathing out---that mine of all knowledge that is divided according to various branches, that is the cause of the distinctions of gods, animals, men, class and stage of life, etc., and that is known as the Rig Veda, etc.—as we know from scriptural passages such as "from that great being this was breathed forth which is the Rig Veda."

# Chapter 15

# **Causal Sentences**

# 2.25 Introduction

Sentences that express a reason, cause, or motive are essential for interpretation and discussion and so are common in almost every commentary. The constructions most often used to state a reason are ablative and instrumental phrases, clauses formed by *iti*, clauses introduced by a relative pronoun or adverb, locative absolute constructions, and clauses introduced by the causal particle *hi*. For example, each of the following may convey the idea "because Indra is king":

indrasya rājatvāt	(ablative phrase)
rajêndra iti	(iti clause)
yasmād rajêndraķ	(relative pronoun)
indre rājñi sati	(locative absolute)
rājā hîndrah	(causal particle hi)

. ر:

Note that commentators often vary these constructions in successive sentences and occasionally combine two or more of them in the same sentence.

SBh 1.1.2 kartavye hi vişaye nânubhavâpeksâstîti śruty-ādīnām eva prāmāņyam syāt. puruşâdhīnâtma-lābhatvāc ca kartavyasya. For if the object [of the knowledge of brahman] were something to be accomplished, then there would be no dependence upon direct experience; consequently the *śruti* and so on would be the only means of knowledge. And because the origination of something to be accomplished depends upon man. (See also: Coulson 170-71; Speijer 365-68; Whitney 303.)

# 2.26 Ablative and instrumental phrases

### 2.26.1 Ablative causal phrases

Nouns in the ablative or (less often) the instrumental case may express a reason or cause. Abstract nouns in  $t\bar{a}$  or tvam are especially common in this construction (see Section 2.8). The noun may be a single word or a compound. The reason may be expressed by the noun alone or by a nominal phrase of which the noun is a member. The noun or nominal phrase may form a complete statement by itself, or it may stand at the beginning, middle, or end of a longer sentence.

SBh 1.1.1 paśvâdibhiś câviśeṣāt and because there is no difference from cows, etc.

SBh 1.1.4 ato vedāntānām ānarthakyam akriyârthatvāt therefore the Vedānta texts are meaningless because they are not concerned with action

SBh 1.1.7 bhrtye tu svāmi-bhrtya-bhedasya pratyakṣatvād upapanno gauņa ātma-śabdah but in the case of a servant the word ātman may be used in a secondary sense because the distinction of master and servant is obvious

In the first example above the nominal phrase *paśvâdibhiś câviśeṣāt* is a complete sentence; the conclusion, stated earlier in the passage, must be supplied by the reader. In the second example the compound *akriyârthatvāt* substantiates the assertion made in the prior portion of the sentence. In the third example the nominal phrase *svāmi-bhṛtya-bhedasya pratyakṣatvād*, inserted in the middle of the sentence, gives a reason for the assertion made by the rest of the sentence.

## 2.26.2 Series of causes

In sentences citing a series of causes, the more removed cause is generally placed before the nearer cause. Two conventions of case are common: either the whole series of causes is put in ablative cases, or the ablative and instrumental cases alternate (the ablative being used for the nearest cause). Examples of the former:

SBh 1.1.4 kriyā-samavāyâbhāvāc câtmanaḥ kartṛtvânupapatteḥ and because the self cannot be an agent since there is no inherence of action (in it)

SBh 1.1.4 heyôpādeya-śūnya-brahmâtmatâvagamād eva sarva-kleṣa-prahāṇāt puruṣârtha-siddheḥ because of the accomplishment of the aim of man, which results from getting rid of all afflictions, which in turn results only from the knowledge that brahman who is devoid of anything to be avoided or sought after is the self

Examples of the latter:

「「「「」」

SBh 1.1.2 na, indrivâvişayatvena sambandhâgrahaņāt no, because the connection is not grasped, since [brahman] is not an object of the senses

SBh 1.1.4 sarva-gatatvena nityâpta-svarūpatvāt sarveņa brahmaņah because brahman is by nature always present to everyone, since it is omnipresent

Note the importance of reading the entire Sanskrit sentence before beginning to translate. Often the most natural English translation reverses the order of the Sanskrit, beginning with the nearest cause and proceeding to the most removed cause.

### 2.26.3 ca in causal series

In sentences citing several causes it is essential to understand correctly the conjunction ca. ca may connect the sentence to the preceding sentence, as in the first example in Section 2.26.2. In this case ca is placed after the first word of the sentence. ca may also connect causes within the sentence, and here one must distinguish between coordinate causes (connected by ca) and serial causes (where one is the cause of the other). Sometimes the meaning of ca cannot be determined by its position alone and must be determined by context.

#### 2.27. iti clauses

#### Chapter 15. Causal Sentences

SBh 1.1.23 tad ayuktam śabda-bhedāt prakaranāc ca samśayânupapatteh that is not right since a doubt is not possible because of the difference of words and because of the topic

SBh 1.1.1 na tāvad ayam ekantenâvişayo 'smat-pratyaya-vişayatvād aparokşatvāc ca pratyag-ātma-prasiddheh first of all, it [the self] is not a non-object in an absolute sense, because it is the object of the notion of I and because the inner self is well known since it is directly experienced

In the first example above *ca* connects *śabda-bhedāt* and *prakaraņāc*; both are causes of *samśayânupapatteh*. In the second example, the particle *ca* connects *asmat-pratyaya-viṣayatvād* and *aparokṣatvāt pratyag-ātma-prasiddheh*; here *aparokṣatvāt* is the cause of *pratyag-ātma-prasiddheh*.

# 2.26.4 Verbal nouns expressing cause

Verbal nouns formed by adding the suffixes *a*, *ana*, *ana* or *ti* to the root are often used as the final member of a compound that expresses a reason or cause. Often the resulting nominal construction is most conveniently translated into English by a clause.

In some cases the same meaning may be expressed by replacing the verbal noun with an abstract noun formed by adding  $t\bar{a}$  or tva to the past passive participle.

*kārya-drstatvāt* because of the seen-ness of the effect: because the effect is seen, because we see the effect

In the following examples note the use of the words *abhāva*, *upapatti*, *asiddhi*, and *sambhava*.

SDS p. 3 dehâtirikta ātmani pramāņâbhāvāt because there is no proof of a self separate from the body

SBh 1.1.22 nâisa doso bhūtâkāsasyâpi vāyv-ādi-kramena kāranatvôpapatteh this fault does not apply because the elemental ether also can be a cause through the series beginning with air SBh 1.1.4 śarīra-sambandhasyâsiddhatvād dharmâdharmayor ātma-krtatvâsiddheḥ because it is not established that dharma and adharma are performed by the self, since the connection of the body [with the self] has not been established

SBh 1.1.5 api ca pradhānasyânekâtmakasya pariņāma-sambhavāt kāraņatvôpapattir moreover the heterogeneous pradhāna can be the cause because of the possibility of transformation (in it)

Some of the most common of such verbal nouns are listed below, along with the meanings that they often have in this construction.

abhāva; -abhāvāt abhyupagama; -abhyupagamāt upapatti; -upapatteḥ darśana; -darśanāt prasiddhi; -prasiddheḥ

yoga; -yogāt sambhava; -sambhavāt siddhi; -siddheḥ absence; because there is no acceptance; because we accept that fitness; because it is reasonable that seeing; because we see that reknown; because it is generally known that fitness; because it is reasonable that possibility; because it is possible that establishment; because it is established

Other terms used in the same construction include *smarana* and *śravana* (see Section 2.41) and *prasanga* and *āpatti* (see Section 2.50.3).

that

# 2.27 *iti* clauses

One of the many uses of *iti* is to cite a reason or cause (see Chapter 16 for other uses). In this case *iti*, whose basic meaning is "thus, in this way," comes to mean "thinking thus, on these grounds, for this reason." Usually the words quoted by *iti* precede the assertion or conclusion that they substantiate.

SBh 1.1.2 anyeşam api bhāva-vikārāņām trişv evântar-bhāva iti janma-sthiti-nāśānām iha grahaņam the other transformations of state also are included among just these three; therefore the text mentions birth, subsistence, and destruction

 $k\bar{a}rya$ -darśanāt because of seeing the effect: because the effect is seen, because we see the effect

Sometimes *iti* is followed by a pronoun or adverb that serves to distinguish this use of *iti* from other common uses. For example, *iti tasmāt, iti tena, ity ataḥ*, "therefore" (lit. "from that" where "that" refers to the content of the *iti* clause).

SBh 1.1.4 tasmān na pratipatti-vidhi-visayatayā śāstra-pramāņakatvam brahmaņah sambhavatīty atah sva-tantram eva brahma śāstra-pramāņakam therefore it is impossible that brahman should be made known by the śāstra as the object of an injunction to know; consequently brahman is made known by the śāstra as an independent entity only

This construction may be made even plainer by the addition of a word such as *kāraņa*; e.g., *iti tasmāt kāraņāt* "for that reason." *iti krtvā*, which often means "thinking thus," may also mean "therefore, for this reason."

GBh 2.10 kṣātram karma yuddha-lakṣaṇam

guru-bhrātr-putrâdi-himsā-lakṣaṇam atyantam krūram api sva-dharma iti krtvā nâdharmāya the action of a warrior, characterized by battle and by violence to his teachers, brothers, sons, etc., though extremely cruel, is his own dharma; therefore it does not lead to adharma

# 2.28 Relative pronouns and adverbs

A reason or cause may be expressed in a clause introduced by a relative pronoun or adverb such as *yasmāt*, *yena*, *yat*, *yataḥ*, or *yāvatā*. The relative in this case functions as a conjunction and has the meaning "because." Such a clause may form a complete sentence that substantiates a previous assertion.

SBh 1.1.14 itaś ca prācuryârthe mayaţ, yasmād ānanda-hetutvam brahmaņo vyapadišati śrutir eşa hy evânandayatîti. And for this reason the suffix maya means abundance. Because the scriptural passage "for he alone causes bliss" shows that brahman is the cause of bliss.

Or the clause may substantiate an assertion made earlier in the same sentence.

GBh 2.14 sukha-duhkhe punar niyata-rūpe yato na vyabhicaratah but pleasure and pain are constant in nature because they do not vary

When the reason precedes the conclusion, the conclusion is usually introduced by a correlative pronoun or adverb such as *tasmāt*, *tena*, *tat*, *ataḥ*, *tataḥ*, or *tāvatā*. The pronoun or adverb functions as a conjunction and has the meaning "therefore."

PP p. 258 yasmād brahmâvāptih purusârthah tena tat jāānenâptum istatamam since obtaining brahman is the aim of man, therefore it [brahman] is what one most desires to obtain through knowledge

Occasionally the relative pronoun or adverb is placed at the end of the clause expressing the reason and the correlative is omitted.

SBh 1.1.6 tathā pradhānam api mahad-ādy-ākāreņa niyamena pravartate yasmāc cetanavad upacaryate in the same way the pradhāna also acts according to rule in the form of the mahat and so on; for this reason it is figuratively said to be intelligent

SBh 1.1.4 nâpi saṃskāryo mokṣo yena vyāpāram apekṣeta nor is mokṣa something to be purified, by reason of which it would depend on action

One way to understand this construction is to supply the correlative before the conclusion: the first example becomes ... yasmāt tasmāc cetanavad upacaryate "because ..., therefore it is figuratively said to be intelligent"; the second example becomes ... yena tena vyāpāram apekṣeta "because ..., therefore it would depend on action." See Chapter 20 for further discussion of pronouns and pronomial adverbs.

# 2.29 Locative absolute

Commentators often use the locative absolute construction to express causal and conditional clauses. Usually the locative absolute precedes the main assertion of the sentence. The basic significance of this construction, i.e., "this being the case," must be variously interpreted according to the context. Note: included here are constructions involving a single abstract noun in the locative case. While these are not strictly locative absolutes, they are used in the same way (such locatives are regarded as *nimitta-saptami* "locative of motive" by the Indian grammarians, though they may be converted to locative absolutes without changing the meaning by adding *sati*).

### 2.29.1 Expressing a cause

The locative absolute may express a reason or cause, in which case it means "given the fact that, since."

SBh 1.1.1 dehêndriyâdişv aham-mamâbhimāna-rahitasya pramātŗtvânupapattau pramāņa-pravŗtty-anupapatteh because the pramāņas could not function, since one who does not identify "I" and "mine" with the body, senses, etc. cannot be a knower

### 2.29.2 Expressing a condition

The locative absolute may express a condition, in which case it means "if, when."

SBh 1.1.1 teşu [mumukşatvâdişu] hi satsu prāg api dharma-jijfiāsāyā ūrdhvam ca śakyate brahma jijfiāsitum jñātum ca for when these [the desire for mokşa, etc.] exist, it is possible to desire to know and to know brahman, both after and even before the desire to know dharma

The condition may be hypothetical, in which case an optative often appears in the conclusion.

SBh 1.1.2 sati hîndriya-vişayatve brahmana idam brahmana sambaddham kāryam iti grhyeta for if brahman were an object of the senses then this [world] would be perceived as connected with brahman as its effect

#### 2.29.3 Concessive locative absolute

The addition of *api* gives the locative absolute the concessive meaning "even though" (see Section 2.54.2).

SBh 1.1.4 na ca pariņisthita-vastu-svarūpatve 'pi pratyaksâdi-visayatvam brahmaņah and, even though it has the nature of an accomplished thing, brahman is not an object of perception, etc.

### 2.29.4 Abbreviated locative absolute

The locative absolute often appears in abbreviated forms such as *tatrâivam* sati, evam sati, and *tathā sati*. Each of these means "this being the case," where "this" refers to some previously stated condition. Note that the locative absolute in general consists of a "subject" and a "predicate," i.e. words that would become the subject and predicate if the locative absolute were made into an independent sentence. For example, in *rāme gate* "Rāma being gone" the subject is *rāme* and the predicate is *gate* (the independent sentence being *gato rāmah*). In these abbreviated forms the predicate consists of the participle *sant* and the adverbs *evam* or *tathā*, while the subject is either omitted or represented by *tatra* (equivalent here to *tasmin*; see Section 2.55).

SBh 1.1.4 evam ca saty athâto brahma-jijñāsêti tad-viṣayaḥ pṛthak-śāstrârambha upapadyate and this being the case, the undertaking of a separate śāstra which has that for its object, with the words athâto brahma-jijñāsā, is appropriate

SBh 1.1.5 tathā sati yan-nimittam īkṣitṛtvam pradhānasya tad eva sarvajñam mukhyam brahma jagatah kāranam if that were the case then the very cause of the pradhāna's being a seer would be the omniscient and primary brahman who is the cause of the world

# 2.30 Causal particle hi

A reason or cause may be introduced by the particle *hi* "since, for." Usually *hi* is placed after the first word of the sentence or clause in which it appears.

#### 214

SBh 1.1.1 sarvo hi puro'vasthite vişaye vişayântaram adhyasyati since everyone superimposes an object on another object that is in front of him

SBh 1.1.1 na hîndriyāny anupādāya pratyakşâdi-vyavahāraḥ sambhavati for without the senses the functioning of perception and so on is not possible

**Chapter 16** 

ALTERNAL ALTERNAL

# Uses of iti

# 2.31 Introduction

The particle *iti* "thus, in this way" has many uses in Sanskrit commentaries. These uses all derive from the basic function of *iti*: it marks the preceding word or words or sentences as a quotation of some sort. "Quotation" is used here in a broad sense; the *iti* clause need not consist of words actually spoken or thought and may describe more generally a reason, purpose, situation, and so on. This section discusses typical constructions employing *iti* and the use of *iti* in quotation, explanation, and comment, as well as constructions employing the infinitive (both with and without *iti*) and the use of *iti* to mark the end of a passage. Quotation marks have been added in the examples to show the extent of each *iti* clause. See Section 2.27 for the use of *iti* to cite a reason or cause and Sections 2.44.3 and 2.45.3 for *ityādi*, *itivat*, and *ityādivat*. The various uses of *iti* in glossing are discussed in Part One. (See also: Coulson 203–05, 230–35; Speijer 383–90, 491–99.)

# 2.32 Common constructions

The words quoted by *iti* usually form a complete statement. If a single word is quoted it normally appears in the nominative case.

"devadatta" iti tam vadanti they call him Devadatta

Generally the *iti* clause stands first in the sentence. In some cases words before the *iti* clause are construed with words following *iti*, and it is possible for one *iti* clause to include another; see Section 2.23 for such constructions. The words immediately after *iti* normally indicate the relationship of the *iti* clause to the rest of the sentence. However *iti* is also used alone, without other explanatory words, to mark words spoken (= *ity uktvā*) or thought (= *iti matvā*) or more generally to give an explanation of some sort.

"indra upagacchati" iti matvā palāyate = "indra upagacchati" iti palāyate he flees, thinking "Indra is coming"

Occasionally demonstratives such as *evam* and *etad* are added redundantly to *iti*.

SBh 1.1.12 "tasmāt saṃsāry evânandamaya  $\bar{a}tm\bar{a}$ " ity evaṃ prāpte therefore the self consisting of bliss is just a transmigrating being—the  $p\bar{u}rvapaksa$  being thus established

SBh 1.1.4 tasmād "aham brahmâsmi" ity etad-avasānā eva sarve vidhayah sarvāņi cêtarāņi pramāņāni therefore all the injunctions and all the other pramāņas come to an end in the realization "I am brahman"

Here ity evam  $pr\bar{a}pte = iti pr\bar{a}pte$  and ity etad-avas $\bar{a}n\bar{a} = ity avas \bar{a}n\bar{a}$ .

# 2.33 Quotation

#### 2.33.1 *iti* with source

*iti* may quote the actual words of a person or text, or the views held by a particular person or school of thought. Usually *iti* is followed by a word or phrase giving the source of the quotation. A verb meaning "to say," "to think," etc. may be added but is not necessary.

SBh 1.1.2 "śrotavyo mantavya" iti śrutih the scriptural passage "[the self] is to be heard, is to be thought about"

SBh 1.1.5 "pradhāna-puruṣa-saṃyoga nityânumeya" iti sānkhyā manyante the Sānkhyas believe that the connections of the pradhāna with the puruṣas must necessarily be inferred

### 2.33.2 *iti* with ablative

Often the word giving the source of a quotation is put into the ablative case. Here the ablative does not signify that the words are quoted from that source; rather, it means "because ... says ...."

This construction is very common in appeals to scriptural authority.

SBh 1.1.5 yat tu jñānam manyase sa sattva-dharmah "sattvāt sañjāyate jñānam" iti smrteh but what you think is knowledge is actually the quality of sattva because smrti says that knowledge arises from sattva

### 2.33.3 iti with partial quotations

A commentator may wish to refer to a long passage without quoting the passage in its entirety. In this case *iti* is often used to quote selected sentences from the passage. These quoted sentences may be connected by expressions such as the following:

... ity ārabhyâha ... iti beginning thus ... the text goes on to say ...

tathā... iti similarly the text says ...

ante ca ... iti and the text concludes ...

... ityādi ... ityantam bhāşyam the bhāşya beginning with ... and ending with ...

#### An example:

SBh 1.1.2 kim punas tad vedānta-vākyam yat sūtreņêha lilaksayisitam. "bhrgur vai vāruņir varuņam pitaram upasasāra. adhīhi bhagavo brahma" ity upakramyâha "yato vā imāni bhūtāni jāyante. yena jātāni jīvanti. yat prayanty abhisamvišanti. tad vijijñāsasva." tasya ca nirņaya-vākyam "ānandād dhy eva khalv imāni bhūtāni jāyante. ānandena jātāni jīvanti. ānandam prayanty abhisamvišanti" iti.

But which Vedānta text does the  $s\bar{u}tra$  wish to indicate here? Beginning with "Bhṛgu Vāruṇi indeed approached his father Varuṇa and said 'Sir, teach me brahman'," the passage goes on to say "from which indeed these beings are born; by which, when born, they live; into which, when dying, they enter---desire to know that." And its concluding sentence is "from bliss only, of course, these beings are born; by bliss, when born, they live; into bliss, when dying, they enter."

# 2.34 Explanation

An *iti* clause is often used to explain a particular term or concept. Sometimes the *iti* clause anticipates what a person might think or say in the situation being described.

SBh 1.1.1 "aham idam mamêdam" iti naisargiko 'yam loka-vyavahārah there is this natural worldly practice expressed in the thought "this is I, this is mine"

SBh 1.1.1 sarvo hy ātmâstitvam pratyeti na "nâham asmi" iti for everyone knows the existence of the self; no one thinks "I am not."

Occasionally the *iti* clause follows the term that it explains.

SBh 1.1.1 na câyam asti niyamah "puro'vasthita eva vişaye vişayântaram adhyasitavyam" iti nor is there an invariable rule that an object can be superimposed only on another object that is in front of us

SBh 1.2.1 punar apy anyāni vākyāny aspasţa-brahma-lingāni sandihyante "kim param brahma pratipādayanty aho svid arthântaram kim cid" iti but other passages, in which the characteristic marks of brahman are not clearly set forth, give rise to doubt: do they refer to the highest brahman or to something else?

Here the meanings of *niyamah* and *sandihyante* are specified by the subsequent *iti* clauses.

# 2.35 Comment

## 2.35.1 Comment with iti

An author may comment on a statement by adding *iti* and a word or phrase that conveys the desired comment. The passive (third person singular) and

the past passive participle (neuter singular) are most often used in this construction. Note that these two are identical in meaning; e.g., *iti niścitam* = iti niścīyate. Occasionally one finds the synonymous construction *iti niścitam bhavati*.

SBh 1.1.7 tasmāc "cetana-vişaya ihâtma-śabda" iti niścīyate therefore it is settled that the word ātman here refers to a conscious entity

SBh 1.1.11 tasmāt "sarvajñam brahma jagatah kāraņam nâcetanam pradhānam anyad vā" iti siddham therefore it is established that the omniscient brahman is the cause of the world, not the unconscious pradhāna nor something else

SBh 1.1.14 yo hy anyān ānandayati sa pracurânanda iti prasiddham bhavati for it is well-known that one who causes others to be blissful must himself abound in bliss

A gerundive, an adjective, or a substantive may also be used with *iti*.

SBh 1.1.4 "yasya tu yathā-pūrvam samsāritvam nâsav avagata-brahmâtma-bhāva" ity anavadyam it is irrefutable that one who is a transmigrating being as before has not comprehended that brahman is his self

PP p. 163 "nâpi jñānântaram utpannam" iti viśesah nor does another cognition arise—this is the difference

Note that some of the phrases commonly used in glossing, e.g., *ity arthah* (see Section 1.10) or *ity anvayah* (see Section 1.44), employ the same construction.

### 2.35.2 Commonly used comments

Some of the most common comments with *iti* are listed below. In some cases more than one form is used; e.g., *iti viruddham, iti virudhyate*, and *iti virodhah* are all used to mean "this is a contradiction."

*ity anavadyam* this is irrefutable *ity abhyupagatam* this we accept

> Universität Hamburg · Asien-Afrika-Institut Abteilung für Kultur und Gaschichte

2.36. Constructions with the infinitive

ity abhyupetam	this we accept
ity avasitam	this is settled
ity upapannam	this is reasonable
iti gamyate	this can be inferred
iti drstam	this is seen (in ordinary experience)
iti niścitam	this is settled
iti prasiddham	this is generally known
iti bādhitam	this is ruled out
iti bhavisyati	this might be
iti yuktam	this is reasonable
iti viruddham	this is a contradiction
iti sambhāvyate	this is possible
iti siddham	this is established
iti sugamam	this is clear
iti spastam	this is clear
,	

# **2.36** Constructions with the infinitive

The following constructions with the infinitive are used both to govern an *iti* clause (compare comments with *iti* in Section 2.35) and by themselves.

### 2.36.1 Infinitive with sak and yuj

The infinitive is often used with the passive or the gerundive of  $\dot{s}ak$  "to be able," or with the passive or the past passive particle of yuj "to fit, to be suitable."

*iti śakyate vaktum = iti śakyam vaktum* this can be said, it is possible to say this

*iti yujyate vaktum = iti yuktam vaktum* this can reasonably be said, it is reasonable to say this

Note that the infinitive in Sanskrit has only one form, and acquires an active or pasive sense according to the mode of the accompanying verb.

devadattah śaknoti ghatam kartum it is possible for Devadatta to make a pot (active)

devadattena śakyate ghatam kartum it is possible for a pot to be made by Devadatta (passive)

In the examples below the infinitive has a passive sense and may be rendered in English by a passive infinitive (e.g., "to be made").

SBh 1.1.25 katham punaś "chando'bhidhānān na brahmâbhihitam" iti śakyate vaktum but how can it be said that brahman is not denoted because the meter is denoted?

SBh 1.1.4 ata sa [ātmā] na kena cit pratyākhyātum śakyo vidhi-śeṣatvam vā netum therefore it [the self] cannot be denied by anyone, not can it be made subordinate to an injunction

SBh 1.1.1 "vişayini ... vişayasya tad-dharmānām câdhyāsah tad-viparyayena vişayinas tad-dharmānām ca vişaye 'dhyāso mithā" iti bhavitum yuktam it is reasonable that superimposition of the object and its qualities on the subject and conversely superimposition of the subject and its qualities on the object are false

SBh 1.1.7 na tu nirdosam sāstram apramāņam kalpayitum yuktam but it is not reasonable to imagine that the faultless sāstra is not a pramāņa

Note that *sakyam* and *yuktam* sometimes appear as impersonal indeclinables in this construction; e.g., in the second example above *sakyo* agrees with *sa*, but the sentence would also be correct if *sakyam* were substituted for *sakyo*.

### 2.36.2 Infinitive with *arh*

In other states of the second

The infinitive may be combined with the verb arh "to be obliged to" to give the sense "ought, should." The infinitive of  $bh\bar{u}$  is particularly common in this construction; e.g., *iti bhavitum arhati* "this must be the case."

SBh 1.1.12 para evâtmânandamayo bhavitum arhati the [self] consisting of bliss must be the highest self

220

i

2.37. End of a passage

SBh 1.1.4 nârhaty evam bhavitum kārya-vidhi-prayuktasyâiva brahmaņah pratipādyamānatvāt it does not have to be so, since brahman is taught only in so far as it is connected with injunctions to action

## 2.36.3 Infinitive expressing purpose

The infinitive sometimes expresses aim or purpose.

*rāmam drastum agacchat* he went to see Rāma (i.e., in order to see Rāma)

This sense of purpose may be conveyed by an infinitive governing an *iti* clause.

PP p. 30 "nyonya-dharmāś ca" iti pṛthag dharma-grahaṇaṃ "dharma-mātrasyâpi kasya cid adhyāsa" iti darśayitum the separate mention of the word dharma in the phrase anyonya-dharmāś ca serves to show that there may also be superimposition of some quality only

# 2.37 End of a passage

*iti* sometimes marks the end of a section of text. This usage is particularly common in passages expressing a  $p\bar{u}rvapaksa$  view (see Chapter 19).

SBh 1.1.1 tat punar brahma prasiddham aprasiddham vā syāt. yadi prasiddham na jijñāsitavyam. athâprasiddham nâiva śakyam jijñāsitum iti. ucyate... But that brahman must be either well-known or not well-known. If it is well-known then one need not desire to know it; if it is not well-known then it is impossible to desire to know it. We reply ...

A much longer example is found in *SBh* 1.1.4, where a *pūrvapakṣa* discussion of many lines is concluded as follows:

tasmāt pratipatti-vidhi-visayatayâiva śāstra-pramāņakam brahmâbhyupagantavyam iti. atrâbhidhīyate... Therefore brahman is to be accepted as proven by the *sāstra* only in so far as it is the object of the injunction to know. Here we reply ... Compare this construction with the common use of *iti* to mark the end of a chapter or an entire work.

iti śrīmac-chārīraka-mīmāmsā-bhāṣye śrī-śankara-bhagavat-pāda-kṛtau prathamâdhyāyasya prathamah pādah thus ends the first pāda of the first adhyāya of the commentary on the Śrīmac-chārīraka-mīmāmsā composed by Śrī-śankara-bhagavat-pāda

# Chapter 17

# References

# 2.38 Introduction

The most common reference to another person or text is a direct quotation followed by *iti* and the name of the person or text quoted (see Section 2.33). This section discusses other terms and constructions that are often used in making references. These will be taken up according to the nature of the reference: a commentator may refer to the text or the author of the text he is commenting on, to his own work, to various recognized authorities, and to persons holding a particular view. The traditional terms  $m\bar{u}la$  and  $m\bar{u}lak\bar{a}ra$  are used in this section to refer to the text being commented on and the author of the text being commented on, respectively. See Section 2.57.2 for the use of *etat* and *idam* to refer to preceding and subsequent passages of a work.

. بو<sup>2</sup>

# 2.39 Text and author of text

### 2.39.1 Verbs of taking

#### erbs of taking-(

Carling and

Derivatives of the roots grah "to grasp," parigrah "to embrace," and  $up\bar{a}d\bar{a}$  "to take" may indicate what is specifically mentioned or referred to in the  $m\bar{u}la$ . For example:  $up\bar{a}d\bar{a}na$  "mention of or reference to in the  $m\bar{u}la$ ," grh $\bar{t}a$  "mentioned or referred to in the  $m\bar{u}la$ ."

SBh 1.1.22 tasmād ākāśa-śabdena bhūtâkāśasya grahaņam therefore by the word "ether" there is reference in the mūla to the elemental ether

SBh 1.1.20 tac ca paramêśvara-parigrahe ghațate and that makes sense if the reference in the mūla is to the Supreme Lord

SBh 1.2.1 sarvam khalv idam brahmêti sva-śabdenâiva brahmôpāttam brahman is directly mentioned by its own name in the passage "of course all this is brahman"

erbs of taking—)

### 2.39.2 prakrta, aprakrta

The past participle *prakrta* "being the subject of discussion" may refer to what is being talked about in a passage of the  $m\bar{u}la$ . Conversely *a*-prakrta points to what is not the subject of discussion.

SBh 1.1.26 itaś câivam abhyupagantavyam asti pūrvasmin vākye prakrtam brahmêti and for this reason we must accept the fact that brahman is the topic of the previous passage

SBh 1.1.12 kim ihânandamaya-sabdena param eva brahmôcyate yat-prakrtam satyam jñānam anantam brahmêti does the word ānandamaya here refer to the highest brahman that is the subject of the passage "brahman is true, knowledge, endless"?

### 2.39.3 Author of text commented on

The *mūlakāra* may be referred to directly by name, by an honorary title such as *bhagavān*, or by a descriptive term such as *sūtrakāra* "author of the *sūtras*" or *bhāsyakāra* "author of the *bhāsya*."

SBh 1.1.2 kim-lakṣaṇaṃ punas tad brahmêty ata āha bhagavān sūtrakāraḥ but what is the definition of that brahman? the blessed author of the sūtras explains (in the following sūtra)

### 2.39.4 Author as understood speaker

的國家的關於總統國家的目的

金融

のないないです。

The  $m\bar{u}lak\bar{a}ra$  is often the unexpressed subject (or agent, in a passive construction) of a verb in the third person singular. Commentators use such verbs (especially in the *avatarana* or brief introduction to a verse or other portion of the  $m\bar{u}la$ ) to explain the intentions of the  $m\bar{u}lak\bar{a}ra$  in a given passage. For example,  $\bar{a}ha$  "the  $m\bar{u}lak\bar{a}ra$  says" often introduces a quote from the  $m\bar{u}la$ .

PP p. 135 tat-sādhanârtham āha pratyag-ātma-prasiddher iti in order to substantiate that, the mūlakāra says "because the inner self is well-known"

GBh 2.23 kasmād [ātmā] avikriya evêty āha why is [the self] changeless? the mūlakāra explains (in the following verse)

Other phrases employing  $\bar{a}ha$  include *ity āsankyâha* "having raised this doubt the *mūlakāra* says" (see Section 2.49.4), *dṛṣṭântam āha* "the *mūlakāra* gives an example," and *uttaram āha* "the *mūlakāra* replies." Note that *āha* sometimes means "an objector says" and serves to introduce an objection (see Section 2.49.6). *ucyate* "is said by the *mūlakāra*" may introduce a quote from the *mūla*, especially where this answers a question or objection.

GBh 2.48 yadi karma-phala-prayuktena na kartavyam karma katham tarhi kartavyam ity ucyate if action is not to be performed by a person impelled by the fruit of action, then how is it to be performed? the mūlakāra explains (in the following verse)

Note that *ucyate* is also used to introduce the commentator's own reply to an objection (see Section 2.50.1). Some other common verbs used in this construction are listed below (he =  $m\bar{u}lak\bar{a}ra$ ).

*ārabhate* he begins *āśańkate* he raises a doubt *udāharati* he cites as an example *upasaṃharati* he summarizes darśayati he shows

pariharati he refutes

prapañcayati he explains in detail

varñayati he describes

The future tense refers to a passage of the *mūla* subsequent to the one presently under consideration; e.g., *vakṣyati* "the *mūlakāra* will state," *darśayisyati* "the *mūlakāra* will show."

SBh 2.1.6 tarkâpratișțhāṇād iti ca kevalasya tarkasya vipralambhakatvam darśayiṣyati and the sūtrakāra will show the fallaciousness of mere reasoning in the sūtra "tarkâpratisțhāṇāt"

### 2.39.5 Participles of verbs of saying

Participles of verbs meaning "to say," "to show," etc. may refer to the *mūlakāra*; e.g., *ukta* may mean "stated by the *mūlakāra*," *vakṣyamāṇa* may mean "to be stated by the *mūlakāra*," and *vivakṣita* may mean "intended by the *mūlakāra*" (lit., "what he means to say").

SBh 1.1.1 brahma ca vakṣyamāṇa-lakṣaṇam janmâdy asya yata iti and the definition of brahman will be stated by the sūtrakāra in the sūtra "from which there is the origin and so on of this (world)"

SBh 1.1.31 tri-vidham iha brahmôpāsanam vivakṣitam a three-fold meditation on brahman is intended here (i.e. in this passage) Note that such participles may also refer to the commentator's own work (see Section 2.40.2).

## 2.39.6 Instrumental of present participle

The instrumental of a present participle may refer to the  $m\bar{u}lak\bar{a}ra$  in passive constructions where the agent (the  $m\bar{u}lak\bar{a}ra$ ) is not actually mentioned.

SBh 1.1.3 kim-artham tarhîdam sūtram yāvatā pūrva-sūtra evâivam-jātīyakam sāstram udāharatā sāstra-yonitvam brahmaņo daršitam then what is the purpose of this sūtra, in as much as by adducing such a sāstra on the previous sūtra it was shown that the sāstra is the source of brahman?

PP p. 137 nanu brahma-vidyām anartha-hetu-nibarhaņīm prati-jānatâvidyânartha-hetuh sūcitā objection: by asserting that knowledge of brahman destroys the cause of evil, it is shown that ignorance is the cause of evil

Here the present participles *udāharatā* and *pratijānatā* properly refer to *sūtrakāreņa*, the unexpressed agent of *darśitam* and *sūcitā*, respectively. Note that neither the Sanskrit nor the English translation requires mention of the word *sūtrakāra*.

### 2.39.7 Auto-commentary

**建立**346 图

When an author writes a commentary on his own work (a so-called "autocommentary") he normally refers to the *mūlakāra* as if the *mūlakāra* were another person. Thus the third person is used (not the first person, as one might expect), and the other terms and constructions described above are employed where appropriate.

# 2.40 References to the commentary

### 2.40.1 First person plural

A commentator often refers to his own work with a verb in the first person plural. The past and future tenses refer to preceding and subsequent passages of the commentary.

SBh 1.1.4 nityam aśarīratvam akarma-nimittatvād ity avocāma we said earlier that bodilessness is eternal because it does not have action as its cause

SBh 1.1.1 yathā câyam arthah sarveşām vedāntānām tathā vayam asyām śārīraka-mīmāmsāyām pradaršayişyāmah and how this is the meaning of all the Vedānta texts we shall show in this śārīraka-mīmāmsā

2.41. Recognized authorities

### 2.40.2 Participles

Participles of verbs meaning "to say," "to show," etc. may refer to the commentator's own work. The past participle *ukta* is particularly common in such references; e.g., *ukta* "previously stated," *yathôkta* "as previously stated," *ukta-nyāyena* "in the manner previously stated."

#### GBh 2.21 nanûkta evâtmano 'vikriyatvam

 $sarva-karm \hat{a}sambhava-k \bar{a}ra na-visesa h$  objection: it was previously said that the particular cause of the impossibility of all actions (in the self) is the fact that the self does not change

#### SBh 1.1.5 brahma ca sarvajñam sarva-śakti

*jagad-utpatti-sthiti-nāśa-kāraņam ity uktam* and it has been said that *brahman* is the omniscient, omnipotent cause of the origin, subsistence, and destruction of the world

Note that such participles may also refer to the mūlakāra (see Section 2.39.5).

## 2.40.3 Relative clause

A relative clause introduced by *yat* and concluded by *iti* may quote some earlier statement of the commentary. This construction is especially common where an earlier point is raised again in the course of discussion. For example, *yat tûktam* ... *iti* or simply *yat tu* ... *iti* may mean "what was said previously, i.e., ... ."

SBh 1.1.4 yat tu heyôpādeya-rahitatvād upadeśânarthakyam iti nâişa dośah but as for what was said before, i.e., that the teaching is purposeless because it is free of anything to be avoided or sought after—that fault does not apply.

When the correlative *tat* is used it refers to the entire statement quoted by *yat* and *iti*.

SBh 1.1.5 yat tûktam sattva-dharmena jñānena sarvajñam pradhānam bhavişyatîti tan nôpapadyate but what was said before, i.e., that the pradhāna may be omniscient because knowledge is a quality of sattva—that is unreasonable. Note that the same construction (but not with *ukta*) may be used to quote something other than the commentator's own work, e.g., a rival interpretation.

## SBh 1.1.4 yad api ke cid āhuķ

*pravrtti-nivrtti-vidhi-tac-cheşa-vyatirekena kevala-vastu-vādī veda-bhāgo nâstîti tan na* and what some say, i.e., that no portion of the Veda declares a mere thing separate from injunctions to act or to refrain from acting or from something subordinate to such injunctions—that is wrong

## 2.40.4 Reference to earlier topic

An abbreviated reference to earlier commentary may be made by mentioning the topic of the earlier discussion.

 $v\bar{a}yun\bar{a} vy\bar{a}khy\bar{a}te$  this is explained by  $v\bar{a}yu$ : this is explained by what we said concerning  $v\bar{a}yu$ 

# 2.40.5 agre

的。如此是一個的時代。目的目的。如此在各時代的

The adverb *agre* "in front of" means "later on, below" when used in reference to the commentator's own work.

evam agre vaksyate thus it will be explained later on

# 2.41 Recognized authorities

Special terms may be used to refer to recognized authorities. The choice of such terms varies according to the subject matter of the work, the particular commentator, and the nature of the authority referred to. A few examples are given below.

Derivatives of *āmnā* refer to what is taught in the sacred or traditional texts: *ity āmnāyate* "this is taught in the sacred texts," *āmnāta* "taught in the sacred texts."

Derivatives of drs' refer to what is seen in ordinary experience: *iti* drsyate = iti drstam "this is seen in ordinary experience," *iti drstam loke* 

"this is seen in the world," -*darśanāt* (final member of compound) "because it is seen that...."

The word *nyāya* refers to a popular maxim: *iti nyāyāt* "because of this popular maxim."

Derivatives of *prasidh* refer to what is generally known in the world: *iti prasidhyate = iti prasiddham* "this is generally known," *iti prasiddham loke* "this is generally known in the world," *-prasiddheh* (final member of compound) "because it is generally known that..."

Derivatives of *smr* refer to what is taught in *smrti: iti smaryate* "this is taught in *smrti,*" -*smaranāt* (final member of compound) "because *smrti* teaches that...."

Derivatives of *śru* refer to what is taught in *śruti: iti śrūyate* "this is taught in *śruti,*" -*śravanāt* (final member of compound) "because *śruti* teaches that..."

# 2.42 Genitive in reference to particular views

Persons who hold a particular view may be referred to by a word in the genitive case, where the genitive is construed with a word or phrase that expresses the view in question. The reference may name the persons directly.

SBh 1.1.4 yathā pṛthivy-ādi jagan-nityatva-vādīnām yathā ca sānkhyānām guņāh like the earth and so on for those who believe that the world is eternal, and like the gunas for the Sānkhyas

Or a general reference may be made to anyone who holds such a view.

SBh 1.1.4 yasya tûtpādyo mokşas tasya mānasam vācikam kāyikam vā kāryam apekşate for whomever mokşa is something to be produced, for him it depends on a mental, verbal, or bodily effect

Commentators normally gloss such a genitive with the word *mate* "in the view." For example, in the examples above *sānkhyānām* would be glossed *sānkhyānām mate* "in the view of the Sānkhyas," while *yasya* would be glossed *yasya mate* "in the view of whomever."

Chapter 18

# **Examples and Comparisons**

# 2.43 Introduction

Certain terms are commonly used to introduce examples and comparisons. *bahuvrīhis* in *ādi* often give specific examples of a more general term. Comparisons may be drawn by adverbs formed with the suffix *vat* or by phrases containing *iva* or *yathā*. Such comparisons may give a corroborating example (e.g., "this situation is exemplified in this instance") or may comprise the main statement (e.g., this situation is like that situation"). See Section 2.53.7 for the use of *tathā hi* and *tad yathā*. (See also: Coulson 49–50, 195, 214–15; Speijer 229.1, 241, 430, 470 a; Whitney 1107.)

÷٢,

# 2.44 Bahuvrīhis in ādi

### 2.44.1 Compounds in -ādi

A REPUBLIC MARKED AND A REPUBLIC

A CHARGE

 $\bar{a}di$  m. "beginning" is very common in *bahuvrīhi* compounds whose literal meaning is "of which ... is the first, beginning with ..." Sometimes the prior member of the compound is in fact the first element of a series and the compound is best translated literally. In such cases a *bahuvrīhi* in *anta* m. "end" often expresses the final element of the series.

SBh 1.1.19 *sira-ādisu pucchântesv avayaves-ktesu* when the parts beginning with the head and ending with the tail have been mentioned

More often the prior member of the compound is simply a characteristic or prominent member of the group that is denoted by the compound as a whole. Here a less literal translation of  $\bar{a}di$  is more appropriate.

indrâdayo devāh the gods, Indra and so on; Indra and the other gods

 $\bar{a}k\bar{a}s\hat{a}d\bar{n}i$  bhūtāni the elements, ether and so on; the elements, such as ether

Here bahuvrihis in  $\bar{a}di$  give examples of the general terms they modify: Indra is a particular god and ether is a particular element. Often the general term is unexpressed and the bahuvrihi is used as a substantive.

```
āgatā rāmâdayah Rāma and the others came
```

Here  $r\bar{a}m\hat{a}dayah$  mentions a particular individual (Rāma) and denotes (in addition to Rāma) the set of individuals associated with Rāma. Note that the *bahuvrīhi* is often singular in this construction, even though it refers to more that one thing.

SBh 2.1.24 nahi vāyur ākāšo vâuṣṇyâdinā balād dadhi-bhāvam āpadyate for neither air nor ether is compelled by heat and so on to turn into curds

*bahuvrīhis* of the same meaning as those in  $\bar{a}di$  may be formed by  $\bar{a}dika$  (see Section 2.15), the related word  $\bar{a}dya$ , and other words meaning "beginning" such as *prabhrti*.

indrâdayo devāh = indrâdikā devāh = indrâdyā devāh = indra-prabhrtayo devāh

### 2.44.2 Series of compounds

Several *bahuvrīhis* in  $\bar{a}di$  may be used in a single statement in order to make the sentence logically complete. Often it is not necessary to translate each  $\bar{a}di$  by "etc." or "and so on," and a more natural translation can be obtained by considering the purpose of the  $\bar{a}dis$ .

2.44. Bahuvrīhis in ādi

SBh 2.1.1 sarvajñah sarvêśvaro jagata utpatti-kāranam mrt-suvarnâdaya iva ghata-rucakâdīnām the omniscient Lord of all is the cause of the origin of the world, just as clay and gold, for instance, (are the causes of the origin) of a pot and a gold ornament

SBh 1.1.1 paśvâdayah śabdâdibhih śrotrâdīnām sambandhe sati śabdâdi-vijñāne pratikūle jāte tato nivartante when there is contact of sense objects such as sound with sense organs such as the ear and when an unpleasant cognition of sound, etc. has arisen, cows and other animals move away from that

In the first example above the  $\bar{a}dis$  simply show that other related examples could be named if desired, and so the  $\bar{a}dis$  may be rendered by "for instance." In the second example each *bahuvrīhi* in  $\bar{a}di$  implies a more or less well defined set of items (e.g., *śrotrâdi* "the ear and so on" refers to the five sense organs), and so the  $\bar{a}di$  compounds may be rendered by the general terms that they imply.

#### 2.44.3 ityādi

The compound *ity-ādi* (lit., "of which the first is thus") may be used in direct quotation to indicate that the quotation is one of several that might be named. In this case *ity-ādi* has the sense "quotations such as...." Like other *bahuvrīhis* in *ādi, ity-ādi* may modify another word in the sentence, or the word that it modifies may be unexpressed, or *ity-ādi* may occur in compound.

SBh 1.1.4 kim-arthāni tarhy ātmā vā are drastavyah śrotavya ity-ādīni vidhi-cchāyāni vacanāni then what is the purpose of passages that resemble injunctions, such as "the self indeed, my dear, is to be seen, is to be heard"?

SBh 1.1.12 evam ākāśas tal-lingād ity-ādisu drastavyam so it is to be regarded in passages such as ākāśas tal-lingāt

SBh 2.1.7 sarvam tam parādād yo 'nyatrâtmanah sarvam vedêtyādi-śravanāt because of śruti passages such as "everything has deserted him who knows everything in something other than the self" 236

Sometimes *evam* is added redundantly to form expressions such as *ity-evam-ādi* (= *ity-ādi*) or *iti câivam-ādi* (= *ity-ādi ca*).

# 2.45 Suffix vat

#### 2.45.1 Adverbs in -vat

The suffix *vat* (neuter accusative singular of the possessive suffix *vant*) often forms adverbs meaning "like, as." Such adverbs may be made from any noun, e.g., *dehavat* "like the body." The grammatical relationship between the noun and the other words of the sentence is not indicated in the form of the adverb and must be understood from the context. The correct use of *vat* when it means "like, as" is limited by Pāṇini 5.1.115–116 to constructions that can be paraphrased in one of the following ways:

*brāhmaņavad vartate = brāhmaneņa tulyam vartate* he acts like a Brahmin (paraphrase by instrumental of noun with *tulyam*—similarity of action must be denoted)

mathurāvat śrughne prākārah = mathurāyām iva śrughne prākārah the wall in Śrughna, like the one in Mathurā (paraphrase by locative of noun with iva)

devadattavad yajñadattasya gāvah = devadattasyêva yajñadattasya gāvah Yajñadatta's cows, like Devadatta's (paraphrase by genitive of noun with iva)

However in actual practice adverbs in *vat* are often used incorrectly, especially in popular literature; e.g.,  $m\bar{a}trvat para-d\bar{a}r\bar{a}n paśyati = m\bar{a}taram iva para-d\bar{a}r\bar{a}n paśyati$  "he looks on another's wife as he looks on his mother" (paraphrase by accusative of noun with *iva*).

## 2.45.2 With standard of comparison

Adverbs in *vat* often express a comparison where the noun to which *vat* is added is the standard of comparison (*upamāna*).

SBh 1.1.4 sa ca dagdhêndhanâgnivat svayam evôpaśamyate and that extinguishes itself, like a fire whose fuel has been consumed

2.45. Suffix vat

Here dagdhêndhanâgnivat = dagdhêndhanâgnir iva (note that vat is used correctly here, since dagdhêndhanâgnivat svayam ev "paśamyate can be paraphrased by dagdhêndhanâgninā tulyam svayam ev "paśamyate, and the point of comparison is the act of extinguishing itself). Often the suffix vat is added to bahuvrīhis in ādi to indicate that the noun mentioned in the bahuvrīhi is one of several that might be named for the sake of comparison.

*dehâdivat* like the body and so on; as is the case, for example, with the body

Sometimes the noun to which *vat* is added is construed with another word of the sentence (see Section 2.14 for a parallel construction in which the prior member of a compound is construed with a word outside the compound).

SDS p. 2 tebhya eva dehâkāra-parinatebhyah kiņvâdibhyo mada-śaktivac caitanyam upajāyate consciousness arises from just those [four elements] which are transformed into the form of the body, just as the power of intoxication (arises) from yeast and so on

### 2.45.3 itivat, ityādivat

法律法

The adverbs *itivat* "as is the case in this quotation" and *ityādivat* "as is the case in quotations such as this" are used when direct quotations are presented for comparison.

SBh 1.1.4 nâpi viśista-kriyā-yoga-nimittam vāyur vāvā samvargah prāna vāvā samvarga itivat nor is [knowledge of the self as brahman] based on connection with a particular action, as is the case in the passages "wind indeed is the absorber," "breath indeed is the absorber"

SDS p. 6 mama śarīram iti vyavahāro rāhoh śira ityādivad aupacārikah the expression "my body" is figurative, as is the case in expressions such as "Rāhu's head"

# 2.46 iva and yathä

Comparisons are often expressed by *iva* "like, as" or *yathā* "like, as." If the standard of comparison (*upamāna*) is one word, *iva* is placed after it; if it consists of several words, *iva* is often placed in the middle of them. *yathā* usually precedes the standard of comparison and is particuarly common when the standard of comparison is expressed by a verbal clause.

SBh 2.1.4 nanu dharma iva brahmany apy anapeksa āgamo bhavitum arhati objection: the sacred texts must be independently authoritative with regard to brahman also, just as (they are) with regard to dharma

SBh 1.1.4 svåtma-dharma eva sams tiro-bhūto mokṣaḥ kriyayâtmani samskriyamāne 'bhivyajyate yathâdarśe nigharṣaṇa-kriyayā samskriyamāne bhāsvaratvam dharmaḥ mokṣa is a hidden quality of the self that is manifested when the self is purified by action, just as the quality of brightness (is manifested) when a mirror is purified by the act of rubbing

Comparisons may be made more explicit by using the correlatives  $yath\bar{a} \dots tath\bar{a}$  (or *evam* or *ittham*).... If the subject under discussion is compared to something else, then the former is normally treated as the subject of comparison (*upameya*) and placed in the *tathā* clause.

SBh 1.1.4 yathä svargâdi-kāmasyâgnihotrâdi-sādhanam vidhīyata evam amŗtatva-kāmasya brahma-jñānam vidhīyate just as means such as the agnihotra are enjoined for one who desires heaven and so on, so knowledge of brahman is enjoined for one who desires immortality

Sometimes this construction is abbreviated by replacing the  $tath\bar{a}$  clause with tadvat, lit., "like that," i.e., "so it is here (in the case under discussion)."

SBh 1.1.1 yathā rājāsau gacchatīty ukte sa-parivārasya rājāo gamanam uktam bhavati tadvat just as the sentence "there goes the king" implies the going of the king with his retinue, so it is here

# Chapter 19

# **Terms of Discussion**

# 2.47 Introduction

Discussion of the meaning of the text being commented on usually takes the form of a dialogue between the commentator and one or more opponents who raise objections to his interpretation. The parts of the dialogue are marked by certain conventional terms that may be roughly divided into those used to present an objection and those used to reply to an objection. Certain terms also mark an alternative interpretation in cases where the commentator is willing to admit another view.

(1)

# 2.48 Dialectic style

## 2.48.1 Oral debate

Analysis of the meaning and implications of the text being commented on (as opposed to analysis of the words of the text, discussed in Part One) usually appears as a dialogue whose form is based, both historically and stylistically, on oral debate. The view of the commentator and his school is known as the *siddhānta*, lit., "the demonstrated conclusion." Opposing views are included under the term *pūrvapaksa*, lit., "the prior view." The *pūrvapaksa* may be the established doctrine of another school, or it may be any question or doubt anticipated by the commentator. The dialogue that leads to proof of the *siddhānta* may take several forms. Often a problem is posed, a *pūrvapaksa* is put forth, and a debate develops be-

2.48. Dialectic style

#### Chapter 19. Terms of Discussion

tween the *pūrvapakṣin* (the upholder of the *pūrvapakṣa*) and the *siddhāntin* (the upholder of the *siddhānta*, i.e., the comentator) that eventually leads to confirmation of the *siddhānta*. Sometimes the *siddhānta* is stated first, only to be attacked by the *pūrvapakṣin* and defended in the ensuing debate. In longer arguments the *siddhānta* may be confronted with a series of *pūrvapakṣas*. In this case a commentator often permits one *pūrvapakṣin* to debate another (or several others) before the *siddhānta* is finally established. Such passages deserve careful reading since the reply to an objection may not represent the commentator's own view but may be the reply of one *pūrvapakṣin* to another. Note that when the text is a collection of *sūtras*, some *sūtras* may express a *pūrvapakṣa* but lack any of the terms described in Section 2.49 that ordinarily mark an objection. However *pūrvapakṣa sūtras* (both with and without the special terms that mark an objection) are usually clearly identified as such by the commentators.

### 2.48.2 Pūrvapaksa and siddhānta

The terms pūrvapaksa and siddhānta are occasionally used directly.

SBh 1.1.4 tu-sabdah pūrvapaksa-vyāvrtty-arthah the word tu (in the sūtra) serves to rule out the pūrvapaksa

When the  $p\bar{u}rvapaksa$  is the well known view of another school it may be identified as such.

SBh 1.1.5  $k\bar{a}n\bar{a}d\bar{a}s$  tv etebhya eva vākyebhya īsvaram nimitta-kāraņam anumimate 'nūms ca samavāyī-kāraņam the followers of Kanāda on the other hand infer from these same passages that the Lord is the efficient cause and the atoms are the material cause

More often the  $p\bar{u}rvapaksa$  and *siddhānta* are identified by the terms discussed in Sections 2.49 and 2.50.

The word paksa "side, alternative" may refer to any particular view.

SBh 1.1.5 tāni [vedānta-vākyāni] pradhāna-kāraņa-pakķe 'pi yojayitum śakyante those [Vedānta passages] can also be interpreted on the view that the pradhāna is the cause

## 2.48.3 Direct address

The influence of oral debate on the dialectic style of Sanskrit commentaries is seen most clearly when the *siddhāntin* and the *pūrvapakṣin* address one another directly. Second person pronouns and second person verbs are not uncommon in such passages.

SBh 1.1.1 yuşmat-pratyayâpetasya ca pratyag-ātmano 'vişayatvam bravīşi and you (i.e., the siddhāntin) say that the inner self which is free of the notion of you is not an object

SBh 1.1.5 idam tāvad bhavān prastavyah katham nitya-jāāna-kriyātve sarvajāatva-hānir iti first of all, you (i.e., the pūrvapaksin) are to be asked this: why should [brahman's] always being engaged in cognitional activity cause it to lose its omniscience?

### 2.48.4 Example of dialectic style

The following passage is an example of the dialectic style and illustrates some of the terms discussed in Sections 2.49 and 2.50.

PP pp. 100-101 tenântaḥkāranôparāga-nimittam mithyâivâhankārtrtvam ātmanah sphatika-maņer ivôpādhāna-nimitto lohitimā. Therefore the self's thinking "I am the agent," caused by the coloring (of the self) by the mind, is false, just as the red color in a crystal caused by proximity [to a red *japā* flower] is false

katham punah sphatike lohitimno mithyātvam. But how is the red color in the crystal false?

ucyate yadi sphatika-pratisphālitā nāyana-raśmayo japā-kusumam upasarpeyus tadā višista-sannivesam tad eva lohitam grāhayeyuh. na hi rūpa-mātra-nisthaś cāksusah pratyayo drsta-pūrvah. nâpi svâśrayam anakarsad rūpa-mātram pratibimbitam kva cid upalabdha-pūrvam. We reply: if the ocular rays reached the japā flower after being deflected by the crystal, then they would cause us to grasp that red color that inheres in the thing it qualifies (i.e., the flower). For visual cognition has never been seen to consist of color alone. Nor has mere color ever been found to be reflected without reference to its substrate.

240

nanv abhijātasyêva padma-rāgâdi-maņer japā-kusumâder api prabhā vidyate tayā vyāptatvāt sphatiko 'pi lohita ivâvabhāsate. Objection: light exists in the japā flower and so on also, just as it does in noble gems such as a ruby; the crystal also apears to be red because it is pervaded by that (light).

tathâpi svayam alohito mithyâiva lohita ity āpadyeta. Even so the fact remains that what is not red in itself (i.e., the crystal) falsely appears to be red.

*atha prabhâiva lohitâvabhāsate na sphațika iti.* Then the light alone appears red, not the crystal.

*sauklyam api tarhi sphațike prakāseta.* In that case whiteness also would appear in the crystal.

atha prabhayâpasāritam tad iti cet. Now if it is objected that it (i.e., the whiteness) is driven out by the light.

sa tarhi nīrūpaḥ katham cākṣuṣaḥ syāt. na ca rūpi-dravya-samyogād vāyor api tathātva-prasangāt. na prabhā-nimittam lauhityam tatrôtpannam uttara-kālam api tathā-rūpa-prasaṅgāt. How in that case could (the crystal) which would have no color be visible? Nor (is the crystal visible) because of a connection with another substance possessing color, since the same would apply to air also. Nor is redness produced in it by the light, since that would involve (the crystal's) retaining the same color (i.e., red) even after (the japā flower is removed).

# 2.49 Objection

The following terms are commonly used to raise an objection. Normally the objection is a  $p\bar{u}rvapaksa$ , but occasionally some of the same terms are used for a counterobjection by the *siddhāntin*. *iti* often marks the end of the objection (see Section 2.37).

#### 2.49.1 nanu and atha

The most common term for introducing an objection is the particle *nanu* "but, objection:." *nanu* stands at the beginning of the first sentence of the objection, and is especially common in objections consisting of several sentences.

*PP p.* 84 *nanu vişayinas cid-ātmanah kathaņ vişaya-bhāvah* but how can the subject whose nature is consciousness be an object?

PP p. 55 nanu na vyāpakam idam laksaņam svapna-sokâdāv asambhavāt objection: this definition does not extend far enough since it does not include dreaming, grief, and so on

The particle *atha* "now, then, but" introduces a new idea or a new topic (see Section 2.53.1) and may introduce an objection. *atha* also stands at the beginning of the sentence.

PP p. 171 atha ko'yam tarko nāma but what is this so-called reasoning?

PP p. 333 atha śabda-janyasyâiva jñānasyâbhyäso vidhīyata iti then repeated thinking of the cognition produced by śabda is enjoined

## 2.49.2 Interrogatives

Objections are often phrased as direct questions introduced by interrogatives such as *katham*, *kva*, or *kim*. Especially common is *katham punar* "but how?"

SBh 1.1.1 katham punah pratyag-ātmany avişaye 'dhyāso vişaya-tad-dharmānām but how can there be superimposition of the object and its qualities on the inner self which is not an object?

kim combined with the instrumental has the idiomatic meaning "what is the use of  $\dots$ ?" This construction may be used to find fault with an opponent's statement.

SBh 1.1.4 yadi nāmôpadistam kim tava tena syād iti even if [an ordinary thing] is taught [in the Veda]—what is the use of this to you?

2.49. Objection

#### 2.49.3 iti cet

An objection, particularly a short one, may be marked by the concluding words *iti cet*, lit., "if thus," i.e., "if this objection is raised, if you say this." Usually the objection concluded by *iti cet* is followed in the same sentence by the response to the objection.

SBh 1.1.4 aśarīratvam eva dharma-kāryam iti cen na tasya svabhāvikatvāt if it is objected that bodilessness itself is produced by dharma, then we reply no, since that (bodilessness) is natural

SBh 1.1.4 avişayatve brahmanah sāstra-yonitvânupapattir iti cen nâvidyā-kalpita-bheda-nivrtti-paratvāc chāstrasya if you say that if brahman is not an object it will be impossible for the sāstra to be its source, then we reply no, because the sāstra is concerned with discarding the distinctions fabricated by avidyā

### 2.49.4 ity āśankya

Derivatives of the root *āśank* "to suspect (wrongly)" often mark an interpretation that appears to be reasonable but is incorrect. For example: *ity āśankya* "having raised (or anticipated) this doubt," *āśankyate* "is suspected, is wrongly held to be true."

PP p. 248 tatra krama-niyamârtho 'tha-śabda ity āśańkyâha phala-jijñāsya-bhedāc ca the word atha in the passage signifies order of succession—having raised this doubt, the author of the text replies "because of the difference of fruits and objects of inquiry"

SBh 1.1.1 ata eva na brahma-śabdasya jāty-ādy-arthântaram āśańkitavyam therefore one should not suspect that the word brahman has another meaning such as class

SBh 1.1.3 janmâdi kevalam anumānam upanyastam ity āśańkyeta tām āśańkām nivartayitum idam sūtram pravavrte śāstra-yonitvād iti one might suspect that the origin and so on [of the world] were put forth [in the sūtra] merely as an inference—in order to rule out this doubt, the sūtrakāra has undertaken this sūtra "śāstra-yonitvāt"

## 2.49.5 prāpta and āksipta

The past participle  $pr\bar{a}pta$  "obtained," in the sense of "obtained at first view, concluded superficially," often marks a  $p\bar{u}rvapaksa$ . Thus  $pr\bar{a}pta$  is "obtained as the  $p\bar{u}rvapaksa$ " and *iti*  $pr\bar{a}pte$  is "this being obtained as the  $p\bar{u}rvapaksa$ ."

SBh 1.1.12 kim tāvat prāptam brahmaņo 'rthântaram amukhya ātmânandamayah syāt first, what is the pūrvapaksa? the [self] consisting of bliss must be something other than brahman, not the primary self

SBh 1.1.4 tasmān na brahmaņah śāstra-yonitvam iti prāpta ucyate therefore the śāstra is not the source of brahman—the pūrvapakṣa being thus established, the sūtrakāra replies (in the following sūtra)

The past participle *āksipta* "objected, put forth as an objection" may be used in the same way.

SBh 2.1.6 tasmād vilaksaņam evêdam brahmaņo jagad vilaksaņatvāc ca na brahma-prakrtikam ity āksipte pratividhatte therefore this world is completely different from brahman and because it is different it cannot have brahman as its material cause—this objection being raised, the sūtrakāra replies (in the following sūtra)

## 2.49.6 Verbs of speaking

The third person of verbs meaning "to speak" may introduce an objection, e.g.,  $\bar{a}ha$  "an objector says," *apara*  $\bar{a}ha$  "another says," *atra ke cid*  $\bar{a}huh$ "on this point some say." But the same expressions may also introduce another interpretation acceptable to the commentator (see Section 2.51.2), and  $\bar{a}ha$  without an expressed subject may refer to the author of the text being commented on (see Section 2.39.4). The meaning of these expressions is usually clear from the context.

PP p. 199 āha mā bhūj jñāna-dvayam an objector says: let there not be two cognitions

244

#### Chapter 19. Terms of Discussion

SBh 1.1.6 atrâha yad uktam nâcetanam pradhānam jagat-kāranam *īkṣitrtva-śravanād iti tad anyathâpy upapadyate* here an objector says: what was said before, i.e., that the unconscious *pradhāna* cannot be the cause of the world because *śruti* teaches that [the cause] is a seer—that can be explained in another way

### 2.49.7 Clauses beginning with *na ca*...

A brief objection may be expressed in an *iti* clause introduced by *na* and followed by a gerundive, e.g., *na ca* ... *iti mantavyam* "nor should it be thought that ...," *na ca* ... *iti vācyam* "not should it be said that ...," or "*na ca* ... *iti codanīyam* "nor should it be urged that ..." (see Section 2.23). The reason for the prohibition usually follows.

SDS p. 8 na ca vyāpti-jñānam sāmānya-gocaram iti mantavyam vyaktyor avinābhāvâbhāva-prasangāt nor should it be thought that knowledge of concomitance has the generic properties as its object, since that would involve the absence of concomitance of the two individuals

# 2.50 Reply

The following terms are commonly used to answer an objection. In most cases it is the *siddhāntin* who replies, but occasionally some of the same terms are used when the  $p\bar{u}rvapaksin$  responds to an objection.

### 2.50.1 Verbs of speaking

The most common term for introducing the *siddhāntin*'s reply to an objection is *ucyate*, lit., "it is said," i.e., "this is said in answer, we reply." Similar expressions include *atrâbhidhīyate* = *atrôcyate* "here we reply, on this point we reply" and *brūmaḥ* "we reply." These expressions are particularly common when the *siddhānta* is introduced after a long *pūrvapakṣa* or after a debate between two or more *pūrvapakṣins*. The reply itself may be formulated in the commentator's own words or it may contain a quote from the text being commented on. In the latter case it may be more appropriate to render *ucyate* by "the author of the text replies" (see Section 2.39.4).

#### 2.50. Reply

SBh 1.1.28 tatra prasiddher väyuh prāna iti prāpta ucyate prāna-sabdam brahma vijneyam the word prāna means air because it is well-known in that (meaning)—the pūrvapakṣa being thus established, we reply that the word prāna must be understood to denote brahman

SBh 1.1.4 atrâbhidhīyate na karma-brahma-vidyā-phalayor vailakṣaṇyāt here we reply no, because of the difference between the fruit of the knowledge of karma and the fruit of the knowledge of brahman

SBh 1.1.19 tatra katham brahmanah sva-pradhānatvam śakyam vijnātum prakrtatvād iti brūmah how can it be known that brahman is the chief thing in the passage? we reply: because it is the subject of discussion

# 2.50.2 tan na and other denials

The reply to an objection may begin with an explicit denial of it, e.g., tan na "not that," tad asat "that is not right," tad ayuktam "that is not reasonable," nâitad evam "this is not so." Or a prohibition may be placed within an iti clause, e.g., tan mā bhūd iti "lest that be the case," mā bhūd āśankêti or tan mâśankîti "lest this doubt arise." The pronoun tat in such expressions may refer to a preceding yat (see Section 2.40.3) or may simply refer to the objection in general.

SBh 1.1.31 nâitad evam upāsā-traividhyāt this is not so because of the threefoldness of meditation

GBh 2.18 anyathā pṛthivy-ādivad api nityatvam syād ātmanas tan mā bhūd iti nityasyânāšina ity āha otherwise the self's eternality would be like that of the earth and so on—lest that be the case, the author of the text uses the words "eternal" and "unperishing"

Sometimes the opponent's view is rejected in a more dramatic fashion.

SDS p. 7 tad etan mano-rājya-vijrmbhanam so this [i.e., the argument you have presented] is just building castles in the air

### 2.50.3 prasanga and āpatti

Derivatives of the roots *prasañj* "to become attached to" and *āpad* "to fall into misfortune" are often used to reject a proposition by pointing to its undesirable consequences. For example: *prasanga* and *āpatti* "undesirable implication," *prasajyate* and *āpadyate* "is implied as an undesirable consequence."

SBh 1.1.22 na câyam  $\bar{a}k\bar{a}sa-sabda$  ubhayoh s $\bar{a}dh\bar{a}ranah sakyo vij<math>\bar{n}\bar{a}tum$ anek $\hat{a}rthatva-prasang\bar{a}t$  nor can the word  $\bar{a}k\bar{a}sa$  here be understood to have both (meanings) equally, since that would involve (a single word's) having more than one meaning

SBh 1.1.4 yady ātmā sváśraya-kriyayā vikriyetânityatvam ātmanaḥ prasajyeta if the self were altered by an action inhering in it then impermanence of the self would result

SBh 1.1.1 tato jñātam evêty ajijñāsyatvam punar āpannam then [brahman] must be known and so [brahman's] not being an object of inquiry would again result

When an assertion is made, the reason for rejecting its alternative is often expressed in a phrase containing *prasariga* or *āpatti* and introduced by *anyathā* "otherwise."

SBh 1.1.19 tenânandamayasya brahmatvam anyathā prakṛta-hānâprakṛta-prakriyā-prasaṅgāt therefore the [self] consisting of bliss is brahman, since otherwise we would make the mistake of abandoning the subject of the passage and accepting what is not the subject of the passage

#### 2.50.4 doşa

The word *doṣa* "fault, defect" refers to a flaw in an argument or doctrine. The reply to an objection often begins *na doṣaḥ* or *nâiṣa doṣaḥ* or *nâyaṃ doṣaḥ* "this fault (does) not (apply)," i.e., "the fault proposed by the objector does not hold true for our doctrine." Similarly *na duṣyati* means "does not harm our position." SBh 1.1.12 yat tûktam annamayâdy-amukhyâtma-pravāha-patitatvād ānandamayasyâpy amukhyatvam iti nâsau dosa ānandamayasya sarvântaratvāt but as for what was said before, i.e., that the [self] consisting of bliss also cannot be the primary [self] because it belongs to a series of secondary selves beginning with the [self] consisting of food—this fault does not apply, because the [self] consisting of bliss is the innermost of all

SBh 1.1.4 ato 'vidyā-kalpita-samsāritva-nivartanena nitya-muktâtma-svarūpa-samarpaņān na mokṣasyânityatva-doṣaḥ therefore the fault of impermanence does not belong to mokṣa because by discarding the appearance, fabricated by avidyā, that (the self) is a transmigrating being, (scripture) teaches that the nature of the self is eternally free

### 2.50.5 syāt and bhavet

The optatives *syāt* and *bhavet* "might be" may be used to state that the objection might have been valid if the objector had not overlooked an important point, e.g., *syād etat* "this might be the case," *syād etad evam* "this might be so." The reason why the objection is not valid then follows.

SBh 1.1.4 syād etad evam yadi rajju-svarūpa-śravaņa iva sarpa-bhrāntih samsāritva-bhrāntir brahma-svarūpa-śravaņa-mātreņa nivarteta na tu nivartate this might be so, if the error of transmigratoriness were dispelled by merely hearing about the true nature of brahman, just as the error of the snake (is dispelled) when one hears about the true nature of the rope; but it is not dispelled

SDS p. 7 syād etat. syād esa mano-ratho yady anumānâdeh prāmāņyam na syāt. asti ca prāmāņyam . This might be. This would be our wish if inference and so on had no validity; but they do have validity.

Note that in some authors *syāt* may refer forward and may mean "let us suppose the following."

### 2.50.6 yady api and other concessives

If the respondent is willing to grant part or all of an objection, expressions such as *yady api*... *tathâpi*... "even if ... still ...," *kāmam* "granted," *satyam* "true," *bādham* "yes, surely," *astu* "let it be," and *sarvathâpi* "but in any case" may be employed.

SBh 1.1.4 atrâpare pratyavatisthante yady api śāstra-pramānakam brahma tathâpi pratipatti-vidhi-visayatayâiva śāstrena brahma samarpyate here others raise an objection: even if the śāstra is the means of knowing brahman, still brahman is taught by the śāstra only as an object of the injunction to know

PP p. 73 nanu na jīvo brahmaņo 'nya ity uktam. bādham ata evârthāj jīve brahma-svarūpa-prakāsâcchādikâvidyā kalpyate. Objection: it has been said that the jīva is not different from brahman. Yes—for this very reason avidyā is known by implication to be that which obscures the light of the true nature of brahman in the jīva.

SBh 1.1.24 astu tarhy atrivrt-krtam tejah prathamajam in that case let the light (in the passage) be the first-born (light) not yet made tripartite

SBh 1.1.25 asmin pakse brahmâivâbhihitam iti na chando'bhidhānam. sarvathâpy asti pūrvasmin vākye prakrtam brahma. On this view brahman only is denoted and so the meter is not denoted. But in any case brahman is the subject of the previous passage.

### 2.50.7 tarhi

In discussing the implications of a statement or objection the particle *tarhi* "in that case," i.e., "if we accept that to be true," is often used.

PP p. 128 anupakurvann api tarhi sahakāry apekṣyeta in that case the auxillary would be required even though it does not render assistance

SBh 1.1.19 annamayādīnām api tarhi brahmatva-prasangah in that case the [self] consisting of food and the other [selves] also would have to be brahman

# 2.51 Alternative interpretation

#### 2.51.1 Second interpretations

After giving one interpretation of the text being commented on or of some point raised in discussion of the text, a commentator may introduce a second interpretation with particles such as  $athav\bar{a}$  or  $yadv\bar{a}$ , both of which have the sense "or else, on the other hand." Frequently (but not always) the second interpretation is favored by the commentator. In the first example below a second interpretation of the  $s\bar{u}tra$  " $s\bar{a}stra-yonitv\bar{a}t$ " is introduced; in the second example a second interpretation is given of the word *jyotis* in the  $s\bar{u}tra$  "*jyotis* caranâbhidhānāt".

. ر:

SBh 1.1.3 athavā yathôktam rg-vedâdi śāstram yonih kāraṇam pramāṇam asya brahmano yathāvat-svarūpâdhigame or else the śāstra consisting of the Rig Veda and so on as described above is the source, i.e., the cause, i.e., the authority for knowledge of the true nature of that brahman

SBh 1.1.24 yadvā nâyam jyotih-sabdas cāksur-vrtter evânugrāhake tejasi vartate 'nyatrâpi prayoga-darsanāt or else the word jyotis here does not denote the light that aids the function of the eye, since we see that it is used in other meanings also

### 2.51.2 Citing others

Expressions such as *apara āha* "another says" and *atra ke cid āhu*<sup>h</sup> "on this point some say" often introduce an objection (see Section 2.49.6). The same expressions introduce an alternate interpretation (usually not the favored one) when the interpretation is allowed to stand unrefuted by the commentator.

# Chapter 20

# **Particles and Pronouns**

# 2.52 Introduction

In Sanskrit prose, particles play an important role in connecting sentences and in altering the meaning of a sentence. This section discusses some of the particles used for these purposes, and also the use of pronomial adverbs as pronouns, some peculiarities in the use of correlative pronouns and adverbs, and uses of the four common demonstrative pronouns. (See also: Coulson 61–63, 76, 171–77, 191–95, 207–14; Speijer 261–90, 394–490.)

9

# 2.53 Particles connecting sentences

The following particles often connect a sentence (or clause) to what has been said before, especially when they appear at the beginning of the sentence (or clause). Frequently this connection is crucial to the meaning of the passage. Note that particles such as kim ca, lit., "and what?" are properly rhetorical questions but are best rendered in English by less literal translations. See Section 2.30 for the use of the causal particle hi.

### 2.53.1 atha

30 20

atha "now, then, but" introduces a new topic or a new idea.

SBh 4.1.1 trtīve 'dhyāye parâparāsu vidyāsu sādhanâśrayo vicārah prāyeņâtyāgāt. athêha caturthe phalâśraya āgamişyati . In the third adhyāya discussion was centered primarily on the means (of knowledge) with regard to the higher and lower knowledges. Now in this fourth  $(adhy\bar{a}ya$  there will be discussion) of the fruits (of knowledge).

Compare the common use of *atha* to begin a chapter or an entire work. *atha dvitīyo 'dhyāyaḥ* now begins the second *adhyāya* Often the new idea introduced by *atha* is an objection (see Section 2.49.1). In conditional sentences *atha* may be used with *yadi* to propose an alternative, i.e., *yadi* ..., *atha* ... "if ... but if ... ."

PP p. 123 yadi na [artha-kriyām] kuryād asal-lakṣaṇa-prāpter na paramârtha-vastv atha kuryān na tarhi sthāyī if it does not perform [useful activity] then it is not a real thing since it fulfills the requirements of a non-existent thing; but if it does perform [useful activity] then it is not permanent

Occasionally atha is used in place of yadi or cet to state a condition.

SBh 1.1.4 atha prakaraṇântara-bhayān nâitad abyupagamyate tathâpi sva-vākya-gatôpāsanâdi-karma-paratvam if this is not accepted for fear of introducing a new topic, then even so (it must be admitted that the Vedānta texts) are concerned with actions such as meditation contained in those very texts.

#### 2.53.2 tatra

*tatra* often refers to a recently mentioned quotation or fact in the sense "in that passage" or "in that case" or "with regard to that" and so on. When *tatra* refers to a recently mentioned substantive it is the equivalent of the locative of the pronoun *tat* (see Section 2.55).

SBh 1.1.1 tatrâtha-śabda ānantaryârthaḥ parigrhyate the word atha in the passage is used to signify immediate succession

SBh 1.1.23 vāyu-vikārasya pañca-vṛtteḥ prāṇasyôpādānaṃ yuktam. tatra hi prasiddhataraḥ prāṇa-sabdaḥ . It is reasonable that the reference is to the five-fold prāṇa that is a modification of air, since the word prāna is better known in that sense. When a number of items have been mentioned in a previous sentence, *tatra* may begin the next sentence with the meaning "of these things just mentioned." In this case *tatra* is the equivalent of *teşu* and is regularly glossed as *teşu madhye* "among these."

PP p. 257 bahavaś ca sambandha-viśesah. tatrâvaśyam anyatamah pratipattavyo 'nyathā vyavahārânupapatteh . And there are many kinds of relationship. Of these one must necessarily be accepted because otherwise discourse would be impossible.

PP p. 90 tatra na tāvat prathamah kalpo jada-svarūpe pramāņa-phale višvasyânavabhāsa-prasangāt of these, first of all, not the first alternative, because if the fruit of pramāņa were insentient then nothing at all would appear

#### 2.53.3 tāvat

 $t\bar{a}vat$  may be used adverbially in the meaning "first, first of all." In this sense it usually introduces the first of a series of points or alternatives and may be combined with *tatra* (see the last example above). Generally  $t\bar{a}vat$  is placed after the first word of the sentence.

SBh 1.1.22 kutah samsayah. ubhayatra prayoga-darsanāt. bhūta-visese tāvat suprasiddho loka-vedayor ākāsa-sabdo brahmaņy api kva cit prayujyamāno dršyate. Why is there a doubt? Because we see its use in both meanings. First, the word ākāsa is well known in both worldly discourse and in the Veda in the sense of a particular element; but in some cases we see that it is used to denote brahman also.

SBh 1.2.1 na tāvad anīyastvam jyāyastvam côbhayam ekasmin samāśrayitum śakyam virodhāt first of all, smallness and greatness cannot both reside in one thing, because of the contradiction

### 2.53.4 atra and iha

*atra* and *iha* often refer to a recently mentioned quotation or fact in the sense "in this passage" or "in this case" or "with regard to this" and so on. If the reference is to a recently mentioned substantive, *atra* and *iha* are equivalent to the locative of the pronoun *idam* (see Section 2.55).

SBh 1.1.22 kim punar atra yuktam bhūtâkāsam iti but what is right in this case? [the word  $\bar{a}k\bar{a}sa$  refers to] the elemental ether

SBh 1.1.23 ata iha prāṇa-śabdena katarasyôpādānam yuktam iti bhavati samśayah therefore there is a doubt: which of the two (meanings) is signified by the word prāṇa in this passage?

Similarly ity atra may be "in this passage (quoted by iti)."

SBh 1.1.4 yathā ca puruso vāva gautamâgnir yosā vāva gautamâgnir ity atra yosit-purusayor agni-buddhir manasi bhavati and just as in the passage "man indeed, O Gautama, is a fire; woman indeed, O Gautama, is a fire" there is a mental meditation upon man and woman as fire

The expressions *ihâpi* "here too, in this case also" and *iha tu* "but here, but in this case" show that the situation just described is similar to or different from the subject of discussion.

PP p. 249 *iha tu nitya-nirvṛttam puruṣa-vyāpārânapekṣam brahma jijñāsyam* but in this case the object of inquiry is *brahman*, eternally accomplished and independent of man's action

*iha* may also signify "in this world" (= *iha loke*).

### 2.53.5 kim ca, etc.

kim ca "moreover," api ca "moreover," anyac ca "and another thing," aparam ca "and a further thing," tathā ca "and similarly," and the like all serve to add something to what has been said before.

PP p. 49 kim ca rajata-rūpeņa cet pariņatā šuktih ksīram iva dadhi-rūpeņa tadā dosāpagame 'pi tathâivâvatistheta moreover, if shell were transformed into silver as milk is transformed into curds then it would remain just so (i.e., as silver) even when the defect is removed

SBh 1.1.4 api ca brāhmaņo na hantavya iti câivam-ādyā nivŗttir upadiśyate moreover, abstinence from action is taught in passages such as "a Brahmin is not to be killed"

#### 2.53.6 tu, etc.

tu, kim tu, param tu, punar, and the like share the meaning "but" and often introduce a slightly different idea or (more strongly) a modification or correction of what has been said before. A more emphatic antithesis is introduced by kim tarhi "but rather, instead."

SBh 1.1.4 na tu tathā brahmaņa upāsanā-vidhi-sesatvam sambhavati but brahman cannot thus be subordinate to injunctions to meditate

SBh 1.1.2 tasmāj janmâdi-sūtram nânumānôpanyāsârtham. kim tarhi vedānta-vākya-pradaršanârtham . Therefore the sūtra beginning with the words janmâdy does not serve to put forth an inference, but rather to point out a Vedānta text.

### 2.53.7 tathā hi and tad yathā

Clarification or illustration of what has been said before is introduced by  $tath\bar{a}$  hi (lit., "for thus") and tad yath $\bar{a}$  (lit., "that is like this"). Both have the sense "that is to say" or "for instance."

SBh 1.1.1 prāk ca tathā-bhūtâtma-vijñānāt pravartamānam śāstram avidyāvad-viṣayatvam nâtivartate. tathā hi brāhmaņo yajetêtyādīni śāstrāņy ātmani varņâśrama-vayo'vasthâdi-viśeṣâdhyāsam āśritya pravartante. And the śāstra, operating before such knowledge of the self has arisen, does not go beyond having as its object those affected by avidyā. That is to say, passages of the śāstra such as "a Brahmin should sacrifice" operate on the assumption that there is superimposition on the self of particulars such as class, stage of life, age, state, and so on.

SBh 1.1.1 adhyāso nāmâtasmims tad-buddhir ity avocāma. tad yathā putra-bhāryâdişu vikalesu sakalesu vâham eva vikalah sakalo vêti bāhya-dharmān ātmany adhyasyati . We have said that superimposition is the notion of one thing in another thing. For instance, a man superimposes external qualities on his self if he thinks "I am ill" or "I am well" when his son, wife, and so on are ill or well.

2.54. Particles of emphasis

### 2.53.8 Words for "therefore"

A conclusion may be introduced by a pronoun or adverb such as *tasmāt*, *tena*, *tat*, *tataḥ*, *ataḥ*, or *itaḥ*. The pronoun or adverb functions as a conjunction and has the meaning "because of the preceding, for this reason, therefore."

SBh 1.1.1 tasmād avidyāvad-visayāny eva pratyaksâdīni pramānāni sāstrāni ca therefore the pramānas such as perception and the sāstras have as their object those affected by avidyā

PP p. 140  $b\bar{a}dham ukta-laksanavidya pratyag-drśy api sambhaven$ nâitāvatā tat-sambhavah sidhyate. tena nidarśanīyah sah. Granted thatavidyā as defined above is possible even in the case of the inner self; still,this much does not establish the actual existence of it (i.e., avidyā).Therefore that is to be demonstrated.

SDS p. 3 tac caitanya-visista-deha evâtmā therefore the self is just the body qualified by consciousness

The force of the pronoun or adverb may be strengthened by addition of the particle *eva* (see Section 2.54.1); e.g., *tasmād eva* or *ata eva* "for this very reason." The reason itself may be expressed in a clause introduced by a relative pronoun or adverb; see Section 2.28 for such constructions.

# 2.54 Particles of emphasis

#### 2.54.1 eva

eva lays stress on the immediately preceding word and may be rendered as "just, only, merely." It conveys the sense "just this—no more, no less."

SBh 1.1.2 na vastu-yathâtmya-jñānam purusa-buddhy-apekşam. kim tarhi vastu-tantram eva tat. Knowledge of the true nature of a thing does not depend upon the notions of man; instead, it depends only on the thing itself.

SDS p. 3 anganâdy-ālinganâdi-janyam sukham eva purusârthah the aim of man is just the pleasure that arises from such things as embracing

women This meaning is strengthened in the expression *eva kevalam* "just this and no more."

SBh 1.1.1 brahma-codanā tu puruṣam avabodhayaty eva kevalam but the scriptural imperatives concerning brahman only instruct a man and no more

When combined with the pronoun *tat*, *eva* means "the very one, the same one."

SBh 1.1.19 tathā sati tad eva brahmânandamaya ātmâvayavī tad eva ca brahma puccham pratisthâvayavah this being the case, that very brahman would be the self consisting of bliss, the whole, and that same brahman would be the tail, the foundation, a part

PP p. 107 atrâha bhavatu na vastv-antaram. tad eva tad iti tu na kṣamyate. Here the objector says: we grant that [the image in a mirror] is not a different thing [from the object reflected]; but we cannot accept that it is the same thing (i.e., that the two are identical).

# 2.54.2 api

*api* has two common meanings when modifying the immediately preceding word: "also, too" and "even, though." The intended meaning is usually clear from the context.

SBh 1.1.5 nan-ktam sarva-jñāna-śaktimattvena sarvajñām bhavişyatîti. tad api nôpapadyate . Objection: it was said that [the pradhāna] might be omniscient because it possesses the capacity for all knowledge. That also is not reasonable.

SBh 1.1.1 nanu sesa-sasthi-parigrahe 'pi brahmano jijfiāsa-karmatvam na virudhyate objection: even if the genitive of general relation is used in the text, brahman's being the object of the desire to know is not contradicted

With regard to the first meaning, note the use of api (either alone or in combinations such as api ca) at the beginning of a sentence to connect the sentence to what has been said before (see Section 2.53.5). With regard to

#### 2.56. Correlative pronouns and adverbs

#### Chapter 20. Particles and Pronouns

the second meaning, note the use of api with the locative absolute to mean "even though" (see Section 2.29.3) and the common expressions yady api "even if" and *evam api = tathâpi* "even so, still, nevertheless" (see Section 2.50.6).

# 2.55 Pronomial adverbs used as pronouns

Pronomial adverbs formed with the suffixes *tas* and *tra* (also the adverbs *kva* and *iha*) may refer to persons and things as well as places and are sometimes equivalent in both construction and meaning to the corresponding pronouns of suitable gender and number in the ablative or locative case.

### 2.55.1 Adverbs corresponding to pronouns

The following adverbs may be used in place of the corresponding pronouns:

kutaḥ	= ablative of kim
kutra	= locative of kim
yataḥ	= ablative of yat
yatra	= locative of yat
tatah	= ablative of <i>tat</i>
tatra	= locative of <i>tat</i>
atah	= ablative of <i>idam</i>
atra	= locative of <i>idam</i>
iha	= locative of <i>idam</i>
kva	= locative of kim

Note that the relative adverbs *yatah* and *yatra* may be correlated with the pronoun *tat* and the demonstrative adverbs *tatah* and *tatra* may be correlated with the pronoun *yat*.

SBh 1.1.1 yatra yad-adhyāsas tad-vivekâgraha-nibandhano bhramaḥ the superimposition of one thing on another is an error based on the failure to grasp the difference of the two SDS p. 31 yad vedyate yena vedanena tat tato na bhidyate whatever is made known by a cognition, that is not different from that (cognition)

PP p. 211 adhīta-vedasya yo 'rtho 'vagatas tatrâiva vipratipattayah santi there are conflicting views with regard to that meaning learned by one who has studied the Veda

In the first example above yatra = yasmin and both yatra and yad correspond to tat (= tayoh); in the second example tatah = tasmat and corresponds to yena; in the third example tatra = tasmin and corresponds to yo '*rthah*.

### 2.55.2 Adverbs as attributives

The adverbs mentioned above may act as attributives of singular substantives in the ablative or locative case.

tato deśāt = tasmād deśāt	from that place
kutaś cit kāraņāt = kasmāc cit kāraņāt	from some cause
atra ghațe = asmin ghațe	in this pot
iha loke = asmin loke	in this world

# 2.56 Correlative pronouns and adverbs

#### 2.56.1 Repeated correlative

The same relative pronoun or adverb may be repeated to give a general sense, e.g., *yo yah* "whoever" and *yatra yatra* "wherever." In this case the correlative pronoun or adverb is also repeated.

SBh 1.1.12 yatra yatra vibhūty-ādy-atiśayah sa sa iśvara ity upāsyatayā codyate in whomever there is pre-eminence of power and so on, he is enjoined as the one to be worshipped as the Lord

SBh 1.1.24 tasmād yad yat kasya cid avabhāsakam tat taj jyotih-sabdenâbhidhīyate therefore whatever illuminates something is denoted by the word jyotis Chapter 20. Particles and Pronouns

### 2.56.2 Multiple pairs

Sanskrit often employs more than one pair of correlatives in the same sentence. The correlatives (or all but one) usually have an indefinite sense. There is no requirement that the demonstratives appear in the same order as the relatives, and either may occur in compound. Such sentences are often difficult to translate into English since English rarely uses more than one pair of correlatives in the same sentence. It may be helpful, especially upon first reading, to represent each pair of correlatives symbolically by letters such as "A," "B," and so on.

SBh 1.1.1 yatra yad-adhyāsa tat-krtena doseņa guņena vâņu-mātreņâpi sa na sambadhyate when there is superimposition of A on B, B is not connected even in the slightest degree with any fault or virtue created by A; or, when there is superimposition of one thing on another, the latter is not connected even in the slightest degree with any fault or virtue created by the former

SDS p. 25 yad yadā yat karoti tat tadā tatra samartham when A causes B at time C, A is samartha of B at time C; or, whenever one thing causes another, the former is samartha of the latter at that time

In the first example above the correlatives are yatra = sa (B) and yad = tat (A); in the second example the correlatives are yad = tat (A),  $yad\bar{a} = tad\bar{a}$  (C), yat = tatra (B).

### 2.56.3 yathā with tathā

A clause introduced by *yathā* may express the object of a verb meaning "to say," "to show," and so on. The correlative *tathā* is normally used if the *yathā* clause precedes the main verb. In this construction *yathā* means literally "in what way" and can often be translated by "how."

PP p. 137 yathâkāśasyâkṣa-vyāpāram antarâpy aparokṣatā tathā darśayiṣyāmaḥ we will show how the ether is directly known even without the functioning of the eye

SBh 1.1.5 yathā tu tarkenâpi brahmaņa eva kāraņatvam nirvodhum śakyate na pradhānâdīnām tathā prapañcayişyati na vilakṣanatvād asyêtyevamādinā but the sūtrakāra will explain in sūtras such as na vilakṣaṇatvād asya how by reasoning also the causality of brahman alone can be established, and not that of the pradhāna and so on

Note that in both examples a sentence of equivalent meaning can be obtained by omitting  $yath\bar{a}$  and replacing  $tath\bar{a}$  with *iti*.

### 2.56.4 yat meaning "that"

The pronoun *yat* is sometimes used as a conjunction to introduce a subordinate clause. In this case *yat* means "which is the fact that" or simply "that." A correlative to *yat* is sometimes used, and *iti* is sometimes added redundantly to the end of the *yat* clause.

SBh 1.1.4 alankāro hy ayam asmākam yad brahmâtmâvagatau satyām sarva-kartavyatā-hāniḥ kṛta-kṛtyatā cêti for this is our glory, that when there is knowledge of the self as brahman all duties cease and everything is accomplished

PP p. 317 ato 'lankāra eva sūtrānām yad anekârthatā nāma therefore it is indeed an embellishment of sūtras that they have more than one meaning

In the first example the yat clause refers back to the correlative pronoun ayam and is concluded by *iti*. In the second example the yat clause refers to the preceding word alarikāra. Sentences of equivalent meaning can be obtained by converting the yat clauses to *iti* clauses (i.e., by omitting the yat in both examples and adding *iti* to the end of the second), and in fact such subordinate clauses are far more often formed with *iti* than with yat (see Chapter 16). Note that when such a clause is formed with *iti* it tends to precede the word or words to which it refers, but when it is formed with yat it tends to follow.

### 2.56.5 Repeated anya

The notion "the one ... the other ..." or "some ... others ..." may be expressed by repetition of the pronoun *anya*.

#### Chapter 20. Particles and Pronouns

SBh 1.1.23 yathā pituh pitêti prayoge 'nyah pitā şaşthi-nirdişto 'nyah prathama-nirdiştah just as in the expression "the father of the father" one father is indicated by the genitive case and the other is indicated by the nominative case

Similar expressions include ke cit (or eke) ... apare (or anye) ..." some ... others ... ."

# 2.57 Demonstrative pronouns

# 2.57.1 Scale of proximity

The uses of the pronouns *idam*, *etat*, *adah*, and *tat* are distinguished by the Indian grammarians in the following *kārikā*:

idamas tu sannikṛṣṭaṃ samīpatara-varti câitado rūpam adasas tu viprakṛṣṭaṃ tad iti parokṣe vijānīyāt

One should know that *idam* refers to what is near, *etat* to what is closer at hand, *adah* to what is remote, and *tat* to what is not visible to the speaker.

The words idam, etat, and adah are normally deictic pronouns, i.e., they point out a person or thing and distinguish it from others of the same class. Often they point to something in the speaker's presence. In this case idam and etat designate something near the speaker and usually correspond to the English demonstrative "this." Their areas of meaning tend to overlap, though etat is more emphatic and is often translated by "here" (e.g., esa devadattah "here is Devadatta"). adah designates something remote from the speaker and possibly not visible to him, and usually corresponds to the English demonstrative "that." These three pronouns may also point to something just mentioned or about to be mentioned; in particular etat often points to what has just been said and idam to what is about to be said (see Section 2.57.2 tat is properly an anaphoric pronoun, i.e., it refers back to a preceding word in the context. It is therefore the usual correlative of the relative pronoun yat when the relative clause precedes the main clause (when the relative clause follows the main clause the correlative is often idam or etat or is omitted altogether). When tat is used in apposition with a substantive it is usually translated by the definite article "the" (e.g., sa brāhmaņaḥ "the Brahmin"); when it appears alone it is usually translated by the appropriate personal pronoun (e.g., saḥ "he" or "it"). In cases other than the nominative and accusative, *idam* may be used anaphorically as a synonym of *tat* (e.g., *asya* = *tasya* "his" or "its"). *idam* is enclitic when used in this way.

### 2.57.2 Reference to earlier and later passages

In conversation and discussion *etat* usually refers back to what has already been said while *idam* usually refers ahead to what is about to be said.

GBh 4.20 tadīyam karmâkarmâiva sampadyata ity etam artham darśayisyan āha being about to reveal this meaning, i.e., that the action of such a man becomes non-action, the author of the text says (the following verse)

GBh 8.19 samsāre vairāgya-pradarśanârtham cêdam āha and in order to demonstrate dispassion toward samsāra the author of the text says this (i.e., the following verse)

In the first example *etam artham* refers back to the preceding *iti* clause; in the second example *idam* refers ahead to the verse of the text about to be quoted.

#### 2.57.3 tad etat and tad idam

*tat* is sometimes used in apposition with *etat* or *idam* to make an emphatic reference to something recently discussed; e.g., *tad etat* or *tad idam* "this one here (*etat* or *idam*) that we have been talking about (*tat*)."

SBh 1.1.1 tam etam evam-lakṣaṇam adhyāsaṃ paṇḍitā avidyêti manyate this superimposition as defined above scholars regard as avidyā

SDS p. 6 tad etat sarvam samagrāhi all of this is summarized (in the following verses)

GBh 1.1 tad idam gītā-śāstram

samasta-vedârtha-sāra-sangraha-bhūtam durvijneyârtham this

#### Chapter 20. Particles and Pronouns

 $g\bar{t}d-s\bar{a}stra$  which is a summary of the essence of the meaning of the entire Veda and whose meaning is hard to grasp

When *tat* is used in the predicate and the subject is *etat* or *idam*, the sense is "this one (*etat* or *idam*) is the one that we have been talking about (*tat*)" or "this is the same one."

so 'yam brāhmanah this is the Brahmin; or, this is the same Brahmin

Note that the pronoun subject (*ayam*) may be placed in the middle of the predicate (*sa brāhmaņaḥ*); see Section 2.21.2.

# Appendix

# **Useful Reference Works**

The following reference works may be useful in reading Sanskrit commentaries and other works composed in the bhāsya style.

9

 Abhyankar, Kashinath Vasudev and J. M. Shukla. A Dictionary of Sanskrit Grammar. 2nd rev. ed. Gaekwad's Oriental Series 134. Baroda: Oriental Institute, 1977.

Dictionary of Sanskrit grammatical terms, including both technical terms of Pāṇinian grammar and more general grammatical terms.

Apte, Vaman Shivaram. The Student's Guide to Sanskrit Composition. 24th ed. Pune: Anath Vidyarthi Griha Prakashan, 1960.
 Halpful observations on suprov the uses of particular and more set.

Helpful observations on syntax, the uses of particles, and many specific points of grammar and composition.

3. Böhtlingk, Otto. *Pāņini's Grammatik.* 2nd ed. 2 vols. 1887. Kyoto: Rinsen Book Company, 1977.

Translation of Pānini's grammar; the second volume includes several indices that are very useful for looking up technical terms found in commentaries.

4. Cardona, George. *Pānini: A Survey of Research*. New Edition. Delhi: Motilal Banarsidass, 1998.

A critical survey of research on Pānini and Pāninian grammar, which provides not only a detailed bibliography but a reasoned treatment of work to date on every relevant topic.

- Appendix
- Cardona, George. *Pānini: His Work and Its Traditions*. Volume 1: Background and Introduction. Second revised and enlarged edition. Delhi: Motilal Banarsidass, 1997.

A very thorough and careful presentation of Pānini's approach and techniques, providing a rich and invaluable description of the entire system.

6. Coulson, Michael. Sanskrit: An Introduction to the Classical Language. Oxford: Teach Yourself Books-Hodder and Stoughton, 1976.

Sanskrit primer that provides additional discussion and examples for some of the material covered in our Part Three, along with many helpful observations on other grammatical points (see the table of contents). A brief but excellent introduction to commentarial techniques may be found in the discussion of Mallinātha's commentary on *Kumārasambhava* 6.84–95 on pp. 258–267.

 Devasthali, G. V. Anubandhas of Pāņini. Publications of the Centre of Advanced Study in Sanskrit, Class B, No. 2. Pune: University of Poona, 1967.

Explanations of the meanings of the indicatory letters used in the technical terms of Pāṇinian grammar, together with indices of grammatical elements both with and without the indicatory letters.

8. Hartmann, Peter. Nominale Ausdrucksformen im Wissenschaftlichen Sanskrit. Heidelberg: Carl Winter Universitätsverlag, 1955.

Detailed and scholarly treatment of nominal constructions in the bhāşya style of Sanskrit prose.

 Jacobi, Hermann. "Über den nominalen Stil des wissenschaftlichen Sanskrits." Indogermanische Forschungen 14 (1903): 236–251. Rpt. in: Hermann Jacobi. Kleine Schriften. Wiesbaden: Franz Steiner Verlag GmbH, 1970. 1: 6–21.

Good general introduction to the bhāṣya style of Sanskrit prose; includes an exemplary passage, with translation, from the *Vedānta*-paribhāṣā.

 Jhalakīkar, Mahāmahopādhyāya Bhīmācārya. Nyāyakośa or Dictionary of Technical Terms of Indian Philosophy. Rev. by Mahāmahopādhyāya Vāsudev Shāstrī Abhyankar. 4th ed. Bombay Sanskrit and Prakrit Series 49. Pune: Bhandarkar Oriental Research Institute, 1978.

Dictionary of technical terms of Indian philosophy, with special emphasis on Nyāya terminology. A difficult work, but useful for students who have some experience in reading Sanskrit philosophical texts. In Sanskrit.

11. Kale, Moreshwar Ramchandra. A Higher Sanskrit Grammar. Rpt. Delhi: Motilal Banarsidass, 1972.

The only grammar of Sanskrit in English that gives a full presentation of the  $P\bar{a}n$ inian system of grammar. Unfortunately it has no index.

12. Katre, Sumitra M. Astādhyāyī of Pāņini. Texas Linguistics Series. Austin: University of Texas Press, 1987.

A complete English translation of Pānini's grammar, with useful indices. The text is presented not in its traditional form but in markedup Roman transliteration with sandhi dissolved and hyphens and accents added.

13. Renou, Louis. *Histoire de la langue Sanskrite*. Les Langues du Monde 10. Lyon: Editions IAC, 1956.

A short history of Sanskrit commentaries and a good general introduction to the bhāṣya style of Sanskrit prose may be found on pp. 133–146.

14. Speijer, J. S. Sanskrit Syntax. 1886. Delhi: Motilal Banarsidass, 1973.

Comprehensive treatment of Sanskrit syntax that provides additional discussion and examples for some of the material covered in our Part Three (see the table of contents and the Sanskrit index). Speijer often refers to Pāņini on specific grammatical points.

- Appendix
- 15. Whitney, William Dwight. Sanskrit Grammar. 2nd ed. 1889. Cambridge, MA: Harvard University Press, 1971.

Comprehensive treatment of Sanskrit and Vedic morphology from a Western viewpoint; very little discussion of Pāninian grammar or of Sanskrit syntax.

# **Commentarial Passages Quoted**

and the state of the

GBh 1.1 188, 263
GBh 2.10 189, 208
GBh 2.14 209
GBh 2.16 190
GBh 2.18 245
GBh 2.20
GBh 2.21 228
GBh 2.23 225
GBh 2.48 225
GBh 4.20 263
GBh 8.19 263
Kāśikā 1.2.9164
Kāśikā 1.2.26164
Kāśikā 3.1.7 44
Kāśikā 3.1.8 46
Kāśikā 3.1.1047
Kāśikā 3.1.1147
Kāśikā 3.1.1247
Kāśikā 3.1.15
Kāśikā 3.1.17
Kāśikā 3.1.1848
Kāśikā 3.1.1948
Kāśikā 3.1.2048
Kāśikā 3.1.21
Kāśikā 3.1.2245
Kāśikā 3.1.2345
Kāśikā 3.1.2445
Kāśikā 3.1.25
Kāśikā 3.1.2646
Kāśikā 3.2.8364
Kāśikā 3.3.1151
Kāśikā 3.3.1251
Kāśikā 3.3.1351

Kāśikā 5.1.567
Kāśikā 5.1.1067
Kāśikā 8.1.54164
Kir.M. 13.6971
Kum.C. 1.10 142
Kum.C. 1.46 116
Kum.C. 7.5352
Kum.M. 1.10
Kum.M. 1.23 58
Kum.M. 1.5061
Kum.M. 1.1114, 129
Kum.M. 1.2 32, 103, 146
Kum.M. 1.4 15, 80, 121
Kum.M. 1.5
Kum.M. 1.6 129
Kum.M. 1.7 55
Kum.M. 1.10 142
Kum.M. 1.11 . 23, 126, 133, 141
Kum.M. 1.13 107
Kum.M. 1.1524, 27
Kum.M. 1.16 109, 139
Kum.M. 1.18
Kum.M. 1.20 59
Kum.M. 1.23 45
Kum.M. 1.24
Kum.M. 1.25 129
Kum.M. 1.26
Kum.M. 1.2781, 145
Kum.M. 1.28 122
Kum M. 1.32 167
Kum.M. 1.35 16
Kum.M. 1.39 25
Kum.M. 1.41
Kum.M. 1.43 23

# Commentarial Passages Quoted

Kum.M. 1.44 100
Kum.M. 1.46 23, 81, 116
Kum.M. 1.47
Kum.M. 1.48
Kum.M. 1.51
Kum.M. 1.53 129
Kum.M. 1.55 80
Kum.M. 1.58
Kum.M. 1.60 95
Kum.M. 2.1
Kum.M. 2.2
Kum.M. 2.4 54
Kum.M. 2.7 44
Kum.M. 2.10 160
Kum.M. 2.1175, 164
Kum.M. 2.17 110
Kum.M. 2.20
Kum.M. 2.21 31
Kum.M. 2.22 129
Kum.M. 2.23
Kum.M. 2.25 100
Kum.M. 2.27 27
Kum.M. 2.28 117
Kum.M. 2.29
Kum.M. 2.30 99
Kum.M. 2.32 103
Kum.M. 2.35
Kum.M. 2.41 125
Kum.M. 2.4599, 100
Kum.M. 2.46 56
Kum.M. 2.47 30, 107, 136
Kum.M. 2.5161, 167
Kum.M. 2.56 163
Kum.M. 2.57
Kum.M. 2.59 31
Kum.M. 2.64 30
Kum.M. 3.1 135

Kum.M. 3.5
Kum.M. 3.10 133
Kum.M. 3.14 44
Kum.M. 3.19 166
Kum.M. 3.2373
Kum.M. 3.26 31
Kum.M. 3.27 105
Kum.M. 3.29 107
Kum.M. 3.30 108
Kum.M. 3.36 104, 105
Kum.M. 3.37 61
Kum.M. 3.39 108
Kum.M. 3.40 54
Kum.M. 3.41
Kum.M. 3.44 17
Kum.M. 3.50 118
Kum.M. 3.53 105
Kum.M. 3.54 134
Kum.M. 3.57 66
Kum.M. 3.59 31
Kum.M. 3.64 71
Kum.M. 3.65 80
Kum.M. 3.67 24
Kum.M. 3.6824, 83
Kum.M. 3.72 136
Kum.M. 3.74 166
Kum.M. 3.75 131
Kum.M. 3.76 79
Kum.M. 4.3 134
Kum.M. 4.9
Kum.M. 4.1024, 100
Kum.M. 4.13 83
Kum.M. 4.14 104
Kum.M. 4.17 65
Kum.M. 4.18
Kum.M. 4.21 89
Kum.M. 4.26 51

Kum.M. 4.27
Kum.M. 4.42–4.43 162
Kum.M. 4.45 114
Kum.M. 5.1
Kum.M. 5.36 167
Kum.M. 5.55 14
Kum.M. 5.73 25
Kum.M. 5.86
Kum.M. 6.93 112
Kum.V. 1.1
Kum.V. 1.4
Kum.V. 1.1062, 118, 142
Kum.V. 1.14124
Kum.V. 1.15 26, 53
Kum.V. 1.1962
Kum.V. 1.25
Kum.V. 1.2727
Kum.V. 1.27 [M 28] 98
Kum.V. 1.3727
Kum.V. 1.37 [M38] 99
Kum.V. 1.45 [M 46] 116
Kum.V. 1.56
Kum.V. 1.56 [M 57] 98, 124
Kum.V. 1.5815
Kum.V. 1.59
Kum.V. 1.60
Kum.V. 2.1
Kum.V. 2.3
Kum.V. 2.5
Kum.V. 2.7
Kum.V. 2.9160
Kum.V. 2.17 78, 111
Kum.V. 2.18
Kum.V. 2.26
Kum.V. 2.28
Kum.V. 2.30111
Kum.V. 2.33

Kum.V. 2.42
Kum.V. 2.44
Kum.V. 2.47
Kum.V. 2.51
Kum.V. 2.57
Kum.V. 3.8 100
Kum.V. 3.10
Kum.V. 3.18
Kum.V. 3.27 104
Kum.V. 3.44
Kum.V. 3.46 100
Kum.V. 3.52
Kum.V. 3.56101
Kum.V. 3.63
$Kum.V.\ 3.69\ldots\ldots .80$
Kum.V. 4.1
Kum.V. 4.9
Kum.V. 4.13
Kum.V. 4.35 82
Kum.V. 4.36
Kum.V. 5.10
Kum.V. 5.14
Kum.V. 5.22
Kum.V. 5.29 [M 30]71
Kum.V. 5.44
Kum.V. 5.63 14
Kum.V. 6.3 62
Kum.V. 6.43 55, 82
Kum.V. 6.87 [6.88] 54
Kum.V. 6.92
Kum.V. 7.53 52
Kum.V. 7.57
Kum V. 1.49 [1.50]
Megh.M. 1
Megh.M. 4
PP p. 49254

## Commentarial Passages Quoted

PP p. 55......241 R PP p. 73......248 PP p. 84.....241 PP p. 90......253 R PP p. 107 ..... 257 R PP p. 123 ..... 252 R PP p. 128 ..... 248 PP p. 135 ..... 225 R PP p. 137 ..... 227, 260 РР р. 140 ..... 256 R PP p. 171 ..... 241 R РР р. 199 ..... 243 R PP p. 211 ...... 259 R PP p. 248 ..... 242 R PP p. 249 ..... 254 РР р. 257 ..... 253 PP p. 317 ..... 261 R РР р. 333 ..... 241 R PP pp. 100–101......239 R **PP** p. 79.....196 R PP p. 163 ..... 217 PP p. 183 ..... 198 PP p. 258 ..... 209 PP p. 356 ..... 199 Ragh.H. 2.3 ..... 58 Ragh.H. 3.55 [M 54] ..... 101 Ragh.H. 5.58 [M 5.57].....78 Ragh.M. 2.10 ..... 40, 42 Ragh.M. 4.72.....61 R Ragh.M. 5.72.....61 R Ragh.M. 1.1 15, 20, 74, 125, 145 R Ragh.M. 1.2 . 15, 20, 44, 55, 113 R Ragh.M. 1.3.....63, 65

Ragh.M. 1.416, 20, 29, 140
Ragh.M. 1.5 93, 160
Ragh.M. 1.5–1.9162
Ragh.M. 1.6
Ragh.M. 1.723, 28, 29, 44
Ragh.M. 1.810, 26, 75, 133
Ragh.M. 1.9
Ragh.M. 1.11122
Ragh.M. 1.12 80, 82, 99, 107
Ragh.M. 1.13 77, 84, 106, 133
Ragh.M. 1.14 22, 161
Ragh.M. 1.15129
Ragh.M. 1.16 22, 103
Ragh.M. 1.17 14, 21, 124
Ragh.M. 1.1832, 91, 129, 161,
167
Ragh.M. 1.2021, 54, 72
Ragh.M. 1.21
Ragh.M. 1.23 62, 136
Ragh.M. 1.24 . 11, 16, 21, 22, 99
Ragh.M. 1.25 22, 25, 70
Ragh.M. 1.2621
Ragh.M. 1.2720
Ragh.M. 1.28
Ragh.M. 1.2920
Ragh.M. 1.30 83, 130
Ragh.M. 1.3163
Ragh.M. 1.32 165, 166
Ragh.M. 1.3546, 123, 163
Ragh.M. 1.3620, 73, 160
Ragh.M. 1.38 53
Ragh.M. 1.40 145, 167
Ragh.M. 1.4363
Ragh.M. 1.4429, 56, 78, 95
Ragh.M. 1.49 64, 123
Ragh.M. 1.50124
Ragh.M. 1.51 82, 140

Ragh.M. 1.53 80
Ragh.M. 1.5411
Ragh.M. 1.55 73, 129
Ragh.M. 1.58 105, 139
Ragh.M. 1.59 26, 31
Ragh.M. 1.60
Ragh.M. 1.6159
Ragh.M. 1.63
Ragh.M. 1.66
Ragh.M. 1.67113
Ragh.M. 1.69165
Ragh.M. 1.7165
Ragh.M. 1.72
Ragh.M. 1.78167
Ragh.M. 1.81 84, 131
Ragh.M. 1.83103, 107, 161
Ragh.M. 1.84113
Ragh.M. 1.85 60, 63
Ragh.M. 1.8665
Ragh.M. 1.8873
Ragh.M. 1.89166
Ragh.M. 1.90 16
Ragh.M. 1.91 80, 95
Ragh.M. 2.3 59
Ragh.M. 2.4 54
Ragh.M. 2.9 115
Ragh.M. 2.10
Ragh.M. 2.1747
Ragh.M. 2.2076
Ragh.M. 2.22130
Ragh.M. 2.23 131
Ragh.M. 2.29
Ragh.M. 2.3074
Ragh.M. 2.54 51
Ragh.M. 2.5572
Ragh.M. 2.6147
Ragh.M. 2.65 54

Ragh.M. 2.70
Ragh.M. 2.71 60
Ragh.M. 2.72 53, 58
Ragh.M. 2.73
Ragh.M. 3.565, 75
Ragh.M. 3.10
Ragh.M. 3.19 121
Ragh.M. 3.35 47
Ragh.M. 3.41 115
Ragh.M. 3.45111
Ragh.M. 3.51 51
Ragh.M. 3.54101
Ragh.M. 3.55134
Ragh.M. 3.57122
Ragh.M. 3.59122
Ragh.M. 3.6262
Ragh.M. 4.1
Ragh.M. 4.4
Ragh.M. 4.5 100, 105
Ragh.M. 4.6 65
Ragh.M. 4.12 53
Ragh.M. 4.3171
Ragh.M. 4.33 28
Ragh.M. 4.35 77
Ragh.M. 4.40
Ragh.M. 4.4253
Ragh.M. 4.45
Ragh.M. 4.61 108
Ragh.M. 4.7474
Ragh.M. 4.8775
Ragh.M. 5.2 56, 76, 78
Ragh.M. 5.3 79
Ragh.M. 5.9 72
Ragh.M. 5.15 73, 124
Ragh.M. 5.19 68
Ragh.M. 5.25 135
Ragh.M. 5.32

# Commentarial Passages Quoted

Ragh.M. 5.36
Ragh.M. 5.40
Ragh.M. 5.4550
Ragh.M. 5.52
Ragh.M. 5.57
Ragh.M. 5.61
Ragh.M. 5.6251
Ragh.M. 5.6963
Ragh.M. 5.73 47, 108
Ragh.M. 5.7577
Ragh.M. 11.8
Ragh.M. 11.1065
Ragh.M. 11.11
Ragh.M. 12.6853
Ragh.M. 12.71163
Ragh.M. 12.73115
Ragh.M. 12.7464
Ragh.M. 12.7565
Ragh.M 4.464
Ragh.M 4.85 58
SBh 1.1.4 182
SBh 1.1.1 177, 180–182, 189,
193, 197, 199, 204, 206,
210-212, 216, 219, 220,
226, 227, 233, 236, 239,
241, 242, 246, 252, 255-
258, 260, 263
SBh 1.1.2 180, 182, 190, 203,
205, 207, 210, 214, 215,
224, 255, 256
SBh 1.1.3 187, 200, 226, 242,
249
SBh 1.1.4181,
196, 197, 199, 204, 205,
207–209, 211, 214, 217,
219, 220, 227-230, 233-

236, 238, 241–243, 245–
248, 252, 254, 255, 261
SBh 1.1.5198, 207, 211, 214,
215, 228, 238, 239, 257,
260
SBh 1.1.6209, 244
SBh 1.1.7181, 204, 217, 219
SBh 1.1.11 217
SBh 1.1.12214, 219, 224, 233,
243, 247, 259
SBh 1.1.14 208, 217
SBh 1.1.19190, 231, 245, 246,
248, 257
SBh 1.1.20 224
SBh 1.1.22206, 223, 246, 253,
254
SBh 1.1.23 206, 252, 254, 262
SBh 1.1.24 248, 249, 259
SBh 1.1.25 219, 248
SBh 1.1.26 224
SBh 1.1.28 245
SBh 1.1.31 226, 245
SBh 1.2.1 216, 224, 253
SBh 2.1.1 233
SBh 2.1.4 177, 198, 236
SBh 2.1.6226, 243
SBh 2.1.7 233
SBh 2.1.24
SBh 2.2.17 176
SBh 4.1.1
SDS p. 2 235
SDS p. 3 256
SDS p. 6235, 263
SDS p. 7245, 247
SDS p. 8 244
SDS p. 25
SDS p. 31

SDS p. 2	181
SDS p. 3	180, 206
Śiśu.M. 1.41	71
Śiśu.M. 14.4	71

÷

f

# **Grammatical Citations**

Coulson 46–49 195	Pā. 2.1.51109–112
Coulson 49–50 231	Pā. 2.1.52 108
Coulson 61–63 251	Pā. 2.1.55106, 107
Coulson 76251	Pā. 2.1.56 107
Coulson 107–11 185	Pā. 2.1.57 ff 102
Coulson 140–42 175	Pā. 2.1.60 105
Coulson 154–58 175	Pā. 2.1.61 103
Coulson 170–71 204	Pā. 2.1.64 113
Coulson 171–77 251	Pā. 2.1.68 103
Coulson 191–95 251	Pā. 2.1.69 104
Coulson 195231	Pā. 2.1.72 123
Coulson 203–05 213	Pā. 2.2.1–2101
Coulson 207–14 251	Pā. 2.2.3–5 101
Coulson 214–15 231	Pā. 2.2.4 123
Coulson 230–35 213	Pā. 2.2.18 113, 114, 116
	Pā. 2.2.19 119
Pā. 1.2.64–1.2.73 145	Pā. 2.2.23-2.2.28 127
Pā. 1.2.68	Pā. 2.2.25134, 135
Pā. 1.4.58 113	Pā. 2.2.26 135
Pā. 1.4.60 116	Pā. 2.2.27 135
Pā. 1.4.61–79116	Pā. 2.2.29 125
Pā. 2.1.5–2.1.20	Pā. 2.2.31 101
Pā. 2.1.6	Pā. 2.3.25 180
Pā. 2.1.7	Pā. 2.4.1 111
Pā. 2.1.13	Pā. 2.4.2 126
Pā. 2.1.22–2.1.48	Pā. 2.4.29 112
Pā. 2.1.22–2.2.22	Pā. 3.1.5-3.1.3243
Pā. 2.1.24	Pā. 3.1.6–3.1.743
Pā. 2.1.34-35100	Pā. 3.1.7 44
Pā. 2.1.37	Pā. 3.1.8 46
Pā. 2.1.47 103	Pā. 3.1.8–3.1.9
Pā. 2.1.49 104	Pā. 3.1.10 47
Pā. 2.1.49–2.1.72 102	Pā. 3.1.11
Pā. 2.1.50108, 109	Pā. 3.1.12–3.1.13 47

Pā. 3.1.15	. 48
Pā. 3.1.17	. 48
Pā. 3.1.18	. 48
Pā. 3.1.21	. 47
Pā. 3.1.22–3.1.24	. 45
Pā. 3.1.25	. 43
Pā. 3.1.26	. 45
Pā. 3.1.91–3.4.117	. 49
Pā. 3.1.101	. 66
Pā. 3.1.108	. 59
Pā. 3.1.115	. 66
Pā. 3.1.124	. 79
Pā. 3.1.134	114
Pā. 3.1.136	. 42
Pā. 3.1.145–147	. 57
Pā. 3.2.78	. 64
Pā. 3.2.79	. 64
Pā. 3.2.83	
Pā. 3.2.84-3.2.122	
Pā. 3.2.99	
Pā. 3.2.134-3.2.179	
Pā. 3.3.1–2, 3.4.75	
Pā. 3.3.126	
Pä. 3.4.68	
Pā. 4.1.1572	-
Pā. 4.1.76–5.4.160	
Pā. 4.1.92	
Pā. 4.1.137	
Pā. 4.1.17074	•
Pā. 4.2.49	
Pā. 4.2.129	
Pā. 4.3.11	
Pā. 4.3.17	
Pā. 4.3.23	
Pā. 4.3.74	-
Pā. 4.3.79	
Pā. 4.3.81	124

Da / 2 90 10.
Pā. 4.3.82         124           Pā. 4.3.143         ff.         124
Pā. 4.3.148 ff
Pā. 4.4.1
Pā. 4.4.45
Pā. 4.4.76
Pā. 4.4.77 1
Pā. 4.4.92
Pā. 4.4.98
Pā. 4.4.99
Pā. 4.4.105
Pā. 5.1.5
Pā. 5.1.9
Pā. 5.1.10
Pā. 5.1.36
Pā. 5.1.42
Pā. 5.1.105
Pā. 5.1.115
Pā. 5.1.115–116 234
Pā. 5.1.117 78
Pā. 5.1.119175
Pā. 5.1.125 75
Pā. 5.1.128
Pā. 5.1.12910, 75
Pā. 5.1.130 75
Pā. 5.2.36
Pā. 5.2.37124, 194
Pā. 5.2.42
Pā. 5.2.121 80
Pā. 5.2.124 80
Pā. 5.2.134 68
Pā. 5.3.7
Pā. 5.3.57 82
Pā. 5.3.67
Pā. 5.3.103
Pā. 5.4.25
Pā. 5.4.38

Pā. 5.4.50
Pā. 5.4.77
Pā. 5.4.78 84
Pā. 5.4.91112, 114
Pā. 5.4.151 84
Pā. 5.4.15384, 131
Pā. 5.4.154 188
Pā. 6.1.94 122
Pā. 6.3.73–77117
Pā. 6.3.101 ff 112
Pā. 6.3.107 113
Pā. 6.3.109 121
Pā. 6.4.174
Pā. 7.4.32 83
Speijer 16–17195
Speijer 31 195
Speijer 193.3 185
Speijer 197185
Speijer 213 b185
Speijer 228–31 185
Speijer 229.1 231
Speijer 235–39 175
Speijer 241 231
Speijer 261–90 251
Speijer 36568 204
Speijer 383–90 213
Speijer 394–490 251
Speijer 430231
Speijer 470 a231
Speijer 491–99 213
NR: 202 204
Whitney 303 204
Whitney 60743
Whitney 77543
Whitney 996 ff
Whitney 1000–1025

Whitney 1026–104044	
Whitney 1040 43	
Whitney 1041–105245	
Whitney 1041 b 43	
Whitney 1053–106846	
Whitney 1056 43	
Whitney 1090 ff 116	
Whitney 1107 231	
Whitney 1143–120149	
Whitney 1202–124567	
Whitney 1222 h 185	
Whitney 1237 175	
Whitney 1239 175	
Whitney 1252–1258125	
Whitney 1253.1125	
Whitney 1253.2126	
Whitney 1262–129196	
Whitney 1264–127896	
Whitney 1279-1291 102	
Whitney 1280 102	
Whitney 1281 ff 112	
Whitney 1286–1287119	
Whitney 1288.1117	
Whitney 1292 ff 115	
Whitney 1293-1308 127	
Whitney 1310 115	
Whitney 1312 108	
Whitney 1313 92	

Sansk	rit	Ind	ex
-------	-----	-----	----

. .

eg 1

-a, 10, 11, 71, 74-76, 83, 206 -aka, 73 agre, 229 aC, 84 aÑ, 10, 75 aN, 38, 39, 72, 74-77, 82 ata eva, 256 atah, 209, 256, 258 atra, 253, 258, 259 atra ke cid ähuh, 243, 249 atrôcyate, 244 atrâbhidhīyate, 244 atha, 240-242, 251, 252 athavā, 249 adah, 262 adhikarane, 55 -ana, 38, 39, 72 anavadyam, 218 anubandha, xxvi, 16, 37 anuvartate, 164 anuvrtta, 163 anuvrtti, 163, 164 anekapada-bahuvrihi, 132 anena, 25 -anta, 231 -antara, 191 anya, 191, 261 anyac ca, 254 anyathā, 246 anye, 262 anvaya, 154, 157, 159, 217 anvaya-mukhena, 149 anvayamukhī, 147-149, 151, 153-156, 159

# 

apara āha, 243, 249 aparam ca, 254 apare, 262 apādāne, 55 api, 198, 210, 257, 258 api ca, 254, 257 a-prakrta, 224 abhāva, 206, 207 abhūta-tadbhāve, 83 abhyupagatam, 218 abhyupagama, 207 abhyupetam, 218 ayam, 69, 261, 264 -artha, 191 arthah, 217 arh, 219 alaukika-vigraha, 34 avatarana, 151, 224 avasitam, 218 avidyā-hetukatvam, 179 avyayībhāva, 86, 87, 90, 92, 93, 95, 185 asiddhi, 206 astu, 248 asmin, 259 asya vidyate, 79 asyâsti, 79 a-sva-pada-vigraha, 90 ā, 16 āksipati, 27 āksipta, 243

āksepasamādhāna, 5, 171

-ākhyā, 30

2	82
-	04

ān, 16	iti niścīyate, 217		ity āmnāyate, 229	upapada, 37, 42, 57–64, 70, 119,
ātma-māne, 64	iti nyāyāt, 229	•	ity āśańkya, 242	154
-ādi, 22, 186, 190, 231-233, 235	iti prasiddham, 218, 230		ity āśankyâha, 225	upapannam, 218
-ādika, 232	iti prasidhyate, 230		ity āśayah, 25	upamāna, 234, 235
-ādya, 232	iti prāpte, 243		ity uktvā, 214	upamānapūrvapada, 106
āpatti, 207, 246	iti bādhitam, 218	2 1.	ity upapannam, 218	upamita-samāsa, 107
āpad, 246	iti bhavitum arhati, 219		ity evam prāpte, 214	upameya, 236
āpadyate, 246	iti bhavişyati, 218		ity-ādi, 213, 233, 234	upādā, 223
āmnā, 229	iti bhāvaḥ, 23–25		ity-ādi ca, 234	upādāna, 223
āmnāta, 229	iti matvā, 214		ityādivat, 213, 235	upādhi, 57, 73
āśańk, 242	iti yāvat, 23, 25		ity-evam-ādi, 234	-
āśankyate, 242	iti yuktam, 218		idantā, 182	-ū, 84
āha, 225, 243	iti viruddham, 217, 218		idam, 52, 54, 55, 77, 182, 223,	atraéana 81 145 145 146
	iti virudhyate, 217		253, 258, 262–264	ekaścza, 81, 145, 145, 146
-ika, 73, 74	iti virodhah, 217		-in, 64, 68	ekašesa-vrtti, 35 eka 262
icchati, 44	iti vivaksitah, 25		inI, 68, 80	eke, 262
it, 37	iti vivaksyate, 26		indrasya, 179	-eņya, 73
itaḥ, 256	iti śeșah, 160, 163	,	iva, 64, 89, 231, 234–236	etat, 214, 223, 262–264 etena, 25
itaC, 78	iti śrūyate, 230		iş, 44	-eya, 71
itaretara dvandva, 145	iti sambhāvyate, 218	· · ·	iha, 253, 254, 258, 259	eva, 82, 193, 256, 257
itaretara-dvandva, 125	iti siddham, 218	· ·	iha tu, 254	evan prāpte, 214
iti, 15, 52, 56, 57, 61–63, 79, 173,	iti sugamam, 218		iha loke, 254	evam sati, 211
179, 195, 198–200, 203,	iti sūcitah, 26	2	ihâpi, 254	evan, 211, 214, 234, 236
207, 208, 213–218, 220,	iti sūcyate, 26		K '	evani, 211, 214, 234, 250 evam api, 258
221, 223, 228, 240, 244,	iti spastam, 218		-ī, 72, 74, 84	evani api, 258
245, 254, 261, 263	iti smaryate, 230		ī-, 83	-ka, 82, 83, 179, 185, 188
iti kŗtvā, 208	itivat, 213, 235		-īna, 71, 73, 74	Ka, 37, 38
iti gamyate, 26, 218	ittham, 236	$\lambda_{i}$	īyasUN, 82	ka, 188
iti cet, 242	ity ataḥ, 208			katham punah, 241
iti câivam-ādi, 234	ity atra, 254		ukta, 226–228	katham-bhūta, 148
iti tasmāt, 208	ity anavadyam, 217		ukta-nyāyena, 228	kathambhūtinī, 147, 148, 150, 151,
iti tātparyaḥ, 25	ity anvayaḥ, 159, 217	- 	ucyate, 225, 244	154–156, 159, 163
iti tena, 208	ity abhiprāyaḥ, 25	1	uņ, 49	katham, 241
iti drstam, 218, 229	ity abhyupagatam, 217		uņ-ādi, 49	kaP, 84
iti drsyate, 229	ity abhyupetam, 218	:	uttarapada-dvigu, 110	karane, 54, 62
iti dhvanyate, 26	ity arthaḥ, 23–26, 217		uttaram āha, 225	karane LyuȚ, 39
iti niścitam, 217, 218	ity avasitam, 218		upapatti, 206, 207	kartari, 52–54

. althe

kartari krt, 49, 60, 61, 64 kartary upamāne, 64 karmaņi, 52, 54, 56 karmaņi krt, 61 karmadhāraya, 36, 86–89, 91, 102, 106, 186, 187, 189 kalāpaka, 161 -kalpa, 191 kalpaP, 83, 191 kasmāt, 259 kāmam, 248 -kāra, 15 kāraka, 52, 60, 61 kāraņa, 208 kārikā, 1 kāvya, 2 kim ca, 251, 254 kim tarbi 255	-garbha, 192 guṇaḥ, 176 guru, 176 gurutā, 176 gurutvam, 176 guru-bhāvaḥ, 176 gṛhīta, 223 gminI, 80 grah, 223 Nas, 34		DUkr $\tilde{N}$ , 39 DUkr $\tilde{N}$ karaņe, 39 dhaK, 11 dha $\tilde{N}$ , 71 Na, 71 namul, 51 nic, 45 nij-anta-dhātu, 45 Nya, 74 NyaT, 79 -t-, 72 tac-chīla, 57, 58 tat, 67, 77, 164, 185, 189, 190, 209, 228, 245, 252, 256- 259, 262-264 tat-, 63	tad ayuktam, 245 tad asat, 245 tad idam, 263 tad etat, 263 tad etat, 263 tad eva, 257 tad yathā, 231, 255 tad-artham, 78 tad-arham, 78 taddhita, 10, 34, 35, 39, 67, 68, 74, 83, 191, 194 taddhita-vṛtti, 35 taddhitārtha-dvigu, 109 taddhitânta, 67 tadvat, 236 tan na, 245 tan mā bhūd iti, 245 tayaP, 78 -tayā, 180
kalāpaka, 161	0		namul, 51	tad-artham, 78
-kalpa, 191			•	tad-arham, 78
kalpaP, 83, 191	÷ · ·	يە يىل	•	taddhita, 10, 34, 35, 39, 67, 68,
kasmāt, 259	•			74, 83, 191, 194
	<b>Q</b>			
	•		inya1,79	taddhitārtha-dvigu, 109
	guru-bhāvaḥ, 176		-t 72	÷
	gṛhīta, 223			
•	gminI, 80			
•	grah, 223			
•				
•	Nas, 34	1		- · ·
kim tarhi, 255	ŇīР, 72, 74			-
kiṃ tu, 255			tat tathā, 40	tayoḥ, 259
kim-višista, 148	ca, 88, 205, 206		tat tathā-bhūtam, 40	taraP, 82
kim, 185, 190, 241, 258	cikitsati, 43	•	tat tathôktam, 40	tarhi, 248
ku-, 112	cintayati, 43		tatah, 209, 256, 258, 259	-tas, 81, 258
kutah, 258, 259	cur-ādi, 43		tatpurusa, 58–60, 68, 70, 83, 86,	tasIL, 81
kutra, 258	cet, 252		87, 90, 92, 96, 121, 154,	tasmai hitam, 67
kulaka, 161, 162	corayati, 43	ł	157, 176, 186, 187, 194	tasmāt, 209, 256, 259
kr, 51, 83	cvi, 47, 83		tatra, 211, 252, 253, 258, 259	tasmād eva, 256
krt, 11, 35, 38, 42, 45, 46, 49, 50,			tatra bhavam, 72, 73	tasmin, 211, 259
52-64	-jana, 192		tatra sādhuņ, 73	tasya, 179
krd-anta, 49	-jāta, 193		tatrâivam sati, 211	tasya karma, 75
krd-vrtti, 35	jugupsate, 43		tat-śīla, 57	tasya bhāvah, 75
ke cit, 262			tatpurușa, 176, 178, 179, 185, 186,	tasya rājā, 76
kevala-samāsa, 88, 89	țīkā, 147, 149		188, 189, 191–193	tasya vikārah, 76
krt, 61	țyuL, 72	1	tathā, 167, 211, 236, 260, 261	tasya samūhah, 76
•			tathā ca, 254	tasyêdam, 77
kva, 241, 258	thaK, 74		tathā sati, 211	tasyâpatyam, 74
KviP, 38	țhaÑ, 73		tathā hi, 231, 255	-tā, 175–183, 188, 204, 206
kha, 71	Da, 11	<b>.</b>	tathâpi, 258	tācchīlya, 63

tācchīlye, 64	na ca mantavyam, 244	punah punah, 45	brūmaḥ, 244
tārakâdi, 78	na ca vācyam, 244	purohitâdi, 76	• · · · · · · · · · · · · · · · · · · ·
tāvat, 253	na dusyati, 246	-pūrva, 193	bhagavān, 224
tāvata, 253	na doşah, 246	pūrva-pakṣa, 238	bhavişyati, 57, 218
tāvatā, 209	nanu, 240	pūrvapaksa, 220, 237, 238, 240,	bhavet, 247
-ti, 39, 206	napumsakâikaśesa, 146	243, 244	-bhāvaḥ, 176
tin, 49	nahi, 199	pūrvapaksin, 238, 239, 244	bhāve, 50, 51, 59, 65
titiksate, 43	nañ, 16	pũrvapaksin, 239	bhāve kṛt, 59, 60
tu, 255	nañ-tatpurusa, 36	prsodarâdi, 121	bhāșya, 171, 172
tena, 209, 256	-nāma, 30	prakrta, 224	bhāṣyakāra, 224
teşu, 253	nāma, 28, 30	prajñâdi, 82	bhū, 83, 219
tesu madhye, 253	nāma-dhātu, 46	pratīka, 150	-bhūta, 165, 166, 189, 198
-tra, 81, 258	nitya, 91	prathamânta-vigraha, 41	bhūta, 198
-tva, 175-183, 188, 204, 206	nitya-samāsa, 90, 91	-prabhrti, 232	bhūte, 56
-tvāt, 180	nipāta, 68	prabhrti, 232	bhṛśam, 45
-tvena, 180	nipātana, 65, 68	prasanga, 207, 246	-bheda, 194
	nipātita, 36, 65, 68	prasajyate, 246	
darśana, 207	nimitta-saptami, 210	prasanj, 246	matUP, 80, 81
-darśanāt, 229	niścitam, 217, 218	prasiddham, 218, 230	mate, 230
darśayişyati, 226	niścīyate, 217	prasiddhi, 207	matvarthīya, 68, 79
dig-bahuvrīhi, 135	nivrtta, 164	-prasiddheh, 230	matvā, 214
dig-bahuvrīhi, 135	nivrtti, 164	prasidh, 230	madhyamapadalopa-bahuvrihi, 42
dīdāmsate, 43	nyāya, 229	-	-maya, 124, 208
dṛś, 229	nâitad evam, 245	prasidhyate, 230	mayaT, 124, 208
drśyate, 229	nâișa dosah, 246	prādi-tatpurusa, 113, 114	mayūravyamsakâdi, 123
dṛṣṭam, 218, 229	nâyam dosah, 246	prādi-bahuvrīhi, 130	mahākāvya, xxvi
doșa, 246	nayanı doşan, 240	prādhānya, 86	mahā-balatvāt, 179
dvandva, 85–91, 125, 176, 178,	pakșa, 27, 238	prāpta, 243	mahākāvya, 2
185, 186, 189	pathati, 16	prāpte, 214	mahākāvya, 2
dvandvâikaśesa, 145	padaccheda, 4, 9, 13	bahuvrīhi, 30, 40, 41, 77, 79, 83–	-mātra, 193, 194
dvigu, 108	padapāțha, 13	89, 91, 127, 128, 130,	mātraC, 78, 82, 194
	padārthokti, 4, 9	132, 134, 136, 157, 178,	-mātrā, 193
dhātu-lopa, 114	param tu, 255	179, 185–188, 190–193,	mīmāmsate, 43
dhvanyate, 26	parigrah, 223	231–233, 235	mūla, 147–151, 154, 156, 157, 223–
na, 16, 244	parinisthita-vibhakti-vigraha, 41		226
na ca, 199, 244	pāțha, 16	bāḍham, 248 bādhitam, 218	mūlakāra, 223–228
na ca codanīyam, 244	punah, 255		va 11 70 76
na ca couantyant, 244	punai, 200	bībhatsate, 43	-ya, 11, 70–76

.

yaK, 76 yan, 45 yan-anta-dhātu, 45 vaT. 11, 34, 70-72, 74-76, 78, 82 yat, 164, 185, 190, 208, 228, 245, 258, 259, 261, 262 vat tu. 228 yat tûktam, 228 vatah, 208, 258 yatra, 258, 259 yatra yatra, 259 yathā, 167, 231, 235, 236, 260. 261 vathā tathā, 167 yathôkta, 228 yadi, 252 yady api, 248, 258 vadvā, 249 vasmāt, 208 yasmin, 259 yasya, 230 vasya mate, 230 yāvatā, 208 yu, 38 yuktam, 218, 219 yugma, 161 yugmaka, 161, 162 yuj, 218 yuvâdi, 75 yena, 208, 259 yo yah, 259 yoga, 207 yojya, 160 rūpakasamāsa, 108 laukika-vigraha, 34

laksana-labdha, 36 laksayati, 27 LyuT, 38, 39 vaktavvam, 200 vakşyati, 226 vaksyamāna, 226 -vat, 231, 234, 235 vatI, 71, 79 -vant, 234 vartate, 164 vartamäne, 56 vākyayojanā, 5, 9 vākya-vrtti, 35, 187 vigraha, 4, 9, 21, 33, 35, 86, 91 vinI, 80 vibhakti-tatpurusa, 87, 186, 189 vibhakti-samjñaka, 81 vibhaktisamiñaka, 68 viruddham, 217, 218 virudhyate, 217 virodhah, 217 vivaksita, 226 vivaksitah, 25 vivaksyate, 26 -viśesa, 30, 194 viśesana, 161 vuÑ, 73 vrtti, 35 vrddhi, 10, 49, 176 śak. 218 śakyam, 219 śakyo, 219 -śabda, 15 śāstra, 1, 2 śāstrīva-vigraha, 34

-śīla, 58, 63 śīlam asya, 63 śīšāmsate, 43 śukla, 176 śuklatvam, 176 śravaņa, 207 -śravaņāt, 230 śrut, 230 śruti, 230 śrūyate, 230

-sa, 43 samkhyā-bahuvrīhi, 134 -samjñā, 30 samdānitaka, 161 sambadhvate, 160 sambandha, 160 sambandhanīya, 160 sambandhin, 77 sat, 165, 166, 198 sati, 166, 210, 211 satī, 165 satyam, 248 satya-vädī, 57 san, 35, 43, 44, 165 san-ādi, 49 san-ādy-anta-dhātu, 43 san-ādy-anta-dhātu-vrtti, 35 sant. 198, 211 sann-anta-dhātu, 44 samasânta, 81 samāsa-vrtti, 35, 187 samāsânta, 83 samāhāra, 90 samāhāra-dvandva, 36, 126 samāhāra-dvigu, 111 sampradāne, 54

ţ

-sambandhin, 78 sambhava, 206, 207 sambhāvyate, 218 sarvathâpi, 248 saha-bahuvrihi, 131 siddham, 218 siddhānta, 237, 238, 244 siddhāntin, 238-240, 244 siddhi, 207 sU. 34 su-, 112 sugamam, 218 sup, 67 sup-sup-samāsa, 89 sūcitah, 26 sūcyate, 26 sūtra, 1, 3, 163, 164, 238 sūtrakāra, 224, 226, 227, 242 strī. 67 spastam, 218 smarana, 207 -smaranāt, 230 smŗ, 230 syāt, 247 svād etat, 247 syād etad evam, 247 svärthika, 81 svårthika, 68

hi, 203, 211, 251 hita, 67 he, 167

# **English Index**

ablative and instrumental phrases, 206 abstract nouns, 177-185 causal, 182, 183 common constructions, 179-182 formation, 177, 178 in ablative case, 182, 183 in accusative case, 183, 185 in compound, 180 in dative case, 183 in genitive case, 184 in instrumental case, 182-185 in locative case, 184 in nominative case, 183 predicative instrumental, 184, 185 significance, 178, 179 with subjective genitive, 179-182 with two genitives, 182 with verbs of motion, 185 abstract suffixes, 177 accompanying words, see upapada compounds adhikarane krt, see primary derivation, in kāraka meanings ādi in glosses, 22 adverbs, 169 pronominal, 260-264 Alankāracūdāmani of Hemacandra, 3 alternative interpretations, 251 citing others, 251

giving second views, 251 alternatives given by commentators, 156 Amarakosa of Amarasimha, 10 Amarasimha, 10, 29 analysis, 4, 5, 12, 19, 33–42 and inflection of the original, 41.42 general principles, 35-39 glossing within, 40 in ordinary language, 34, 37, 153 in technical language, 34, 37-39.153 insertion of, 21, 39-42 interruption of, 41, 42 obscuration of, 37, 40 types of, 34 Annambhatta, 2, 161 antecendent of pronoun supplying, 32 anubandhas, see indicatory phonemes anuvrtti, see words carried over anva repeated, 263, 264 apādāne krt, see primary derivation, in kāraka meanings apposition, 20 arranging the words of the text, 149-160 anvayamukhī approach, 150, 151 insertion of explanatory material, 151–156

kathambhūtinī approach, 149, 150 overview, 149 variations on the two main approaches, 156-160 Arunagirinātha, 159 atha in objections, 243 attibutive noun, 199 attributive adjectives, 167 attributive nouns with bhūta, 200 with sant, 200 attributive substantives, 167, 168 author references to, see references authorities references to, see references autocommentaries, 2, 3 auto-commentary, 229 avyayībhāva compounds, see compounds, avyavībhāva bahuvrīhi compounds, see compounds, bahuvrīhi Bhagavadgītā, 168 Bhāmaha, 2 bhāsya style, 173-266 introduction, 173-175 Bhatti, 2 Bhattikāvya of Bhatti, 2 bhave krt, see primary derivation bhavisyati krt, see primary derivationwith temporal restrictions -bhūta marking a karmadhāraya, 191,

192

bhūte krt, see primary derivationwith temporal restrictions Böhtlingk, O. von, 38 bold type, xxvi, 11, 14, 151 bound forms, 92, 119–121, see upapada compounds Brahmasūtrabhāsya of Śańkara, xxvi

са

in series of causes, 207, 208 carat, see circumflex accent Cardona, George, 38 Cāritravardhana, 158 causal sentences, 205-214 with ca in causal series, 207, 208 with hi, 213, 214 with iti clauses, 209, 210 with ablative and instrumental phrases, 206 with locative absolute, 211-213 with relatives, 210, 211 with series of causes, 206, 207 with verbal nouns, 208, 209 causative verbs, 45, 46 circumflex accent. xxvi citation of words, 14-16 using *iti*, 15 using -sabda or -kāra, 15 using indicatory phonemes, 16 using inflected indeclinable, 15, 16 colon, xxvii commentaries abundance of, 1-3

as vehicles of originality, 2 assistance they offer, 1, 3-5 five services of, 3, 9, 10 importance of, 1-3 reasons for, 1-3 upon commentaries, 2, 3 commentary references to in commentary, see references comparisons, 233-238 usind itivat and ityādivat, 237 using iva, 238 using vat, 236, 237 using yathā, 238 complex formations, 4, 5, 12, 19, 33 - 42modes of, see modes of complex formations compound nouns, see compounds compounds, 4, 14, 85-145, 187-196 avyayībhāva, 86, 87, 92-95 irregular, 95 obligatory, 93-95 optional, 93 bahuvrīhi, 88, 127-137 anekapada, 132, 133 dig, 136 prādi, 131 saha, 131, 132 samkhyā, 135, 136 appositional, 128–133 in *ādi*, 233–236 irregular, 136, 137 marked by -ka, 190 negative, 130 nonappositional, 133, 134

of reciprocal combat, 136 special, 135, 136 dvandva, 87, 88, 125-127 itaretara, 126 samāhāra, 126, 127 marked by tat, 191 dvigu, 109–112 samāhāra, 111, 112 taddhitârtha, 110 uttarapada, 110, 111 gati-tatpurusa, see tatpurusa with inseparable first member karmadhārava, 87, 102-106 marked by -bhūta, 191, 192 of comparison, 106-108 kevala-samāsa, 88, 89 prādi-tatpurusa, see tatpurusa with inseparable first member tatpurusa, 87, 96-125 dvigu, see dvigu gati, see tatpurusa with inseparable first member karmadhāraya, see karmadhāraya mayūravyamsakādi, 123, 124 prādi, see tatpurusa with inseparable first member prsodarādi, 121-123 upapada, 119–121 case, see tatpurusa, dependent dependent, 96-102 irregular, 121-125 negative, 118, 119 with inseparable first member, 112–118

additional meaning, 89, 90 aggregation, 90 clarification of connection, 90 comparison, 89 basic categories, 86-89 containing pronouns, 192 ending in -antara, 193 ending in -artha, 193 ending in -garbha, 194 ending in *-gata*, 194 ending in -jana, 194 ending in -jāta, 195 ending in -kalpa, 193 ending in -mātra, 195, 196 ending in -pūrva, 195 ending in -viśesa, 196 ending in special terms, 193-196 general principles, 85–92 irregular, 92 longer, 137-145 analyzing, 187-189 choice of options, 142-145 general principles, 138-141 special techniques, 141, 142 obligatory, 90-92 asvapadavigraha, 90, 91 bound forms, 92, 119-121 conventional meanings, 91 optionality of, 90 relation to outside words, 189, 190 connective particles, 253-258 construction indicating, see indicating the construction

293

in *-am*. 51 continuatives, 51 correlatives, 261-264 multiple, 262 repeated, 261 Coulson, Michael, xxv, xxvii cvi formations, 47, 83, 84 definition marking as, 29 definitions, 28-30 offered by commentator, 28, 29 quoted from dictionaries, 29, 30 demonstratives, 264-266 level of proximity, 264, 265 referring to earlier and later passages, 265 denominative verbs, 46-48 "becoming something", 47 "behavior like", 47 "behavior toward", 47 "desire to have", 46 "doing or making something", 48 "feeling something", 48 "making something so", 47 "making use of something", 48 derivation primary, see primary derivation secondary, see secondary derivation derivative verbs, 43-48

continuative

types of, 43, 44 desiderative verbs, 44 Devasthali, G. V., 38 Dhātupātha appendix to Pānini's grammar, 39 dialectic style, 239-242 example of, 241, 242 dictionaries formats of, 29 in Sanskrit, 10, 12, 29, 30 Sanskrit-to-English, 4 diminutive suffixes, see secondary derivation, svârthika suffixes discussion, 239-251 direct address in, 241 double meanings, see puns dvandva compounds, see compounds, dvandva dvigu compounds, see compounds, dvigu ekaśesa formations, 147, 148

male and female, 147 neuter, 148 ellipsis, xxvi emphatic particles, 258–260 enclitics glossing of, 20, 21 euphonic combination, *see* sandhi examples, 233–238 using  $-\bar{a}di$ , 233–236 using *vat*, 236, 237 extension meaning by, 27

five services, 3, 9, 10

footnotes, 3 formulas of analysis, *see* analysis future participle, 57

genitive case in reference to views, 232 gerund in English, 50 in Sanskrit, see continuatives gerundive, 65 glosses location of, 20, 21 glosses, expansions of, 22, 23 glossing, 4, 5, 19-32 by equivalent forms, 21 types of, 10–12 with participles, 21 with periphrastic constructions, 21 habitual action, see primary derivation halfpenny, 101 Hemacandra, 3 Hemādri, 37, 42 hi in causal sentences, 213, 214 honorifics. 82 implication, see extension implications explicit statement of, 22 implied contrasts explicit statement of, 22, 23 indeclinable inflected, 15, 16 indeclinables

glossing of, 31

indicating the construction, 5, 149-169 adjectives and adverbs, 163 arranging the words of the text, see arranging the words of the text connections between statements. 164 connections between verses. 163 direct statement of connections, 161–165 marking features of syntax, 166-169 attributive adjectives, 167 attributive substantives, 167, 168 locative absolute, 168, 169 subject and predicate, 166. 167 with sambandha, etc., 162 with viśesana, 163 with yojya, 163 words carried over. see words carried over words supplied, see words supplied indicatory phonemes, xxvi, 16, 37-39 infinitives, 51 Ingalls, Daniel H. H., xxv instrumental case of present participle, 228 intensive verbs, 45 interpretations alternative. see alternative interpretations

introduction to section of commentary, 153 *iti* at end of passage, 222, 223 common constructions, 215.

in analysis of *krt* formations, 52, 56, 57 in causal sentences, 209, 210 in comment, 218–220 in explanation, 218 in quotation, 216–218 scope of, 200, 201 uses of, 215–223 with infinitive, 220–222 *iti cet*, 244 *ityādi* in examples, 235, 236

#### -ka

marking a bahuvrīhi, 190 Kale, M. R., 38 Kālidāsa, 2, 10, 150 kārkake krt, see primary derivation karane krt, see primary derivation. in kāraka meanings kārikā works, 1 karmadhāraya compounds, see compounds, karmadhārava karmani krt, see primary derivation, in kāraka meanings kartari krt, see primary derivation, in kāraka meanings Kāśikāvrtti of Jayāditya and Vāmana. 46 kāvya poems, 2 Kāvyālankāra of Bhāmaha, 2

Kāvyānušāsana of Hemacandra, 3 Kešavasvāmin, 29 kevala-samāsa compounds, see compounds, kevala-samāsa Kiparsky, Paul, 91 Kiraņāvalī of Udayana, 2 krt suffixes, see primary derivation Kumārasambhava of Kālidāsa, 138, 141, 142, 150, 152, 157

locative absolute, 168, 169, 184 abbreviated, 213 concessive, 213 expressing a cause, 212 expressing a condition, 212 in causal sentences, 211–213 locative case in dictionaries, 29 in giving the meaning of a root, 39 in giving the meaning of suffixes, 39 in glossing particles, 31

#### Māgha, 153

*mahākāvya* poems, xxvi, 2, 154 Mallinātha, xxvi, 10, 11, 37, 40, 42, 123, 139, 142–145, 151, 155–157, 167, 168 memorization, 2 modes of complex formations, 35 *mūla* (root text), 149–153, 156– 159

na ca in objections, 246

namul gerund, see continuative in -am nanu in objections, 243 Nārāyaṇapaṇḍita, 159, 161 nominal compounds, see compounds nominative case in analyses, 41, 56, 62, 64 in definitions, 10 in dictionaries, 29 nouns abstract, see abstract nouns null suffix, 37, 38 Nyāyakośa of Bhīmācārya Jhalakīkar, 3 objections, 242–246

replies to, 246-250 answering, 5 answering of, 173 using *āksipta*, 245 using iti cet, 244 using ityāśanya, 244 using na ca, 246 using nanu or atha, 243 using prapta, 245 using interrogatives, 243 using verbs of speaking, 245, 246 omission of words from the root text. 156 oral argument, 1 oral debate, 5, 239, 240 oral instruction. 5 original text, see mūla (root text) Padapātha text of the Veda, 13

Padārthasangraha of Praśastapāda, 2

Pāninian grammar, xxi, xxii, xxvi, 5, 9, 10, 12, 16, 19, 33-39, 43, 46, 49, 50, 52, 56-59, 64, 67, 68, 73. 74, 77, 78, 83, 88, 89, 91, 94, 123, 152, 161, 193, 196, 236 paraphrases more removed, 23-27 paraphrasing, 4, 19-32 with anena, 25, 26 with iti yāvat, 25 with ity arthah, 23, 24 with ity bhavah, 24 with other tags, 25 Parāśarapurāna, 3 parentheses, xxvii participles, 51, 64, 65 in references, 230 particles, 253-260 connecting sentences, 253-258 giving the meaning of, 31 of emphasis, 258-260 partitive genitive, 101 past active participle, 56, 62 past passive participle, 56, 63, 65 patronymics, 74 pejoratives, 82 phrase mode, 35 pick-up pronoun, 28, 41, 42 plural in references, 229 possessive suffixes, see secondary derivation Praśastapāda, 2 predicate nominative, 197-199 predicative accusative, 198

predicative instrumental, see abstract nouns preliminary view, see pūrvapaksa present participle, 65 present passive participle, 56 primary derivation, 11, 42, 49-66 habitual action, 57, 58 in kāraka meanings, 52-55 adhikarane, 55 apādāne, 55 karane, 54 karmani, 52, 54 kartari, 52, 53 sampradāne, 54 in the sense of the action, 50. 51, 59, 60, 65 irregular forms, 65, 66 other restrictions of meaning, 57, 58 verbal participles, see participles with accompanying words, see upapada compounds with temporal restrictions. 55-57 future. 57 past, 56 present, 56, 57 primary nominal derivation, see primary derivation pronoun pick-up, see pick-up pronoun relative, see relative pronouns pronouns, 260-266 adverbs used as, 260, 261 correlative, 261-264 demonstrative, 264–266

### English Index

glossing of, 32 in compounds containing pronouns, 192 pronouns subjects, 198 proper names glossing of, 30 prose word order, 150, 156 puns glossing of, 27, 28 pūrvapaksa, 240 quotation marks, 15 Raghuvamśa of Kālidāsa, 10 Rāma Govinda Siddhāntavāgīśa, 158 references, 225-232 to authorities, 231, 232 to commentary, 229-231 to earlier topic, 231 to text or author, 225-229 to views, 232 relative pronouns, 151 in analysis of krt formations, 52 in analysis of taddhita formations, 80 in causal sentences, 210, 211 in references, 230 repetition of text, 14, 15 replies, 246-250 using dosa, etc., 248, 249 using prasanga, etc., 248 using tarhi, 250 using concessives, 250 using denials, 247 using optatives, 249

using verbs of speaking, 246, 247 root text, *see mūla* (root text) root-forming suffixes, 43, 49

Samāsacakra, 36, 150 sampradane krt, see primary derivation, in kāraka meanings sandhi, xxvi, 4, 13 Śańkara, xxvi Sanskrit texts modern editions, 13 Sanskrit tradition fondness for commentaries. 1, 2 Indian curriculum, 2 peculiar features, 1 pedagogical concerns, 2, 3 richness of commentaries, 1 śāstra works, 1, 2 secondary conjugations, see derivative verbs secondary derivation, 10, 35, 67-84 case-substitute suffixes, 81 compound formulas, 78, 79 cvi. see cvi formations dependent formulas, 68-74 ablative, 71, 72 accusative, 70 dative, 71 instrumental, 71 locative, 72-74 genitive formulas, 74-77 connection in general, 77 tasya bhāvah, 75 tasyâpatyam, 75

nominative formulas, 77, 78 possessive suffixes, 79-81 straightforward possession, 80 with additional meaning, 80, 81 svârthika suffixes, 82-84 diminutives, 82 samāsânta suffixes, 84 secondary nominal derivation, see secondary derivation secondary usage, see extension semicolon, xxvii sentences causal, see causal sentences long, 201-203 separation of words, see word division series of causes, 206, 207 siddhānta, 240 Śiśupālavadha of Māgha, 144, 145, 153 skeleton sentence, 149, 150, 161 special items glossing of, 30-32 species names glossing of, 31 speech referring to, 26-28 Speijer, J. S., xxv, xxvii square brackets, xxvi, xxvii, 11 standard of comparison, 236, 237 students beginning, 4, 168, 173, 174 frustrations of, 4 in traditional curriculum, 2, 36

memory of, 30 of the sacred lore, 26, 68 ponies for, 13 problems of, 3 tedious life of. 5 subcommentaries, 2, 3 subject and predicate, 166, 167 subject at hand, 226 subjective genitive, 179-182 suffixes Pānini's treatment of. 37-39 suffix-formations resembling compounds, 125, 193, 196 suggestions conveying of, 26 sūtra works, 1 sūtra works, 1, 3, 165, 166 svârthika suffixes, see secondary derivation synonyms simple, 20, 40 taddhita suffixes, see secondary derivation Tarkadīpikā of Annambhatta, 2, 161 Tarkasangraha of Annambhatta, 2.161 tat marking a dvandva, 191 tatpurusa compounds, see compounds, tatpurusa technical terms glossing of, 28

temporal restrictions, see primary

derivation

tenth-class verbs, 43

### English Index

text references to, *see* references text being commented on, *see mūla* (root text) "therefore" words for, 258 thesaurus, 29 topic of discussion, 226

Udayana, 2

un-ādi suffixes, 49 uncommon words glossing of, 28 Upanisads, 1 upapada compounds, 58–64 with kartari krt, 60, 61 with krt in another kāraka meaning, 62, 63 with other conditions of meaning, 63, 64 with passive krt formation, 62 with temporal restriction, 62, 63 uppercase letters, xxvi

v, 225, 226 Vaišeșikasūtra of Akṣapāda, 2 Vallabhadeva, 142–145, 150, 154, 157, 159 variant readings, 16, 17 vartamāne kṛt, see primary derivationwith temporal restrictions vat in comparisons, 236–238 Vedānta, xxvi Vedāntasūtra of Bādarāyaṇa, 1 verbal endings, 49 verbal nouns

in series of causes, 208, 209 verbal participles, see participles verbs of saying, 228 verbs of speaking in objections, 245, 246 in replies, 246, 247 verses in syntactic connection, 163 vibhaktisamjñaka suffixes, see secondary derivation, casesubstitute suffixes views references to, see references vigraha, see analysis Viveka of Hemacandra, 3 vocatives, 151, 169 vrddhi derivatives, 178

Whitney, W. D., xxvii word division, 4, 13-17 specification of, 14 word order, 197-203 scope of *iti*, 200, 201 with attributive noun, 199 with predicate nominative, 197-199 with pronouns subjects, 198 words citation of, see citation of words repetition of, see repetition of text words carried over, 165, 166 words implied, xxvii words supplied, xxvi, 165, 166

yat as conjunction, 263